AD-A015 245

VULNERABILITY MODEL. A SIMULATION SYSTEM FOR ASSESSING DAMAGE RESULTING FROM MARINE SPILLS

ENVIRO CONTROL, INCORPORATED

PREPARED FOR COAST GUARD

JUNE 1975

DISTRIBUTED BY:

ومحرب بقديقان بمخداه





THIS DOCUMENT IS BEST QUALITY AVAILABLE. THE COPY FURNISHED TO DTIC CONTAINED A SIGNIFICANT NUMBER OF PAGES WHICH DO NOT REPRODUCE LEGIBLY.

> REPRODUCED FROM BEST AVAILABLE COPY

VULNERABILITY MODEL

Report No. CG-D-137-75

5

2

5

01

AD A

A Simulation System for Assessing Damage Resulting from Marine Spills



June 1975

FINAL REPORT

NATIONAL TECHNICAL INFORMATION SERVICE US Department of Commerce spin-inglided, VA 22151

Document is available to the public through the National Technical Information Service, Springfield, Virginia 22151

Prepared for

BEPARTMENT OF TRANSPORTATION UNITED STATES COAST BUARD Office of Research and Development Washington, D.C. 20500

And the second second

		• •	rennical Kabatt D	ocumentation rege
1. Report No.	2. Government Accession No.	3. 0	Recipient's Catalog N	•.
CG-D-136-75				
4. Title and Subtitle			Report Date	
VULNERAFILITY MODEL: A S	Simulation System for		Tune 1075	
Assessing Damage Resulting	ng from Marine Spills		June 1975	
		0.1		9n (88 4
		te. 1	Performing Organizatio	n Report No.
7. Author's) Norman A. Eisenber	rg; Cornelius J. Lynd	h;		
Roger J. Breeding				
9. Performing Organization Nome and Add		10	Work Unit No (TRAIS	57
Enviro Control, Inc.			DOT-CG-33.377	-A
1530 East Jefferson Stree	et	[II	Contract or Grent No.	
Rockv111e, MD 20852				
		·	Type of Report and P	eriad Covered
12. Spansoring Agency Name and Address				
U. S. Coast Guard				
Office of Research and De	evelopment		Final Report	
Washington, DC 20590		14	Sponsoring Agency C	ele
summarized here, describ- preliminary test results This first stage of mode of an operational comput the philosophy, concepts the modeling, as now ope increase in precision, o hensive tool for assessi	es the research backgr associated with the is l development consiste er simulation, thereby , and approaches perta rational, are subject r both. Ultimately, is ng damage resulting f	cound, compu first stage ad of the de demonstrat ining to th to enhancem the model is com marine s	tational tech of developmen sign and impl ing the feasi e VM. Certai ent by augmen intended to pills.	niques, and t of the VM. ementation bility of n aspects of tation, be a compre
17. Key Words	18. 0.,	PRICES S	UBJECT TO C.	ANGE
Vulnerability Model Damage assessment Marine Spills	Doc thr tio	ment is ava ough the Nat n Service, S	ilable to the ional Technic pringfield, V	e public sal Informa- /A 22161
19. Security Classif. (of this concit)	20, Security Classif, Inf th	1 0000)	21. No. of Poges	22. Price
Unclassified	Unclassified	- F - F -		
				i
	di			

· · ·

· 7.

:

:

:

•

EXECUTIVE SUMMARY

.

The Vulnerability Model (VM) is a computerized simulation system for assessing damage that results from marine spills of hazardous materials: the final report, summarized here, describes the research background. computational techniques, and preliminary test results associated with the first stage of development of the VM. This first stage of model development consisted of the design and implementation of an operational computer simulation, thereby demonstrating the feasibility of the philosophy, concepts, and approaches pertaining to the VM. Certain aspects of the modeling, as now operational, are subject to enhancement by augmentation, increase in precision, or both. Ultimately, the model is intended to be a comprehensive tool for assessing damage resulting from marine spills. The research and development effort reported here was performed under Department of Transportation United States Coast Guard Contract number DOT-CG-33377-A. The entire documentation for the effort has five levels of reporting detail: (1) this executive summary presents a nontechnical overview of the VM purpose, structure, and operation, briefly describes methods of damage assessment, indicates the research background supporting the VM, and summarizes preliminary test results; (2) the main body of the final report provides the technical aspects of the VM structure, development, and use; (3) to allow a smooth flow to the presentation in the main report, more detailed information, such as involved mathematical derivations, complex flow charts, and case-history details, is relegated to the several appendixes; (4) a user's guide is provided as a separate document giving the details of the computer programming and the operation of the VM; (5) finally the Coast Guard has been issued the computer tapes and card decks required to set up and run the VM.

In recent years, industrial expansion, new technologies, and a centralization of chemical production have led to increased bulk transport of hazardous chemicals on U. S. waters. Steadily increasing transport of fuels, both conventional and nuclear, is required to meet the demands of the current energy crisis. An increase in the quantity and variety of hazardous chemicals transported in bulk is required to meet the demands of a U. S. chemical industry that is expanding at a rate many times that of population growth [S1].

To address the problems posed by the increase in transport of hazardous materials, the U. S. Congress has passed the Ports and Waterways Safety Act of 1972 that charges the U. S. Coast Guard with providing for the safety of "ports, harbors, waterfront areas, and navigable waterways of the United States." To discharge these duties related to safety, the U. S. Coast

[[]S1] Luckritz, R.T. Hazardous materials spill prevention in the bulk marine carriage of dangerous cargoes, pp. 18-24. In Control of Hazardous Material Spills, Proceedings of the 1974 National Conference on Control of Hazardous Material Spills. Am. Inst. Chem. Eng., New York, 1974.

Guard engages in such facets of spill control as prevention, containment operations, cleanup, and restoration. Although the response activities, such as containment, receive such publicity, the most productive spill control measure appears to be prevention, since spills of hazardous materials are difficult to detect, contain, control, or mitigate, and the damage caused by such spills is difficult to reverse. In order to assure the safety of marine transport by spill prevention, the Coast Guard is authorized to regulate the movement of bulk cargoes and the design and operation of vessels carrying them. The prevention of all accidental spills, however, appears unrealistic, because the only known way to preclude accidents entirely is to abstain from the activity giving rise to them.

.

Since cessation of marine transportation is impractical, some risk is unavoidable due to the threats posed by mechanical failure, by unusual environmental conditions (severe storm, tidal wave), and by human fallibility. Although the risk cannot be completely eliminated, appropriate regulations can reduce risk. However, substantial increases in safety rarely come cheaply. The reduction in risk attributable to any regulation must be weighed against the increased operation costs or capital expenditures such regulation requires. The need to perform cost-benefit analyses of Coast Guard actions has led to a research and development program to establish a Risk Management System. The VM is an important component of this system.

Background of Risk Management System [S2]

Development of the Risk Management System began in the spring of 1971. It involves analytic development in three separate fields:

- Spill-Risk Analysis (Spill Analysis) assessment of risks that a vessel or facility spill will occur and assessment of the effectiveness of spill prevention regulations
- Vulnerability Analysis (Public Damage Assessment) assessment of the threats to people, property, and the environment due to spills from a vessel or marine facility
- System Cost Analysis (Economic Impact) assessment of the cost impacts on the government and consumers of implementing alternative regulatory actions for spill prevention

Figure S.1 depicts a systems approach to risk management which is the goal of the Risk Management System. When fully developed, this methodology will be applicable, both to nationwide regulation and to special circumstances within local port or waterway areas, for assessing the costs and henefits expected from regulatory changes within those areas.

i i

[[]S2] Dunn, W.A., and P.M. Tullier. Spill Risk Analysis Program. Phase II. Methodology Development and Demonstration. Operations Research, Inc., Silver Spring, Md., August 1974. USCG Report No. CG-D-15-75 (NTIS AD 785026).



. . . .

•

கற்று கொழில் கற்றுகளுத்திற்றும்.அன்றுகள் நடித்து பான்னத்து காற்றாலம் கற்றுக்காட்டியார். கற்று

• .-

Interded - Dec

\$

FIGURE S.1. RISK MANAGEMENT - THE SYSTEMS APPROACH

17 M

A STATE OF TAXABLE PARTY.

100

THE REAL PROPERTY OF

1000

1000

Goal of This Effort

a service and a subscription of the service of the

(A) in the second s

The use of deterministic models for assessing damage to vulnerable resources is not a new idea. Such modeling efforts have been used extensively for many years in the areas of civil defense (especially nuclear weapon attack assessments) and war gaming. What is new, however, is the use of these techniques to assess damages resulting from spills of cargoes carried in marine transport. Such modeling efforts can provide the Coast Guard with valuable information on which it can base its regulatory decisions. Thus, this research represents an effort to develop and to demonstrate the utility of techniques specifically designed to support regulatory decision-making.

A secondary use of considerable import is to assist in the planning and selection of response actions. A further benefit in obtaining a workable model at this stage of development is in identifying those areas of modeling must critical in obtaining accurate hazard assessments.

Limitations on the Scope of This Effort

A main consideration in developing a damage assessment tool has been to establish a workable model within a framework flexible enough to allow for future model enhancements. The approach taken was to simulate the spill through a series of separate submodels, many of which had previously been developed for the Coast Guard. In modeling complex physical processes, it has been necessary to make simplifying assumptions. A conservative design philosophy guided the development of models used in the VM; consequently, whenever alternate approximations were available, the approximations were chosen so that the VM does not underestimate damage. The approximations made in model development are recognized, but the modularized structure of the VM facilitates the insertion of additional and improved models as they become available.

In order to obtain a complete working damage assessment model within the time and resources allocated and to make maximum possible use of previous model development work, the scope of this effort was limited. Some of the more important limitations are as follows.

- The portion of the population indoors is considered to be sheltered; no damage is assessed to the indoor populace. Further, it was assumed that fifty percent of the subject population was indoors.*
- Census data were used to determine the location of the vulnerable population; since census data primarily deal with the location of the residence of people, no modeling was effected to deal with the movement of people from home to work, to school, to recreational areas, or to other nonresidential locations.
- Most of the physicochemical models consider that only a single process occurs at one time, so that separate physical events occurring simultaneously are modeled as a sequence of separate events; for example, the VM models spilling, spreading, and burning as separate events, each terminating before the next can begin, even though these events can and do occur simultaneously.

Actually, 80-85 % of the population is normally indoors, but damage to the indoor population is not modeled in this first stage of the VM.

. 1

• Explosion of an unconfined vapor cloud is partially determined by a user input option and patially determined by the simulated characteristics of the vapor cloud; since the decision to simulate an explosion is not entirely based on physical principles, it may properly be argued that the VM will simulate an explosion when, in fact, none could occur.

and the lighter .

- Damage to the environment is not currently assessed; however the concentration of hazardous material in air and water is computed, so that the user may estimate environmental consequences at his option.
- Test runs were performed only for five cargoes.
- Only inhalation toxicity was treated; injury by ingestion of toxic materials was not assessed.
- Secondary damage mechanisms, such as ignition of expecially hazardous establishments (e.g., refineries), were not treated; only the direct consequences of the spill were simulated.
- Response actions (e.g., spill containment, population evacuation, and fire fighting) were not considered.
- Spills of solids and reacting chemicals were not considered.
- Underwater releases were not considered.
- Explosion damage to structures is assessed on the assumption that the structures affected are framed with wood members.

This listing of constraints on the development of the VM to its current state is not intended to discredit the VM. Rather this information is provided to help define just what type of results might reasonably be expected from the VM at this time. Although the VM has several important limitations, it is believed that, as it now stands, the VM is a useful tool for the risk analysis of marine spills; with further development and improvement, the VM should be able to provide even more utility and insight to the considerations of this important problem.

Major Accomplishment

and the second second

ř

r ..

1

ì

L. A

" and Asunat

₹ ._____

and the second secon

The major accomplishment of this effort has been the demonstration that the concept of the VM is suitable for implementation as a functioning tool for use in risk analysis. It has been demonstrated that all of the building blocks required for a vulnerability analysis, viz., the data bases describing the vulnerable resources and physical setting of the spill, the predictive models describing spill development, and the predictive models describing damage to the vulnerable resources, are either currently available or can be obtained. Exercise of the VM for five hazardous cargoes of particular interest, for a variety of spill sizes, and for various environmental conditions has yielded damage estimates that are judgementally credible within the input assumptions and apparently consistent with observations of actual accidental spills. (However, the amount of available quantitative information concerned with damage resulting from accidental spills of hazardous materials is small.)

Application of the VM

The VM is a damage assessment tool. Given a spill scenario by the user, that is, given the characteristics of the spill and the physical setting in which the spill occurs, the VM simulates the physicochemical transformations of the spilled material and estimates the damages inflicted on vulnerable resources by these processes. The VM addresses such questions as: If X tons of substance Y were spilled at location Z with the wind blowing due north, etc., how many people might be killed or injured? How many structures will be damaged? How much water and air pollution will occur and where? The VM is a deterministic model; that is, the probability of occurrence of the various events comprising the spill scenario is not considered. Instead the scenario specified by the user leads invariably to a particular spill development and damage estimate. Other ongoing elements of the USCG Risk Analysis Research Program are concerned with the definition of changes in the probabilities of events comprising the spill scenario as a function of changes in regulatory and operational controls. The probability changes, combined with VM damage estimates, will be used to determine the risk reduction benefit of certain types of Coast Guard safety actions.

Limitations on Use of the Results Derived from the VM

There is a danger that the preliminary results obtained from the VM might be misinterpreted, so care must be exercised in considerations of these results. One point that must always be borne in mind is that the VM is entirely deterministic.

The damage assessed by the model is predicated on a set of circumstances (input conditions) chosen by the user. The probability that such a set of conditions exists is not considered by the VM. The likelihood of a spill is small; this small probability combined with the probabilities of all of the other conditions yields a small overall probability that any given scenario actually occurs. Thus, although the consequences of some simulations are quite dire, the <u>risk</u> (the mathematical expectation) is small; large losses from rare events do not necessarily indicate a high risk activity.

Because the VM is completely deterministic and because no assessment is made of the likelihood of a user-chosen scenario, considerable judgment and experience are required to avoid unreasonable (even impossible) input conditions. For example, it is possible to simulate very large spills, even though no vessel in existence is large enough to carry that much cargo.

Another illustration of possible misuse of the VM is that the user may specify a large spill at a given site, even though ships capable of carrying that large a cargo as a single load cannot navigate the waters required to reach that spill site. It must also be acknowledged, when interpreting results, that the VM is in its first stage of development. Spills have been simulated for only five liquids. Solids, reacting substances, secondary damage mechanisms, damage to indoor (sheltered) populations, etc., have not been modeled. Although such omissions do not automatically mean the results are incorrect, evaluation of results in a general context is not possible without further model development. Finally, interpretation of the results should be tempered with the knowledge that the VM alone cannot determine the risk of marine transport in comparison to other forms of bulk transport or even other human activity involving risk.

Overview of the VM

1. <u>Scope</u>

The VM is a computer simulation designed to provide quantitative measures of the consequences of marine spills of hazardous materials. The simulation starts with a description of the nature of the spill itself, continues through the dispersion of the hazardous material, and ultimately includes assessment of the immediate effects of the spill on surrounding vulnerable resources, namely, people, property, and the environment.

The model is designed so that it may ultimately treat any type of material carried in bulk quantities in marine transportation. These materials may exist in gas, liquid, or solid phase as cargoes and may change phase upon release into the air or water environment. The materials may react, dissolve, or otherwise be admixed with surrounding air and water. Where appropriate, the model treats the mass transfer from a material spilled in or on the water to the air. The logical sequencing in the VM has been designed so that the VM can treat virtually all of the large class of materials carried in bulk in marine transport; however, the computational submodels are not available now to describe the behavior of all of these materials when spilled. With the submodels that are operational, the VM is able to treat spills of many liquids and gases carried in bulk quantities. Many cargoes of particular hazard are carried as bulk liquids which can currently be treated by the VM at this first stage of development. At present, the VM has been exercised for only five cargoes.

2. Simulation Scenario

The simulation requires three types of descriptive data that define: (1) the spill, (2) the physical setting in which the spill occurs, and (3) the vulnerable resources that are subject to the effects of the spill. The spill is described in terms of its location and spill rate, the physical and chemical properties of the spilled material, and the quantity of the spill. The physical setting is described in terms of the geometric configuration of the shoreline(s), hydrologic/oceanographic properties, and meteorological data. Vulnerable resources are described in terms of demographic distribution, property distribution, and land/water use. The geographic area of concern may represent any user-defined location, a rectangular area measuring ten miles in length and five miles in width being typical of anticipated applications. The physical setting and the distribution of vulnerable resources are described in terms of mutually exclusive geographic cells that cover the entire area of concern.

3. Submodels

The VM consists of submodels interconnected by an executive routine, with built-in logic dictating the sequence of submodel processing as a function of the spill development. Among these submodels are simulations of surface spreading, water mixing, air dispersion, conflagration and explosion, and submodels for assessing the effects from the dissemination of the hazardous material on vulnerable resources. Some of the submodels had been designed previously under U. S. Coast Guard sponsorship as part of the CHRIS (Chemical Hazard Response Information System) [S3] model development. Others were designed specifically for the VM. A generalized flow diagram of the model is presented in Figure S.2 and a more detailed description follows.

Service Restantion in a family and

4. Operational Phases

ł

The VM operates in two phases. Phase I simulates the spill itself, the physical and chemical transformations of the spilled substance and its dissemination in space. This phase covers the time period from the initiation of the spill until a user-specified time has elapsed. The time interval between simulation calculations is specified by the user but may be overridden by certain submodels (such as the explosion submodel). A time-history file of the spill sequence simulated during the first phase is retained in some form of computer storage such as magnetic tape or disc.

In Phase II, the computer first superimposes this time-history file upon the vulnerable resources map and then assesses the effects of toxicity, explosion, and/or fire on the vulnerable resources as a function of time. Estimates of deaths and nonlethal injuries to people and of damage to property are presented in tables.

5. Current Status of Development

At present the VM is in a first stage of development. It has been demonstrated that an actual working model is capable of carrying on a simulation from the specification of cargo and spill conditions through to the assessment of damages to vulnerable resources. The quantitative results of the simulation appear to correlate with the picture of events given both by expert judgment and by historical records of accidental spills. This correlation is obtained even though it is recognized that some of the modeling, by necessity, is not at the highest level of sophistication. Furthermore, the cost of a given simulation has been kept within reasonable bounds. Likewise the cost of data preparation required by the VM is not excessive. Interpretation of output requires a knowledge of the basis (not the technical details) of the modeling methods used and the ability to judge the suitability of arbitrary user inputs. Once the proper use

[[]S2] Raj, P.P.K., and A.S. Kalelkar. Assessment Models in Support of the Hazard Assessment Handbook. Arthur D. Little, Inc., Cambridge, Mass., January 1974. USCG Report No. CG-D-65-74 (NTIS AD 776617).



- --

.

.

.

FIGURE S.2. GENERALIZED FLOW MAGRAM OF THE VULNERABILITY MODEL

í x

of the VM is learned, there is virtually no cost involved in output interpretation, since the computer results are presented in an easy-to-understand, user-oriented way.

Phase I Submodels

This section describes those submodels that are presently included in Phase I of the VM. Because the air dispersion submodel which was originally part of HACS* [S4] has been extensively revised, and because the fire and explosion submodels were developed almost entirely in this program, these submodels are discussed in greater detail than other Phase I submodels. Some of these other Phase I submodels, which were developed previously under USCG sponsorship, were modified somewhat for inclusion in the VM.

The submodels used in the VM treat the physicochemical processes affecting the spilled hazardous material; at the present time, the processes simulated by the VM may be classified as follows:

- 1. Cargo venting
- 2. Spill development
- 3. Air dispersion
- 4. Combustion (fire and explosion)

For certain of these processes different submodels are used, depending upon the nature of the spilled substance. Other processes, such as fire and explosion, consist of a sequence of dissimilar events, so the computer simulation consists of a sequence of submodels. The operational sequence of these various submodels is shown in Figure S.3.

The Phase I submodels for various physical and chemical processes treated by the VM are connected by an executive and storage routine. The sequence of processing depends upon the cargo and the development of the situation. The first submodel used is always the tank venting submodel, which calculates the rate of escape of the cargo in both gas and liquid phases as a function of initial tank conditions and the size and location of the /ent or rupture. If only gas is vented, the simulation will pass directly to the air dispersion submodel from the venting submodel. If the cargo is vented as a liquid, it may remain in that phase or it may change to gas phase. If the cargo does change to a gas, some of it may flash directly to a gas when the tank pressure is released, but in most cases the bulk of the gas will be released more slowly by evaporation or boiling.

For many liquid spills, a pool of the cargo will form on the surface of the water, but a cargo with a liquid-phase density greater than that of

^{*} Hazard Assessment Computer System

[[]S4] Hazard Assessment Computer System (HACS) User Manual. Arthur D. Little, Inc., Cambridge, Mass., December 1974.

FIGURE S.3. OVERALL FLOW DIAGRAM FOR THE VM.



water will sink at once. At present there are five submodels for treating a spilled liquid:

- 1. Spreading and evaporation of an immiscible, floating, cryogenic liquid
- 2. Spreading and evaporation of an immiscible, floating liquid with high vapor pressure

ą.

- 3. Sinking and boiling of an immiscible liquid
- 4. Mixing, advection, and dilution of a miscible liquid in a tidal river, nontidal river, or still water
- 5. Mixing, dilution, and evaporation of a miscible liquid with high vapor pressure

Of course there are possibilities other than these five, but these five do cover a large number of hazardous cargoes frequently carried in bulk quantities. Liquefied natural gas, for example, is treated by the first submodel, gasoline by the second, and liquid chlorine by the third. The fourth submodel does not give an escape of gas to the atmosphere, so the simulation may stop with the calculation of the cargo concentration in the water or the simulation may proceed to calculate evaporation. The other submodels give a rate and duration for the gas evolution, and the simulation proceeds to the air dispersion model.

Air Dispersion

This submodel calculates the concentration of the cargo in gas phase in the air from the time the gas is released into the atmosphere until a fire or explosion occurs or until the maximum time stipulated for the simulation is reached.

The submodel is based on the Gaussian distribution, which is a theoretical solution to the partial differential equation governing diffusion. The dispersion coefficients used in the Gaussian distribution are obtained from the analysis of many observations of plumes from tall stacks and of puffs of smoke or some other tracer. Plumes result from continuous releases, whereas puffs result from instantaneous releases. Puffs are three-dimensional Gaussian distributions in which the dispersion coefficients depend upon the distance traveled by the puff center-of-mass. The plume is a twodimensional Gaussian distribution in which the dispersion coefficients depend upon the distance downwind from the source to the observation point. The values of dispersion coefficients obtained from experimental observations are parameterized on the basis of the atmospheric stability or the turbulence class.

At present, the submodel will select the plume (continuous source Gaussian distribution) if the release time of the spilled material into the air is relatively long; for short release times, the puff (instantaneous source Gaussian distribution) is selected. Since there is no model extant that adequately treats spills with intermediate release times, either a puff or plume model is used until more adequate models are available. Because the Gaussian models were based on data gathered from dilute plumes and puffs, their use for very concentrated cases, as in the VM, is an extrapolation. When the puff model is used close to a very large spill of material which vaporizes quickly, the standard Gaussian model gives concentrations which are higher than the density of pure cargo vapor at ambient atmospheric temperature and pressure. The puff model has been modified to preclude this by allowing a region of pure cargo vapor surrounded by a region where the concentration decreases in the Gaussian fashion. The pure cargo vapor concentration is at the proper density for ambient conditions. This modified distribution is used only when the regular puff model would give unrealistically high concentrations. A STATE COLUMN

2

all hitsen traffie

مدينية (يستنقا ألمان العلم المرارية). ومعالمة المرارية عام المرارية - المالية المرارية المراجعة عامية المالية المستركمة المالية المالي

ŧ

Fire and Explosion

٩

This group of submodels determines whether a flammable cargo will be ignited and then determines the physical characteristics of the resulting combustion (fire, explosion, or both). Four types of fire and explosion phenomena are modeled in this section of the VM; they are:

- 1. Ignition
- 2. Explosion
- 3. Flash fire
- 4. Pool burning

The modeling of the phenomena of fire and explosion proceeds in three temporal phases. First the decision of whether, when, and where ignition occurs is made. Subsequent to ignition either an explosion or a flash fire is modeled. Following either of these events, the burning of flammable liquid on the water surface, if any liquid remains, is modeled; currently, burning from a vessel venting flammable fuel is not modeled.

The <u>ignition</u> submodel determines whether ignition occurs, which ignition source originates the ignition, and at what time during the simulation the ignition takes place. All ignition sources are assumed to be located at the grid cell centers. The user predetermines whether a given ignition source will cause fire or explosion. The user also specifies the strength of the ignition source. Since combustion will occur only over a certain range of fuel-air ratios, the decision that combustion occurs will be made only if the vapor concentration in a given cell is within the flammability limits for the substance under consideration and the given cell contains an ignition source of strength sufficient to ignite the spilled substance.

The <u>explosion</u> submodel calculates the peak overpressure and the dynamic impulse generated by the explosion of a flammable fuel-air (cargo-air) mixture. In addition to these variables that are required for damage assessment, the explosive yield and TNT equivalent are also determined. The well-known scaling laws for condensed phase explosions are assumed to hold. Only that portion of the fuel-air mixture with a concentration between the explosive limits is permitted to contribute to the explosive yield. For that part of the fuel-air mixture richer than stoichiometric, but leaner than the upperexplosive-limit concentration, only that fraction of fuel for which there is sufficient oxygen for complete burning contributes to the explosive yield. The <u>flash fire</u> submodel calculates the effective radiation intensity level and the effective radiation duration resulting from the flash fire. The flash fire is considered to be the rapid combustion without detonation of the premixed fuel-air layer within the flammable concentration limits. The heat generated essentially instantaneously by combustion is assumed to be lost from the combustion layer entirely by radiation; thus the radiation loss as a function of time may be calculated. The time varying radiation level is represented by a fixed radiation level (the effective radiation level) and an effective duration. This submodel assumes that only that portion of the fuel-air mixture within the flammable limits burns and then only to the extent permitted by the local oxygen concentration. The radiation from the flash fire is assumed to affect only those portions of space inside the burning layer.

The <u>pool burning</u> submodel calculates the duration and magnitude of thermal radiation emitted by a burning pool of flammable cargo; the radiation level is calculated for any desired point in space. This submodel is comprised of several other submodels:

- 1. Flame size
- 2. Thermal radiation from flames
- 3. Radiation view factor between an inclined cylindrical flame and an arbitrarily criented surface in space
- 4. Burning time

The <u>flame size</u> submodel calculates the height, diameter, and angle of inclination of a flame from a burning pool; the wind blowing across the pool surface causes the flame to be inclined with respect to the normal to the pool surface. The formulas used to calculate flame height, diameter, and inclination angle are empirical expressions obtained by curve fitting experimental laboratory data and extrapolating to the larger scale occurrences possibly resulting from marine spills of hazardous materials.

The <u>thermal radiation from flames</u> submodel calculates the radiant heat flux incident on a receptor at some distance from a burning pool. The flame from the burning pool is modeled as a cylindrical radiator of uniform temperature; this constitutes the major assumption of this submodel. The cylindrical radiator is allowed to be inclined with respect to the vertical. The atmospheric transmissivity and flame emissivity are assumed to be one, i.e., the atmosphere is not allowed to absorb radiant energy and the flame is treated as an ideal black body radiator.

The <u>radiation view factor between an inclined flame and an arbitrarily</u> oriented surface in space submodel calculates, on a normalized basis, the view factor between a cylindrical, inclined flame and a receptor; the receptor is assumed to be oriented with respect to the flame so that it receives the maximum possible radiation flux. The view factor is a purely geometrical property of the spatial arrangement of the flame and receptor. Thus no physical assumptions, except those basic to radiant heat transfer, are required by this model.

The <u>burning time</u> submodel calculates the length of time the pool will continue to burn after it is ignited. It is assumed that the pool is extinguished when all of the fuel remaining in the pool at the time of ignition is burned up. It is further assumed that fuel leaves the pool only by burning but not by evaporation or by water mixing. It is also assumed that spreading stops when ignition occurs; although spreading will not cease entirely after ignition, the spreading rate will certainly be reduced because mass is being lost by combustion.

÷.

After the spill development is simulated and the results are stored by grid cell and time interval, these data along with the information describing the vulnerable resources are used to assess the damage which may result.

Phase II - Damage Assessment

As it currently stands, the VM models damage to vulnerable resources from four physical events:

- 1. Air dispersion of a toxic gas
- 2. Explosion
- 3. Flash fire
- 4. Pool burning

Damage to vulnerable resources is conversionantly discussed in terms of (1) toxic injury, (2) explosion damage, and (3) fire damage. The assessment algorithms for toxic damage are highly dependent on the type of substance spilled; whereas the assessment algorithms for fire and explosion are independent of the type of substance spilled, though the values of the variables used in the algorithm do depend on the type of substance spilled.

The vulnerable resource "people" may be affected by inhalation of toxic vapor, by thermal radiation from a flash fire, and by peak overpressure or impulse from an explosion or by some combination of these. The vulnerable resource "structures" may be affected by thermal radiation from a flash fire or burning pool and by peak overpressure or impulse from an explosion. The vulnerable resource "environment", defined as air and water for the purposes of the VM, may be affected by the spilled substance in its vapor or liquid phase or by a reaction product. In most cases, the percent of the vulnerable resources affected within a given grid cell is calculated and then applied to the numbers of vulnerable resources present, giving the total numbers of vulnerable resources affected (for a given time period and given grid cell). Figure S.4 shows the specific types of injury and damage assessed for toxic gases, fire, and explosion. The code name of the algorithm used for computing the portion of the resource affected is listed under the column heading "Function." The factor or factors calculated from the simulations of the physical events which are used by each algorithm to assess damage are shown in the adjacent column.

For most of the assessment functions, the damage or injury is related to the causative factor by means of probit equations, resulting in the

Damage Causing	Vulnerable	Type of Injury or Damage		Cause of	Outdoors (Unsheltered)		
Event	Resource			Injury or Damage	Function	Factor(s)	
TOVICITY		Death		Toxic vapor:	11	Concentration and	
TUXICITY	People	Nonlethal injury		concentration or	T2	time	
L		Irrita	tion	cumulative dose	T3	Concentration	
		Death		Direct blast	El	Peak overpressure	
				Impact	E2	Impulse	
			Eardrum rupture	Direct blast	E3	Peak overpressure	
		Nonlothal	Bone fracture	Impact	E4	Impulse	
FYPLASTAN	People	Tojury	Puncture wounds	Flying fragments	E5	Impulse	
		Injury	Multiple injury	Two or more of the above	E6	Impulse or peak over- pressure and impulse	
	Channehuman	Structural damage		Direct blast	51	Peak overpressure	
	Structures	Glass breakage		Direct blast	S2	Peak overpressure	
	People	Death		Thermal radiation	B1*	Duration and magnitude of thermal radiation	
POOL BURNING		First degree burn			B2*		
	Structures	Ignition		Thermal radiation	B3	Duration and magnitude of thermal radiation	
FLASH FIRE	Recelle	Death		Thormal radiation	Fl	Effective duration and	
	reopte	First degree burn		Inermal raulation	F2	thermal radiation	
	Structures	Ignition		Thermal radiation	F3	Effective duration and effective magnitude thermal radiation	

FIGURE S.4. PHASE II DAMAGE ASSESSMENT

* These functions are essentially the same as F1 and F2; however as indicated in the adjacent column, slightly different arguments are used. These functions have not yet been implemented as part of the computer program, although to do so will be very easy.

XVÍ

calculation of the fraction of the population in each cell which is affected. Four assessment procedures, however, use simple threshold criteria; they are: (1) B3 ~ ignition of structures from pool burning, (2) F3 - ignition of structures from flash fire, (3) F2 - nonlethal injury from a flash fire, and (4) T3 - irritation from inhalation of toxic gases.

.

Provisions have been made in the VM to prevent double counting in three different situations. Double counting is used in this context to mean the inclusion of an element of some vulnerable resource (e.g., a person or a building) in more than one category of damage or injury. Three situations arise in which double counting will occur unless provisions are made to prevent it. The situations are as follows.

- 1. A single damage mechanism from one event simultaneously causes injuries of differing severity (e.g., inhalation of toxic gas may cause death, nonlethal injury, or irritation).
- 2. Two or more damage mechanisms from one event simultaneously cause injuries of the same severity (e.g., an explosion can kill people either by direct blast effects or by impact).
- 3. Different events at different times cause damage to the same resource, but the first event so severely damages some portion of the resource that further damage is irrelevant (e.g., persons killed by toxic gas cannot be further injured by a subsequent explosion).

1. Toxic Injury

Toxic injury is assessed only for the vulnerable resource "people." The toxic damage caused by irritant gases can in general be classed into three categories:

a. Death

- b. Nonlethal injury
- c. Irritation

The category of injury sustained by exposed resources depends, in general, upon both the duration of exposure and the concentration level experienced. This dependence is nonlinear; dose, the product of concentration level and duration, is not the appropriate variable to assess response to irritant gases. For example, for concentrations over the lethality threshold, doubling the concentration level does not halve the time required to produce the same death rate; instead, as the concentration level increases, the time to produce a given injury level decreases at a disproportionately rapid rate. Toxic damages to the vulnerable resource "environment" is not assessed in the VM at this time. The VM predicts the concentration of a toxic substance in the air and water. Comparison of these predicted concentrations to air and water quality standards would appear to be an attractive method for assessing damage to the environment, but such an approach has many pitfalls, among which are: (1) difficulty of implementation; (2) multiple standards for the same substance; (3) chronic, instead of acute, exposure standards; (4) statement of quality standards in a form other than levels not to be exceeded. Because of these problems, the condition of exceeding air or water quality standards has not been slopted as a viable procedure for assessment of environment damage in the VM. It was determined that an assessment of damage to people and property would be considered as the original tasks of the VM. It is anticipated that environmental damage will be considered at a later date.

2. Explosion Damage

Explosion damage is assessed to the vulnerable resources "people" and "structures." Personnel experience explosion damage in two categories, (a) death and (b) nonlethal injury. It is customary to categorize explosion damage to personnel in three categories, depending in the causative mechanism of damage; thus:

- a. primary damage direct blast effects (interaction between the blast wave and personnel only, with no other intervening or associated factors)
- b. secondary damage damage from missiles and fragments
- c. tertiary damage damage from translation and subsequent collision with an obstacle; i.e., impact

יייבוי) לאווע אוויילאיז או אינייי, או**ל האול**ון, ואלאאר איניאין אינייינע אוויילא אוויילא אוויינע אוויילא אוויינע

In the VM, death is assessed for primary damage manifested as lung hemmorhage or for tertiary damage manifested as skull and body bone fractures. Nonlethal injuries are assessed for all three damage categories, including the secondary damage of puncture wounds from missile penetration. In addition, injury resulting from two or more damage mechanisms is assessed in a separate category, multiple injury. Structures experience explosion damage in two categories: (a) serious structural damage and (b) window glass breakage. The physical variable that determines the extent of explosion damage in the VM is the peak overpressure.

3. Fire Damage

Damages from fire are assessed to the vulnerable resources, "people" and "structures." The damage assessed to structures is ignition. The damages assessed to personnel are: (a) death and (b) nonlethal burns. For all types of damages, two parameters have been found to be significant: (a) level of thermal radiation and (b) duration of the thermal radiation; therefore a variable combining these two parameters is used for assessment purposes. The assessment of deaths from flash fire is based on data obtained primarily from studies of the effects of nuclear weapons. The radiation from a flash fire is time varying. To be compatible with the damage assessment procedure, this time varying radiation is parameterized by calculating an effective pulse intensity and an effective pulse duration. Because of the uncertainty in determining the degree of the burns and the effects of clothing and shielding, the VM does not make a quantitative assessment of nonlethal burn injuries. Instead, it informs the user that the threshold for causing first degree burns to exposed skin has been exceeded. The assessment of fire damage to structures is based on studies of the ignition of wood. Factors influencing wood ignition are: (a) radiation intensity level, (b) duration of radiation exposure, (c) wood type, and (d) the presence or absence of a pilot flame near the irradiated wood. Duration and level of radiation intensity are factors computed by Phase I submodels. Wood type is not treated explicitly; average values are used. For flash fire the presence of a pilot flame is assumed; for pool burning pilot flames are assumed to be absent.

Input/Output

Four types of input data are required to run the VM:

- 1. The physical and chemical properties of the cargo, which are stored in a library file; also a 'ibrary file of default values for missing property values or missing user-supplied data
- 2. Demographic and other local information for the region of interest, also stored in a library file
- 3. Data defining the spill and the air and water conditions at the time and location of the spill, and locations and strengths of ignition sources
- 4. Operating parameters and override values

The physical and chemical properties of the cargo are stored in the chemical properties file; this file, designed for use with HACS [S4], has been adopted for use with the VM. In addition to obvious information about each material such as molecular weight and density at standard temperature and pressure, the properties file contains some information about flammability, toxicity, flame temperature, and other properties of the material. For properties such as vapor pressure, viscosity, and thermal conductivity which vary with the temperature, the constants in the equations that give these quantities as functions of temperature are stored in the properties file.

The information about the distribution of population and buildings was taken from census data, and, at present, each census tract constitutes a cell. Those census tracts that included any navigable water were divided into the land portion and the water portion. Each cell is identified by a number and represented by a grid point. For regularly shaped tracts with a uniform housing density, the grid point was chosen to be approximately the center of the tract. For irregularly shaped tracts, or those which included a significant amount of uninhabited area, a point representative of the settled area was chosen to represent the census tract.

The third category of input information required is that which defines the spill and the air and water environment in which it occurs. These data include the location of the spill, the substance in the tank, the tank temperature, pressure and dimensions, and the size and location of the rupture or vent. To define the environmental conditions, such items as air temperature, pressure, humidity, wind velocity, and stability class are needed, in addition to water temperature, salinity, pH, and current or tidal conditions. The fourth category of data includes information which the user provides for each simulation and which does not fit in the previous category. For example, the user must specify what portion of the available grid is to be used for this simulation. The user may wish to override the default value for an internal parameter such as the time step used in the air dispersion submodel. Or, if chemical properties of the spilled cargo are known to the user but have not yet been placed in the properties file, the user may specify the values of these properties in order to override the default values which would otherwise be used.

للمامير وموسات للمديق والالكالماناتي ساولهم بمسألم اللاغات يدهدها

Contractory of the

ļ

The transfer from the simulation of physical events to the damage assessment portion of the VM is by means of files which store the values of physical quantities. The largest files are those which store the concentration of the cargo in the air and in the water as functions of time and grid point. For "one-time" events such as fires or explosions, only one value of each physical quantity used in damage assessment need be stored for each grid point. For explosions, the values of the peak overpressure and the dynamic impulse are given. For a flash fire, the effective radiation intensity and duration are given, and for pool burning the radiation intensity and burning time are given. The effects of toxic gases are assessed directly from the variation of concentration with time. After the spill development is simulated and the results are stored by grid cell and time interval, these data along with the information describing the vulnerable resources are used to assess the damage which may result.

After the Phase I simulation results have been processed in Phase II to give damage assessments, key features of the Phase I simulation, as well as the results of the Phase II damage assessment, are printed out in a user-oriented, easily interpreted format. Both in the cell-by-cell printout and in the summary printout, damage is separated into classes by receptor (i.e., people or structures) and by causes (i.e., toxicity, explosion or fire).

Because the final damage assessment is presented in an easy to understand, user-oriented manner, further processing of the data or lengthy analysis is obviated. Of course, it is assumed that the user is knowledgeable about the cargo type, the geographical area, and the modeling assumptions to the extent that the results will not be misinterpreted.

Test Results

Test runs of the VM were made for the five cargoes considered under a variety of spill and environmental conditions; these test runs were performed in order to:

- Show the feasibility of the VM concept
- Test the computerized logic deciding the sequence of submodel execution
- Demonstrate the plausibility of the damage assessment compiled
- Test the sensitivity of the computed results to various input parameters

Sizes of spills simulated ranged from 0.6 m^3 (160 gallons) to 50,000 m^3 (13 million gallons). Wind direction, wind speed, spill size, cargo type, ignition source type, and atmospheric stability class were the input parameters varied in the course of the testing.

The test results gave spill development behavior and damage estimates which are not inconsistent with the small number of observed accidental spills. In general, spills of the toxic irritant gases produced simulated injuries and deaths that were an order of magnitude larger than the damages to people resulting from the explosion of an equal amount of spilled flammable material. Injury and death from deflagration (flash fire) of a flammable spilled material were considerably less than the injury and death produced by the detonation (explosion) of the same amount of flammable spilled material.

Uses of the VM and Suggested Improvements

At present, the VM is in its first stage of development; even in this developmental stage the VM is a useful tool. Among the uses of the VM are:

- o Demonstration of the feasibility of implementing a simulation system for assessing damage resulting from spills of hazardous materials
- Aid in planning future R&D efforts by pinpointing areas where modeling needs improvement
- o Aid in planning regulatory actions
- o Aid in planning programs for response measures

Although it is useful in its present state of development, the VM requires further improvement before it will be able to realize its full potential for utility as a risk analysis tool. Improvements to the VM can be divided into two classes:

- 1. Improvements to specific submodels or the addition of submodels to account for additional phenomena
- 2. Improvements to the overall VM structure

Work should be undertaken to expand and improve the spill development (Phase I) modeling. The VM should be expanded to include new submodels developed for the Coast Guard; seven additional submodels have been developed that are suitable for inclusion in the VM. These are:

- 1. Release and migration of heavy insolubles on river beds
- 2. Heating, rupture of the container, and release of pressurized cargo in a fire
- 3. The release of three specific reactive chemicals
- 4. Release, spread, dispersion, and fire hazard due to a continuous release of cold, liquefied gases
- 5. Water dispersion of chemicals with finite solubility

xxi

6. Heating and rupture of tanks on sunken barges carrying cryogenic materials

7. Release of cold and soluble chemicals underwater

in the effects of the set of the

HANNO'S CONTRACTOR

The VM should be exercised for various additional cargoes, especially chemicals of the type to which these new submodels apply. Two areas of fire and explosion modeling currently in the VM should be improved. A more sophisticated treatment of the flash fire phenomena should be achieved. At present, the radiation characteristics at the site of combustion are calculated, but radiation levels at points distant from the site of combustion are not calculated. Consequently, no damage is assessed, except in those cells containing burning fuel at the cell center. An improved flash fire submodel should permit assessment of damage at points distant from the site of combustion. This improved calculation of radiation levels will continue to treat the variation of radiation with time but will improve the treatment of radiation variation in space. Another area of fire and explosion modeling that requires improvement is the treatment of ignition sources. The present model classifies ignition sources on the basis of strength by using the concept of flash point in a rather unconventional manner. Also the decision as to whether conflagration or detonation results if an ignition occurs is made a priori by the user, rather than being a computed decision in the simulation. At this time, it is doubtful that a model can be formulated to make the choice between conflagration and detonation except at a disproportionately high effort and cost.

On the other hand, other aspects of the modeling of ignition phenomena could be improved. A treatment is desired which would consider ignition potential as a true function of area, rather than the current treatment which considers ignition sources to be concentrated at the center of a grid cell. By describing ignition sources in a manner other than locating the sources at discrete sites (the cell centers), a more realistic assessment of time to ignition and distance between ignition point and spill may be obtained. The ability to grade ignition sources according to strength is a desirable property of the VM. However, a gradation system that is based on flashpoint, although useful, lacks rigor in the correspondence between the physical phenomena modeled and the mathematical description used in the VM. However, no solution may exist to the ignition gradation problem, which is both physically more realistic and manageable in terms of computation and data preparation; therefore, the problem of ignition source gradation may be researched, but a modification to this portion of the VM may not be advisable unless a tractable solution is found.

Certain aspects of current damage assessment procedures should be improved. The VM should be modified to assess deaths and injuries suffered by indoor sheltered populations due to all relevant damage mechanisms. Consideration should be given to the varying degrees of shelter afforded by different types of structures. For injury caused by inhalation of toxic gases, assessment of injury to the indoor population will involve considerations of seepage of the toxic substance into structures. The assessment techniques for damage caused by inhalation of toxic gases evolved during the first stage of development of the VM require that the time history of

xxii

concentration of the gas be known. For injury caused by explosion, assessment of injury to the indoor population will involve the degree of shelter afforded by the placement of the people in the building. For example, persons in front of glass walls or large windows facing the blast will have a much greater chance of injury from flying fragments than persons not so located. Buildings with fewer and smaller windows will not present this hazard but may cause high velocity jets to issue from those windows facing the blast, and these high velocity jets of air may cause serious translation damage. Injury to persons inside buildings as a result of flash fire or pool burning is much less likely to occur as a primary damage mechanism than injury to sheltered populations from explosion or toxic substances. Whether a person inside a building is harmed directly by burning of the spilled material depends mainly on whether that person is in front of a large window or is otherwise <u>in</u> the building but "unsheltered" by the building.

The data file of vulnerable resources should be expanded to account for those portions of the population that are found in the different classifications of structures for different times of the day and different days of the week. It may be desirable to categorize the population into three location classes: (1) at home, (2) at work, and (3) in transit. A fourth, and possibly highly significant class, may be "at a recreation facility," e.g., at a stadium, playground, beach, or fairground. Movement of the population, especially regulating recreational location, will also depend on the time of year. The "at work" location of the younger segment of the population will normally be at schools; in most parts of the country, schools are closed or only partially utilized in the summer.

A better treatment of the ignition of structures should be brought about. A more precise method to account for shielding from thermal radiation is to be devised. The current criterion that 25% of the structures are ignited in a cell subject to radiation sufficient to cause ignition is to be replaced by a criterion that has a variable percentage ignited and the percentage ignited is to be calculated on the basis of physical principles.

The damage assessment procedures in the VM should be expanded to account for damage mechanisms not currently modeled. Among the additional damage mechanisms that should be accounted for in the VM are the following:

- 1. The spread of fires initiated by the burning of the spilled cargo should be considered. This will involve both fire spread from building to building and the ignition of new sources of fuels, as might be found in a nearby refinery.
- 2. The inhalation of toxic combustion products should be examined.
- 3. The ingestion of water containing toxic concentrations of pollutant should be considered. A specific level of injury or percent of population injured is not possible since the quantity and rate of ingestion are highly variable and unpredictable. What is required is some indication of the toxic hazard presented by a given concentration of spilled substance.
- 4. The phenomenon of a roiling fireball should be examined, and modeled if it is to be a significant damage mechanism. The roiling fireball is distinguished from the flash fire by: (1) combustion in the

flash fire is rapid compared to combustion in the fireball; (2) the burning mixture in the fireball is rich and is supported by turbulent diffusion, whereas the burning mixture in the flash fire is premixed to within the flammable limits.

- 5. Injury by asphyxiation should be addressed.
- 6. Further consideration should be given to the significant ways in which diffuse explosions differ from conventional explosions, both in physical characteristics and in damage phenomena; the possibility that a deflagration may "shock up" to produce a significant blast overpressure should be considered.

Methods to make the VM more accurate and efficient by modifying the present grid cell structure should be examined. One possible approach is to compute the precise region of damage from a spill-based coordinate system and then to process only those demographic data pertinent to the impacted area. Such an approach may also permit the plotting of isodamage contours. To restructure the VM in this manner will require that the computation of physical phenomena at cell centers be abandoned; consequently, the Phase I subroutines may require revision to compute the locus of a constant value of a physical parameter (say overpressure) rather than to compute the value of a parameter at a given point. ۲۰۰ میکند. میکند. به میکند. از میکند. از میکند و میکند را بازی از میکند. از میکند. بازی میکند. بازی میکند. میکند میکند. میکند میکند میکند. بازی میکند از میکند میکند. از میکند و میکند را بازی میکند. از میکند میکند میکند میکند و میکند.

with a train

And a state of the second s

Conclusion

The listing of such a large number of recommendations was intended to point the way toward areas related to the VM for which additional effort is believed to have the greatest potential for benefit. This list of recommendations is not, nor is it intended to be, an indictment of the VM. To the contrary, the VM, in this its first stage of development albeit crude in certain aspects, is believed to be a useful, practical tool for use in the risk analysis of marine spills. Further development of the VM will make this already functional tool more useful, more precise, and more widely applicable.

REFERENCES

- [S1] Luckritz, R.T. Hazardous materials spill prevention in the bulk marine carriage of dangerous cargoes, pp. 18-24. In Control of Hazardous Material Spills, Proceedings of the 1974 National Conference on Control of Hazardous Material Spills. Am. Inst. Chem. Eng., N.Y., 1974.
- [S2] Dunn, W.A., and P.M. Tullier Spill Risk Analysis Program. Phase II. Methodology Development and Demonstration. Operations Research, Inc., Silver Spring, MD, August 1974. USCG Report No. CG-D-15-75 (NTIS AD 785026).
- [S3] Raj, P.P.K. and A.S. Kalelkar. Assessment Models in Support of the Hazard Assessment Handbook. Arthur D. Little, Inc., Cambridge, Mass., January 1974. USCG Report No. CG-D-65-74 (NTIS AD 776617).
- [S4] Hazard Assessment Computer System (HACS) User Manual. Arthur D. Little, Inc., Cambridge, Mass., December 1974.

ACKNOWLEDGMENTS

Ċ,

The authors wish to acknowledge the contributions made to the development of the Vulnerability Model by other members of the ECI staff. John Morton provided toxicological expertise and is primarily responsible for Appendix E. Dr. A. Hadermann, Dr. Timothy Kao, Mr. J. Horowitz, Mr. I. Takacs, and Mr. R. M. Rowley provided assistance in their areas of particular expertise, which are, respectively, physical chemistry, transport phenomena and flow chart development, water quality, air quality, and computer systems.

د...<u>این از مر</u>جع بیور <mark>باد باد این گاریا</mark> د درار <mark>الباد</mark> در مصحه بر برست.

والطبيف الكشمين اللها مشتهدهم الالطاران الشكاك كالرائي كالقادي مقارضا والمستعدية

1

and a start of the second of a second state of the second

Dr. John A. Brown, of John Brown Associates, Inc., performed the historical survey reported in Chapter 9 and Appendix F. Most of the computer programming was done by Mr. J. Weirman, Operations Research, Inc.

The authors gratefully acknowledge the assistance provided them by Mr. Neal FitzSimons of the Defense Civil Preparedness Agency.

The program was performed for the Office of Research and Developmenc, U. S. Coast Guard, under the supervision of Mr. John Dwyer. The guidance and insights provided by Mr. Dwyer were most helpful in the successful completion of this work; the comments and advice of other USCG staff members, particularly Dr. John Gardenier, Dr. John Cece, and Lt. Michael Taylor, are greatly appreciated. The direction provided by the members of the Risk Analysis Advisory Board is sincerely acknowledged. TABLE OF CONTENTS

فالفظاف اللاعد بدرنا وسمرغ حافا كطافي معتد ماؤهانا يتحرك والمحرف فأكارا أللم مراخاتك لأخد

1

INTRODUCT ION	Page 1			
CHAPTER 1 General Description of the Vulnerability Model	7			
CHAPTER 2 Description of Submodels Used In Phase I of the Vulnerability Model	15			
CHAPTER 3 Air Dispersion	33			
CHAPTER 4 Fire and Explosion Submodels	49			
CHAPTER 5 Input/Output Data For Phase I	67			
CHAPTER 6 Phase II Assessment Procedures	75			
CHAPTER 7 Examples of Computer Runs For An Urban Area	97			
CHAPTER 8 Sensitivity Analysis	129			
CHAPTER 9 The Historical Survey Subtask	137			
CHAPTER 10 Conclusions and Recommendations	151			
REFERENCES	161			
APPENDIXES				
A - Phase I Flow Charts With Narrative	165			
Bl - Modification of the Gaussian Puff Model	185			
B2 - Gaussian Plume Air Dispersion Mode	1 195			
B3 - Scale Analysis	201			

xxvii

APPENDIXES, continued

۰. . ۱

ł

1

С	1	-	Ignition Source Considerations Page	207
С	2	-	Calculation of the Amount of Fuel Consumed	211
С	3	-	Equivalency of Change in Helmholtz Free Energy and Change in Enthalpy For Estimating the Yield of Vapor- Air Explosions	221
С	4	-	Correction to Heats of Combustion	225
С	5	-	Damage Potential of Confined Explosions	226
D		-	Damage to Vulnerable Resources From Fire and Explosion	229
E		-	Inhalation Toxicology of Chlorine and Ammonia	257
F	,	-	Case Studies	269
F	1	-	Cyclohexane Cloud Explosion, Flixborough, England - 1 June 1974	270
F	2	-	Ethylene Cloud Explosion, Longview, Texas - 25 February 1971	277
F	3	-	Propane Cloud Explosion Decatur, Illinois - 19 July 1974	279
F	4	-	Vinyl Chloride Monomer Cloud Explosion, Climax, Texas - 29 June 1974	283
F	5	-	Propylene Cloud Explosion, East St.Louis, Illinois - 22 January 1972	285
F	6	-	Propane Cloud Detonation, Franklin County, Missouri – 9 December 1970	28 9
F	7	-	Crude Oil Vapor-Mist Explosion, Hearne, Texas - 14 May 1972	29 7
F	8	-	Natural Gas Cloud Explosion and Fire, Houston, Texas - 9 September 1969	299

•

:

xxviii

.

APPENDIXES, continued

·····

i

F 9	-	LPG Tank Car Explosions Crescent City, Illinois - 21 June 1970 Page	301
F1 0	-	LPG Tank Car Explosions, Laurel, Mississippi — 25 January 1969	305
F11	-	Vinyl Chloride Monomer Tank Car Explosions, Houston, Texas - 19 October 1971	309
F12	-	Propylene Tank Truck Explosion, New Jersey Turnpike - 21 September 1972	313
F13	-	Propane Cloud Fireball, Lynchburg, Virginia - 9 March 1972	317
F14	-	Natural Gas Liquids Vapor Cloud Fire, Austin, Texas - 22 February 1973	321
G	-	Adjustments to Prevent Double Counting	325

e isterioù e e

i

•

التأبيل ليقدمهم تجا

LIST OF TABLES

معامدة مرتبي للما متحصيف ومعتيون الربية ووهاديوا لياريب

_

.......

Table No.	Caption	Page
CHAPTER 3 3-1	Suggested Estimates for σ , σ	39
• -	y's	•••
3-2	Relation of Turbulence Types to Weather Conditions	39
3-3	Selected Values of $\sigma_{\mathbf{y}}$	43
CHAPTER 5		
5-1	Constants Currently Listed in the Chemical Properties File	68
5-2	Geographical/Demographic Dats Input for Phase I	70
5-3	Phase I Output Data for Each Cell	74
CHAPTER 6		
6-1	Phase II Damage Assessment	76
6-2	Relationship Between Percentages and Probits	80
6-3	Summary of Probit Equations Based on Information in Appendix D	81
6-4	Data Used in Deriving Equations for Estimating Deaths from Chlorine and Anhydrous Ammonia Gas	86
6-5	Summary of Nonlethal Injuries Received by Accidental Inhalation of Chlorine	88
6-6	Summary Data for Nonlethal Injuries from Chlorine Inhalation	88
CHAPTER 7		
7-1	Spill Sizes	98
7-2 a	Spill Definition and Environmental Data	101
7 - 2b	Selected Chemical and Physical Properties of the Cargo	103
7-2c	Geographical Data	104
7-2d	Spill and Ignition Data	105
7-2e	Concentrations in the Cells (as Computed at the Cell Centers) for Each Time Step	106

CHAPTER 7, continued

141

للمعتقدة وهناهما

1

7-2f	Concentrations in the Cells (as Computed at the Cell Centers) for Each Time Step Page	≥ 107
7-2g	Results of the Flash Fire and Explosion	109
7-2h	Cell by Cell Assessment of Injury and Damage Caused by Explosion	110
7-21	Continuation of Cell by Cell Assessment of Injury and Damage Caused by Explosion and Summary for All Cells	111
7-2j	Cell by Cell Assessment of Injury and Damage Caused by Flash Fire and the Summary for All Cells	112
7-3	Designation Denoting Source of Value for the Variable	113
7-4	Test Runs for the Urban Grid	114
7-5	LNG Spills	116
7-6	Chlorine Spills	119
7-7	Ammonia Spills	122
7-8	Immediate Pool Burning	125
CHAPTER 8 8-1	Summary of Toxic Effects	131
8-2	Summary of the Effects of Explosions and Flash Fires	134
CHAPTER 9 9-1	Additional Mishap Incidents Mentioned During Interviews and Recommended for Specific Exploration with the Respective Respondents Listed	138

xxxi

LIST OF ILLUSTRATIONS

I !

e

2

Figure No.	Legend	Page
EXECUTIVE SUMMARY	Risk Management - The Systems Approach	111
S.2	Generalized Flow Diagram of the Vulnerability Model	ix
S.3	Overall Flow Diagram for the VM	×i
S.4	Phase II Damage Assessment	xvi
INTRODUCTION		
I-1	Risk Management - The Systems Approach	2
CHAPTER 1		
1-1	Generalized Flow Diagram of the Vulnerability Model	9
1-2	Types of Submodels Currently Implemented in Phase I and Phase II	11
CHAPTER 2		
2-1a	Flow Chart for Phase I Detailing Sub- program Sequencing and Branching Decisions for Cargo Venting and Spill Development in Water	16
2-16	Flow Chart for Phase I Detailing Sub- program Sequencing and Branching Decisions for Air Dispersion and Fire and Explosion	17
CHAPTER 3		
3-1	Horizontal Dispersion Coefficient, ^G y, vs. Downwind Distance from Source for Pasquill's Turbulence Types Used for the Plume Model	37
3-2	Vertical Dispersion Coefficient, σ_z , vs. Downwind Distance from Source of Pasquill's Turbulence Types Used for the Plume Model	20
	oped for the stame hodet	30
3-3	Correction Factor for Meander of Wind	45

· • •

xxxii
CHAPTER 4		
4-1	Flow Chart of Fire and Explosion Submodels	50
4-2	Thermal Radiation from a Prismatic Volume of Gas	61
4-3	Emissivity of Water Vapor	62
4-4	Variation of $t_{1/2}$ with β	65
CHAPTER 5 5-1	Example of a Tract Map for a Portion of an Urban Area	71
CHAPTER 6 6-1	Phase II Flow Chart	78
6-2	Selected Contour Lines for Chlorine Lethality Equation (6-2a)	85
CHAPTER 7		
7-1	Relative L ocation of the Grid Cells for the Test Run	99

Ì

xxxiii

INTRODUCTION

This report describes the development of a computerized simulation model designed to estimate the consequences of a marine spill of a hazardous chemical. This Vulnerability Model (VM) has been developed for the U. S. Coast Guard under contract DOT-CG-33377-A. The model simulates the physical phenomena associated with spills and the responses of vulnerable resources to the various damage-inducing mechanisms resulting from the spill. The VM produces estimates of the total losses incurred in terms of deaths, injuries and value of property damage. It also identifies the time and location of the losses and the damage mechanism which caused them.

<u>ىلىلىكى بى بەركىمى مەركىكىلىك تەركىكى يەركىنى بىلىكى بىلىكى بىلىكى بىلىكىكىكى بىلىكىكىكى بىلىكىكىكى بىلىكى بىلى</u>

100

In recent years, new technologies and an expanding search for energy sources have led to increased bulk transport of hazardous chemicals on U. S. waters. The Ports and Waterways Safety Act of 1972 charges the Coast Guard with providing for the safety of "ports, harbors, waterfront areas and navigable waterways of the United States." To meet this responsibility, the Coast Guard is authorized to regulate the movement of bulk cargoes and the design and operation of vessels carrying them. Substantial increases in safety rarely come cheaply. The reduction in risk attributable to any regulation must be weighed against the increased operating costs of capital expenditures such regulation requires. The need to perform cost-benefit analyses of Coast Guard actions has led to a research and development program to establish a Risk Management System. The VM is an important component of this system.

The Risk Management System (Figure I-1, [1]) consists of three submodels linked together to relate various system parameters to the risks and costs of operating the system. The first submodel is spill analysis, which estimates the change in the likelihood of spill accidents induced by regulating the values of some system parameters. Next is the damage assessment phase in which the VM estimates the consequences of a spill occurring for a given set of conditions. When the consequences of an accident are combined with the likelihood of the accident occurring, a measure of the system risk is produced. It then remains to measure the economic impact of the regulation by varying the inputs to the cost submodel. The merit of the regulation can then be evaluated by comparing the reduction in risks to the increase in costs.

The main objective in developing a damage assessment tool has been to establish a workable model within a framework flexible enough to allow for future model enhancements. The approach taken was to simulate the spill through a series of separate submodels. Many of these submodels had previously been developed for the Coast Guard. In modeling complex physical processes, it has been necessary to make simplifying assumptions. A conservative design philosophy guided the development of models used in

Dunn, W. A., and P. M. Tullier. Spill Risk Analysis Program. Phase II. Methodology Development and Demonstration. Operations Research, Inc., Silver Spring, Md., August 1974. USCG Report No. CG-D-15-75 (NTIS AD 785026).



AV DU DUDU DUDU

FIGURE I-1. RISK MANAGEMENT - THE SYSTEMS APPROACH

the second s

الحجاب المحجر بالعاديس إنتواب

the VM; consequently, whenever alternate approximations were available, the approximations were chosen so that the VM does not underestimate damage. The approximations made in model development are recognized, but the modularized atructure of the VM facilitates the insertion of additional and improved wodels as they become available. A subsidiary objective in developing the VM is to aid in the assessment of the effectiveness of response measures (e.g., evacuation, chemical neutralization or adsorption, and firefighting). The effectiveness of various response measures is determined by simulating a spill and appropriately manipulating those parameters of the model affected by the postulated response. Use of the VM for evaluating response measures is deferred pending further model development.

والمتعاطية والمستعمل والمستعمل والمستقصين والمستعمل والمستعمل والمستعمل والمستعمل والمستعمل والمستعمل والمستعمل

There is a danger that the results of the VM may be misinterpreted, so caution and care must be exercised when considering the results. One point that must always be borne in mind is that the VM is entirely deterministic. The damages assessed by the model are predicated on a set of circumstances (input conditions) chosen by the user. The probability that such a set of conditions exists is not considered by the VM. The likelihood of a spill is small; this small probability combined with the probabilities of all of the other conditions yields a small overall probability that any given scenario actually occurs. Thus, although the consequences of some simulations are quite dire, the risk (the expected value) is small; large losses from rare events do not necessarily indicate a high risk activity. Because the VM is completely deterministic and because no assessment is made of the likelihood of a user-chosen scenario, considerable judgment and experience are required to avoid unreasonable (even impossible) input conditions. For example, it is possible to simulate a very large spill, even though no vessel in existence is large enough to carry that much cargo. Another instance of the possible simulation of an unrealistic scenario is the specification of a certain size spill at a location unreachable by a cargo ship large enough to carry the quantity of material specified as spilled. A further example is that the user may specify an explosion-generating ignition source, thereby causing an explosion to be simulated, for a situation in which the chemical properties of the spilled substance or the conditions at the time of ignition make the likelihood of explosion extremely small. It must also be acknowledged, when interpreting results, that the VM is in its first stage of development. Spills have been simulated for only five liquids. Solids, reacting substances, secondary damage mechanisms, damage to indoor (sheltered) populations, etc., have not been modeled. Although such omissions do not invalidate the results, comparison of results in a general context (for different chemicals, different locations, etc.) should be made with caution until further development improves the accuracy and confidence level of the model. Finally, interpretation of the results should be tempered with the knowledge that the VM alone cannot determine the risk of marine transport in comparison with other forms of bulk transport or even with other human activity involving risk.

In spite of some current limitations, the VM can perform many useful functions. By using it, the relative potential consequences of transporting different commodities can be assessed in a physically based, consistent, quantitative manner. Damage estimates resulting from a simulated spill of one cargo may have a different degree of accuracy from the results of another cargo, since a different set of submodels, each with its own precision, is used to obtain the result. Nevertheless, the relative potential consequences of transporting different cargoes are given. The sensitivity of the damage estimates to the location of the spill can also be shown with the present model. Damage estimates for locations with greatly different topographic features may possess different degrees of accuracy, since some submodels give more accurate simulations for one type of topography than another. Regardless, the relative potential consequences at different locations can be demonstrated. This type of information should be very useful in determining where facilities to load and unload vessels should be optimally located.

This final report marks the end of the first phase in the development of the VM. This model is but one of several elements in a continuing search for better methods to identify, assess, and control risks and to improve safety associated with the transport of hazardous materials. It is recognized that the subject matter of this report is emotionally sensitive and that the interim findings presented are subject to misinterpretation or misuse. However, it is concluded that the potential benefits to be derived from presenting these findings far outweigh the possible harm that cou'd occur if the results are misinterpreted, misused, or misrepresented by persons with differing motives. Constructive interest, suggestions, and criticisms of this research effort, the methods used, and the interpretation of results are solicited.

The repetitive statement of so many cautionary notes is not, nor is it intended to be, an indictment of the VM. To the contrary, the VM, in this its first stage of development, is believed to be a useful, practical tool for use in the risk analysis of marine spills. Further development of the VM will make this already functional tool more useful, more precise, and more widely applicable.

Chapter 1 of this report presents an overview of the VM design. Chapter 2 consists of a summary of the several submodels used in the VM, some of which had been developed previously under USCG sponsorship; other submodels were developed as part of this current project. Chapter 3 describes air dispersion modeling, and Chapter 4 describes fire and explosion modeling; both of these areas of modeling have received considerable attention during this effort and are, therefore, given detailed description. The VM operates in two stages, referred to as Phase I and Phase II. Phase I simulates the spill itself and the distribution of the spilled sub-cance in space, whereas Phase II uses data from Phase I to assess injuries/damage to vulnerable resources. Chapter 5 summarizes the type of input and output data associated with Phase I, and Chapter 6 summarizes the assessment procedures used in Phase II, including flow diagrams. Chapter 7 describes test computer runs of the VM . Chapter 8 presents a sensitivity analysis of the VM for input variable changes. Chapter 9 summarizes the conclusions of a historical survey performed in this effort to help validate the VM. Chapter 10 gives conclusions and recommendations. Appendix A contains flow diagrams and a related description of Phase I. Appendix B gives details of the air dispersion modeling. Appendixes 0 and D give details of the fire/explosion submodels and their damage assessment procedures, respectively. Appendix E describes the approach taken in deriving the toxicity damage assessment procedures. Append'x F is a set of case studies

compiled in the historical survey subtask. Appendix G details procedures to avoid double counting in damage assessment. Instructions on data preparation and model execution are found in a separate document, the "Vulnerability Model User's Manual."

.

٠,

فليسديد والظنافينية فالمعاولا عدد

-

CHAPTER 1

GENERAL DESCRIPTION OF THE VULNERABILITY MODEL

Introduction

And the second second second

2

This chapter presents a general overview of the Vulnerability Model (VM) and lists some of the constraints necessarily imposed upon model design and development in order to produce an operational package within the resources available. Details of what is summarized here are presented in the following chapters and in the appendixes.

Vulnerability Model

1. Scope

The VM is a computer simulation designed to provide quantitative measures of the consequences of marine spills of hazardous materials. The simulation starts with a description of the nature of the spill itself, continues through the dispersion of the hazardous material, and ultimately includes an assessment of the immediate effects of the spill on surrounding vulnerable resources, namely, people, property, and the environment.

The VM was originally conceived to be, and is ultimately intended to be, a damage assessment tool capable of treating spills of virtually all cargoes carried in marine bulk transport. However, in order to obtain a complete, functional simulation system within the resources allocated, the version of the VM operational at present was developed with fewer capabilities and less completeness than the comprehensive model originally envisioned. A comprehensive version of the VM should treat all cargoes whether in solid, liquid, or gas phase. The model should also treat other physicochemical processes affecting cargoes, including change of phase upon release into air or water, reaction, dissolution, admixture, and mass transfer between water and air. The version of the VM operational at present treats only fluid cargoes and is able to simulate only some of the physicochemical processes that a more complete model could simulate. Currently, the VM has been exercised for only five cargoes: anhydrous ammonia, chlorine, gasoline, liquefied na^{*}ural gas, and methanol.

A main consideration in developing a damage assessment tool has been to establish a workable model within a framework flexible enough to allow for future model enhancements. The approach taken was to simulate the spill through a series of separate submodels, many of which had previously been developed for the Coast Guard. In modeling complex physical processes, it has been necessary to make simplifying assumptions. A conservative design philosophy guided the development of mcdels used in the VM; consequently, whenever alternate approximations were available, the approximations were chosen so that the VM does not underestimate damage. The approximations made in model development are recognized, but the modularized structure of the VM facilitates the insertion of additional and improved

Preceding page blank

، مم طلاع الم أن التقليم التكمية عن الكان منه التليمة التحميم التكامية عن قالة الم

i

and the state of the barrenses

models as they become available. However, because the ultimate goal of the VM is a general use damage assessment tool, the use of chemicalspecific models has been avoided.

المحجر بمعد الانتار

2. <u>Simulation Scenario</u>

.......

THE PARTY OF THE P

1

÷,

- - -

Ē,

i,

-

and the second second second

The simulation requires three types of descriptive data that define: (1) the spill, (2) the physical setting in which the spill occurs, and (3) the vulnerable resources that are subject to the effects of the spill. The spill is described in terms of its location and spill rate, the physical and chemical properties of the spilled material, and the quantity of the spill. The physical setting is described in terms of the geometric configuration of the shoreline(s), hydrologic/oceanographic properties, and meteorological data. Vulnerable resources are described in terms of demographic distribution, property distribution, and land/water use. The geographic area of concern may represent any user-defined location, a rectangular area measuring 10 miles in length and 5 miles in width being typical of anticipated applications. The physical setting and the distribution of vulnerable resources are described in terms of mutually exclusive geographic cells that cover the entire area of concern.

3. Submodels

The VM consists of submodels interconnected by an executive routine, with built-in logic dictating the sequence of submodel processing as a function of the spill development. Among these submodels are simulations of surface spreading, water mixing, air dispersion, conflagration and explosion and submodels for assessing the effects from the dissemination of the hazardous material on vulnerable resources. Some of the submodels had been designed previously under U. S. Coast Guard sponsorship as part of the CHRIS (Chemical Hazard Response Information System) [2] project development. CHRIS is a chemical hazard response information system embodied in a set of field manuals; the models in CHRIS are presented in a format for hand calculation of spill development. Some of these same models have also been incorporated into HACS (Hazard Assessment Computer System) [3], which is designed for headquarters use. The models in HACS are presented in a format for computer calculation of spill development. Several of the submodels in the VM, including all of the damage assessment procedures, were designed specifically for the VM. A generalized flow diagram of the model is presented in Figure 1-1.

^[2] Raj, R. R. K., and A. S. Kalelkar. Assessment Models in Support of the Hazard Assessment Handbook. Arthur D. Little, Inc., Cambridge, Mass., January 1974. USCG Report No. CG-D-65-74 (NTIS AD 776617).

^[3] Hazard Assessment Computer System (HACS) User Manual. Arthur D. Little, Inc., Cambridge, Mass., December 1974.



مايتا سايت بالمرابع والمراقع والمحربة والمتحرية والمحد والمحديث والمحديث والمحافظ والمحرفة والمحرفة والمحدود والمحدث والمحد

FIGURE 1-1. GENERALIZED FLOW DIAGRAM OF THE VULNERABILITY MODEL

4. Operational Phases

The VM operates in two phases. Phase I simulates the spill itself, the physical and chemical transformations of the spilled substance and its dissemination in space. This phase covers the time period from the initiation of the spill until a user-specified time has elapsed. The time interval between simulation calculations is specified by the user but may be overridden by certain submodels (such as the explosion submodel). A time-history file of the spill sequence simulated during the first phase is retained on magnetic tape, disk, or other semipermanent computer storage media.

In Phase II, the computer first superimposes this time-history file upon the vulnerable resources map and then assesses the effects of toxicity, explosion and/or fire on the vulnerable resources as a function of time. Estimates of deaths and nonlethal injuri s to people and of damage to property are provided in tables.

A schematic of the types of submodels currently implemented in Phase I and Phase II is given in Figure 1-2. Details of the Phase I submodels as implemented currently are given in Chapters 2, 3, and 4 and Appendixes B and C. Details of the Phase II submodels as implemented currently are given in Chapter 6 and Apperdixes D, E, and G. Generalized flow diagrams of Phase I submodels for a comprehensive version of the VM, i.e., flow diagrams showing spill development for virtually all marine cargoes, are presented with an explanatory narrative in Appendix A. These flow diagrams and discussion are presented to guide the development of new models and to indicate the planned structure of a comprehensive VM; the material presented in Appendix A should not be interpreted as representing the current state of development of the VM.

5. Constraints

This section lists some of those constraints to model development and application required to produce an operational model within the resources available. It should be noted that some of the limitations listed here may be relaxed or removed in the event of further model development.

- (a) Several submodels applicable to marine spills of hazardous chemicals have been developed under contract to the USCG and are described in the CHRIS documentation [2]. Some of these have been programmed in FORTRAN and are currently integrated into HACS [3], also under contract to the USCG. Concurrent with development of the VM, the CHRIS and HACS documentation has been reviewed with the intent of using as many available submodel designs and computer routines as feasible in order to avoid duplicating research, development and computer programming. Some of these submodels are being used in the VM with little or no modification.
- (b) Secondary damage mechanisms, such as ignition of specially hazardous establishments (e.g., refineries), were not treated; only the direct consequences of the spill were simulated. Treatment of secondary effects, such as fire storms, additional spills caused by the primary spill, hazards from damage to key facilities (such a gas distribution on water supply system), was deferred for future development.



FIGURE 1-2. TYPES OF SUBMODELS CURRENTLY IMPLEMENTED IN PHASE I AND PHASE II

- (c) The portion of the population indoors was considered to be sheltered, and no damage was assessed to this group. Further, it was assumed that fifty percent of the subject population was indoors.
- (d) Explosion damage to structures was assessed on the assumption that the structures affected are framed with wood members; treatment of explosion damage to more substantial structures was deferred for future development.
- (e) Only inhalation toxicity was treated; injury by ingestion of toxic materials was not assessed.
- (f) Micrometeorological effects (e.g., airflow patterns around buildings) were not modeled because to do so would necessitate so fine a degree of resolution in the model that input preparation and computer run time would become prohibitively expensive. Consequently, modeling to account for the net effects of micrometeorology has been developed and is described in Chapter 3; however, the computer programming for this modeling has not been implemented. In addition, a uniform wind velocity was assumed to prevail over the entire area of concern.
- (g) Certain aspects of underwater modeling of chemicals were beyond the scope of this stage in the development of the VM. Initially, vessel spills were to be modeled only for hull ruptures at or above the waterline. Underwater releases were not considered.
- (h) Hazardous spills occurring on land that remain on land would require substantial modification of the model. Spills occurring on land but reaching the water could be treated as special cases of spills on water during subsequent development of the VM.
- (1) Of the various hazards resulting from spills, air and water pollution received lowest priority in order to avoid duplicating work now in progress at EPA. Damage to the environment was not assessed; however, the concentration of hazardous material in air and water was computed, so that the user may estimate environmental consequences at his option.
- (j) Explosions of nonchemical origin, such as those that can occur when cryogens are released on water, were not considered in this model.
- (k) The model was designed primarily to assess risk to the public rather than to those persons experiencing prolonged or unusual levels of exposure to hazards (such as emergency personnel).
- (1) Explosion of an unconfined vapor cloud is partially determined by a user input option and partially determined by the simulated characteristics of the vapor cloud; since the decision to simulate an explosion is not entirely based on physical principles, it may properly be argued that the VM will simulate an explosion when, in fact, none could occur.

(m) Test runs were performed only for five cargoes: anhydrous ammonia, LNG (liquefied natural gas), chlorine, methanol (wood alcohol), and gasoline; an analysis of inhalation toxicology was performed only for ammonia and chlorine.

the strength of the second second

- (n) Response actions (e.g., spill containment, population evacuation, and fire fighting) were not considered.
- (o) Spills of solids and reacting chemicals were not considered.
- (p) Census data were used to determine the location of the vulnerable population; since census data primarily deal with the location of the residence of people, no modeling was effected to deal with the movement of people from home to work, to school, to recreational areas, or to other nonresidential locations.
- (q) Most of the physicochemical models consider that only a single process occurs at one time, so that separate physical events occurring simultaneously are modeled as a sequence of separate events; for example, the VM models spilling, spreading, and burning as separate events, each terminating before the next can begin, even though these events can and do occur simultaneously.

6. Current Status of Development

. « رُحيان الله الينجيرية و هر أنه بايد النبية المركزة المالية المالية، ريد الله الله الله المراجد ا

At present, the VM is in a first stage of development. It has been demonstrated that an actual working model is capable of carrying on a simulation from the specification of cargo and spill conditions through to the assessment of damages to vulnerable resources. The quantitative results of the simulation appear to correlate with the picture of events given both by expert judgment and by historical records of accidental spills. This correlation is obtained even though it is recognized that some of the modeling, by necessity, is not at the highest level of sophistication. Furthermore, the cost of a given simulation has been kept within reasonable bounds. Likewise the cost of data preparation required by the VM is not excessive. There is virtually no cost involved in output interpretation, since the computer results are presented in an easy-to-understand, useroriented way.

Although the results are presented in what is thought to be a clear, forthright manner, there does exist a danger that the results of a given simulation may be misinterpreted. The VM is a deterministic model. It predicts, with what is thought to be a reasonable degree of realism, the consequences of a situation specified by the user's set of input variables. It certainly does <u>not</u> predict what are the absolutely certain, not even the probable, outcomes of certain activities of marine transportation; this is because the VM is a tool for damage assessment, which is only a part of the larger problem of risk analysis. At least five other considerations, not within the current scope of the VM, are required to perform risk analysis.

(1) The probability of the various events (the spill, the wind direction, the material spilled, etc.) predicating a given spill scenario must be considered; the likelihood, perhaps even the possibility, of certain events simulated cannot be determined without further research.

(2) The risk of damage to vulnerable resources presented by some facet of marine transport should be judged relative to risks presented by other endeavors. For example, the risk associated with a particular maritime activity may be relatively acceptable if it is less than the risk associated with some other form of bulk transport, such as rail, truck, or pipeline transport.

(3) The VM is not providing a cost/benefit analysis of the regulatory control of marine transport activities; neither is it determining risks, including economic risk, engendered by discontinuance of these maritime activities. For example, to discontinue shipment of chlorine could result in a serious water quality crisis; to stop the importation of LNG could worsen the energy crisis.

(4) In modeling complex physical processes, it has been necessary to make simplifying assumptions. A conservative design philosophy guided the development of models used in the VM; consequently, whenever alternate approximations were available, the approximations were chosen so that the VM does not underestimate damage. This conservative model design philosophy has a tendency to yield high estimates of damage; thus, if a spill scenario were actually repeated many times, it is likely that only a few occurrences of such a scenario, if any, would yield damage as large as that predicted by the VM simulation. On the other hand, it has not been deemed feasible or appropriate at this stage of VM development to define quantitatively the "worst case" of a spill and its consequences.

(5) The levels of risk acceptable to our dynamic society are not well defined, are changing, and are not truly within the province of the VM; judgments regarding these levels of risk are to be determined by the policy makers and public.

The preceding strong caveat is an attempt to prevent misinterpretation of the results of VM simulations. By no means is it intended to discredit the VM. It is believed that, as it now stands, the VM is a useful tool for the risk analysis of marine spills; with further development and improvement the VM should be able to provide even more utility and insight into considerations of this important problem. A STATE OF A

CHAPTER 2

DESCRIPTION OF SUBMODELS USED IN PHASE I OF THE VULNERABILITY MODEL

Introduction

This chapter describes those submodels that are presently included in Phase I of the VM. Because the air dispersion submodel which was originally part of HACS has been extensively revised, and because the fire and explosion submodels were developed almost entirely in this program, these submodels are discussed in some detail in Chapters 3 and 4, respectively. The other Phase I submodels were developed previously under USCG sponsorship. Some of these submodels were modified somewhat for inclusion in the VM. Detailed descriptions of these submodels are contained in the CHRIS documentation. The Phase II submodels (damage assessment) are treated in Chapter 6 and associated appendixes.

The submodels used in the VM treat the physicochemical processes affecting the spilled hazardous material; at the present time, the processes simulated by the VM may be classified as follows:

- 1. cargo venting
- 2. spill development in water
 - A. surface spreading
 - B. water mixing
 - C. sinking and boiling
- 3. air dispersion
- 4. fire and explosion

For certain of these processes different submodels are used, depending upon the nature of the spilled substance. Other processes, such as fire and explosion, consist of a sequence of dissimilar events so that the computer simulation consists of a sequence of submodels.

The selection of submodels, the flow of data, and the sequencing of submodel execution are controlled by internal logic embodied in the Phase I executive subprogram. The decisions made by the Phase I executive are based on properties of the cargo, user inputs, results computed by Phase I subprograms, or some combination of these. A flow chart indicating the sequence of subprogram execution and the important decisions determining branching is given in Figures 2-la and 2-lb. This flow chart represents the structure of Phase I as it is currently programmed in the operational VM; the analogous flow chart for a comprehensive version of the VM, which version would be the ultimate product of further development, is presented





and a second the second of the second sec

يحافظ المستعد المراجع والمسالية ومعداد معا

Contractor and Contractor and a second second



FIGURE 2~15. FLOW CHART FOR PHASE I DETAILING SUBPROGRAM SEQUENCING AND BRANCHING DECISIONS FOR AIR DISPERSION AND FIRE AND EXPLOSION

17

1.2.1.2.1.1.

with an explanatory narrative in Appendix A. The following consists of brief discussions of the computational submodels used in Phase I.

1. Rate of Escape of Fluid Cargo [2, Ch. 2]

Contraction of the second second

.

~

.

PK4

Ļ

This submodel provides a time history of tank conditions and venting rates of gas or liquid subsequent to a rupture in the tank wall. The venting of fluids is described by equations applicable to ideal fluid flow; i.e., the flow is assumed to be inviscid. Low flow rates (caused by small vent holes, low driving pressures, or high fluid density) and high cargo viscosity are two factors that will tend to make the inviscid assumption invalid. Since viscous effects reduce the venting rate, this assumption will tend to cause venting rate, and therefore combustion and toxic damage, to be moderately overestimated. For gas venting it is assumed that the perfect gas law holds; although this is a rather standard engineering assumption, the fact that many cargoes of interest experience wide swings in pressure and temperature during venting may introduce considerable divergence from the perfect gas law, especially for certain cargoes. Departure from the perfect gas law is expected to influence damage estimates only slightly; whether damage is overestimated or underestimated by use of this approximation is not clear at this time and may well depend upon the cargo, tank pressure, and tank temperature. Equilibrium thermodynamic relationships are assumed to be valid descriptors of the nonequilibrium venting processes; this assumption, however, is virtually universally in engineering and is not expected to produce any significant error. It is assumed that all of the liquid and gas inside the tank is at a uniform temperature during each step of the venting process; this assumption is difficult to justify, but at this time it has not been determined to what extent accuracy is compromised by the errors arising from this assumption. Rapid venting will tend to prevent the attainment of thermal equilibrium inside the tank. The thermodynamic process experienced by the fluid in the tank is assumed to be either adiabatic or isothermal. This common engineering assumption represents the limiting cases for polytropic processes; however, the actual thermodynamic process experienced in venting may be some other polytropic process or may not even be describable as a polytropic process. These assumptions about the thermodynamic process may yield considerable error in the estimate of venting time and total mass vented; however, further study is required to quantify the extent of the error. In any event, assumptions that underestimate venting rate or total mass vented will tend to underestimate damage. Evidently, the more mass of hazardous substance escaping from the vessel, the more damage it is capable of producing. Damage will be greater for higher venting rates because less time is then available to reduce the concentration of the hazardous substance to acceptable levels. This is true regardless of whether these levels are related to toxic damage or flammability limits.

ા આ પ્રેસિસિસિસિસિસિસિસિસિસિસિસિસિસિસિસિસિસિસ્ટ પ્રેડીસિસ્ટિસિસ્ટ વિશેષ વિશેષ વિશેષ વિશેષ વિશેષ વિશેષ વિશેષ વિ

ALC: NO

.....

1.4

i

Required input data are:

- initial tank conditions (temperature, pressure), tank volume (and approximate geometry), and initial mass content;
- physical and chemical properties of the spilled substance;
- size and locarion of the rupture.

If the tank contains only gas or if the rupture is above the liquidvapor interface, only gas is vented. If the rupture is below the liquidvapor interface, only liquid is vented; however, there is a provision for changing from the venting of liquid cargo to the venting of gaseous cargo, if and when the liquid level falls below the level of the vent.

This is an acknowledged deficiency in modeling that may underestimate damage for releases of high vapor pressure cargoes (such as LNG and chlorine), when the vent hole is placed near the top of the tank, but below the initial liquid level in the tank.

For vented gas, the flow may be choked. When the pressure inside the tank is sufficiently large with respect to atmospheric pressure (as a function of the specific heat ratio for the gas), the flow that will occur at the hole is said to be choked flow. The functional dependence of flow rate on tank pressure depends upon whether the flow is choked or not.

This submodel is employed in the first calculation block on the main flow chart presented in Appendix A. This submodel should be accurate for many fluid cargoes, except highly viscous cargoes carried in containers with internal pressures near atmospheric pressure. (The model is not valid if the Reynolds number is too small). This model is also less valid for very high flow rates, when the mechanical dynamics as well as the thermodynamics of the fluid in and exiting from the container become important. The degree of error involved by neglecting these dynamic effects has not yet been quantified, but neglect of dynamic effects probably tends to underestimate damage.

2. Spill Development in Water

Ì

Submodels simulating the physical processes affecting the halardous material spilled in the water are classified in the categories of (A) surface spreading, (B) water mixing, and (C) sinking and boiling.

A. Surface Spreading

The surface spreading submodel treats two cases of an immiscible liquid on the water surface: (1) simultaneous spreading and evaporation of a cryogenic liquid on water and (2) simultaneous spreading and cooling of a high vapor pressure chemical.

(1) <u>Simultaneous Spreading and Evaporation of a Cryogenic Liquid</u> on Water [2, Ch. 9]

This submodel estimates the spread rate, time required for complete evaporation, and the maximum extent of spread of a cryogenic liquid floating on the water surface. It is assumed that the spill occurs instantaneously, the spread area is continuous at every instant, and the heat for evaporation comes primarily from water. It is further assumed that the properties of the spilled liquid do not change during the spread. The input data required include the properties of the liquid and the heat transfer coefficient between the liquid and water. Since the heat necessary to evaporate the cryogen comes from water, the water may freeze and an ice sheet may be formed under the spreading liquid. The submodel treats this possibility. The assumptions that the spill is instantaneous and that the spreading is continuous are made for model development purposes only and do not reflect what might be realistically expected to occur in an actual spill. The violent agitation of the interface between the water and the spilled cryogenic liquid due to the rapid boiling of the cargo at this point may keep a solid ice sheet from forming under the spill and may make the spill pool break up into many small pools. The assumptions that the spill is instantaneous and that the spreading is continuous tend to underestimate the extent of spreading and the amount of the cryogen evaporated; thus the damage caused by the spill may be moderately underestimated.

كالماديد والمكافعات ومطاوعهم والمعالم

のないというないであるというできょうないできたが、このできたのできたからないというできょうというというできょう

This submodel was developed primarily for liquid natural gas but will adequately treat spills of other liquefied hydrocarbons. Although some ammonia may go into solution in the water, this submodel is being used in the case of a liquid ammonia spill at this time. Liquid anhydrous ammonia is not normally carried in refrigerated containers; treating it as a cryogen is thus decidely unconventional. However, the proc3sses of venting and evaporation will tend to cool the ammonia, so that its spill development in water will be similar to that of a cryogen. When the development of the VM was undertaken, no submodel was available in HACS which considered the spread, dissolution, and evaporation of a liquid of finite solubility such as anhydrous ammonia. Furthermore the dissolution model in HACS does not consider evaporation. The analysis performed in the A. D. Little study [4] considered the dissolution of liquid NH₃ in a highly idealized context with no surface spreading. This analysis was thermodynamically based and is not easy to incorporate into a model that considers the diffusion of the ammonia into the water and the transport by currents. Furthermore, the philosophy in the VM is to avoid chemically specific submodels. Therefore, the submodel for an insoluble cryogen was used. Although use of this submodel for a spill of liquid ammonia is not altogether appropriate, it is the best choice at this time.

(2) <u>Simultaneous Spreading and Cooling of a High Vapor Pressure</u> Chemical [2, Ch. 10]

This submodel is used to estimate the extent of spread and the evaporation rate of a high vapor pressure, lighter-than-water liquid spilled on water. In order to construct a mathematically tractable submodel, spreading and evaporation are estimated independently. The submodel utilizes basic concepts of spread and evaporation caused by a vapor pressure difference between the liquid surface and the atmosphere.

The following assumptions are made.

- (i) All of the liquid is spilled instantaneously.
- (ii) The spreading is independent of evaporation.
- (111) Entire liquid mass is at a single temperature (mixed mean temperature) at every instant of time; that is, there are no thermal gradients in the liquid mass itself.

 ^[4] Raj, P. P. K., J. Hagopian, and A. S. Kalelkar. Prediction of Hazards of Spills of Anhydrous Ammonia on Water. Arthur D. Little, Inc., Cambridge, Mass., Jan. 1974. D.O.T. Report No. CG-D-74-74 (NTIS AD 779400).

- (iv) Liquid and water properties are constant.
- (v) The mass-transfer coefficient is constant.
- (vi) The vapor concentration of the cargo in the distant atmosphere is zero.
- (v11) The temperature of the liquid when first spilled is the same as that of the water temperature.

where a magine in its provide a second

「「「「「「「」」」」」

سلتكاليا وحجافط الماكم

Assumptions (iv) and (vi) appear to be valid and do not seem to induce any significant errors. Assumption (vii) may not be correct at all times but probably has a small effect on damage compared to other assumptions. Assumptions (ii) and (iii) may introduce significant errors in damage assessment, but the error will in general overestimate damage. The effect of assumptions (i) and (v) may be to underestimate damage in some cases.

Input data required to run the submodel include the physical and thermal properties of the spilled substance and the saturated vapor pressure-temperature relationship for the substance.

This model was developed with extremely high vapor pressure substances such as diethyl ether and ethyl acetate in mind, but it should also serve for propylamine, pentane, ethyl bromide, and other petroleum derivatives. The VM also uses this submodel for gasoline.

B. Water Mixing

Two submodels for the mixing of a miscible liquid with water have been selected for initial inclusion in the VM: (1) mixing of a neutrally buoyant liquid and (2) mixing of a highly soluble, high vapor pressure liquid.

(1) Mixing and Dilution of a Water-Miscible Liquid [2, Ch. 4]

This submodel estimates the concentration, over time, of a water-miscible chemical spilled on water. Classical diffusion equations are used, strictly applicable to neutrally buoyant solutes (liquids and solids that dissolve in water). Both instantaneous and continuous spills are considered. Calculations are dependent upon the state of the water surface -calm water, tidal river, or nontidal river.

It is assumed that there is no rapid settling of the liquid due to high density of the spilled chemical, and it is assumed that no heat transfer, chemical reaction, or phase changes take place (i.e., it is assumed that the total mass of the liquid which is mixing with water remains a constant). As long as these constraints on the nature of the spilled cargo are adhered to, this submodel should yield reasonably realistic results; however, spills of cargoes that are not neutrally buoyant may behave in a manner significantly different from that predicted by this submodel. The geometry of the water region, stream and tidal velocities, total mass of liquid spilled, and the location of the spill are required input.

This submodel is very detailed in its treatment of the currents and density gradients in the water but does not consider evaporation. It is used in the VM to calculate the concentration of methyl alcohol in the river.

(2) <u>Mixing and Dilution of a High Vapor Pressure, Highly Soluble</u> Chemical [2, Ch. 11]

.

This submodel estimates the vaporization rate as well as the area and duration over which the evaporation takes place for the spill of a high vapor pressure, highly soluble liquid on water. The submodel is basically that of mixing and dilution in a river of uniform velocity over a cross section. For the navigable rivers of primary interest in the VM, the assumption of a uniform velocity profile is quite reasonable since the boundary layers on the channel sides and bottom are small compared to the stream dimensions. The effect of the approximation on damage assessment is difficult to determine without further study. It is first assumed that the entire liquid spilled goes into solution in water and the concentration is then estimated. The vapor pressure (on the water surface) is then calculated and the vaporization rate is estimated.

The basic assumptions are as follows.

- (i) The air is saturated with vapor just above the water surface.
- (ii) The chemical spilled reaches the temperature of the water instantly.
- (iii) To estimate the water dispersion (and hence surface concentration), it is assumed that the entire mass of the liquid spill initially goes into solution with water.
- (iv) An instantaneous spill at a point is assumed for calculating the water dispersion.

Of these assumptions, the last (iv) is least justifiable and most likely to produce error in the damage assessment; for the case of relatively long spill release times, the model may tend to underestimate the evaporation rate and thereby underestimate the damage caused by the dispersed vapor. The other assumptions, (i), (ii), and (iii), seem to be suitable for the level of accuracy required.

Required inputs include the mass of liquid spilled, saturated vapor pressure relationship (at water temperature), characteristics of the river, and the mass-transfer coefficient for surface evaporation.

This submodel is an extension of the previously described submodel to include the calculation of evaporation rates. In order to concentrate on this, only simple water conditions are considered. This submodel is appropriate for spills of methyl alcohol, diethylamine, or trimethylamine.

C. <u>Boiling of Heavy Liquids with Boiling Temperatures Less than</u> <u>Ambient [2, Ch. 12]</u>

This submodel estimates the rate of boiling for immiscible liquids having densities greater than that of water and having boiling points below ambient water temperature. Boiling and sinking occur at the same time. The basic assumptions are as follows.

- (i) The liquid spilled breaks up into small drops instantaneously, and these drops attain terminal velocities in a very short time with very little evaporation.
- (ii) All of the drops formed are of the same size.
- (iii) The drop cluster formed has high porosity; that is, the intering distances are large enough so that, as a first approximition, the effect of other drops on the motion of any single drop in the cluster can be neglected. In short, it is assumed that the motion of each drop is independent from all others.
- (iv) The critical Weber number is 8; that is, any drop moving at a velocity greater than that for which the Weber number is 8 breaks up into smaller drops
- (v) Forced convection heat and mass transfer results are assumed to apply

A STATUTE AND IN THE REAL PROPERTY OF

The last assumption, (v), seems quite acceptable. Assumption (iv) may not be quite true since boiling simultaneous with sinking is liable to affect the stability of the droplets, thereby changing the critical Weber number; however, the error introduced into the damage assessment by this slightly to moderately inappropriate assumption is expected to be small. Assumptions (i), (ii), and (iii) are all subject to challenge. The spilled liquid may tend to stay together, rather than break up into widely dispersed, uniform droplets as assumed. Nevertheless, these assumptions will all tend to produce a rate of vapor evolution higher than that which will actually occur. Consequently, the damages caused by the air dispersion of the evolved vapor are liable to be overestimated.

The density and surface tension of the spilled liquid, its boiling temperature at atmospheric pressure, the latent heat of vaporization, and the temperature, density, specific heat, and viscosity of the water are needed for these calculations.

This submodel is appropriate for Freon 114 and some other halogenated-hydrocarbons. It may be used for liquid chlorine spills; however, since chlorine is slightly soluble in water, some chlorine will be lost by going into solution.

3. Atmospheric Dispersion

This submodel calculates the concentration of the cargo in gas phase in the air from the time the gas is released into the atmosphere until a fire or explosion occurs, or until the maximum time stipulated for the simulation is reached. If the cargo vents as a gas, or the gas is generated by a liquid cargo which is denser than water and which has a boiling point higher than the ambient water temperature, the source of the vapor is taken to be a point source. If the gas is liberated by evaporation from a pool of liquid cargo on the water surface, the point source is removed to a virtual position five pool diameters upwind.

The submodel is based on the Gaussian distribution, which is a theoretical solution to the partial differential eduation governing diffusion problems. The dispersion coefficients used in the Gaussian distribution are obtained from the analysis of many observations of plumes from tall stacks and of puffs of smoke or some other tracer. Plumes result from continuous releases, whereas puffs result from instantaneous releases. Puffs are three-dimensional Gaussian distributions in which the dispersion coefficients depend upon the distance traveled by the puff center-of-mass. The plume is a two-dimensional Gaussian distribution in which the dispersion coefficients depend upon the distance downwind from the source to the observation point. The values for dispersion coefficients that have been compiled are based on data in which the distances (distance travelled for the puff - distance downwind for the plume) range from approximately 100 meters to several kilometers; for calculations involving short or long distances extending beyond the range covered by experiment, the VM uses values for the dispersion coefficients extrapolated from the empirical values. Extrapolation far back from 100 meters toward zero may yield inaccurate concentration values; a model, other than the Gaussian plume model, is probably more appropriate for small distances (~ 10 meters).

The dispersion coefficients are parameterized on the basis of the atmospheric stability or the turbulence class. The atmospheric stability is related to the amount of mixing in the lowest several hundred meters of the atmosphere, which is strongly dependent upon the variation of the temperature and wind velocity as functions of height and, near the ground, upon the surface roughness. The temperature structure is, in turn, dependent on the amount of heat from the sun reaching the ground and on the absorption properties of the ground. Criteria for determining the stability class based on these factors are given in standard references (see for example, Slade [5] as explained in Chapter 3).

At present, the submodel will select the plume (continuous source Gaussian distribution) if the total release time is longer than five time steps, although the user may specify that the puff (instantaneous source) Gaussian distribution be used regardless of the release time. It is presumed that the time steps are chosen to be related to certain advection times as explained elsewhere. Furthermore, the plume model is not used for wind speeds of less than 2 meters per second, because it is not valid for light winds. The plume model does not allow for diffusion in the direction of the wind, so the discontinuous changes in concentration at the upstream and downstream ends of the modeled plume are more abrupt than in reality. The puff model has the puff center leaving the source position when the gas liberation begins, but the total mass in the puff increases as the mass of gas liberated increases. Neither of these aspects of the model is wholly satisfactory, but no analytic models for short plumes or long puffs are currently available.

^[5] Slade, D. H. (ed.). Meteorology and Atomic Energy 1968. U. S. Atomic Energy Commission, Oak Ridge, Tenn., July 1968. (NTIS TID-24190)

Implicit in the Gaussian models is the assumption that the wind velocity is not a function of time or position. For the plume model, the effects of surface roughness and the meandering of the plume with time may be approximated by adjustments to the dispersion coefficients. However, one value of the surface roughness must be taken for the entire area of interest. The meandering adjustment changes the time-average concentration but does not attempt to model the statistical fluctuations of the concentration within that averaging period. Buoyancy effects are not considered in either the plume or the puff model.

Because the Gaussian models were based on data gathered from dilute plumes and puffs, their use for very concentrated cases, as in the VM, is an extrapolation. When the puff model was used close to a very large spill of material which vaporizes quickly, the model originally gave concentrations which were higher than the density of the cargo vapor at ambient atmospheric temperature and pressure. The puff model has been modified to preclude this by allowing a region of pure cargo vapor surrounded by a region in which the concentration decreases in the Gaussian fashion. The pure cargo vapor concentration is at the proper density for ambient conditions. This modified distribution is used only when the regular puff model would give unrealistically high concentrations.

The air dispersion submodel is presently being used only for the case of cargo vapors in the air, even though this submodel is also applicable to cases in which the cargo material is suspended in very fine solid or liquid particles. Currently, loss mechanisms are not included, nor are reactions taken into account. The possibility of the formation of a fog by the cargo droplets or by the cooling of the air to form water droplets is not considered. The effects of precipitation upon the concentration are not considered, except insofar as the precipitation may affect the stability class.

The data inputs required for this submodel are: wind speed, stability class, spill location, pool diameter, and the rate of cargo vapor liberation. A flag may specify if the puff model is to be used regardless of the gas escape rate. This submodel is discussed in more detail in Chapter 3 and Appendix B.

4. Fire and Explosion

This group of submodels determines whether a flammable cargo will be ignited and then determines the physical characteristics of the resulting combustion (fire, explosion, or both). Four types of fire and explosion phenomena are modeled in this section of the VM; they are: (A) ignition, (B) explosion, (C) flash fire, and (D) pool burning. The modeling of the phenomena of fire and explosion proceeds in three temporal phases. First the decision of whether, when, and where ignition occurs is made by the internal computer logic based on user inputs, results computed by submodels, and the properties of the cargo. Subsequent to ignition, either an explosion or a flash fire is modeled, depending on the type of ignition source specified by the user, as explained below. Following either of these events, the burning of flammable liquid on the water surface, if any liquid remains, is modeled; currently, burning from a vessel venting flammable fuel is not modeled. In addition, the user may specify that ignition occurs at the spill site so that only pool burning results. The treatment of (A) ignition, (B) explosion, and (C) flash fire is presented in greater detail in Chapter 4; a more complete evaluation of the acceptability of the assumptions used to develop these models is given there.

and a state of the second s

A. Ignition

This submodel determines whether ignition occurs, which ignition source originates the ignition, and at what time during the simulation the ignition takes place. All ignition sources are assumed to be located at the grid cell centers. This assumption is not realistic, especially in an urban area, but is an expendient measure used to avoid a detailed and costly specification of the boundaries of each grid cell. In some cases, this assumption will cause explosion and fire damage to be overestimated, since the size and travel of the flammable cloud will be greater for delayed ignition at the cell center than for ignition at a cell boundary. In other cases, however, ignition will not be simulated when, in fact, it would occur, because the concentration of the vapor at the cell center is below flammable limits, even though the concentration is within flammable limits elsewhere in the cell. The user predetermines whether a given ignition source will cause fire or explosion. This a priori determination of the nature of combustion may produce the simulation of an explosion when, in fact, an explosion is unlikely or even impossible. This expediency is justified by the lack of a general theory to predict the combustion behavior of unconfined flammable vapor clouds. In addition to lack of agreement about the conditions under which unconfined vapor clouds ignite, a technical determination of combustion behavior is further inhibited by the difficulty in specifying certain parameters known to influence combustion behavior. For example, an initiating detonation of sufficient strength may induce the detonation of a contiguous unconfined vapor cloud; however, the strength, time, and location of an initiating detonation that arises, say, from the seepage of the flammable vapor into a confined space (such as a building enclosing electrical equipment) are very difficult parameters to compute or even estimate. The user also specifies the strength of the ignition source. The classification of ignition sources according to potency is based on the NFPA classification of flammable liquids [6]. Flammable liquids are classified according to ease of ignition by flashpoint; the higher the flashpoint of a liquid the more difficult that liquid is to ignite. As discussed in more detail in Chapter 4, the use of the concept of flashpoint to grade ignition sources is a rather unconventional technique, adopted in the VM so that the user has the option of changing a simulation by only type of flammable substance spilled, yet obtaining combustion in one case and not in the other. Ignition sources are so designated that the NFPA class of liquids they can ignite is known; of course, all liquids more flammable than the designated class are also assumed to be ignited by that same ignition source.

This submodel uses basic physicochemical principles to determine the ignition event. There are three requirements for combustion: (1) fuel, (2) oxidizing agent, and (3) ignition source. The user will specify whether a given grid cell contains an ignition source. The fuel is provided

^[6] Tyron, G. H. (ed.). Fire Protection Handbook, 13th ed. National Fire Protection Association, Boston, 1969.

by the dispersed flammable vapor, whereas the oxidizing agent is provided by the oxygen in the air. Since combustion will occur only over a certain range of fuel-air ratios, the decision that combustion occurs will be made only if the vapor concentration in a given cell is within the flammability limits for the substance under consideration and the given cell contains an ignition source of strength sufficient to ignite the spilled cargo. The user must use judgment in specifying an ignition source and its strength at a given location. Rural areas should have fewer ignition sources per unit area, whereas urban areas should have a greater concentration of ignition sources. Ignition sources specified for residential and recreational areas should be less potent than those specified for heavy industrial areas containing facilities, such as welding shops and smelters, that are very powerful sources of ignition.

The input data required for this submodel are:

- type (fire or explosion) of ignition source for each grid cell;
- potency class of each ignition source;
- flashpoint of the spilled substance;
- upper and lower flammability limits of the spilled substance;
- concentration of the air-dispersed cargo for the time and grid cell location under consideration.

B. Explosion

This submodel calculates the peak overpressure and the dynamic impulse generated by the explosion of a flammable cargo-air mixture. In addition to these variables required for damage assessment in Phase II, the explosive yield and TNT equivalent are also determined.

This submodel assumes the following.

- The exploding mass acts like a condensed phase explosive (high explosive) located on the water or land surface.
- The explosive yield is given by the product of the heat of combustion per unit mass an. the total mass of fuel participating in the explosion.
- Only that portion of the fuel-air mixture with a concentration between the explosive limits can contribute to the explosive yield. For that part of the fuel-air mixture richer than stoichiometric, but leaner than the upper-explosive-limit concentration, only that fraction of fuel for which there is sufficient oxygen for complete burning contributes to the explosive yield.
- The well-known scaling laws for explosions are assumed to hold.

This model requires as input the time of ignition, the parametric values determining the concentration in space at the time of ignition, the heat of combustion of the fuel (cargo), the location of the explosion epicenter, and the atmospheric temperature and pressure.

C. Flash Fire

🚛 da kalen in 1975 haarde een statster in 1975 het de sta statster in 1975 het de statster in 1975 he

> This submodel calculates the effective radiation intensity level and the effective radiation duration resulting from the flash fire. The flash fire is considered to be the rapid combustion without detonation of the fuel-air layer premixed within the flammable concentration limits. The heat generated essentially instantaneously by combustion is assumed to be lost from the combustion layer entirely by radiation; thus the radiation loss as a function of time may be calculated. In order to fit in with the computational procedure in Phase II, this time-varying radiation level is represented by a fixed radiation level (the effective radiation level) and an effective duration.

> > This submodel assumes the following.

- Only that portion of the fuel-air mixture within the flammable limits burns and then only to the extent permitted by the local oxygen concentration.
- The energy released by combustion is mixed uniformly throughout the combustion layer.
- The heated combustion layer loses energy entirely by radiation.
- The emissivity of the layer is taken to be unity.
- The effective duration is three times the time required to reach the temperature given by the average of peak temperature and ambient temperature.
- The radiation intensity is that level emitted by the layer when its temperature is the average of the peak temperature and ambient temperature.
- The radiation from the flash fire is allowed to affect only those portions of space inside the burning layer.

The input data required for this submodel include the time of ignition, the parametric values determining the spatial concentration at the time of ignition, the heat of combustion of the fuel (cargo), and the ambient temperature.

D. Pool Burning

This submodel calculates the duration and magnitude of thermal radiation emitted by a burning pool of flammable cargo; the radiation level is calculated for any desired point in space. This submodel is comprised of the following submodels:

- (1) flame size;
- (2) thermal radiation from flames;
- (3) radiation view factor between an inclined flame and an arbitrarily oriented surface in space;
- (4) burning time.

The pool burning submodel calls submodels (1), (2), and (4) above once for the entire grid structure; submodel (3) is called for each grid point. The burning time [submodel (4)] is the same for all grid cells. The flame size calculation (1) is used as input to the thermal radiation calculation (2) which in turn is used with the view factor calculation (3) to give the radiation intensity at the selected spatial location. In the following discussion, these four submodels, comprising the pool burning submodel, are described in more detail.

(1) Flame Size

This submodel calculates the height, diameter, and angle of inclination of the flame from a burning pool; the wind blowing across the pool surface causes the flame to be inclined with respect to the normal to the pool surface. The details of this submodel are given in the CHRIS documentation [2, Ch. 6].

The formulas used to calculate flame height, diameter, and inclination angle are empirical expressions obtained by curve-fitting experimental data. The major assumption for these models then is that relationships obtained under laboratory conditions may be extended to larger scale occurrences in the field. Although large-scale events will probably behave very much like small-scale laboratory experiments, provided all experimental conditions other than size are duplicated, the fact is that the conditions in the field are different from those in the laboratory. Perhaps the most significant difference is that in the laboratory an effort is made to keep wind velocity constant, whereas in the field the wind gusts, changing both speed and direction. At this time, the effect of these discrepancies on damage assessment is not known.

The data required for this submodel are:

- liquid burning rate
- pool diameter
- wind velocity
- (2) Thermal Radiation From Flame

This submodel calculates the radiant heat flux incident on a receptor at some distance from a burning pool. A detailed discussion of this submodel is given in the CHRIS documentation [2, Ch. 7].

The flame from the burning pool is modeled as a cylindrical radiator of uniform temperature; this constitutes the major assumption of this submodel. The cylindrical radiator is allowed to be inclined with respect to the vertical. The atmospheric transmissivity and flame emissivity are assumed to be one, i.e. the atmosphere is not allowed to absorb radiant energy and the flame is treated as an ideal black body radiator. These assumptions about atmospheric transmissivity and flame emissivity are not correct; however the degree of error induced in the damage assessment by the use of these assumptions is unknown. The assumption that the flame is a cylindrical radiator of uniform temperature is more realistic and probably has a negligible effect on the estimation of damage. The input data required by this submodel are:

- flame height, diameter, and inclination angle;
- location of the receptor with respect to the burning pool;
- the adiabatic flame temperature.

(3) <u>Radiation View Factor Between an Inclined Flame and an</u> <u>Arbritrarily Oriented Surface in Space</u>

This submodel calculates on, a normalized basis, the view factor between a cylindrical, inclined flame and a receptor; the receptor is assumed to be oriented with respect to the flame so that it receives the maximum possible radiation flux. As discussed in many standard texts on heat transfer, for example Eckert and Drake [7], the view factor is a purely geometrical property of the spatial arrangement of the flame and receptor. Thus no physical assumptions, except those basic to radiant heat transfer, are required by this model.

The input data required by this submodel are the same as for (2) above, namely:

- flame height, diameter, and inclination angle;
- the location of the receptor.
- (4) Burning Time

This submodel calculates the length of time the pool will continue to burn after it is ignited.

It is assumed that the pool is extinguished when all of the fuel remaining in the pool at the time of ignition is burned up. It is further assumed that fuel leaves the pool not by evaporation or water mixing, but only by burning. The burning time is calculated by the following:

$$t_{b} = \frac{V_{p}}{A_{p} r_{b}}$$

where

t_b = pool burning time (s); V = volume of fuel remaining in the pool at the time of ignition (m³);

- A_{D} = area of the pool at the time of ignition (m²);
- r_{h} = burning rate of the fuel (m/s).

^[7] Eckert, E. R. G., and R. M. Drake. Heat and Mass Transfer. McGraw-Hill, New York, 1959.

The assumption, implicit in the above, that spreading stops after ignition, is subject to dispute. However, at the present time, no analysis is extant which treats the spread of a burning pool of spilled cargo. Certainly spreading rate will be reduced after ignition, because mass is being reduced by combustion.

The data required for this submodel are the volume of cargo remaining in the pool at the time of ignition, the area of the pool at ignition time, and the burning rate of the cargo.

1

いたのではないないないないない いたいてい

CHAPTER 3

AIR DISPERSION

In order to affect the vulnerable resources, the spilled material must be transported away from the spill site and dispersed in the air and water in the vicinity of the spill. This dispersion is the result of a number of natural phenomena. The most important of these phenomena, however, are the convection of the material with existing currents in the air and water and the diffusion of the material from a small volume of high concentration to form a larger volume of low concentration. These mechaniams, over which man has very little control, seriously increase the hazard from the spill and hinder the cleanup efforts.

Although there is no intention to undervalue the serious consequences of the cargo's dispersion in the water, the effects of the hazardous material in the water are neither immediate nor violent in most cases. There may be sufficient time to shut down the water intakes to water supply systems, close shellfish beds to harvesting, or warn the populace against eating fish caught from a certain section of the river. But the effects of the dispersion of the cargo in the air may be of immediate significance. A fire or explosion may devastate part of a city or town, perhaps killing many people, or toxic vapors could cause numerous casualties in a matter of minutes or hours. Thus the air dispersion submodel is of particular importance, because air dispersion is much faster than water dispersion, combustion can take place in air but not in water, and people can be injured very rapidly by inhalation of toxic gases or by asphyxia.

The air dispersion submodel calculates the location and concentration of the cargo in the air from the time that is escapes until the time that there is no further interest in its dispersion. At this time, the dispersion of the cargo in the air is treated only if the cargo is in the gas phase. The air dispersion of particulates and aerosols is not currently treated, although the same Gaussian models may be applied, as is or modified to include loss mechanisms and reactions. The escape of the gas may be by direct venting from a tank or by means of evaporation or boiling from a pool of spilled liquid cargo. The case of a liquid cargo which is denser than water and has a boiling point lower than the ambient water temperature is also considered. In this case, the gas is liberated at the water surface. Mechanisms which might be investigated for future inclusion include loss by chemical reactions, scavenging by aerosols, rainout (loss of the vapor by going into solution in the raindrops), and impingement upon vegetation.

Preceding page blank

Basis of the Submodel - Gaussian Distribution

Almost every model of atmospheric dispersion employs one of two approaches, or, occasionally, some combination of the two. These approaches are the Gaussian distribution and the finite-difference model. These approaches are quite different in use and in their basic assumptions. The Gaussian models are semi-empirical and are dependent upon the determination of certain parameters from experiments. The finite-difference models operate from basic laws of fluid dynamics. CHRIS and HACS have chosen to utilize Gaussian models, and the VM is currently using Gaussian models.

The finite-difference models operate by the repetitious application of basic equations at many points in space for a number of time steps. There are no analytic means by which the concentration at the 30th time step can be calculated - the program must proceed through the first 29 time steps first. Other drawbacks of the finite-difference models are that they require a regular grid and that special precautions must be taken at the edges of the grid to ensure that the presence of the edges does not affect the results. The accuracy of the results is primarily dependent upon the fineness of the mesh.

Once the grid network is set up and the date storage and manipulation problems are solved, the finite-difference model is very flexible insofar as physical phenomena are concerned. The statements of the basic equations occupy only a small part of the program, and features such as buoyancy and the change of wind velocity with time or position may be easily incorporated. The case in which the wind speed is zero or very close to zero causes no special problems. Removal and generation terms may also be added to the equations in a fairly straightforward manner. But the finite-difference models occupy more storage space and take much more computer time than the Gaussian models.

CHRIS and HACS use Gaussian air dispersion models. Although Raj and Kalelkar [2] discuss both the plume (continuous source) model and the puff (instantaneous source) model, only the puff model was programmed for HACS at the time HACS was received. Because of the use of Gaussian models by CHRIS and HACS and because of attempts to economize on both computer storage and running time, Gaussian models are used in the VM as well.

Since the Gaussian models are largely empirical, one must always be careful that they are not used to simulate conditions which are widely different from those under which they are validated. Specifically, the diffusion coefficients have been obtained for plumes and puffs which are diffuse and for which the tracer gas or smoke has a density close to the density of air. For the case of a vapor which has a density quite different from that of air, or for the case in which the vapor is very concentrated, say. more than several percent, the Gaussian models would have to be applied with caution. Since the model does not operate directly from basic physical principles, the modification of the model to account for effects such as buoyancy is not straightforward and may require experimental validation. In addition to speed of operation, the Gaussian models have two advantages in being analytic expressions. First, the time step can be chosen on the basis of the times at which an output is desired, and it is not necessary to evaluate the expression at intermediate times to obtain a concentration at a desired time. Second, the concentration can be calculated for any arbitrary location and only for that location. The finitedifference model requires that the concentration at all grid points be calculated at each time step. The concentration in a Gaussian model may be found for any point at any time without calculating that for other points and times. And the Gaussian models do not require any regular grid system. This feature is quite valuable in the VM where to facilitate model testing the present grid system is based on census tracts, which are irregular in density and distribution.

Description of the Submodels

The basic mathematics for both the puff and the plume is contained in Chapter 5 of [2]. The concentration at some point (x, y, z) at time t is given for the puff model by

$$C(x,y,z,t) = \frac{2M}{(2\pi)^{3/2} \sigma_x \sigma_y \sigma_z} \exp \left[-\frac{(x-Ut)^2}{2\sigma_x^2} - \frac{y^2}{2\sigma_y^2} - \frac{z^2}{2\sigma_z^2}\right] (3-1)$$

and for the plume model by

$$C_{c}(x,y,z,t) = -\frac{2Q}{(2\pi)U\sigma_{y}\sigma_{z}} \exp \left[-\frac{y^{2}}{2\sigma_{y}^{2}} - \frac{z^{2}}{2\sigma_{z}^{2}}\right]$$
(3-2)

The following nomenclature is used:

σ_x, σ_y, σ_z = diffusion coefficients (m) C_I = concentration for an instantaneous release, i.e., puff (kg/m³) C_c = concentration for a continuous release, i.e., plume (kg/m³) M = mass of vapor liberated (kg) U = wind speed (m/s) Q = rate of vapor liberation (kg/s) x, y, z = Cartisian coordinates with the origin at the source of the air dispersion material. The wind is taken to blow toward the positive x-direction. The vertical coordinate is z. The cross-wind coordinate is y.

Note that, as explained in the following discussion, the σ 's in equation (3-1), which depend on the distance travelled by the center of the puff, are different from the σ 's in equation (3-2), which depend on the distance (x) downwind from the source.

For the venting of the cargo in the gaseous state, the coordinate system has the origin at the spill location. For the release of vapor by evaporation from a pool of liquid cargo on the water surface, the coordinate system has the origin five pool diameters upwind of the pool center. The positive x axis is oriented in the direction toward which the wind is blowing. The z = 0 plane is the water surface, and for venting gas it is assumed that the vent is at the water surface. Unless specifically directed otherwise, the VM evaluates the concentration at z = 1 m. At present, the rate of vapor liberation, Q, is the average release rate. It is a simple modification to incorporate a variable release rate by evaluating Q at time t_e , where $t_e = t - x/U$; t_e is the time at which the vapor observed at the point, (x, y, z), at the time, t, was released. This has not been done because of the difficulty of calculating the amount of vapor which burns or explodes when the release rate is variable. Modification of this part of the model to allow for variable release rates would be valuable, if priorities in further work permit.

Diffusion Coefficients

The diffusion coefficients in (3-1) and (3-2) are parameters which have been determined by experiments. They are strong functions of the stability of the atmosphere - the amount of turbulence present and the variation of the temperature with height. Different experimenters have classified the stability of the atmosphere in different manners, so that direct comparison of one parameterization of the diffusion coefficients with another is not always possible. For the plume model, the set of coefficients derived by Pasquill [8], [9] has found wide acceptance. Figures 3-1 and 3-2 show these curves. The figures have been taken from Chapter 3 of Meteorology and Atomic Energy [5], and the reader is referred to this chapter for a discussion of atmospheric stability. These curves have been numerically approximated in HACS subroutine JHHDC. The values of the diffusion coefficients used for the puff model have been taken from Chapter 4 of Meteorology and Atmomic Energy [5] and are shown in Table 3-1. There has been much less work on puffs than there has been on plumes, so the coefficients for the puff model are not as widely used as the plume coefficients.

يديك كالشرخ زمت

A summary of the stability classes is shown in Table 3-2. These are the classes used in the parameterization of the dispersion coefficients for the plume model. The parameterization for the puff coefficients is only for three classes of stability. The program is set up to allow for six stability classes in case the plume model is used, but, if the puff model is used, designation of classes A, B, or C will result in the unstable parameterization being used, classes D and E will use the neutral figures, and class F refers to the very stable curve.

^[8] Pasquill, F. The estimation of the dispersion of windborne material. Meteorology Mag. 90:33-49, 1961.

^[9] Pasquill, F. Atmospheric Diffusion. Van Nostrand, London, 1962.



FIGURE 3-1

HORIZONTAL DISPERSION COEFFICIENT, σ_y , VS. DOWNWIND DISTANCE FROM SOURCE FOR PASQUILL'S TURBULENCE TYPES USED FOR THE PLUME MODEL

(From page 102 of [5])




VERTICAL DISPERSION COEFFICIENT, σ_z , VS. DOWNWIND DISTANCE FROM SOURCE FOR PASQUILL'S TURBULENCE TYPES USED FOR THE PLUME MODEL

(From page 103 of [5])

Parameter	Conditions	100 m	4000 m	Approximate power function
σ _y (m)	Unstable Neutral Very stable	10.0 4.0 1.3	300 120 35.0	$0.14(x)^{0.92} \\ 0.06(x)^{0.92} \\ 0.02(x)^{0.89}$
σ ₂ (m)	Unstable Neutral Very stable	15.0 3.8 0.75	220 50.0 7.0	0.53(x) ^{0.73} 0.15(x) ^{0.70} 0.05(x) ^{0.61}

TABLE 3-1

SUGGESTED ESTIMATES FOR σ_y , σ_z (From page 175 of [5])

The values for σ given in the columns headed 100 m and 4000 m are observed values. The approximate power function given in the last column relates the value of σ to distance travelled by the puff center (x) for the cases indicated; these functions are obtained by curve fitting the observed data.

	Night Ime conditions
B— Moderately unstable conditions C—Slightly unstable conditions	E Slightly stable conditions F Moderately stable conditions
A — Extremely unstable conditions	D— Neutral conditions*

				Nighttime conditions		
Surface wind	Day	time insolat	ion	Thin overcast $or \ge \frac{4}{3}$	≤¾ cloudiness	
speed, m/sec	Strong	Moderate	Slight	cloudinesst		
<2		A-B	B			
2	A – B	В	С	E	F	
4	B	B – C	С	D	E	
6	с	C-D	D	D	D	
>6	С	D	D	D	D	

Applicable to heavy overcast, day or night.

tThe degree of cloudiness is defined as that fraction of the sky above the local apparent horizon which is covered by clouds.

TABLE 3-2

RELATION OF TURBULENCE TYPES TO WEATHER CONDITIONS

(From page 101 of [5])

In Figures 3-1 and 3-2, we note that the abscissa extends from 100 m to 100 km, but this does not mean that the curves for the diffusion parameters are valid over that entire distance. These curves were derived from data obtained from about 100 m downwind of the source to several kilometers. In a few cases, data were obtained between 5 and 20 kilometers. Although the figures do not imply anything about use of these curves for distances shorter than 100 m, it might be inferred from Figures 3-1 and 3-2 that these curves are to be used for distances up to 100 km. This inference is not correct. For example, for stability class A at 20 km, $\sigma_z = 3000$ m. This length is greater than the height of the inversion base on all days, except the hottest sunny summer days. Thus, even for distances as short as 20 km, the Gaussian model must be used with care for certain stability classes.

On days for which unstable atmospheric conditions prevail, at distances sufficiently far from the source the value of σ_z will predict mixing above the inversion base, which in fact will not occur. For stable regimes, the detailed layer structure sometimes has to be taken into account. Fortunately, the VM is usually not concerned with distances greater than 10 km, and the curves for the dispersion coefficients are trustworthy for all stability classes up to 10 km. For short distances, the curves may safely be extrapolated to 50 m, but their use for distances of less than 50 m is questionable. Certainly their extrapolation to 10 m is unwarranted. والتلاق بما تبدين يطيئوا للاخ المسميسة المسلطانية بالطاري يسمسان فتشمير التقارب يصميلك بتناه فارحت محافلان الرحمان فأظهاف سندما والخليم

and it the state multiple set in such

The same caveats on use apply to the puff dispersion coefficients in Table 3-1. The values of σ_y and σ_z are given for 100 m and 4000 m, because these are the limits of the range over which the given power function approximation is strictly accurate. The VM uses these power functions for the range from 50 m to 10 km, when necessary.

Choice Between Puff Model or Plume Model

٩,

As stated above the plume model is strictly applicable only to continuous releases, whereas the puff model is strictly applicable only to instantaneous releases. The spills to be simulated in the VM will generally release material into the air over some finite time; i.e., the release time will be neither infinite, as required for the plume, nor infinitesimal, as required for the puff. Since no model was readily available for use in the VM to treat the realistic case of a finite release time, it was decided that the puff model would be used for those cases in which the release time is short, whereas the plume model would be used for those cases in which the release time is long. To accomplish this, however, a precise, quantitative definition of "short" and "long" release times must be made.

For "long" release times, the prevailing wind will disseminate the airborne material in a cone-shaped plume: subsequent to the end of the release, the entire plume will be translated in the windward direction with minimal change in length measured along the wind axis. For a plume, the main cause for dissemination of material in the direction of the wind is advection by the wind. For "short" release times the wind does not have sufficient time to disseminate the airborne material, so a puff, expanding as it travels, is formed. Although the wind transports the puff as a whole in the windward direction, the wind does not spread the material to any great extent. However, turbulent diffusion will cause the puff to spread in the windward direction. Evidently then the key to choosing between the puff and the plume is whether advection or diffusion is the predominant mechanism for spreading the material in the windward direction.

The plume model will be preferable to the puff model if the diffusion in the x direction (the windward direction) is small with respect to the length of the plume. The length of the plume will be Ute, where U is the wind speed and te is the time it takes for complete evaporation of the gas. Thus, when the last bit of gas is released at time te, the first bit, released at t = 0, will be at $x = Ut_e$. The scale of the diffusion in the x direction is given by σ_x , where the puff data are used to evaluate σ_x as a function of distance. Evaluating σ_x at $Ut_e/2$, we compare σ_x and Ut_e . If Ut_e is greater than $S\sigma_x$, then the plume is long with respect to the diffusion in the x direction which will occur at each end, and the plume model is preferable. If Ut_e is smaller than $2\sigma_x$, then the diffusion in the x direction should not be ignored and the puff model should be used. If Ut is between $2\sigma_x$ and $5\sigma_x$, then neither model is entirely appropriate, and either may be used at the discretion of the person using the VM. Since the plume model equation is undefined for zero wind speed and is inaccurate for low wind speeds, the use of the plume model is not recommended when U is less than 2 m/s.

and the second section of a

Choice of Time Step

The choice of an appropriate duration for the time step is also related to the wind transport time H/U. H is the scale of the region of interest and might range from 0.5 km for a small spill to 5 km or more for a large spill. In order to get some detail from the VM, the time step should be a fraction of H/U, say, (H/U)/10 to (H/U)/5. Further, no matter what values H and U have, the time step chosen should not be so large that the puff may completely pass by a grid cell during the time step. For the puff distribution to be evaluated at a point within $0.5\sigma_x$ of the peak value, the time step, Δt , must be less than σ_x/U , since it will take σ_x/U to move the puff a distance σ_x . As σ_x will vary with the time that the puff has been traveling from the source, a typical value of σ_x for the chosen stability class should be selected. For wind speed of about 5 m/s or less, time steps between 0.5 and 2 min are generally appropriate.

As presently implemented, the plume model uses the average escape rate for the duration of the vapor release, so the concentration at any point will be constant for a time equal to the release time. Thus the selection of a time step, Δt , is not quite as crucial as it is in the puff model, where a concentration close to the maximum concentration may be missed entirely if Δt is too long. If the plume model is appropriate and the time step is less than (H/U)/5, then the release of the gas should extend over several Δt or longer. The choice of time step and the decision of whether or not the plume model is appropriate are discussed in greater detail in Appendix B3.

Modification of the Puff Model

The dispersion coefficients for the puff model were derived for dilute concentrations; therefore, when the puff model is applied close to a large spill, it may calculate a concentration of the cargo gas which is greater than the density of pure cargo gas at ambient atmospheric temperature and pressure. This is unrealistic, of course, and the Gaussian puff dispersion model has been modified to preclude this event. If the regular puff equation would give a concentration at the puff center which exceeds the density of pure gas at ambient conditions, a modified distribution is used. This distribution has a hemisphere of pure cargo gas centered about the puff center, and from the surface of the hemisphere the concentration decreases in Gaussian fashion, with dispersion coefficients appropriate to the distance the puff has traveled from the spill. The radius of the hemisphere is determined by the necessity of conserving mass. As the puff is translated downwind, the radius of the hemisphere of pure vapor will shrink to zero, whereupon we have the regular Gaussian puff distribution. The details of this modification are contained in Appendix B1.

Effects of Surface Roughness

The dispersion coefficients have been derived, for the most part, from experiments conducted over flat grasslands. The effects of buildings in urban and suburban areas are to enhance the mixing processes, mostly by the greater mechanical turbulence induced but partially from the different thermal characteristics of manmade surfaces such as cement and asphalt. The plume model may be adjusted to account for these surface effects, but at present there is no way to adjust for a change in surface roughness. Thus one class of surface roughness which typifies the entire region over which the plume travels is required.

In a paper to be published in the Journal of the Air Pollution Control Association (kindly made available to us in advance by the author), N. E. Bowne suggests modifications of the standard diffusion parameters which will more accurately depict the spread of a plume over urban and suburban areas. Selected values of σ_y are presented in Table 3-3 to show that the dispersion over built-up terrain may be several times greater than it is over rural areas. These corrections for surface effects have not yet been implemented in the computer simulation.

Meandering of the Plume

The assumption that the wind is a constant with respect to time, location, and height does not mean that the random or statistical fluctuation of the wind direction about a mean value has to be omitted entirely. This is also known as meandering of the plume, and changes in the wind direction which have time periods shorter than the observation or averaging period may be treated in a statistical manner. Thus synoptic changes which typically have time scales on the order of one to several hours are excluded. But fluctuations with periods of seconds and minutes are amenable to statistical treatments. These short-period variations are of interest because they will spread the plume from the spill over a wider area than will the theoretical, but never observed, constant wind. The amount of cross-plume dispersion (the spread in the direction perpendicular to the mean wind direction) in the Gaussian model is controlled by the value of the parameter σ_v . The longer the measuring period, the larger σ_v should be to account for increased spreading of the effluents by random changes in the wind direction during this period. The values for σ_v and σ_z given in Slade [5] and used by Raj and Kalelkar [2] are the values originally published by Pasquill from measurements having a duration of roughly 10 minutes. For this reason, the concentrations calculated using these diffusion coefficients will likewise be the average concentration for a 10-minute duration. For time averages other than 10 minutes, the value of the diffusion coefficient. $\sigma_{\mathbf{v}}$, changes because of ever present random fluctuations in wind direction.

Dispersion Coefficients, m

Bernetten ander son ander son ander an

ومتحفظة جازيا كالمستنا ومعيمين والمكر المنكر التربي والمناور والمنافقة والمترافية والمرور

ومروقتهم والمحاورة الكالكان ومرارك والمرارك

Distance Downwind, km

.

	Stability Category	<u>0.1</u>	1	10	100
Sigma y, rural	A B C D E F	30 22 14 9 7 .	200 160 110 70 55 32	1500 1200 800 550 400 270	11000 8000 6000 4100 3000 2000
Sigma y, urban	A B C D E F	58 45 38 32 26 21	310 230 190 150 110 75	1900 1500 1100 780 500 390	11006 8000 6000 4000 2500 1900
Sigma z, rural	A B C D E F	14 10 7 5 3 2	400 100 63 31 20 14	3000 1400 500 130 78 48	3000 2300 1600 500 200 100
Sigma 2, suburban	A B C D E F	15 12 10 8 7 6	400 100 62 39 26 20	2900 1300 490 250 140 80	3000 2400 1600 800 440 180
Sigma z, urban	A B C D E F	26 20 17 15 13 12	700 280 150 90 45 31	3000 2500 1500 700 200 100	3000 2500 1700 1200 500 220

TABLE 3-3 (from N. E. Bowne)

SELECTED VALUES OF $\sigma_{\mathbf{v}}$

Dispersion coefficients for three types of regions showing the influence of surface roughness and other gross measures of micrometeorological factors For very short durations (seconds), the wind direction will change very little, so the average concentration for this period will be a maximum. For periods on the order of an hour or two, the fluctuating wind direction will move the plume about, and the average concentration over an hour or more will be much lower. The corrections to σ_v to account for the averaging time have been given by Turner [10], who summarized the work of Stewart, Gale, and Crooks [11] and Cramer [12], and found that concentrations decrease with sampling times of from 3 seconds to 30 minutes according to a one-fifth power law function ($C \propto t^{-2}$). Nonhebel [13] reported correction coefficients for σ_v for periods of as long as 24 hours.

रिक्रम् स्टब्स् क्रिक्सि किंग्स्ट्रिय किंग्स्ट्रिय किंग्रिक्सि किंग्रिक्सि किंग्रिक्सि किंग्रिक्सि कि

はたいかい マンファイ

The t^{-.2} power correction factor is plotted in Figure 3-3. Extrapolation of this curve for periods of less than 10 seconds and longer than 1 hour must be used cautiously, as the power law was designed to fit the data in the range from about 1 minute to about thirty minutes. The correction factor from Figure 3-3 may be used to adjust the plume concentration on axis by multiplying the concentration calculated for 10 minutes by the correction factor. For concentrations off the axis, the value of σ_y for 10 minutes must be divided by the correction factor, and the plume concentration calculated as usual with the adjusted σ_y .

Although the correction of one calculated concentration is no problem, the situation is more complicated for a number of sequential calculations. Let us compare 10 sequential calculations for a period of 1 minute with a single calculation for 10 minutes. Figure 3-3 shows that the value on axis calculated for a 1-minute averaging period will be 1.6 times the value for the 10-minute averaging period. If the wind direction does not change, the average for the ten 1-minute periods will be the same as for any individual 1-minute period and will be 1.6 times the average for one 10-minute period. This is incorrect, of course, and is due to the fact that the wind did not vary in direction from one 1-minute period to the next as it would in the physical world. For sequential applications of the Gaussian plume equation, then, if the correction factor is to be applied to σ_v to account for the length of the time step, then the wind direction must also be changed in a random manner about an average direction. Modification of the air dispersion submodel to account for the duration of the averaging period and the statistical fluctuations of the wind direction was felt to be inappropriate at this stage in the development of the VM, since departures from accurate simulation of the physical world in other ways were considered more serious.

- [12] Cramer, H. E. Engineering estimates of atmospheric dispersal capacity. Amer. Ind. Hyg. Assoc. J. <u>20</u>:183-189, 1959.
- [13] Nonhebel, G. Recommendations on heights for new industrial chimneys. J. Inst. Fuel <u>33</u>:479-513, 1960.

^[10] Turner, D. Workbook of Atmospheric Dispersion Estimates. Environmental Protection Agency, Washington, D.C., Revised 1970. Publication No. AP-26.

^[11] Stewart, N. C., H. J. Gale, and R. N. Crooks. The atmospheric diffusion of gases discharged from the chimney of the Harwell Reactor BEPO. Int. J. Air. Pollution <u>1</u>:87-102, 1958.







. . . .

Averaging Period



CORRECTION FACTOR FOR MEANDER OF WIND

Peak Concentration vs. Average Concentration

This meandering of the plume with time will result in an average concentration which is lower than the maximum values obtained instantaneously during the averaging period. In the assessment of toxic effects, the nonlinearity of the concentration-response curve is not so great that this is a serious matter. For flammable vapor clouds, on the other hand, the variation of the instantaneous concentration from the average may mean that vapor cloud would ignite, even though its average concentration was outside the range of flammable concentrations. If the development of the VM continues, it is urged that consideration be given to allowing ignition to occur when the average concentration is outside the flammable concentration range but close to the limiting concentration. Once an appropriate peak-to-mean ratio for the dispersion process is decided upon, the mathematics for determining the new ignition range and the modification of the computer program are straightforward.

Buoyancy

į

.

Fluid cargoes which have boiling points below 5°C at atmospheric pressure are usually transported in liquid form, with either pressure or refrigeration or both being employed to keep the cargo in the liquid state. If such a cargo is spilled, the evolved vapor and the air immediately surrounding it will be near the boiling point of the cargo whether or not the cargo was refrigerated. This is because the vapor is given off at the boiling point temperature and brings the surrounding air close to that temperature by bulk mixing. The low temperature of the recently evolved vapor may have a considerable effect upon the density of the cargo gas, and, if this is the case, the dispersion of the vapor will be significantly different from that of a neutrally buoyant gas. The Gaussian dispersion models, of course, were derived for the neutrally buoyant case, so the present air dispersion submodel will not be particularly accurate when the vapor cloud is very cold. The actual cloud will occupy a greater area close to the ground or water surface and will not extend as high as the calculated cloud. Negative buoyancy will also be a significant factor for very dense gases, such as radon and sulfur dioxide, even if transported and vented as gases at ambient temperature.

Positive buoyancy may be a significant factor in the air dispersion of certain materials vented as gases that are lighter than air, such as hydrogen and methane. Some materials carried as cryogenic liquids, such as LNG, may be negatively buoyant soon after release because of cooling, but upon heating by contact and admixture with the atmospher they may become positively buoyant. The transition from negative to positive buoyancy poses formidable problems of analysis.

The means to incorporate the effects of buoyancy into the Gaussian plume and puff models now used in the VM are neither straightforward nor clearly available. Although the Gaussian models are based on a theoretical solution to a diffusion equation, the values for the dispersion coefficients are obtained from experimental measurements. Thus there is no simple, easily justified adjustment that can be made to the model without some experimental verification. The adoption of unvalidated schemes for use in the VM was deemed inappropriate at this time.

One possible approach to the consideration of buoyancy effects is that correction factors for the neutrally buoyant σ_y and σ_z could be derived from comparison of data from actual spills with concentrations calculated for neutral buoyancy. These correction factors could be parameterized functions of time and stability class to take into account the change in buoyancy with time caused by admixture, warming, or both. Another possibility is the use of one of the empirical air dispersion models developed specifically for spills of this kind [4], [14], [15], although a design guide for the VM has been to avoid the use of chemically specific models. Whatever correction method might be used, it should certainly have the feature that as the cargo becomes close to neutral buoyancy by dilution with air or by warming, then the description of the Gaussian model.

^[14] Fay, J. A. Unusual fire hazard of LNG tanker spills. Combustion Sci. Technol. <u>1</u>:47, 1973.

^[15] Feldbauer, G. W., et al. Spills of LNG on Water - Vaporization and Downwind Drift of Combustible Mixtures. Esso Research & Engineering Company, March 1973. Report No. EE61E-72 (Released by the American Petroleum Institute, Re 6232).

CHAPTER 4

FIRE AND EXPLOSION SUBMODELS

The purpose of the fire and explosion submodels is to calculate those physical quantities required by the Phase II damage assessment models to estimate the consequences of explosion, fire, or both. This set of submodels determines the type of event (fire or explosion), where it occurs, and when it occurs and what physical consequences of import to the VM occur.

The processes of fire and explosion will in general occur in two phases. First an extended vapor-air cloud will rapidly burn or explode. Following either of these forms of rapid combustion, relatively slow burning from the surface of the spill will take place. The rapid combustion occurs because the fuel and oxidant are premixed over a relatively large region of space. The slower burning from the surface of the spill is a typical diffusion flame in which combustion occurs in a narrow reaction zone into which both fuel and oxidant diffuse. Because the physical processes, and hence the damage mechanisms for personnel and material, are quite different in the cases of rapid and slow combustion, different models are required for each of the three processes involved: namely, explosion, flash fire, and burning from the surface of the spill.

The relationship of the four fire and explosion submodels used in the VM to model the various processes involved and their sequence are displayed in the flow chart shown in Figure 4-1. First the decision of whether or not ignition occurs is made; of course, if there is no ignition, no further processing of the fire and explosion submodels is performed. If there is ignition, a user option determines whether fire or explosion is to be modeled. In either event, a check is made, after the rapid combustion event is simulated, to determine whether any fuel remains in the pool of spilled material; if any fuel remains, then burning from the pool is modeled. In the following are presented details of the simulation of each of the four events modeled.

The Ignition Submodel

A primary decision to be made in the fire and explosion submodels is whether combustion (either conflagration or detonation) occurs. Three items are required for combustion to occur: (1) fuel, (2) oxidizing agent, and (3) an ignition source. The user will specify whether a given grid cell contains an ignition source. The fuel is provided by the dispersed flammable vapor, whereas the oxidizing agent is provided by the oxygen in the air. Since combustion will occur only over a certain range of fuel-air ratios, the decision in the VM that combustion does occur is made only if the vapor concentration in a given cell is within the flammability range for the substance under consideration and if the given cell contains an ignition source.

A refinement of the model is to specify whether the ignition source causes an explosion or just a fire (provided, of course, that the vapor-air mixture is in the ignitable range). There are several reasons for choosing the occurrence of fire or explosion on an a priori basis. These reasons all deter the formation of a deterministic model for this decision. One reason is that unconfined vapor-air mixtures are normally not considered to be

Preceding page blank





FLOW CHART OF FIRE AND EXPLOSION SUBMODELS

explosive. However, such unconfined vapor-air mixtures can and do explode if ignition is initiated by a detonation [16] [17]. In the context of accidental spills, such initiating detonations rould originate from (1) the detonation of high explosive (say, as a result of a ship collision) or (2) the detonation of a vapor-air mixture which has seeped into an enclosed space (this source is much more likely to occur). In either case, the size and location of the initiating detonation are quite unpredictable, because of the large number of unspecified (and probably unspecifiable) variables that contribute to these properties of the explosion. Furthermore, the basic research to determine the detonatability of unconfined vapor-air mixtures as a function of the strength of the initiating detonation is not complete and in some cases is the subject of current or planned studies. Thus, even if the properties of the initiating explosion were known quantitatively, there is no method currently available that could be used to predict whether the vapor-air plume would explode. والكالمكرك فأولج أستريح المسالات للواق المحافظ المارات بم

•

بالفحاد فريدي والطلية فركونه والريان والمعاملة فالقوطية

7

.

ţ

Another refinement of the model is the gradations permitted in ignition source strength. The user is allowed to specify the ignition potential of a given source so that, for example, a single specification of a set of ignition sources will result in ignition for a highly flammable substance but will not simulate an ignition for a less flammable substance. The gradations used in the VM for ignition sources are based on the NFPA classification system [6, p. 4-8ff.] for flammable substances. The NFPA classification system is based on the concept of "flashpoint." A substance with low flashpoint is very flammable, whereas a substance with a high flashpoint is more difficult to ignite. In the VM, an ignition source is designated as belonging to a given classification based on the flashpoint of substances it is capable of igniting. Strong sources will ignite most combustible substances, those of both low and high flammability. Weaker sources are only able to ignite the most flammable materials. The gradation of ignition sources is discussed in more detail in Appendix Cl. This rather unconventional application of the concept of flashpoint is primarily an attempt to allow the user to specify ignition sources of different strengths; thus two simulations having the same input data, except for the type of substance spilled, will yield ignition in one case but not in the other. As explained in Appendix Cl, a complete treatment of ignition sources would require very complex models and massive quantities of input data to support them.

Amount of Material in Rapid Combustion

For both the explosion and flash fire submodels it is necessary to determine the amount of fuel that burns in the rapid combustion process. Because the flammable clouds have a spatial and temporal variation in concentration of fuel, calculation of the mass that burns is not an especially straightforward procedure. As discussed in Appendix C2, it seems

- [16] Strehlow, R. A. Uncombined vapor-cloud explosions an overview. In Proceedings of the International Symposium on Combustion, 1972.
- [17] Brown, J. A. A Study of the Growing Danger of Detonation in Unconfined Gas Cloud Explosions. John Brown Associates, Inc., Berkeley Heights, N.J., December 1973.

appropriate to allow only that portion of the fuel-air mixture that has a concentration between flammable limits to contribute to the energy yield of the rapid burning phase of combustion. The fuel-air mixture that contributes to the energy yield may be divided into two parts: (1) the mixture with a concentration varying from the lower explosive limit to stoichiometric and (2) the mixture with a concentration varying from stoichiometric to the upper explosive limit. The portion (1) leaner than stoichiometric burns completely; the portion (2) richer than stoichiometric burns incompletely from lack of sufficient oxygen content. The amount (mass) of fuel that burns in the leaner portion is given by the volume integral of the concentration over that portion of the fuel-air cloud. The mass of fuel that burns in the richer portion is given by the volume integral of concentration over that space, with a multiplicative weighting factor inside the integral to account for the absence of sufficient oxygen to support complete combustion. Thus,

$$m_{e} = \int_{V_{3}} C(x, y, z, t) d\tau + \int_{V_{4}} P(C)C(x, y, z, t) d\tau$$
(4-1)

where

and the second se

į.

Ē

:

ALC YARD VANC

8-- A

and the second second

the let

è

4...

And a state of the second second

į

İ

m_= mass of fuel burning

C(x,y,z,t) = concentration of the flammable material (z is the vertical coordinate)

- F(C) = a weighting function, dependent on concentration, giving the fraction of fuel present that can burn
- $V_3 =$ the region in the half space z>0 enclosed by the surfaces: $C(x,y,z,t) = K_1$ and $C(x,y,z,t) = K_2$ (where K_1 is the lower explosive limit concentration K_2 is the stoichiometric concentration)
- V_4 = the region in the half space z>0 enclosed by the surfaces: $C(x,y,z,t) = K_s$ and $C(x,y,z,t) = K_U$ (where K_U is the upper explosive limit concentration)

 $d \tau$ = the element of volume

This formulation, including an evaluation of the weighting function, F(C), is discussed in more detail in Appendix C2. The analytical forms for the participating mass, m_e . that are obtained by performing the integration indicated in equation (4-1) for specific choices of concentration distribution, C(x,y,z,t), are given in Appendix C2 for a simple puff model and are given in Appendixes Bl and B2 for the modified puff and plume distributions of concentration, respectively.

One further detail in these calculations is the time at which the ignition occurs. Rather than use the preset computation interval, it seems more realistic to assume that an ignition occurs at that moment the concentration contour of the lower flammable limit first coincides with the coordinates of the ignition source. To calculate the time of this occurrence, one sets C $(x,y,z,t) = K_L$ and then solves for t. For the simple puff concentration distribution (see Chapter 3), this procedure yields

$$= \frac{x_{i}^{2} - \sqrt{2}\sigma_{x} \{ ln[\frac{2m}{(2\pi)^{3/2}} - \frac{y_{i}^{2}}{\sigma_{x}\sigma_{y}\sigma_{z}K_{L}}] - \frac{y_{i}^{2}}{2\sigma_{y}^{2}} \}^{1/2}}{U}$$

$$U$$

$$(4-2a)$$

where

and for the plume concentration distribution the time of ignition is given by

$$t_{1} = \frac{x_{1}}{U}$$
(4-2b)

In solving equation (4-2a), it must be borne in mind that σ_x , σ_y , and σ_z are functions of time; consequently, an iterative procedure is required to find t. In practice the iteration is begun at time t_1 , the time at which the concentration at some grid location with an ignition source first exceeds (or is exactly equal to) the lower flammable limit. The distance traveled by the puff at the time t_1 is given by

$$x_1 = Ut_1$$

This value of x_1 is substituted into the approximate power functions given in Table 3-1 to yield values for σ_y and σ_z (σ_x is assumed equal to σ_y). Substituting these values of σ and x_1 into equation (4-2a) gives a new value for time, t_2 . The value of t_2 will, in general, be smaller than t_1 , because the concentration at x_1 , at t_1 , will rarely exactly equal the lower flammable limit. The value, t_2 , is used to find another set of σ 's and the whole process is repeated until the differences between successive values of t_4 are small compared to the value. For example, if

$$\frac{t_{i} - t_{i-1}}{t_{i}} < 0.001$$

then the iteration would be stopped, and t_i would be taken to be the time of explosion. Once t_i is determined, all of the σ 's are known.

The Explosion Submodel

A fundamental quantity needed to characterize the explosion of a vapor-air plume is the energy yield. Kinney [18] points out that a good approximation to the energy yield of an explosion is the change in Helmholtz free energy caused by the chemical reaction producing the explosion. However, many authors [17] [19] analyzing the potential yield of a spill of combustible material calculate on the basis of enthalpy change produced by the chemical reaction. This apparent conflict was resolved by determining that, for vapor-air explosions, the difference between the change in the Helmholtz free energy and the change in enthalpy is negligible. The details of this analysis are presented in Appendix C3.

Therefore, the energy yield of a vapor-air mixture exploding is taken to be

$$W = (-\Delta H) \frac{m_e}{M}$$
(4-3)

where

W = explosion yield (kcal)
AH = change in enthalpy with combustion (heat of combustion kcal/
 kg-mole)
m = mass of the exploding fuel (kg)
M = molecular weight of the fuel

Heats of combustion are determined by measuring the amount of heat liberated when a fuel reacts completely and forms definite reaction products. In an explosion (or flash fire), the chemical products formed by the reaction are not necessarily the same as those formed in the laboratory experiments because the elevated temperature causes the reaction products to dissociate. Nevertheless, the degree of accuracy inherent in equation (4-3) is consistent with the accuracy of the models used elsewhere in this program. It should be noted that the heat of combustion, $-\Delta H$, is for final products that include water <u>vapor</u>; most handbooks give $-\Delta H$ for final products that include <u>liquid</u> water. These values for - Δ H must be corrected for the heat of vaporization of water. In addition, it should be noted that heats of combustion are measured for the substance in the normal state at 25°C. If the reaction of interest is for a substance normally a liquid at 25°C, then, unless specified otherwise, the handbook value must be corrected to account for the heat of vaporization of the fuel. These corrections are detailed in Appendix C4.

^[18] Kinney, G. F. Explosive Shocks in Air, p. 11. The MacMillan Co., New York, 1962.

^[19] Strehlow, R. A. Equivalent Explosive Yield of the Explosion in the Alton and Southern Gateway Yard, East St. Louis, Illinois, January 22, 1972. Engineering Experiment Station, College of Engineering, University of Illinois, Urbana, June 1973. Report No. AAE TR 73-3, UILU-ENG-73 05-05.

Once the explosive yield is determined from equations (4-1), (4-2) and (4-3), the physical parameters of the explosion germane to damage assessment can be calculated from the scaling laws stated by Kinney [20]. The explosion scaling laws are based on the principle of geometrical similarity plus certain basic theoretical considerations and experimental observations. An explosion generates a peak overpressure in the surrounding medium which is dependent on the explosive energy per unit mass in the medium. For most explosions, some sort of spherical symmetry holds, so the volume of the medium affected by the explosion is proportional to d^3 , where d is the distance traveled by the wave front. The mass of the medium affected is then proportional to ρd^3 , where ρ is the density of the medium. Thus, in two geometrically similar explosions, the peak overpressure observed at some point will be the same when $W/\rho d$ is the same in both cases, where W is the energy yield of the explosion. Experiments have been performed on reference explosions in reference media (atmospheres) that correlate peak overpressure and other explosion parameters with distance from the site of the explosion. The observations in nonreference explosions can be determined by combining these tabulated results with the scaling laws. Further dimensional analysis will give scaling laws for quantities other than peak overpressure. These scaling laws are summarized by the following equations:

$$d_{g} = \frac{d_{a}(P/P_{O})^{1/3}}{(W'/W_{O})^{1/3}(T/T_{O})^{1/3}}$$
(4-4)

$$t_{a} = \frac{t_{a} (W'/Wo)^{1/3}}{(P/Po)^{1/3} (T/To)^{1/6}}$$
(4-5)

$$I_{a} = \frac{I_{a} (W'/Wo)^{1/3} (P/Po)^{2/3}}{(T/To)^{1/6}}$$
(4-6)

where

d = scaled distance from explosion center (m)

d = actual distance from explosion center (m)

- $P,T = pressure and temperature of the atmosphere in the actual case (bar, <math>^{\circ}K$)
- P_0, T_0 = pressure and temperature of the atmosphere in the case of the reference explosion (P_0 = 1 bar, T_0 = 288.15[°]K)
- [20] Kinney, G. F. Engineering Elements of Explosions. Naval Weapons Center, China Lake, Calif., November 1968. Report No. NWC TP-4654.

W' = effective energy yield of the actual explosion

- W_0 = energy yield of the reference explosion (1 kg of TNT yields 1.12 x 10[°] calories; thus W_0 = 1.12 x 10[°] calories)
- t = actual time (s)
- $t_{s} = scaled time (8)$
- I_{a} = actual impulse (N-S/m²)
- I_{a} = scaled impulse (N-S/m²)

These laws are simplified considerably if the actual explosion is assumed to occur in the same atmosphere as the reference explosion. Since ratios of absolute values for atmospheric pressure and temperature are raised to fractional powers, these factors are close to unity even when the reference and actual atmospheres are not identical. By assuming essentially identical atmospheres, one obtains



$$t_a = t_s \left(\frac{W'}{W_0}\right)^{1/3}$$
 (4-8)

 $I_a = I_s \left(\frac{W'}{W_0}\right)^{1/3}$ (4-9)

To use these scaling laws, reference is made to Table 4-1 on the following page. The scaled distance is computed by equation (4-4) or (4-7) using the computed value of the energy yield. From the scaled distance the tables give overpressure and Mach number directly. The tables also give the scaled time and impulse from which the actual time and impulse may be computed by the use of equations (4-5) or (4-8) and (4-6) or (4-9). Thus, use of the scaling laws and the tabulated reference values will give, for any distance from the explosion center, the overpressure and impulse. These parameters are necessary to evaluate damage in Phase II.

In equations (4-4) through (4-9), the quantity W', the effective yield, is used. The data for the reference explosion tabulated in Table 4-1 are for a spherically symmetric explosion. For an explosion with a center on a rigid surface, the symmetry is hemispherical, i.e., the rigid surface reflects completely all explosive energy impinging

BASIS ONE KILOGRAM

	SHOCK FRONT			SIDE-ON			REFLECTED				
SCALED DISTANCE (METERS)	MACH NUMBER	TRAVEL TIME (MS)	AVERAGE TRAVEL SPEED (N/MS)	PEAK OVERPRESSURE (MARS)		PRES IKPULSE DUR/ (RAX-HS)		UECAY PARAXETE	PEAX OVERPRESSURE R RATIO	INPULSE (DAR-HS)	VELOCITY DURATION (MS)
.94	3.405	•438	2.147	12.36	1.30	9.63	14	•	72.	6.45	•724
.952	3.364	.448	2+124	12.04	1.29	2 •64	0	4.72	69.8	6.34	.723
96	1.119	.655	2.110	11.84	1.28	2 .64	4 .	4.7	68.31	6.33	.723
.93	3.274	.473	2.074	11-34	1.25	7 .05	5	4.7	64.73	6-20	•726
			2 . 19	10.35	1.51	6 .61	.4	4.6	61.30	6.06	.732
1.00	3.210	1441	2.037	10.33	1 21	7	7	h.5 .	58.02	5.91	.742
1.02	3.147	+307	2.004	10.37	1 2 2	0	7	4 . u	54.88	5.75	.755
1-04	3.005	-525	1.9/1	7.74	1.20	L 10(14	0.6	51.70	5-51	.797
1.05	3.021	.547	1.938	5.40	1.440		10 . 5	4 . J	43.63	5.24	.010
1.95	2.95/	+257	1+935	9.04	1.51	/ 4			40110	2110	
•					1 22		. <	1.0	45.71	5.05	- 893
1.10	2.595	.567	1+8/4	0.01	1.27			3.0	43.04	4	.927
1.12	2.835	.608	1.841	8+21	1.441	5 1/2		1.7	40.51	4.91	.957
1.14	2.775	•630	1+810	7+34	1.20	5 •//		1.4	36.19	4.77	994
1.16	2.723	+652	1.779	7.48	1 • 1 • 2	2 • 10	14	3.0	35.07	4.60	1.009
1.13	2.670	+674	1.751	7.15	1.17	• • •	0	2+2	33.71	4.00	1.007
		407	• •••	6.83	1.15	۹.70	A	3.4	33.92	4.6:	1.632
1.20	2.017	•077	1.403	6.54	3.14	n .n/	.4	3.3	32.05	4.57	1.050
1-22	2.571	-/10	1.0.74	428	1.12	7 .11	0	3.1	30.36	4.52	1.070
1.24	2.720	-740	1.648	6.02	1.11		5	3.0	24.77	4.40	1.090
1.20	2.453	•/0J 7ut	1.010	5.70	1.10	2 .63		2.9	27.27	4.41	1.109
7.50	2.441	•/31	7.030	3.10	T = V 1/1		.•	2		••••	
: 30	2 400		1.619	5.55	1.09	ûя2	S	2.0	25.65	4.36	1.128
1.00	2.700	8000 875	1.6.20	5.34	1.07	3 .5	0	2.7	24.54	4.32	1.145
1+32	2.00	e	1.553	5.13	1.0.5	5 .01	5	2.0	23.30	4.27	1.165
1.3	2.324	.343	1 648	7.7	1.05	5 . 114		2.5	22.15	4.25	1.:22
1.30	2.203	.070	1.505	4.74	1.0.0		6	2.4	21.09	4.19	1.200
1.30	2.234	• 0 7 4	1.044		1.04	0.		£			
1.0	2 222	. 917	1.526	4.59	1.03	S	1	2.3	20.10	4.15	1.217
1.440	6+264 2 +03	a. 1	1.509	4.44	1.02	2 .54	6	2.2	19.20	4.11	1.234
1-44	2	• 7 - 1 06 6	1 1 00	4.30	1.01	2 . 0.		2.2	18.35	4.07	1.249
1.44	2.102	• 700 005	1.069	4+67	1 01		2	2.1	17.49	4.02	1.259
1.40	2.232	• > 7 J	1.003	1.00	1.00	- •0/ L _L	ā.	2.0	16.67	3.97	1.271
1.48	2.102	1.024	1	3+77	* 2 3.	- +00	10	6.4	20101		

.

TABLE 4-1

SAMPLE OF TABULATED DATA FOR A REFERENCE EXPLOSION OF 1 kg of TNT [20]

57

and an and the set of
hand the there is a state

• 7

upon it. To account for the additional energy imparted to the upper half plane by the reflective surface, the yield is taken to be twice the yield expected from a spherically symmetric explosion of the same size. Therefore, we take

W' = 2W

(4-10)

where W is the yield computed according to equation (4-3).

The design of the explosion model is based on scaling laws strictly applicable to condensed phase explosions. However, explosions of fuel-air clouds, called diffuse explosions by one source [21], have significant differences from condensed phase explosions. The nature of exploding fuel-air clouds has been the subject of considerable recent research [22], [23], [24], [25]. Unfortunately, scientific research into the basic phenomena of diffuse explosions has not yet proceeded to the point where the results of the research can be incorporated into the VM, that is, no suitable theoretical or semiempirical models for diffuse explosions are extant. One problem area not yet treated satisfactorily is the propagation of combustion waves through regions of nonuniform fuel concentration; some very recent research [26] has begun to address this problem. Another problem to be addressed is "shocking up." Classical combustion wave theory (Chapman-Jouget theory) [27] predicts that the combustion wave in a fuel-air mixture will be either subsonic (deflagrative) or supersonic (detonative). From classical theory, explosions result only when a detonative combustion wave propagates; de-

- [21] Kirk, P. L. Fire Investigation. John Wiley & Sons, Inc., New York, 1969.
- [22] Hawkins, S. J., and J. A. Hicks. A New Explosives Technique for Synthesizing a Wide Range of Pressure Waveforms in Air. Part 1: Approximate Theory of Air Blast from Extended Explosive Charges. Ministry of Technology, Explosives Research and Development Establishment, Waltham Abbey, Essex, Oct. 2, 1968. Report No. ERDE 9/R/68.
- [23] Woolfolk, R. W., and C. M. Ablow. Dependence of the blast wave from an explosion on the energy release rate. In Proceedings of the Fifteenth International Symposium on Combustion, August 1974.
- [24] Strehlow, R. A., L. D. Savage, and G. M. Vance. On the measurement of energy release rates in vapor cloud explosions. Combustion Sci. Technol. <u>6</u>:307-312, 1973.
- [25] Strehlow, R. A., and A. A. Adamczyk. On the Nature of Non-Ideal Blast Waves. Engineering Experiment Station, College of Engineering, University of Illinois, Urbana, April 1974. Report No. AFOSR-TR-0834.
- [26] Karim, G. A., and P. Tsang. Flame propagation through atmospheres involving concentration gradients formed by mass transfer phenomena. Presented at the ASME-CSME Fluids Engineering Conference, Montreal, 13-15 May 1974.
- [27] Lewis, B., and G. van Elbe. Combustion, Flames and Explosions of Gases. Academic Press, Inc., New York, 1951.

tonative combustion yields the shock waves which are so destructive. Deflagrative combustion waves, on the other hand, produce a "whoosh" not a "bang." However, it appears that, if the initial extent of space experiencing deflagrative combustion is large enough, the subsonic pressure waves propagating away from the combustion zone may "shock up" and develop into damage-generating, finite amplitude blast waves.

Other factors not considered by this explosion model include Mach stem formation [18] and confined explosions. Although treatment of explosions resulting from the seepage of flammable vapor into a confined space is not currently implemented in the VM, the damage potential from such explosions is large as demonstrated in Appendix C5.

The Flash Fire Submodel

The major damage mechanism of the flash burning of the vapor cloud is the heat generated by the combustion. This heat may cause ignition of combustible materials within or near the burning cloud. This heat may also cause burn damage to living organisms within the affected area.

The flash burning occurs at a very rapid rate. Following the combustion, the hot gases remaining lose heat by radiation, conduction, and admixture of cooler gas. The cooling of the combustion products also occurs at a relatively rapid rate.

Damage to materiel and personnel from the flash fire is dependent on the amount of heat transferred and the nature of the heat transfer to the vulnerable receptors. The parameters affecting damage, and to some extent the damage mechanisms themselves, are roughly the same for both living and nonliving receptors. For combustible materials, the significant damage criterion is whether or not ignition has occurred. In general, the noncombustible materials will be considered undamaged by the flash fire; the level and duration of heating are expected to be low enough so that damage to noncombustibles, such as buckling of steel beams or calcination of bricks, is not expected to occur. The ignition of combustible materials is, however, expected to be a significant damage mechanism for the flash fire. The ignitability of a combustible item depends upon a plethora of physical parameters.

and a second
STREET STREET

However, as explained in Appendix D, the only parameters used in the VM to determine ignitability are (1) radiation intensity and (2) duration of the radiation. Unfortunately, the fire hazard presented by the flash fire is quite different from the controlled experiments through which ignition and burn criteria are obtained. In the controlled experiments, the radiation intensity was maintained at a constant level; in the flash fire, the temperature of the reaction products, and therefore the radiation intensity therefrom, decreases rapidly with time subsequent to the ignition of the vapor cloud. Since ignition data for this type of radiation-time variation are not available (nor are they likely to be), the approach taken here is to use the data available for a constant radiation level. To use the available data, the variation in the radiation with time that actually occurs during the flash fire must be parameterized by a single radiation level and an effective duration time for that radiation level. Since it has been determined that the significant parameters are the intensity and duration of radiation from the hot combustion products resulting from the flash fire, the time history of thermal radiation from the hot gases should be considered in detail before the precise forms of the parametric intensity level and time duration are chosen. Following the flash fire of a puff of flammable gat an ellipsoidal shell of hot gas remains. As mentioned previously, this layer (shell) of gas loses heat by radiative, conductive, and convective heat transfer processes.

By far the most important heat loss mechanism in this situation is radiation. Consider a prismatic volume of gas losing heat by radiation through one end surface as shown in Figure 4-2. The heat loss from the volume of gas by radiation through that surface, A_r , can be expressed by

$$q = A_r \sigma \left[\varepsilon_g T_g^4 - \varepsilon_a T_a^4 \right]$$
(4-11)

where

q = heat loss by radiation (J/S)

 A_r = area through which the radiative heat loss occurs (m²)

 T_{g}, T_{a} = temperatures of the radiating gas and the environment to which the hot gas radiates, respectively (°K)

 $\varepsilon_{g,a} = \text{respective emissivities of the gas and environment}$ $\sigma = \text{Stefan-Boltzmann constant}^* = 5.67 \times 10^{-8} \text{ J/(}^{\circ}\text{K}^4 - \text{m}^2 - \text{s})$

Radiation from a volume of gas depends to a large extent upon the presence in that gas volume of gas molecules capable of absorbing and emitting infrared radiation. For the combustion products in which we are interested, the significant molecules are CO_2 and H_2O . The emissivity of a gas layer containing such molecules depends upon (1) the layer thickness, (2) the partial pressure of the thermally active species, and (3) the temperature of the gas. Figure 4-3 shows the emissivity of water vapor as a function of temperature and the product of partial pressure and layer thickness. Not shown is the fact that, for water vapor, emissivity is also a function of partial pressure (as well as the product of partial pressure and layer thickness). For CO2, the behavior of emissivity is similar to that shown in Figure 4-3 for water vapor. When both CO_2 and H_2O are present, the total emissivity is not just the sum of the separate emissivities for each species, but a slightly more complicated computation must be performed to arrive at the total emissivity. Regardless of these complicating factors, it should

The authors recognize the dual use of σ for both Stefan-Boltzmann constant and air dispersion coefficient; however, the use of σ in each discipline is so universal, that a change of symbology for this report might cause more confusion than it would prevent.









- CD - 1

FIGURE 4-3

EMISSIVITY OF WATER VAPOR

Shown as a function of temperature with optical thickness (the partial pressure of H2O times the physical thickness) as a parameter [28]

^[28] Kutateladze, S. S., and V. M. Borishanskii. A Concise Encyclopedia of Heat Transfer (translated by J. B. Arthur). Pergamon Press, New York, 1966.

be pointed out that, for the combustion of flammable vapors, the partial pressures of either CO₂ or H_2O are liable to be in the neighborhood of O.1 atmospheres. On the other hand, the thickness of the ellipsoidal shells under consideration is liable to be tens of meters if not indeed hundreds of meters. Consequently, the emissivities that result will (1) be rather high values of the order of 0.5 and (2) not change very much with either temperature or layer thickness. The layer thickness to which the hot gas is radiating is essentially infinite, since the atmosphere is many hundreds of meters thick.

Although radiation is the primary mechanism by which the hot gas layer loses heat to the surroundings, other heat transfer mechanisms are occurring simultaneously. Rather than considering these processes separately, the additional heat loss can be approximated by raising the emissivity of the hot gas layer. Therefore, because of the large thicknesses of both the emitting and absorbing gas layers and because additional methods of heat transfer are to be ignored, we take as a suitable approximation that the emissivities of the hot gas and the absorbing atmosphere are both equal to one. Therefore, equation (4-11) becomes

$$q = A_r \sigma [T_g^4 - T_a^4]$$
 (4-12)

Now the heat loss from the layer of hot gas causes the temperature of the layer to change according to the relation

$$q = -C_p \rho V_r \quad \frac{dT_g}{dt}$$
(4-13)

where

1

and the second second

ć

. .

.

ė

ì

「子子を見てい

 C_p = specific heat at constant pressure $(J/K_g^{-0}K)$ ρ = density of the layer of hot, radiating gas (K_g^{-M}) V_r = volume of the radiating gas layer (M^3)

Although the layer of hot gas is comprised of combustion products, unburnt fuel, and air, it will nearly always be mostly air; therefore, in the computer program at this time, the density of the gas layer is set equal to that of the ambient air.

Equating the heat flows in equations (4-12) and (4-13) and solving the resulting nonlinear differential equation for temperature as a function of time gives

at + c =
$$\frac{1}{2T_a^3}$$
 [arctan $(-\frac{g}{a}) - \frac{1}{2} \ln (\frac{T_g - T_a}{T_g + T_a})$] (4-14)

where

$$c = \frac{1}{2T_{a}^{3}} \left[\arctan(\frac{T_{1}}{T_{a}}) - \frac{1}{2} \ln(\frac{T_{1}-T_{a}}{T_{1}+T_{a}}) \right]$$
(4-15)

and

$$\mathbf{a} = \frac{\mathbf{A}_{\mathbf{r}}\sigma}{C_{\mathbf{p}}\rho \mathbf{V}_{\mathbf{r}}}$$
(4-16)

and

T₁ = the initial temperature of the gas layer immediately after combustion

Equation (4-14) indicates that the temperature of the gas in the heated layer declines very rapidly at first from the initial temperature, T_1 , and then the temperature declines at a diminishingly smaller rate; i.e., $T_g \rightarrow T_a$ only as $t \rightarrow \infty$. Since the time at which the gas temperature reaches the ambient temperature is not finite, let us, in a manner similar to the procedures used in nuclear physics and electronics, consider the time at which the gas temperature is midway between T_1 and T_a ; that is, let us consider the "half life" of the elevated temperature layer. If we take

$$\Gamma_{g} = \frac{T_{1} + T_{a}}{2}$$
(4-17)

then the time at which this temperature is attained, $t_{1/2}$, is given by

$$t_{1/2} = \frac{1}{2aT_a^3} \left[\arctan\left(\frac{1}{\beta}\right) - \arctan\left(\frac{2}{\beta+1}\right) - \frac{1}{2}\ln\left(\frac{\beta+1}{\beta+3}\right) \right]$$
 (4-18)

where

$$\beta = \frac{T_i}{T_a}$$
(4-19)

The variation of $t_{1/2}$ with β is shown in Figure 4-4. As β increases, i.e., as the initial temperature increases, the time required to reach the one-half temperature level decreases. This is because the heat transfer rate is proportionately higher at higher initial temperatures.

Now we are in a position to parameterize the temperature-time variation of the hot gas layer by an effective temperature and effective radiation intensity. For the effective radiation intensity, I_{μ} , we take



а) Х

H

3

 $\beta = T_i/T_a$

FIGURE 4-4

VARIATION OF $t_{1/2}$ with β

Variation in "half-life" time, $t_{1/2}$, as a function of temperature ratio, $\beta = ({}^{T}i/T_{a})$. Note that $t_{1/2}$ is multiplied by the scale factor, 2 $T_{a}^{3}a$ (as given by equation (4-16))

$$I_r = \sigma [T_g^4 - T_a^4]$$
 (4-20)

where T_g is given by equation

A Stationary and a station of the state of the

For the effective duration, t eff, of the radiation we take

$$t_{eff} = 3t_{1/2}$$
 (4-21)

where $t_{1/2}$ is given by equation

The initial temperature is taken to be the adiabatic flame temperature, as is listed in the VM properties file.

To complete the formulation of the problem, the factors A_r and V_r in equation 4-10 must be given. The volume of the half ellipsoidal shell in given by

$$V_r = -\frac{2\pi}{3} \sigma_x \sigma_y \sigma_z (r_U^3 - r_L^3)$$
 (4-22)

and the area of radiation is given approximately by

$$A_{r} = \frac{2\pi}{3} (r_{U}^{2} + r_{L}^{2}) (\sigma_{x}^{2} + \sigma_{y}^{2} + \sigma_{z}^{2})$$
(4-23)

where, as discussed in Chapter 3, σ_x , σ_y , and σ_z are the dispersion coefficients for the puff and

$$r_{U} = \left[2 \ln \left(\frac{2m}{(2\pi)^{3/2}\sigma_{x}\sigma_{y}\sigma_{z}K_{U}}\right)\right]^{1/2}$$

$$r_{L} = \left[2 \ln \left(\frac{2m}{(2\pi)^{3/2}\sigma_{x}\sigma_{y}\sigma_{z}K_{L}}\right)\right]^{1/2}$$

 K_U = concentration at the upper explosive limit (Kg/m³) K_L = concentration at the lower explosive limit (Kg/m³) m = total mass of vapor released (Kg)

The origin of these terms is given in more detail in Appendix C2.

The Pool Burning Submodel

Since this submodel is largely based on models developed previously under USCG sponsorship, it is not necessary to provide detailed descriptions of them in this report. Further details are provided in Chapter 2 and in the CHRIS documentation [2].

CHAPTER 5

INPUT/OUTPUT DATA FOR PHASE I

Introduction

This chapter summarizes the input data required to run the VM and the types of output data to be provided to Phase II for use in assessing injuries to people and damage to property. In addition, the approach to preparing a geographical grid cell structure is described.

There are three types of data required to run the VM:

- (1) chemical property data stored in a library file;
- (2) geographical/demographic data for the general region of interest, also stored in a library file;
- (3) spill and environment definition data, supplied as input by the user for each spill considered.

In the following, detailed consideration of each class of input data is given. The "Vulnerability Model User's Manual" details the mechanics of data input and output.

Chemical Properties Data

The chemical properties file is designed to contain 71 physical and chemical properties constants for more than 400 hazardous materials. The 71 properties that may be stored in the file for each material are listed in Table 5-1. This table was generated by a reporting computer program developed and run under another Coast Guard effort. The USCG currently has an ongoing program to expand the properties file and to fill _ y gaps that now exist.

The chemical properties file provides virtually all of the physical and chemical properties constants related to the hazardous material that are required to run the HACS programs described in the CHRIS and HACS documentation. In some instances, the properties file does not contain values for the required constant. In such cases, the computer executive automatically refers to a default property value file to obtain an estimated value. Since these estimated values are chosen to apply to a large number of substances, the errors induced by using default values can be quite significant; therefore, every effort should be made to provide missing values in the properties file, so that default values are not used. These missing values can be provided as part of the user-supplied input data and will override the default values. TABLE 5-1. CONSTANTS CURRENTLY LISTED IN THE CHEMICAL PROPERTIES FILE

- c.

ALC PLA

معمدها فالمشالك أألهم فالعاشية بقرعه

BEREVIATION	UNITS	DESCRIPTION
HOLUT	NO	NOLECULAR VEIGHT
NBP	¢	NORHAL BOILING POINT
CRIEN	c c	RORPAL FREEZING POINT Critical temp
CRITPR	SH/CHS	CRITICAL PRESSURE
SPCGV	ND GH	SPECIFIC GRAVITY AT A POINT
STATE	с м0	TEM" FOR SPECIFIC GRAVITY STATE OF CHEMICAL ALIQUID - ROLTO
LOI	GH/CH3	CONSTANT. LIG DEN EQUA
L02	* NO	COEFF LINEAR TERM. LIO DEN EQUA'
	~ NO ¢	CORFF SQUARE TERM, LIQ DEN EQUA UPPER TEMP BOUNDA I TO DEN
LOLO	č	LOWER TENP BOUND, LID DEN
LIOVIS	D-5/CH2	LIQUID VISCOSITY AT A POINT
LVILMP	с *ND	TEMP. FOR LID VIS FOUR
L'V2	*ND	COEFF 1/T. LIO VIS EQUA
LVUP	ç	UPPER TEMP. BOUND LIQ VIS
1 1400	CAL/CHSC	LOWER TEMP, BOUND LIQ VIS
LTCTEN	C	TEMP FON LID THER COND
LICI	*ND	CONSTANT LID THER COND EQUA
LTCZ LTCUP	= NO	COEFF LINEAR TERM, LIG THER COND EQUA
LICLO	č	LOWER TEMP BOUND+ LIQ THER COND
LHCAP	CAL/G-C	LIQUID HEAT CAPACITY AT A POINT
LHCIEN	C	TEMP FOR LID MEAT CAP Constant, 110 MEAT CAD COUR
LHCS	ND	COEFF LINEAR TERMA LIQ MEAT CAP EQUA
LHCUP	c	UPPER TEMP BOUND. LIG HEAT CAP
LHCLO	C D/C#	LOWER TEMP BOUND: LIQ HEAT CAP
STTEN	C	TEMP FOR SURFACE TENSION
IFTEN	O/CH	INTERFACIAL TENSION AT A POINT
IFTEN South	C	TEMP FOR INTERFACIAL TENSION
SOLTEN	C	TEMP FOR SOLUBILITY
SOLI	ND	CONSTANT. SOLUBILITY COUN
SOL 2	*ND	COEFF LINEAR TERM. SOLUHILITY EQUA
VP2	*ND	CONSTANT B. VAPOR PRESSURE EDUA
VP3	*ND	CONSTANT C. VAPOR PRESSURE EQUA
VPUP	C C	UPPER TEMP BOUNDA VAPOR PRESSURE
VHCI		CONSTANT, WAPOR MEAT CAPACITY FOUR
VHCZ	ND	COEFF LINEAR TERM, VAPOR MEAT CAP EQUA
VHC3	ND	COEFF SQUARE TERM. VAPOP HEAT CAP EQUA
VHC4	ND	COSEFF CUBED TERMA VAPOR MEAT CAR EQUA
VHELO	č	LOWER TEMP BOUND+ VAPOR HEAT CAP
FUS	CAL/G	HEAT OF FUSION
COMB	CAL/G CAL/G	HEAT OF VAPORIZATION HEAT OF COMBUSTION
DECOM	CAL/G	HEAT OF DECOMPOSITION
SOLHS	CAL/G	HEAT OF SCLUTION
SOL MA	CAL/G CAL/G	HEAT OF HEACTION (FITH WATER)
FLHLD	ND	LOVER FLAMABILITY LIMIT
FLHUP	ND	UPPER FLAMAHILITY LIMIT
HRNRT TOXIN	CM/5 000	BUNNING RATE TOTICITY BY ISHABATION
NHALP	PPH	SHORT TERM INHALATION LIMIT
NHALT	5	SHORT TERM TRACATION
TOXLO	G/RG 1/80	LUBER LINIT JOAICITY BY INGESTION Upper linit toxicity by ingestion
LATEYOX	ND	LATE TOXICITY
ADFLH	c	ADIABATIC FLAME TEMP
NOLKA A105U54	ND	MOLECULAR PATION REACTANIS TO PRODUCTS Stutchionetric all 1- Euch Patio
FLHTEN	c	FLAME TEMPERATURE
POLFRAC	ND	LIMITING VALUE. MOL FRACTION CONC

د، مدمند، سباب، محمد بمحد. ول

.

*Error in dimensional units as reported on this table.

68

ولوغ فالبالا فالده الالخاط التكريك

Geographical/Demographic Data

The geographical/demographic data required as Phase I input are listed in Table 5-2. Not all of the data listed in Table 5-2 are at present used in the operation of the VM; however, space is allowed for these data and, in some cases, the data are entered in the event that further developments of the VM require this information. Those items marked by an asterisk in Table 5-2 are currently used for computations in the VM. In order to input these data, it is necessary to partition the macroregion in which the spill is to be simulated into grid cells for which representative geographical/ demographic data may be supplied. To this end, a major city and its surrounding areas, including a 30-mile segment of a navigable river, were partitioned into more than 400 grid cells. These cells were determined, for simulation purposes, so that it could then be assumed that whatever occurs within a cell does so uniformly throughout the cell. The river cells are numbered in ascending order from upstream to downstream. The approach taken to this partitioning was to equate a census tract, as defined by the U.S. Department of Commerce, to a cell. By definition, a census tract is a "...smal; area into which large cities and metropolitan areas are divided for statistical purposes. Tract boundaries are established cooperatively by a local committee and the Bureau of the Census and are generally designed to achieve some uniformity of population characteristics, economic status, and living conditions"[29]. Those census tracts that include both land and water were partitioned into two or more cell. The river itself was aivided into 84 cells, so that no ccll would be larger than 1 mile in length and so that each cell would be approximately rectangular in shape as required for river flow simulation.

Figure 5-1 is a reduced copy of a portion of the map of an urban area. Tracts on the map are identified numerically (e.g., 215, 234) and are separated from one another by solid, dark lines. For VM simulation purposes, the center of each tract (or cell) is identified by the latitude and longitude at its centermost point. For irregularly shaped tracts, a point is chosen to be representative of the tract area. Thus, the cell is identified only by a representative point, not by a description of its boundaries.

Data for each tract are available from, among other sources, the 1970 Census of Housing Report for the city. The data include, for each tract, total populaton, percent of population under 18 and over 62 years of age, assessed dollar value of houses, etc. Additional data (such as land use, etc.) were used in preparation for testing the VM; several sources of these applicable data are being used. The population statistics represent the number of persons residing in a given cell, not necessarily the number of persons in the cell at any given time. The estimated dollar value of structures is given only for residential property; these values were derived from the responses to census questionnaires. It is a limitation of the VM that the dollar values for damage assessment do not include damage to commercial property.

^[29] U. S. Department of Commerce, Bureau of the Census, 1970 Census of Housing, Block Statistics, September 1971.

and a **later of the second second**

LALE AN A MARINE

For Land Areas

- * 1. Grid number
- * 2. Latitude
- * 3. Longitude
- * 4. 0 (denotes land grid)
- * 5. Total population
- 6. Percent of population under 18 years of age
- 7. Percent of population over 62 years of age
- * 8. Percent of population sheltered

- * 9. Total number of housing units *10. Average assessed dollar value per unit dwelling
- 11. Housing construction material
- 12. Number of schools
- 13. Land use
- 14. Uniformity of the land topography
- *15. Ignition source code

For Water Areas

- * 1. Grid number
- * 2. Latitude
- * 3. Longitude
- * 4. Depth (nonzero)
- * 5. Length of cell
- 6. Direction of water current
- * 7. Ignition source code
- * 8. Speed of current
- 9. Tidal condition
- 10. Water turbulence level
- 11. Water temperature
- 12. Water density
- 13. Salinity

These items are currently used for computing results in the VM.



ملالا ولينابع للزور

Bureau of the Census publications obtained from the GPO include a description of pier facilities containing, among other information, listings of the industries that make use of the pier facilities.

Detailed 7-1/2 minute maps of the entire area selected for the test have been obtained from the United States Geological Survey Map Division in Arlington, Virginia.

The National Climatological Data Service in Asheville, North Carolina, provided meteorological data covering the years 1971 through 1973. Data include daily and monthly temperature readings, wind speed, and wind direction.

Nautical charts and other data have been obtained from the National Ocean Survey Distribution Center in Riverdale, Maryland. These include tide tables, tidal current tables, shipping lanes and river and channel depths.

Nautical charts have been obtained from the National Ocean Survey Distribution Center in Riverdale, Maryland. These include tide tables, tidal current tables, shipping lanes, and river and channel depths.

Spill and Environmental Definition Data

وكالمستركد والمراجع والمسارية والمسارية والمساولان وأستعاده

The third category of input data required for Phase I operation is the detailed information that defines the spill and the environment in which it occurs. There are four subcategories into which this type of input data falls, namely

- 1. definition of the subset of grid cells which are to be included in the simulation
- 2. spill definition data
- 3. environmental data
- 4. user override data

The data comprising item 1 merely constitute a list of the numbers of the grid cells which are to be considered in the simulation. Because of the manner in which the simulation proceeds, these cells need not be contiguous. However a clearer picture of what occurs is obtained if the cells considered fill out an approximately rectangular region.

The data comprising item 2 consist of spill-related data, such as:

spill location
size of spill
substance spilled
temperature and pressure in the cargo container
diameter of the vent or puncture

As with other types of input data, if no value is specified, the default value file will supply a value for the missing item.

The environmental data comprise a description of the physical background in which the spill occurs; included in these data are items such as: air temperature wind velocity humidity atmospheric stability class barometric pressure

the state and strength

The final category of data consists of input items that will override default file values. That is, if the default values for constants missing from the properties file appear undesirable, the user may override the default values with his own inputs.

-

ren i miner

۰. د

1

.

į

÷

Phase I Output to Phase II

- - Z.-

- . . .

The type of data output from Phase I to Phase II depends upon what course of events the simulation follows. The data are presented on a cellby-cell basis. Certain types of data, such as vapor concentration, are delivered as output for each time step. Other types of data, such as thermal radiation intensity, are delivered as output only once because of the nature of the models involved. Table 5-3 lists the types of data generated by Phase I and used by Phase II to make damage assessments. معمولهم ومعالية فأعمدهما المحمد والمتكافئة والمحمد

ţ

The same of the second

Ê

1

177

.

.....

		Dimensional Units	Generating Submodel	Generated Every Time Interval?
1.	Gas concentration	kg/m ³	AD	Yes
2.	Interval duration	8	E	Yes
3.	Liquid concentration	kg/m ³	WM	Yes
4.	Effective thermal radiation intensity	J/m ² -s	FF	No
5.	Effective time duration	S	FF	No
6.	Thermal radiation	J/m ² -s	PB	No
7.	Burning time	8	PB	No
8.	Peak over pressure	N/m ²	EX	No
9.	Dynamic impulse	N-s/m ²	EX	No

and the second
سماد بلغاء كرواؤه والالبه كمد

للعشر فيلهم تكليل ومقصد متعارب الفككال الطميم المارة المارسينية المطبالاتيان العام بمارفتهم

Key to submodel abbreviations

AD - air dispersion
E - executive
EX - explosion
FF - flash fire
PB - puol burning
WM - water mixing

62 . 6.

74
CHAPTER 6

PHASE II ASSESSMENT PROCEDURES

Introduction

Phase I of the VM simulates the spill itself, the physical and chemical transformations of the spilled substance, and its dissemination in space. The history of the spill development simulated in Phase I is stored by grid cell and time interval. These data, constituting what is referred to as the "time-history" file, along with vulnerable resources data are the input to Phase II. Phase II assesses the effects of the spill on vulnerable resources: people, structures, and the environment. This chapter summarizes procedures of Phase II.

Injury and Damage to Vulnerable Resources

The vulnerable resources of interest in the VM, namely "people," "property," and "the environment," are subject to a plethora of potential hazards from spills of marine cargoes. For example, the vulnerable resource "people" may be affected by inhalation of toxic vapor; by burns from thermal radiation from a fire, either flash fire or pool burning; by peak overpressure or impulse from an explosion, manifested as direct impact or fragment injuries; by ingestion of toxic substances; by infiltration of the skin, mucous membranes, or eyes by toxic substances; by asphyxiation from high concentrations of gases not usually considered hazardous; by pulmonary burns resulting from the inhalation of burning or hot gases; by the inhalation of toxic combustion products; by frostbite from cryogenic liquids; by accidential injury resulting from individual or group panic; by complications of injuries as a result of substandard medical treatment caused by damage to or overtaxing of medical facilities and personnel; by injury from secondary events (secondary fires, secondary explosions, vehicular accidents, etc.) that are induced by the initial spill and its immediate effects.

At this time, however, only a limited number of damage mechanisms are simulated by the VM. The vulnerable resource "people" is modeled to be affected by inhalation of toxic vapor, by thermal radiation from a flash fire, and by peak overpressure or impulse from an explosion, or by some combination of these. The vulnerable resource "structures" is modeled to be affected by thermal radiation from a flash fire or burning pool and by peak overpressure or impulse from an explosion. For the vulnerable resource "environment," defined as air and water for the purposes of the VM, there is no direct calculation of damage caused by the spilled substance in its vapor or liquid phase or by a reaction product. Table 6-1 lists the specific types of injury or damage, assessed during Phase II of the VM, that are caused by toxicity, fire, and explosion, respectively. Damage and injury from explosion and fire are treated in detail in Appendix D; injury from toxic vapors is discussed in depth in Appendix E.

Damage Causing	Vulnerable Type of		Cause of	Outdoors (Unsheltered)			
Event	Resource	Inj	ury or Damage	Injury or Damage	Function	Factor(s)	
TOXICITY	People*	Death NonTethal injury Irritation		Toxic vapor: concentration or cumulative dose		Concentration and time Concentration	
			Death	Direct blast Impact	E1 E2	Peak overpressure Impulse	
	People*	* Non-letha: Injury	Bone fracture Puncture wounds	Impact Flying fragments	E3 E4 E5	Impulse	
EXPLUSION			Multiple injury	Two or more of the above	E6	Impulse or peak over- pressure and impulse	
	Structures	Struc	tural damage	Direct blast	<u><u><u>S1</u></u></u>	Peak overpressure	
	People	Death First degree burn			B1 *** B2 ***	Duration and magnitude	
FUUL DURNING	Structures	Ignition		inerina) rautation	B3	of thermal radiation	
	Deenle		Death		F1	Effective duration and	
FLASH FIRE	reopie	First	degree burn	Thormal radiation	F2		
	Structures	Ignition			F3	effective magnitude o thermal radiation	

TABLE 6-1. PHASE II DAMAGE ASSESSMENT

* Comparable assessment format may be provided by age group where data are available.

****** Separate functions for each chemical.

*** These functions are essentially the same as F1 and F2; however, as indicated in the adjacent column, slightly different arguments are used. These functions have not yet been implemented as part of the computer program, although to do so will be very easy.

Overview of Phase II

e e come de la come come come

The function of Phase II is to read the time-history file of the simulation, to compute the damages resulting from the values of physical parameters contained in the time-history file, and then to assess damages and injury to the vulnerable resources impacted by the physical events. An overall flow chart for Phase II is presented as Figure 6-1. In most cases, the percent of the vulnerable resources affected within a given grid cell is calculated and then applied to the numbers of vulnerable resources present, giving the total numbers of vulnerable resources affected (for a given time period and given grid cell). The level of damage calculated at the center of the grid cell is assumed to apply to the entire grid cell and all of the vulnerable resources in it. The algorithm for computing the percent affected is denoted under the column headed "Function" in Table 6-1. The factor or factors computed by Phase I and used by each algorithm to assess damage are listed in the adjacent column of the table. In the case of first-degree burns resulting from the thermal radiation of a flash fire, the percent of vulnerable resource (people) affected in each cell is not computed; instead, a message is printed out indicating that the radiation intensity and duration were sufficient to cause first-degree burns in that cell. For all other damage assessment cases treated in the VM, the percent of the vulnerable resource experiencing the given type of damage is calculated.

For most of the assessment functions indicated in Table 6-1, the percent of vulnerable resource damaged is related to the causative factors computed in Phase I by probit equations (to be explained below). However five assessment procedures viz., B2, B3, F2, F3, and T3, do not use probit equations. As mentioned in the above, no percentage is calculated for F2 or B2, nonlethal injury to people from fire. For B3 and F3, ignition of structures by pool burning and flash fire, respectively, it is assumed that 25% of the structures in a given cell are ignited when the ignition criteria are met at the cell center. As explained further in Appendix D, this ad hoc assumption of 25% was used as an expedient estimate to account for shielding effects without performing a detailed analysis and calculation. Clearly some structures will be shielded from radiation by other intervening structures. For T3, toxic irritation of people, it is assumed that 100% of the subject population is irritated when the concentration criterion for irritation is met. Nonlethal injury from inhalation of ammonia is not simulated, as explained in Appendix E; thus for T2, in the case of ammonia, zero percent of the population is always assessed.

The remaining assessment calculations are based on probit functions [30]. A probit function takes the following form:

$$Pr = a + b \log_{a} V$$

(6-1)

247-2-10-1-1-

いい うちょう ちょうし ひろう

والمتنافع تعليهما والمعالمات المتحقيق فالمقار فالمقاربة المتعاقفة والمتعاقفة والمتنافع فالمتحقق والمتنافع والمتنافية والمتنا والمتنافية والمتنافية والمتنافية

where the dependent variable, Pr, is a measure of the percent of the vulnerable resource affected and the independent variable, V, is some function of the factor that causes injury or damage to the vulnerable resource. The coefficients a (location parameter) and b (slope parameter) are

^[30] Finney, D. J. Probit Analysis, 3rd ed. Cambridge University Press, London, 1971.

:

かかかい かんばん はいかかい プレス といなん いっぱん しゅうしゃ あいせんかい いういい かんしょう しんかい しょうしょう

ショントレート しまずまたい 日本をあたらう

f fil t

t

1

;

1

.



computed by maximum likelihood estimation similar to the computation of coefficients for classical regression equations. The variable Pr is referred to as a probit (probability unit). It is a Gaussian distributed random variable with mean value 5 and variance 1. The percent of the vulnerable resource affected is the percent corresponding to the cumulative distribution of Pr. This correspondence is given in Table 6-2 in which the main entries are probits and the row and column headings give the corresponding percents.

Each of the assessment equations E1, E2, E3, E4, E5, F1, B1, S1, S2, T1, and T2 (see Table 6-1) is a probit equation. For example, equation E1 is the probit equation for estimating deaths from lung hemorrhage resulting from the peak overpressure from an explosion. E1 is given by

 $Pr = -77.1 + 6.91 \log_{e} (P_{p})$

where P_p is the peak overpressure in N/m². A peak overpressure of 1.10 x 10⁵ N/m², say, gives a probit of 3.11. Table 6-2 shows that this probit corresponds to approximately 3% deaths from lung hemorrhage. The coefficients (a = 77.1 and b = 6.91) were calculated from the following data, taken from Table 6-3:

Percent Affected	Probit (Table 6-4)	Peak Overpressure <u>(N/m²)</u>		
1	2.67	1.00×10^{5}		
10	3.72	1.20×10^{2}		
50	5.00	1.41×10^{2}		
90	6.28	1.76×10^{2}		
99	7.33	2.00×10^{7}		

The above probits and the logarithms of the above peak overpressure values were inserted in a computer program that then calculated the values to be used for a and b.

In some cases, a function of the variable that determines injury or damage was used rather than simply the numerical value of the variable. For example, as explained in Appendix D, the probit equation F1 (deaths from burns) is given by:

 $Pr = -14.9 + 2.56 \log_{e} [t (1^{4/3})/10^{4}]$

where t is the effective duration (in seconds) and I is the effective radiation intensity (in J/m^2 seconds). The 10^4 is used solely to reduce the magnitude of the $I^{4/3}$ value. This reduction, in turn, is absorbed in the slope coefficient, b (in this case b = 2.56). The exponent 4/3 of 1 was obtained iteratively to provide a good fit to the data.

The logarithm is conventionally used in probit equations, not for theoretical purposes, but because the logarithm usually transforms the relationship between causative factor and response into a Gaussian function. This frequently is the case when the variable to be computed

X	0	1	2	3	4	5	6	7	8	9
0	_	2.67	2.95	3.12	3.25	3, 36	3.45	3.52	3.59	3.66
10	3.72	3.77	3.82	3.87	3.92	3.96	4.01	4.05	4.08	4.12
20	4.16	4.19	4.23	4.26	4.29	4.33	4.36	4.39	4.42	4.45
30	4.48	4.50	4.53	4.56	4.59	4.61	4.64	4.67	4.69	4.72
40	4.75	4.77	4.80	4.82	4.85	4.87	4.90	4.92	4.95	4.97
50	5.00	5.03	5.05	5.08	5.10	5.13	5.15	5.18	5.20	5.23
60	5.25	5.28	5.31	5.33	5.36	5.39	5.41	5.44	5.47	5.50
70	5.52	5.55	5.58	5.61	5.64	5.67	5.71	5.74	5.77	5.81
80	5.84	5.88	5.92	5.95	5.99	6.04	6.08	6.13	6.18	6.23
90	6.28	6.34	6.41	6.48	6.55	6.64	6.75	6.88	7.05	7.33
-	0.0	0.1	0.2	0.3	0.4	0.5	0.6	0.7	0.8	0.9
99	7.33	7.37	7.41	7.46	7.51	7.58	7.65	7.75	7.88	8.09

الكالكيرية يترييه أفراسيانكانت فالوبالك كالتناقيه و

۰.

TABLE 6-2

RELATIONSHIP BETWEEN PERCENTAGES AND PROBITS

Probits are the three digit numbers in the table. Percents are read along the top and side margin of the table. The vertical column of percents gives the decade; the horizontal column gives the unit. The table entry appearing in the row of the decade value and the column of the unit value is the probit corresponding to that percent. The last two rows in the table provide a finer reading for very high percent, from 99.0 to 99.9. The second to last row is the tenths of percent to be added to 99%. The last row consists of the corresponding probits.

					Data From Which the Probit Equation Was Derived					
Equation	Type of	Variable	Parameters of Pr	obit Equations		Value of		Value of		Value of
Symbol	Injury or vamage		<u>Constanti a</u>	Slope: b	1 Affected	Variable_	3 Affected	Variable	S Affected	Variable
EI	Death from Lung Hemorrhage	Pp	-77.1	6.91	1 10	1.00x105 1.20x10	50 90	1.41x105 1.76x10 ⁵	99	2.00x10 ⁵
62	Death from Impact	J	-46.1	4.82	0 8	18.0x10 ³ 28.6x10 ³	31 63	37.3x10 ³ 45.2x10 ³	96 100	49.7x10 ³ 60.7x10 ³
E3	Eardrum Rupture	₽ _₽	-15.6	1.93	1	16.5x10 ³ 19.3x10 ³	50 90	43.5x10 ³ 84.3x10 ³		
E4	Injuries from Impact	J	-39.1	4.45	1 50	13x103 20x10	90	28x10 ³		
85	Injuries from Flying Fragments	J	-27.1	4.26	1	1024	50	1877	99	3071
51	Structural Damage	₽ _₽	-23.8	2.92	1 50	6.2x10 ³ 20.7x10 ³	99	34.5x10 ³		
52	Glass Breakage	₽ _D	-18.1	2.79	1	1700	90	6200		
FI	Burn Deaths from Flash Fire	te ^{: 4/3} /10 ⁴	-14.9	2.56	1	1099 1073 1000	50 50 50	2417 2264 2210	99 99 99	7008 6546 6149
<u>م</u> ۵۱	Burn Deaths from Pool Curning	t1 ^{4/3} /10 ⁴	-14.9	2.56	1	1099 1073 1000	50 50 50	2417 2264 2210	99 99 99	7008 6546 6149
τı	NH ₃ Deaths	Σ τc ^{2.75}	-30.57	1.385	3 3 3	37.3 90.9 44.6	50 50 50	74.6 204.6 148.6	99 99	411.8 334.4
TI	Cl ₂ Deaths	5 TC ^{2.75}	-17.1	1.69	3 3 3	14.1x104 17.0x104 21.5x10	50 50 50	34.05x10 ⁴ 47.0x10 ⁴ 64.7x10 ⁴	97 97	105.8x10 ⁴ 129.4x10 ⁴
T2	Cl ₂ Injuries	с	-2.40	2.90	1 25	6 10	50 90	13 20		

KEY: $P_p = peak overpressure (N/m²)$

rp = peak overpressure (n/m / J = impulse (N-s/m²) te = effective time duration (s) Ie = effective radiation intensity (J/m²/s) C = concentration (ppm) T = time interval (minutes) t = time duration of pool burning (s) I = radiation intensity from pool burning (J/m²/s)

Paul Ann in

TABLE 6-3

SUMMARY OF PROBIT EQUATIONS BASED ON INFORMATION IN APPENDIX D

1

A rate i to

is a measure of percent. It is desirable that the causative factor and response be related by a Gaussian function, because such extensive theoretical work has been performed on this distribution that the statistical treatment of such distributions reduced to the use of standard methods. For all probit equations used in the VM, the logarithm provided exceptionally good fits to the data used. يا للت أيتهما الجل القرر

.

.

-

Table 6-3 summarizes the probit equations used for Phase II assessment. The first column of this table gives the equation symbol that corresponds to the function given in Table 6-1. The second column of Table 6-3 specifies the type of injury or damage for which the equation is used, and the third column specifies the types and unit(s) of measurement of the variable that affects the vulnerable resource. The next two columns give the location and slope parameters (a and b) of the corresponding probit equation. The last six columns list the data used to compute a and b, the data being given in pairs. Each pair consists of the percent of vulnerable resource affected and the magnitude of the variable that causes this percent to be affected. The above data used for estimating the coefficients a and b for equation El, for example, are listed in the first two rows of the last six columns of Table 6-3.

In applying percent damage, whether derived from a probit equation or otherwise, to the vulnerable resource "people," it has been assumed that half of the total population is unsheltered (outdoors) and thereby subject to damage. Half the population is assumed to be sheltered (indoors) and no deaths or injuries are assessed for this portion of the population, because more complex models are required to assess injuries to people indoors. The VM does not, at this time, attempt to determine the movements and locations of the population as a function of time of day, day of week, and weather conditions, so half the population was arbitrarily placed outdoors. Furthermore, census data have been used to estimate the population distribution in the region of interest. As a consequence, people are modeled to be at their place of residence rather than at work, school, recreation, or in transit.

Double Counting

Provisions have been made in the VM to prevent double counting in three different situations. Double counting is used in this context to mean the inclusion of an element of some vulnerable resource (e.g., a person or a building) in more than one category of damage or injury. Three situations arise in which double counting will occur unless provisions are made to prevent it. The situations are as follows.

- (1) A single damage mechanism from one event simultaneously causes injuries of differing severity (e.g., inhalation of toxic gas may cause death, nonlethal injury, or irritation).
- (2) Two or more damage mechanisms from one event simultaneously cause injuries of the same severity (e.g., an explosion can kill people either by direct blast effects or by impact).
- (3) Different events at different times both cause damage to the same resource, and the first event so severly damages some portion of the resource that further damage is irrelevant (e.g., persons killed by toxic gas cannot be further injured by a subsequent explosion).

The corrections made to damage assessment for double counting and the means by which these corrections are obtained are described in depth in Appendix G. The type of correction used for situation (1) is applied to values resulting from the sets of damage functions denoted by: (E1, E3), (E2, E4), (S1, S2), (T1, T2, T3). The type of correction used for situation (2) is applied to values resulting from the sets of damage functions giving death and injury resulting from an explosion. The type of correction used for situation (3), by the very nature of the correction, is not applied to any particular set of damage estimates; instead this type of correction is used at the termination of the simulation to arrive at the summary of damages. This correction is used for the vulnerable resource "people" when either flash fire or explosion follows inhalation of toxic gases; this correction is used for the vulnerable resources "structures" and "people" when pool burning follows either flash fire or explosion.

Damage Assessment Procedures

. .

1

.....

1 8 · · · ·

E.

!

As it currently stands, the VM can simulate damage to vulnerable resources from four physical events; these events are:

- (1) air dispersion of a toxic gas
- (2) flash fire
- (3) explosion
- (4) pool burning

Damage to vulnerable resources is conveniently discussed in terms of (1) toxic injury, (2) explosion damage, and (3) fire damage. Detailed consideration is given to toxic damage assessment in Appendix E and to fire and explosion damage assessment in Appendix D. The assessment algorithms for toxic damage are highly dependent on the type of substance spilled; whereas the assessment algorithms for fire and explosion are independent of the type substance spilled, although the values of the variables used in the algorithm do depend on the type of substance spilled.

(1) Toxic Injury

Toxic injury is assessed only for the vulnerable resource "people."

At the present time, only inhalation toxicity is treated in the VM. Toxic injury caused by ingestion of poisonous substances is not treated. One difficulty encountered in attempting to model damage caused by ingestion of toxic material is that the amount of toxic substance ingested by a receptor is usually extremely difficult to estimate. The current treatment of inhalation toxicity is restricted to the substances directly spilled; modeling of the inhalation toxicity of combustion products or of other reaction products has been deferred. Of the five substances currently treated in the VM, only chlorine (Cl₂) and anhydrous ammonia (NH₃) are considered to have an inhalation toxicity liable to cause serious consequences. The toxic damage caused by irritant gases in general falls into three categories: (1) death

.....

- (2) sublethal injury
- (3) irritation

The category of injury sustained by exposed resources depends, in general, upon both the duration of exposure and the concentration level experienced. This dependence is nonlinear; dose, the product of concentration level and duration, is not the appropriate variable to assess response to irritant gases. As an example, for concentrations over the lethality threshold, doubling the concentration level does not halve the time required to produce the same death rate; instead, as the concentration level increases, the time to produce a given injury level decreases at a disproportionately rapid rate. This phenomenon is illustrated by the isodamage curves for chlorine lethality shown in Figure 6-2. स्वर्थका स्वर्थका विद्यालय के स्वर्थका
The dependence of toxic gas lethality on concentration and time was found to be described by a nonlinear function of the form

TCn

where C = concentration of the toxic gas

T = time duration of the exposure

n • an exponent

For both chlorine and ammonia, the best value for the exponent, as determined by fitting the data given in Table 6-3, was found to be n = 2.75.

In the VM the concentration is not constant in time, therefore the function TC^n must be replaced by the quantity

∫ CⁿdT

As an approximation to this integral, the VM uses a finite sum; i.e.,

$$\int C^{n} dT \stackrel{\simeq}{=} \sum_{i} T_{i} \cdot C_{i}^{2.75}$$

where T_i is the duration of a time step and

 C_1 is the concentration during that time step at a given location.

Table 6-4 summarizes basic data on chlorine and ammonia inhalation taken from Appendix E. These data were used to generate the probit equations for lethal injuries from the inhalation of the toxic vapors of chlorine and ammonia. The midpoints of the rectangular areas defined by the data were used in the generation of probit equations T1 for chlorine and ammonia. The resulting equations for lethality are:

for chlorine,
$$Pr = -17.1 + 1.69 \log_e V$$
 (6-2a)

and





SELECTED CONTOUR LINES FOR CHLORINE LETHALITY EQUATION (6-2a)

-most in setting

and the address of the second s

	Deaths (%) General High-Risk			Concentr	ation (ppm)
Effect	Population	Population *	Time	Cl ₂	NH3
Negligible	0	0	Any	< 3	< 100
Complaint, no risk	0	0	Any	3-5	100-300
Severe harassment (some risk)	0	25	Any	5-15	300-1000
Severe harassment/risk Lethal Lethal	0 3 50	25 50 100	< 1/2 hr. 1/2-1 hr. 1-2 hr.	15-25	1000-2500
Lethal Lethal Lethal	3 50 97	50 100 100	< 1/2 hr. 1/2-1 hr. 1-2 hr.	25-40	2500-4000
Lethal Lethal Lethal	3 50 97	50 100 100	< 5 min. 5-15 min. 15-30 min.	> 40	> 4000

TABLE 6-4. DATA USED IN DERIVING EQUATIONS FOR ESTIMATING DEATHS FROM CHLORINE AND ANHYDROUS AMMONIA GAS (Details of the data and their sources are given in Appendix E)

* As explained in Appendix E, the high-risk population consists of the very old, the very young, and those with preexisting pulmonary pathology. where

 $V = \sum_{i} T_{i} c_{i}^{2.75}$

Nonlethal injury is taken to mean hospitalization with or without evidence of permanent or lasting impairment. Of major concern in public disasters is the number of people whom the medical services send to hospitals. This number is interpreted as the count of nonlethal injuries for VM purposes, regardless of actual impairment.

Reports of three chlorine accidents [31-34], summarized in Table 6-5, were the primary sources of data in deriving the VM equation for estimating nonlethal injuries from chlorine. The data in Table 6-5 and the dose response considerations discussed in Appendix E can be summarized as presented in Table 6-6. Nonlethal injury can be correlated with concentration level alone, rather than with the nonlinear function of time and concentration, V, that correlates with death from chlorine inhalation. The probit equation for nonlethal injury resulting from inhalation of chlorine is

 $Pr = -2.40 + 2.90 \log_{e} C$

(6-4)

Since nonlethal injury from chlorine inhalation does not heal immediately (as discussed in Appendix E, it may heal spontaneously after a few days), the appropriate value for concentration in equation (6-4) is not the current level, but the maximum level that occurs in the given cell up to the current time.

Nonlethal injuries from the inhalation of ammonia (NH₃) are not assessed by the VM at this time. There seems to be some disagreement among the authorities as to whether hospitalization is appropriate after acute but nonlethal exposure to ammonia vapors. The studies relevant to this problem are few, so it has been impossible within the scope of this study (investigation of published data - no laboratory work) to obtain good estimates of the dosage required to cause injuries which require hospitalization. Therefore, the function T2 giving percent nonlethal injury from ammonia inhalation is simply:

percent injured = 0

- [31] Kowitz, T.A., R.C. Reba, R.T. Parker, and W.S. Spicer, Jr. Effects of chlorine gas upon respiratory function. Arch. Environ. Health <u>14</u>:545-558, 1967.
- [32] Chasis, H., J.A. Zapp, J.H. Whittenberger, J.L. Helm, J.J. Doheny, and C.M. MacLeod. Chlorine accident in Brooklyn. Occup. Med. <u>4</u>:152-176, 1947.
- [33] Joyner, R.E., and E.G. Durel. Accidental liquid chlorine spill in a rural community. J. Occup. Med. 4:152-154, 1962.
- [33] Weill, H., R. George, M. Schwarz, and M. Ziskind. Late evaluation of pulmonary function after acute exposure to chlorine gas. Am. Rev. Resp. Dis. <u>99</u>:374-379, 1969.

87

(6-3)

to see a commendation

والمرابع المرامع والمرابع والمستعمل والمحالية والمحالي

TABLE 6-5.	SUMMARY OF NONLETHAL	INJURIES	RECEIVED	BY	ACCIDENTAL	INHALATION
	OF CHLORINE					

	N	Hospitalized		
Accident	Exposed	Examined	Hospitalized	Examined)
Industrial Transportation ⁸	150	59	11	18
Brooklyn Subway ^b	1000	418	208	50
Morganza, Louisiana ^C	Unknown	100	17	17

^aSee reference [31] in the text.

^bSee reference [32] in the text.

^CSee references [33] and [34] in the text.

TABLE 6-6.	SUMMARY	DATA	FOR	NONLETHAL	INJURIES
	FROM CHI	LORINE	INH	ALATION	

Cl2 (ppm)	Injuries (%)
20	90
13	50
10	25
6	1
	1

Toxic irritation is treated as an all-or-nothing case for assessment purposes. If the concentration equals or exceeds the specified concentration, every person outdoors is assessed to be irritated. Irritation is assumed to cease as soon as the concentration drops below the specified value. From the data in Appendix E, the threshold for irritation by chlorine is taken to be 3 ppm, and for ammonia it is taken to be 100 ppm. Therefore, the assessment functions T3 for chlorine and ammonia may be written as ينفظكمك فالجاف سرا

and an a surface

"Inc.

and he had

percent irritated = $\begin{cases} 100\% \text{ if } C > C_t \\ 0\% \text{ if } C < C_t \end{cases}$

where $C_t = (3 \text{ ppm for chlorine and} (10 \text{ ppm for ammonia.})$

Toxic damage to the vulnerable resource "environment" is not assessed in the VM at this time. The VM predicts the concentration of a toxic substance in the air and water; comparison of these predicted concentrations to air and water quality standards appears to be an attractive method for assessing damage to the environment. Such an approach, however, has many pitfalls; among these are the following.

- (a) This scheme cannot be implemented for many cargoes. Many substances carried in bulk in commerce upon the navigable waters are uncommon as pollutants, and there are no standards for these materials.
- (b) For some cargoes, there may be more than one standard. For water contamination, for example, there might be one standard for drinking water, another for water safe for swimming, a third standard for the protection of fish, and a fourth level, above which shellfish living in these waters are unfit for human consumption.
- (c) Because most air and water quality standards are developed for situations of chronic exposure there is a problem of exposure time. Most air and water quality standards explicitly mention the averaging time and are designed for situations in which the concentration does not change by orders of magnitude in a few minutes. One may easily conceive a situation in which a very high concentration, for example of SO₂, may exist for a few minutes, killing all of the animals at a location without exceeding the three-hour air quality standard. Although the cumulative dosage at a location may be calculated, there is little information on the response of many plants and animals to short exposures of high concentrations of many common cargoes.
- (d) Some quality standards are not simply stated as concentration levels or dose levels not to be exceeded. Other factors may be involved. For example, one water quality standard for ammonia depends on concentration, water temperature, and water pH. Quality standards for other materials involve complex and expensive bioassay procedures.

Because of the problems raised by these factors, the violation of air or water quality standards has not been adopted as a viable procedure for assessment of environmental damage in the VM. Since the USCG has directed that environmental damage assessment is of lower priority than assessment of damage to people and property, no alternative method for assessing environmental damage has been implemented in the VM. However, the VM does provide the time history of pollutant concentration in air and water so the user may use this information to assess environmental damage at his option. The shift of the second

ģ

भाषता को की मिन्द्री होते हैं कि बिनिय की बिनियम का सम्मन्द्र के मिनिय के प्रति कि
(2) Explosion Damage

مان، «المالية»، مقدمة مطالبة المعادية من من الله في المارية في المارية في المارية من من من من من من المارية في

•

1.2.4

Explosion damage is assessed for the vulnerable resources "people" and "structures." Personnel experience explosion damage in two categories, (a) death and (b) nonlethal injury. It is customary to categorize explosion damage to personnel in three categories, depending on the causative mechanism of damage:

- (a) primary damage direct blast effects (interaction between the blast wave and personnel only, with no other intervening or associated factors)
- (b) secondary damage damage from missiles and fragments
- (c) tertiary damage damage from translation and subsequent collision with an obstacle

In the VM, death is assessed for primary damage manifested as lung hemmorhage or for tertiary damage manifested as skull and body bone fractures. Nonlethal injuries are assessed for all three damage categories, including the secondary damage of puncture wounds from missile penetration. In addition, injury resulting from two or more damage mechanisms is assessed in a separate category, multiple injury. Explosion may cause eardrum rupture, bone fracture, or puncture wounds through the independent mechanisms of direct blast, impact, or flying fragments. Since the causative mechanisms are independent, an individual exposed to an explosion may experience injury from a combination of two or even three causes. That fraction of the population injured by more than one mechanism is determined by the double counting procedures described in Appendix G. Structures experience explosion damage in two categories, (a) serious structural damage and (b) window glass breakage. The physical variables that determine the extent of explosion damage are the peak overpressure and the impulse associated with the blast wave; the values for these variables are generated in Phase I of the VM. Table 6-3 summarizes the probit equations used to assess explosion damage and the data upon which the equations are based; Appendix D explains at length the formulation of the assessment procedures.

Death from direct blast damage usually is due to lung hemorrhage, although other injuries may sometimes contribute. The data in Appendix D indicate that a peak overpressure of about 1 atmosphere (= 10^5 N/m^2 = 14.5 psi) is the threshold for fatal injuries, and that 2 atmospheres causes close to 100% fatality. The probit equation for this type of injury, El, is given by:

$$Pr = -77.1 + 6.91 \log_{10} (P_{2})$$
(6-5)

where P_{D} is the peak overpressure.

Death from tertiary explosion damage results when a person is moved by the blast wave and forcibly impacts with the ground, a wall, or some other object. The speed attained by a person subject to a blast wave is more dependent upon the impulse (roughly the integral of overpressure over time for the blast wave) than upon the overpressure associated with the explosion. Consequently, impulse is the variable used to make this assessment. The nature of the impact injury is complicated by the need to consider the person's position when struck by the blast wave, shielding by objects such as walls or buildings, and the distance to, and the nature of, any surfaces which the person might strike. As explained in Appendix D, data prepared for the Defense Civil Preparedness Agency and based on a model considering most of these factors [35] have been used to derive the probit equation E2:

$$Pr = -46.1 + 4.82 \log_{e} J$$

(6-6)

(6-7)

فعطفا كملك ومطرعه ومرافع

مين موقع معاملة المالية المراجع على المراجع على المراجع على المراجع المراجع المراجع المراجع معامل المراجع المرا المراجع
under handen and

متل ليك**الم**ير هي

where J is the impulse.

As discussed in Appendix D, eardrum rupture is by far the most prevalent injury which is an effect of direct blast. This mechanism operates in a straightforward manner, and there is a general consensus on the pressures required to cause damage. The equation E3, used to assess this type of injury, is:

$$Pr = -15.6 + 1.93 \log_{P} Pp$$

The injuries caused by translation followed by abrupt impact upon some surface are mostly lacerations, contusions, broken bones and internal injuries. Damages caused by this mechanism are difficult to assess, as discussed above. For the assessment procedure in the VM, damage criteria established for an idealized model were adjusted to account for nonideal cifects, on the basis of the relationship of ideal and nonideal criteria for death by this mechanism. This is explained more fully in Appendix D. The equation E4, used to assess this injury, is given by:

$$Pr = -3.91 + 4.45 \log_{0.05} J$$

(6-8)

For most accidental explosions, penetration by flying fragments is the most common mechanism which causes nonfatal injuries. Most of the injuries which result are not serious. More injuries are caused by broken glass than by other material, such as gravel. The number of people injured from fragments is very difficult to treat accurately, because the location and exposure of the population are important and the availability of material to break and form fragments also enters into the assessment. At present, a simplified procedure, based on damage from ten-gram glass fragments, is used.

^[35] Longinow, A., G. Ojdrovich, L. Bertram, and A. Wiedermann. People Survivability in a Direct Effects Environment and Related Topics. IIT Research Institute, Chicago, May 1973.

The resulting equation, E5, is given by:

$$Pr = -27.1 + 4.26 \log_{-} J$$

Assessment of injuries from multiple causes is based on the assessment of injury from each individual cause; the procedure used to eliminate double counting in making the assessment is described in detail in Appendix G.

For the purpose of assessing major damage to buildings, a typical wood frame structure has been assumed. Most residential structures are of this type. Since the census records used to provide data on the number and value of property include only residential property, it is consistent to use damage criteria for wood frame structures. If further development of the VM provides an expanded data base that includes nonresidential structures as well as a designation of structural type, the assessment algorithms may be readily modified on the basis of data available in the literature to encompass damage to structures other than wood frame. The response of the structure has been generalized, so the damage assessment is based only on blast wave parameters. Because complex interreactions between the blast wave and structure are not considered here, the blast wave is parameterized solely on the basis of peak overpressure. The difficulty of precisely describing structure-blast wave interactions and the expediency of using a single blast wave parameter are discussed in more detail in Appendix D. The assessment equation used for major structural damage, S1, is given by:

 $Pr = -23.8 + 2.92 \log_e P_p$

This probit equation is used to relate the given peak overpressure to the percent structural damage caused to a building subject to the given peak overpressure; it does not give the percent of buildings in the geographical cell experiencing complete destruction. Of course, as far as the dollar value of the damage is concerned it is irrelevant whether x percent of the buildings in a cell are completely destroyed or whether all buildings in the cell experience x percent damage. Window breakage is a much simpler phenomenon than other structural response to a blast wave environment. Therefore, it is generally agreed that peak overpressure is the significant causative factor; further, the critical levels required to cause given degrees of damage are generally agreed upon. The assessment equation for glass breakage, S2, is given by:

 $Pr = -1.81 + 2.79 \log_e P_p$ (6-11)

This probit equation gives the percent of exposed windows that are broken by the blast. At this time, no assessment of the dollar value of the breakage is made.

(3) Fire Damage

Damages from fire are modeled as affecting the vulnerable resources "people" and "structures." Damage from flash fire is currently modeled to affect both personnel and structures. The current computer version of the VM assesses damage from pool burning only to structures, since it has been

(6-9)

and the second

. status and all all all and the

(6-10)

assumed that personnel would have time to seek shelter or to evacuate in order to avoid injury. However, injury to personnel can readily be modeled by using the same damage criteria (as expressed in a probit equation) as are used for flash fire; the only difference is that actual radiation intensity and duration are used for pool burning, whereas effective values are used for flash fire. The damage assessed to structures is ignition. The damages assessed to personnel are (a) death and (b) nonlethal For all types of damages, two parameters have been found to be burns. significant: (a) level of thermal radiation and (b) duration of the thermal radiation; therefore a variable combining these two parameters is used for assessment purposes. The level of thermal radiation and its duration are computed by the Phase I submodels, for pool burning and flash fire, as is discussed in Chapter 4. The assessment of deaths from flash fire is based on data obtained primarily from studies of the effects of nuclear weapons. These data, presented in Appendix D, indicate that the number of deaths is proportional to the product of the duration of the radiation pulse with the four-thirds power of the intensity. The radiation pulse from a flash fire is not a square wave, of course, so an effective pulse intensity and an effective pulse durat or must be calculated. This is discussed in Chapter 4. The data relating the lethality levels to the radiation dosage are presented in Appendix D. The probit equation Fl, used to assess deaths from flash fire, is given by:

 $Pr = -14.9 + 2.56 \log_{e} \left(\frac{t I^{4/3}}{10^{4}} \right)$ (6-12)

where t is the effective time duration in seconds and I is the effective radiation intensity in $J/m^2/sec$. The factor 10^4 is a convenient scaling constant.

The VM does not make a quantitative assessment of nonlethal burn injuries, because of the uncertainty in determining the degree of the burns and the effects of clothing and shielding. The data in Appendix D indicate that the threshold for first-degree burns is

 $tI^{1.15} = 550,000$

in a strandined of a far second

•

7

i.

Ē

i

where t is in seconds and I is in $joules/m^2/sec$. In the analysis of first-degree burns it was found that an exponent of 1.15 provided a better fit to the data than the exponent 4/3 used for lethality (see page D-24). This is for exposed skin. The function F2, used to assess nonlethal injury from flash fire, is given by:

percent subject to first-degree burns on exposed skin

$$= \begin{cases} 100\% & \text{for } tI^{1.15} > 550,000 \\ 0\% & \text{for } tI^{1.15} < 550,000 \end{cases}$$
(6-13)

The assessment of fire damage to structures is based on studies of the ignition of wood. Factors influencing wood ignition are: (a) radiation intensity level, (b) duration of radiation exposure, (c) wood type, and (d) the presence or absence of a pilot flame near the irradiated wood. Duration and level of radiation intensity are factors computed by Phase I submodels. Wood type is not treated explicitly; average values are used. For flash fire, the presence of a pilot flame is assumed; for pool burning, pilot flames are assumed to be absent. For flash fire, the radiation intensity is considered to be high enough to cause ignition only in that region where a flammable fuel-air mixture exists, i.e., the region where the flash fire burns. The presence of the flash fire provides an open flame, so the data for pilot ignition are used. For pool burning, the irradiated structures are generally too far from the burning pool for those flames to be considered as a pilot. Thus the data for spontaneous (no pilot) ignition are used for pool burning. The damage assessment functions and procedures for ignition of structures from pool burning, Bl, and from flash fire, F3, may thus be stated.

For ignition from pool burning

1. For every grid cell, look up the radiation intensity, $\mathbf{I}_{\mathrm{r}},$ at the cell center.

2. The radiation intensity at the cell center must exceed the value

$$I_{s} = 2.54 \times 10^{4} \frac{\text{Joules}}{\text{m}^{2} - \text{s}}$$

3. The duration of the pool burning, t_b , must exceed the time given by

$$t_{B} = \left[\frac{6.10 \times 10^{4}}{(I_{r} - I_{B})^{4}}\right]^{5/4}$$
(6-14)

4. If $t_b \ge t_s$, then there is ignition

If $t_b < t_s$, then there is <u>no</u> ignition

where in both cases

 I_r = radiation intensity at the cell center

For ignition from flash fire

1. The vapor concentration at grid cell center must be between the limits of flammapility for the spilled substance (this to assure the presence of a pilot flame).

2. The radiation intensity, I_r , must exceed the value

$$I_p = 1.34 \times 10^4 \frac{\text{Joules}}{\text{m}^2 - 8}$$

3. The effective duration of the radiation, t_{eff} , must exceed the value given by:

$$t_{p} = \left[\frac{7.22 \times 10^{5} \frac{J}{m^{2} - s}}{(I_{r} - I_{p})}\right]^{3/2}$$
(6-15)

4. If $t_{eff} \ge t_p$, then there is ignition

If $t_{eff} < t_p$, then there is <u>no</u> ignition

If ignition occurs (pool burning or flash fire), one-fourth of the structures in the cell are assumed to ignite. This assumption, recognized to be somewhat crude, was made to account for the fact that some structures will be shielded from thermal radiation by others. Line and shirt for the second

يتعارضهم والمتعارضة والمتحكمة والمتحمل والمتعارضة والمتعارضة والمتعارضة والمحافظ والمحافظ

An Example

To illustrate some of the assessment techniques discussed in the preceding section, consider the following hypothetical occurrence. The dispersion of a chlorine spill causes the air in a given grid cell to experience the following time history:

Phase I output for grid cell x

Concentration (ppm)	Time (min) at <u>this Concentration</u>
0	5
10	10
30	5
25	15

For these data, the significant assessment variable

$$V = \sum_{i} T_{i} C_{i}^{2.75}$$

has the value V = 168,100, $\log_e V = 12.03$, so the probit has the value, Pr = 3.23, as determined through equation (6-2a). From Table 6-4 we see that a probit of 3.25 corresponds to 4%. Thus we conclude that for this hypothetical exposure history just under 4% of the exposed population in the given cell will be killed by inhalation of toxic vapors.

During the last time interval the concentration is 25 ppm. Since concentration is the appropriate assessment variable for nonlethal injury from chlorine inhalation, we find $\log_e C = 3.22$; using this value in equation (6-4) we find the needed probit, $\Pr = 6.93$. Again, from Table 6-4, it is seen that probits of 6.88 and 7.05 correspond to 97% and 98%, respectively. Therefore we assess 97% of the exposed population in the given cell as having nonlethal toxic injury. However, since the concentration during the preceding time step was the highest experienced in this cell (30 ppm), it is that value that should be used to assess nonlethal injury. For C = 30 ppm, $\log_e C = 3.40$ and $\Pr = 7.46$. Thus 99.3% of the exposed population received nonlethal toxic injury.

Since the concentration during the last time interval exceeds 3 ppm, 100% of the exposed population is also irritated.

Thus the assessment of toxic injury for the last time interval reade:

Dead	-	42
Nonlethal Injury	-	99.3%
Irritation		100%

Correcting for situation (1) double counting, we obtain:

Dead	-	47	-	42
Nonlethal Injury	-	(99.3 - 4)%	-	95.3%
Irritation	-	[100 - (95.3 + 4)]	=	0.72

بجنفا المتنارة سيستري بمتسا

CHAPTER 7

كالشعالية الإليانية وتناسمتهم فالمساوسات

منكالهم مساريه المعماما معروفت اللهم والمتقار تمسارك ممسمو فتنظفهم

and the second of a second second second

EXAMPLES OF COMPUTER RUNS FOR AN URBAN AREA

Introduction

1.10

3

This chapter presents the results of a number of computer runs for the five cargoes used to exercise the model. The four spill sizes for each of these cargoes used in the test computer runs are shown in Table 7-1. The very large, large, and small sizes of spills were specified by the Risk Analysis Advisory Board of the U. S. Coast Guard. The medium-sized spill was chosen by ECI to be close to the geometric mean of the small spill and the large spill, to provide a spill size intermediate between these two widely different cases. Since the very large spill differs from the large spill by a factor of only two in most cases, the computer test runs used the large, medium, and small sizes. The large and very large spills may be expected to have a significant effect on the general population (over and above those directly concerned with the cargo transport - the ships' crews, marine terminal personnel, and fire and rescue personnel).

Below, we first take up one computer run and present the results in detail, with copies of the actual computer printout. Next the runs for LNG, chlorine, and anhydrous ammonia are discussed. For these three cargoes, the effects of spills of different sizes are compared, and, in addition, one other input variable is varied. For LNG, the wind direction was changed slightly, for chlorine the stability class was changed, and for ammonia the time step was varied. Unless otherwise explicitly stated, in all the test runs that follow, the wind is blowing toward 53° at 4 m/s, and the time step is 2 minutes for cases where the effects of toxicity are being considered and 1 minute when an explosion or flash fire is expected. All spills are located at the center of cell 3, a river or harbor cell. The last computer runs discussed are those where pool burning immediately follows the spill. Medium-sized spills of LNG, methyl alcohol, and gasoline are treated in this way. The spills of methyl alcohol and gasoline are very difficult to ignite because their evaporation rates are so slow. Finally, some important features of the VM, which have been delineated by these test runs, are discussed.

Before proceeding to the first computer run, a word about the demographic data and the grid system is in order. The demographic data from 40 census tracts have been used in these runs, and each census tract has been represented by a point. For cells which appear from the pattern of streets to have a uniform population density, the center of the cell has been approximated and used to represent this cell or tract. For cells which include a great deal of unpopulated area such as cemeteries, parks, marshes, lakes, etc., these areas have been excluded, and what appears to be the center-of-mass for the populated area has been chosen. For irregularly shaped tracts, however, the tract center has always been placed within the tract boundaries.

For these test runs, 50 cells were used, 10 river cells and 40 census tracts. The location of these cell centers is shown in Figure 7-1. These

	Very Large	Large	Medium	Small
LNG Liquified Natural Gas	2 ship tanks 50,000 m ³ 20,750 MT	l ship tank 25,000 m ³ 10,375 MT	200 m ³ 83 MT	500 gallons 2 m ³ 0.8 HT
Chlorine (Cl ₂)	l barge load 1,200 short tons 727 m ³ 1,090 MT	1 barge tank 300 short tons 182 m ³ 272 MT	10 m ³ 15 MT	1 short ton 0.60 m ³ 0.90 MT
Anhydrous Ammonia (liquid NH ₃)	2 ship tanks 20,000 m ³ 16,340 MT	l ship tank 10,000 m ³ 8,170 MT	300 m ³ 251 MT	10 m ³ 8.2 MT
Methanol (Methyl Alcohol) (CH ₃ OH)	2 ship tanks 30,000 m ³ 23,800 MT	l ship tank 15,000 m ³ 11,900 MT	120 m ³ 95 MT	300 gallons 1.10 m ³ 0.9 MT
Gasoline	2 ship tanks 4,000 long tons 5,820 m ³ 4,060 MT	1 ship tank 2,000 long tons 2,910 m ³ 2,030 MT	100 m ³ 70 mT	2 long tons 2.9 m ³ 2.0 MT

TABLE 7-1 SPILL SIZES (MT = METRIC TON = 10³ kg)

86

d. Contin Line

n an air air an Airth I can



RELATIVE LOCATION OF THE GRID CELLS FOR THE TEST RUN Cells 1 through 10 are river or harbor cells, which are assumed to contain no people or structures.

and the infinite instantion of the state of States of Mathematical States in the States of States and States

an Chini Attosherika

A CONTRACTOR OF THE OWNER OF THE OWNER.

data were obtained from census statistics for an actual city. In this study, we considered only one phase of a more complete risk analysis - it was assumed here that a spill of a certain size has occurred. A complete analysis would consider the chances of such a spill taking place. Factors such as the number of ships or barges carrying this sort of cargo, their capacity, the type of waters which they are navigating, the other traffic in these waters, the visibility, currents, tides, and so on must be considered in determining the probability that a certain spill will occur in a particular place over a given period. As this project was to consider only the assessment of damage after the spill had occurred, consideration of this probability was not addressed, and any realistic appraisal of a spill must take the probability of occurrence into account. Therefore any analysis of the damage that may result from the transportation of hazardous cargoes by ship or barge cannot be made on the basis of this study alone. In the development of this model, wherever possible, approximations were made in a manner such that the damage would be overestimated. Thus the resulting damage assessments, although credible, may be somewhat high. The injury and damage figures which follow must be interpreted in this light.

ALL CAR

- The Statistics

Ŧ

The Output of an LNG Run

Tables 7-2 a to j present the details of a computer run for a medium-sized spill of LNG. Following printing of the properties file (not reproduced), the next page of output, Table 7-2a, gives spill and environmental information. The first column in Table 7-2a contains the number which is associated with each variable, and which is necessary for understanding the computer program operations, and provides a convenient way to refer uniquely to each variable. The second column contains an abbreviation of the variable name or a brief description of the variable. The third column gives the value of this variable, the fourth column lists the units of the value given for the variable. Table 7-3 explains the designation printed to denote the source of the value for the variable.

The values of the variables are generally given in the CGS system. The letters N.A. mean that the units for that variable are not available in the file of units. If the units column is blank, then the variable has no units. The items listed under spill definition data are self-explanatory except for variable 2006 which is a flag which is set = 1 if adiabatic wall conditions in the tank are to be used and = 0 if isothermal wall conditions are to be used. Variable 2015 is the height of the hole above the water surface, and variable 2009 is the number of increments to be used in calculating the rate of spilling.

For the environmental data, variable 2058 gives the direction toward which the wind is blowing, in degrees. Variable 2017 is the stability or turbulence class, and variable 2014 is the height above the ground at which the concentrations are to be calculated. Variable 5022 is the ambient atmospheric pressure, and variable 5011 is the density of the water (a density different from 1.00 might occur due to salinity). Variable 2018 is a flag indicating whether the spill is restricted to a channel or is

LNG SPILL, MEDIUM, 200M3 = 83MT

-

- ----

SPILL DEFINITION DATA

1001	CHEM NAME	= L	NG			
2007	INITIAL MASS	=	0.830F 48	G	,	USER
2001	TANK VOLUME	2	0.2008 09	CM 3	•	USER
2002	TANK HEIGHT	Ξ	500.	C M	•	USEP
2004	TEME STAPT	=	-151.	С	•	USEP
2005	TANK PRESS	=	0.203F 07	G/CM2	•	USER
2006	ISD=0, ADB=1	=	1		•	USER
2003	HULF HEIGHT	=	100.	CM	,	USER
 2009	HOLE DIAM	=	50.0	CM	•	USER
 2015	HT OVER SHEE	ż	9.0	CM	•	USER
2009	NUM MASS INC	=	200		•	DEFAULT

ENVIRONMENTAL DATA

• • •

- .

i

2016	WIND SPEEC	z	400.	CM/S .	USER
2059	WEND TOWERD	=	50.0	DEG .	USER
2017	ATMOS COND	=	4	•	USFR
2014	CCOSD Z	=	100.	См	USER
2054	ATE TEMP	=	15.0	ċ ·	DEFAULT
5022	PRESSULE AVA	=	0.1015 07	G/C42 .	USER
2072	HATEP TEND	-	20.0	c .	USER
Sell	DENS WATER	=	1.00	G/CM3	USFR
2018	CHL=1, PtD=2	=	2	•••••••	USER
2020	CHAN'IL WINTH	=	0.6108 05	см .	USFR
2-147	STREAM VEL	=	100.	04/5	JEFAULT
2043	01E 00: E H20	Ξ	50.0	CM2/5 .	DEFAULT
2022	FL 0X + C 1 1 + L M2	=	1	•	USE8
2.148	TIDAL VEL	Ξ	50.0	CM/S .	DEEAULT
2049	TIUAL PERIOD	=	0.5185 06	s .	DEFAULT
21151	PHASE LAS	÷	0.0	Š,	DEFAULT
2051	DECTA COREE	-	0.0	•	DEELULT
2052	MARMING FACT	=	0.3000-01	сма .	DEFAULT
2453	DIF CREE V 4	=	0.1005 00	CM2/S .	DEFAULT

TABLE 7-2a

SPILL DEFINITION AND ENVIRONMENTAL DATA

radial in nature. Variable 2043 is the diffusion coefficient of the liquid cargo in water. Variable 2022 is a flag for cryogenic spills to indicate whether the flux of heat from the water to the liquid cargo pool is constant (flux var = 1) or is limited by the formation of ice (flux var = 2). The next five variables are used by the submodel that calculates the mixing and dilution in rivers and tidal estuaries. The Manning factor concerns the roughness of the river bottom. Variable 2053 is the diffusion coefficient of the cargo vapor in air and is estimated by a subroutine if it is not given by the user.

للكشأكر ويدرده

a the set of the set o

and the second state of the second
بالمحقمة القليم

Selected chemical and physical properties of the cargo are printed on the next page, Table 7-2b. Most of these variable names are sufficiently explanatory, but variables 1010, 1011, and 1012 are constants in the vapor pressure equation. The vapor pressure equation is of the form

$$P = 10^{[A - \frac{B}{(T+C)}]}$$

where P is the vapor pressure, T is the temperature, and A, B, C are the constants given, respectively, by 1010, 1011, and 1012.

Table 7-2c shows the page which displays the geographical data. The latitude and longitude columns originally contained the location of each cell center in degrees, minutes, and seconds, but in the sample page the degrees column has been omitted. The first column gives the cell number, by which all further reference to the cell is made. The second column contains identifying information about each cell. An "R" indicates a river or harbor cell. The other numbers may be block number, census tract numbers, or some other means of entering demographic information. The ignition code has been discussed in Chapter 4. Columns x and y give the location of the cell in a coordinate system in which the spill location is at the origin and the positive x axis is in the direction toward which the wind is blowing.

Table 7-2d shows the page containing spill and ignition information. Variables 4001, 4002, and 4003 give the amount of cargo spilled, and variable 4004 gives the time for the spill to take place. Variable 2038 is the average release rate, and variable 5027 gives the maximum dimension of the pool of liquid cargo on the water surface. This dimension is the radius for a circular pool and the upstream-downstream extent for a spill restricted by the channel (see variable 2018 in environmental data). Variable 5010 is a flag which is set = 0 if the puff (instantaneous source) air dispersion model is to be used and which is set = 1 if the plume (continuous source) model is to be used. If the user does not specify the value of the variable, the VM decides whether the puff model or the plume model is more appropriate.

The first six variables in the ignition data list are self-explanatory. Variables 5042 and 5043 give the effective values of the intensity and duration of the flash fire radiation. For an explosion, variable 5044 gives the mass of cargo vapor which exploded, variable 5045 gives the explosive yield in calories, and variable 5047 gives the yield equivalent in (short) tons of TNT. Variable 4019 is the mass of cargo remaining in the pool at the time of ignition. In this case, it is zero so that no pool burn can follow a flash fire or explosion. Thus the last four variables which concern the flame from the burning pool are zero as well.

LNG SPILL, MEDIUM, 200M3 = H3MT

CHEMICAL PROPERTIES DATA

İ

	1002	MOLEC WEIGHT	*	16.0			115 C 9
• • •	1002	ROLL TEM LTO	-	-161	ormete r	1	
	5012	ECCETING OT	-	-192	Č	•	
	1004	DENS I TO AVE	• -	0 400	C / C M A	•	1015 0000 MUD
	1004		-	0+407 () 1616-02	0-5/0-5	1	
	1005		-	0+1416-02	0-5/042	•	
	1007	HEAT CADC ID	-	0.1020-02		٠	
· · · · · · · · · · · · ·	- 1000	SHEE TENCION	-	14 (1	04676-0	•	
	10.00	DEM FILET VOD	-	1747	07 CM	•	
	1010	VOE COEFE A	-	6 61	070,43	1	
	1011		-	330		,	
	- 1011	VPE COLEE C	-	266		+	
	1013	HEAT CADE VE	-	200.	CAL / C - C	,	
· ····	1014	HEAT OF VOS	-	12)		,	- 731 PROP
	1019	FLAME TEMP	=	200.		,	
	1020	MOLE ERACTA	-		L	•	
	1021	ITO DENS B2	-	ሳ 4 2 4	C/CM2	•	CHM 0600
	5013	CONC. LO PCT	-	5 30	07(*)	,	
	5014	CONC. HP PCT	-	16 (1		,	
	5015	LT2/FUEL PAT	-	0.52		*	CONDUTED
	5023	CONC. LO LIM	-	1.3765-04	6./CV3	1	
	5024	CONC. HP LTM	-		6/643	•	COMPUTED
	5025	CONC. STOTCH	-	0.6805-06	G/CM3	•	
•••	5016	HEAT COMBUST	-			•	
	5017	ADELANE TEND	-	1 1 88F 04	c c	•	11557
• • •	5018	BURNING PATE	-	0.2085-01	64/6	•	10050
	5019	NOLES OYYGEN	=	2.30	C 47.5	•	11566
· · • • • •	5020	FLASHPOTHT	=	-161	c	•	1356
	5021	SPC HEAT WAT	-	1 40	C	•	11550
·· ·	5002	MISCINE IND	=	L • 7()		•	11566
	5003	PEACTIVE IND	=	1		•	
- .	5004	TOXICITY 150	=	ر ۱			USER
	- / / +	MINICI ACCA	-				U 3 🗆 🤊

TABLE 7-2b

SELECTED CHEMICAL AND PHYSICAL PROPERTIES OF THE CARGO

	1. COVD-1	CAL - 0273					
			tots is	501.1	15.10		10-11-60 60 0
			1917-12	26,000		MI10 MI30	
ſF	LL #	CELL IN	TON CODE	LATITUES	LONGITUDE	Х, КЧ	¥, K4
	1	441	ı)	5500	1540	-0.819	0.687
	2	P02	()	5800	1520	-0.410	0.344
	3	2113	Q.	5800	1500	U.()	U.0
	4	2414	13	5744	1420	0.424	-1.161
	5	°05	i)	5720	1340	0. 446	-2.321
	5	5 (1E	41	5640	1230	0.257	- 3.444
	7	S 1) 7	0	5620	1320	0.065	-4-085
	8	202	- ()	5550	1310	-0.325	- 4. 906
	9	G () 9	0	55:30	1300	-0.517	-5-611
	10	310	0	5520	1250	-0.511	-6.019
	11	201.00	-4	0100	0830	11.555	-7.446
	12	202.00	-4	-01.00	1020	9.307	-0.556
-	13	203.00	-4	0100	1200	7.258	1,163
	14	2114.00	-4	0130	1320	6.215	3.247
	15	203.00	-4	0200	1500	4.762	5.675
	16	206.00	- 4	0000	1600	1.153	1.869
	17	207.00	- 4	5840	1600	-0.435	1,977
	18	205.00	-4	5830	1530	-0.019	1.225
	19	209.00	-4	5830	1420	1.415	N-021
	20	210.00	-4	5900	1430	1.406	0.302
	21	211.00	-4	5741)	1430	2.599	1.848
	22	212.00	- 44	0014	1420	3,400	2.386
-	27	212.00		0010	1330	4.474	1,527
	24	214.00	-4	0100	1 300	5.030	2.194
	25	215.00	4	4434;	1300	5.435	1.485
	26	716.00	-4	0000	1200	6.068	- 0.250
	27	217.00	-/+	0030	1140	7.074	0.105
	28	218.00	- 4	0121	1100	7.674	-0.914
	29	219.00	-4	1000	1000	8.526	-2.318
	30	22(4	57411	1000	8.129	-2.7.12
	21	221.00	- 4	0070	0200	10.151	-2.376
	3,2	222.00	-/+	0020	0830	10.464	-3.155
	33	223.00	. 4	0030	0740	11.989	-4.015
	24	22+000		5740	0300	10.526	-4.354
	35	225.00	-4	51411	<u>0840</u>	9.767	-4.167
	3E	226.00	- 4	5900	0340	8.974	-5-113
	7 ٢	227.00	-4	5971	0910	8.756	-4.124
	39	223.00	•4	5171	0.940	8.142	-3.6(19
	39	229.00	- 4	5940	1010	6.734	-4.039
	40	230.00	-4	5130	1100	6.702	-1.998
	41	231,00	4	51-30	1100	5.512	-3.417
	47	732.00	- iş	5040	1200	4.481	-2.148
	43	2 13.00	-4	5+20	1220	4.954	-0.859
	44	234.00	- 4	5930	1230	4.859	-0.451
	45	235.04	. 4	51311	1330	3.629	0.550
	46	236.00	- 4	5200	1320	2.034	-0.129
	47	237.00	- 4	5340	1330	2.639	-0.692
	49	230.41	- 4	5920	1400	1.627	-0.559
	49	234.00	- 4	5720	1300	1.264	-2.772
	59	240.04	•••	50711	1220	1.234	-5-115

and the second second second second second second second second second second second second second second second

あったいいまたいたいたいに、 またないない あたいの時間であたいがある

....

-1

TABLE 7-2c

GEOGRAPHICAL DATA

LNG SPILL, MEDIUM, 20043 = 834T

.

•

SPILL OUTPUT DATE

~

		•••		•	• •		
	5009	SPILL CELL	±	٦			
	4001	TOT PASS GAS	=	0-1765 05	r	٠	CUMPUTED
. .	40112	TOT MASS LID	=	0-1.546 04	C C	•	CHAPOTED
	41113	TOT VOL LIC	=	0-15-6 00		•	COMPUTED
	4004	TIME OF PEL	Ξ	62 6		•	COMPUTED
	2039	AVG ESC HATE	-	1 1565 11	3	•	COMPUTED
	5027	110 255 74-16	-	1.1755 JA	5/5	•	COMPLITED
	4025	MAX DIA DIG	-	7. (7° J8	6	٠	COMPUTED
	4016	TINE LA CHAO	-	7.523 114	C.M	•	COMPUTED
··· •	4073	MASS VAD LLD	-	/ () • ¹ ;	S	•	CCMPUTED
	5.11.1	FLAC IS COL	=	0+654i. Ho	G	+	SYSTEM
		- r + 1 24 + C21	Ξ	() ()		•	USEK

IGNITION OUTPUT DATA

•

5					
511411	ILME AT IGA	=	339.	s.	COMPLITED
5907	IGMITA CELL	3	12		CONDUTED
500a	IGNITAL CODE	=	- 4	•	
5041	PUFF CENTER	Ξ	0.1355 05	C N 1	COMPUTED
502A	SIGY . DICORES	=	0.5155 00		COMPUTED
5029	SIGZASTSOFES	-	0.0125 04	64 .	COMPUTED
5042	FI FICE SADS	-	9.2931 (14	CM +	CUMPUTED
5047	FL FIFF AANA	=	2+97	CAL/CM2S,	CUMPUTED
5.4.6		=	4 . 54	s,	COMPUTED
5444	7/35 V50 0 XP	=	0.257= 119	G ,	COMPUTED
2043	OCT VID CXC	Ξ	39.2.	•	COMPUTED
5046	FXPL YICLD	=	0.6575 12	C 41	COMPLITED
5047	モヤエーデントレイ	=	557.	N. C.	CONDUTED
4012	VOL LTD POOL	=	0.0	C M 3	
2018	CHLEL, ETDER	:	,		50000014U
2019	100 018 2001	=	1.323. 44	- M 1	05-4
2020	CHADEL WITCH	=	1 6105 16		SYSTEM
4907	DTAM FLITE	-	1.010 02	1, M +	USER
40.08	ELANE ALCIN	-	·/•·/	(M ,	DEFAULT
4018		-	0.1	24D •	OFFAULT
504.3		=	7. 0	См ,	DEFAULT
2143	ar and the	Ŧ	() • · *	5,	りビデオリビエ

. •

TABLE 7-2d SPILL AND IGNITION DATA

105

.

			LNG SPILL. M	FOLUN. MONS	1.= 83MT					
VAPOP CONCI	ENTRATION.	KG/M3								
					CFLL NUMB	E SU				
	- 	~			~	•			6	<u>의</u>
20-1 -					5	0 0	د د 2 د	0.0		•••
	0.0	0,0	0-0		0.0	0-0	0.0			
2.00	0.0	0.0	0.0	0.0		0.0	0.0	0.0	0.0	
5.63	0.0	0.0	0.0	0.6	0.0	0.0	0.0	0.0	0.0	0-0
		1	1		CFLL NUMBI	× II				
TINE [N I N]		12	1	11	15	16	11	1.0	19	8
1.00	0.0	0.0	0.0	u•0		0.0	0.0	0.0	0.0	0.0
2.00 2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	••
2	0	0.0	0.0	0,0	0.0	0.0	0.0		0.0	0.0
00° •	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0-0	0-0
2°00				0.0	0 • f		0.0	0.0	0-989-04	0.0
5.63	0.0	0.0	0.0	0.0	0.0	0.0	0*0	0.0	0-4065-01	0-0
			-		CFLL NUMB	.			· · · · · · · · · · · · · · · · · · ·	
	12	≈ °		×,		1 1 1	22	58	<mark>م</mark> ار	8
										•••
0 - 4	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5.00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5.63	0.0	0.0	0.0	0.0	0-0	0.0	0-0	0.0	0.0	0.0
					CELL NUMBE	e				
TIME(41%)		~~			35	36	37	38	39	\$
1.00	0.0	0.0	0.0	0.0	0.0	0.4	0.0	0.0	0.0	0-0
(ð.~	0.0	U.U	0.0	0.0	2.0	0.0	0.0	0.0	0.0	0.0
8	0.0	0.0	0.0	0.0	0.0	0.0	••	0.0	0.0	• • •
00.4	0.0	0.0	0.0	0.0	9.0	0.0	0.0	0.0	0.0	0-0
2.00	0.0	0-0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5.63	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
					CELL NUMBE	ER				
TIME (4 N)	7	2		*	\$	40		48	\$	8
1.00	0.0	•••	0.0	0.0	n •0	0.0	0.0	0°0	0-9	0.0
3.2				0-0	0.0		0.0	0.0	0.0	•••
5.5		00	0.0	0.0	0.0	•••	0.0	0°0	3 (9 (•••
						2.0				
2 . .	> < • <) <	2 4	> <	24	> c > {) (20
200	>			2.2	2,2		2,2			

;

CONCENTRATIONS IN THE CELLS (AS COMPUTED AT THE CELL CENTERS) FOR EACH TIME STEP

TABLE 7-2e

					CELL NUMB	Es .				
THF(4[H)	-	2	•	•	~	•	~	•	6	9
1.00	0.0	0.0	0.0	0.0	0.0	0.0	0'0	0.0	0.0	0.0
2.00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0-0	0.0	0.0
8.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	9
4.00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8.5	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
1.01	0.0	0.0	0.0	0.0	0.0	6.0	0.0	0.0	0-0	9
					CELL NUMB					
אבולוטן	1	~	[]	*	15	91	-11		19	8
1.00	0.0	c.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
2.00 ⁻ 2	0.0	0.0	0.0		0.0	0.0	0.0	0.0	0.0	0.0
3.00	0.0	0.0	0.0	0.0	0	0.0	0.0	0.0	0.0	•
4.00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5-00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.100E 03	0.0
5.63	0.0	0.0	D.O	0.0	0.0	0*0	0*0	0.0	0.5986 05	0-0
					CELL NUMB	EA				
ME (MIN)	21	22	≈	*	\$2	26	22	2	\$2	2
00-1	0.0	0.0	0,0	0.0	0.0	0.0	0*0	0.0	0.0	0.0
	0	0			0.0	•••	0	0.0		
00.0	0.0	0.0	•••	0.0	0	•••	•••	•••	•••	
		0.0	0.0		.		0.0		0.0)
0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	
69.0		0.0	0.0	0.0	0.0	0.0	0.0	2.0	9.9	2
					CELL NUMBI					
ME (NIN)	16	~	3	ž	35	97	12		39	3
1.00	0.0	0.0	0.0	0.0	0.0	0.0	0-0	0.0	0.0	0.0
2.00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0
3.00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
00.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5.00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
2.63	0.0	0*0	0.0	0.0	0.0	0.0	0-0	0.0	0-0	0.0
					CELL NUMB	ER				
INE (N(N)	Ŧ	~	ţ	3	\$	99	4	Ş	\$	2
33	•••	0.0	0.0	0.0	0.0	0.0 0	0.0	0.0	0.0	• ••
8.2	0.0		3.0					0.0	0.0	
8.5	0.0	0.0	0.0	0.0	¢•0	0.0	0.0	0	0.0	0.0
8	••	0	0.0	0.0	0.0	0.0	0.0	0. 0	0.0	••
8.5	0.0	0.0	0.0	0-0			0.0			<
							> •>			2.2

. . .

CONCENTRATIONS IN THE CELLS (AS COMPUTED AT THE CELL CENTERS) FOR EACH TIME STEP

and the state of t

والأطار عشقانا

TABLE 7-2f

. •

107

Tables 7-2e and 7-2f list the concentrations in the cells (as computed at the cell centers) for each time step. In this case, the ignition subroutine has calculated that the cloud ignited at 5.63 minutes, so this time is listed in addition to the 1-minute time steps. Table 7-2e gives the concentrations in kg/m³, and Table 7-2f gives them in parts per million (by volume).

Table 7-2g lists the results of both the flash fire and the explosion. For these computer runs, the program was altered so that, if ignition occurred, the results of both were computed in the same run, which saved considerable computer time. The discance listed is the distance from the puff center at the time of ignition. Thermal radiation from a flash fire was calculated only for those cells in which the concentration was above the lower flammable limit. Since the vapor cloud ignited at cell 19, the first cell in its path in this example, there is a non-zero radiation value only for that cell. The liquid pool had completely evaporated prior to ignition, so there was no pool burning in this example. and the second second second second second second second second second second second second second second second

1

Tables 7-2h and 7-2i list the damage and injuries which resulted from the explosion, first for each cell and then a summary for all cells. Table 7-2j shows the analogous information for the flash fire case.

Results of Test Runs

The test runs for the five cargoes used to exercise the VM are listed in Table 7-4. There are six or more runs for LNG, ammonia, and chlorine, but since the clouds resulting from methyl alcohol and gasoline spills did not have high enough concentrations to ignite when they reached the first land cell downwind, for these two cargoes only the case where pool burning was initiated immediately after the spill was run. Methyl alcohol, gasoline, and LNG are not toxic, so the hazard from them is due to fire or explosion. Chlorine is toxic but incombustible, so the hazard from chlorine stems from inhalation only. Ammonia is both toxic and flammable, so all types of damage may result. The concentrations of ammonia which result in death in a minute or so may be considerably below the lower flammable limiting concentration. Ammonia inhalation will cause death in a minute or so at concentrations which are below the lower flammable limiting concentration, so for the cells on the x axis (directly in the path of the cloud) the people are killed by the toxic effects before the explosion occurs. After an explosion or flash fire, pool burning takes place if there is any liquid remaining in the pool. No injuries or damage to structures resulted from the burning pools because the pool in the center of a water cell was more than 1 km from the nearest land cell where the injuries and damage were assessed.

The results for the LNG, chlorine, and ammonia runs are discussed in the following sections, and then the three cases of immediate pool burning are taken up. Finally, a few points concerning certain features of the VM are discussed.

LNG Spills

The results of the simulations of six spills of LNG are shown in Table 7-5. For each of the three sizes of spills, the VM was run with the

LNG SPILL, HEDIUM, 200M3 . 83HT

• • • •

1.0

المكمل التمال يوانيه بالمسلقا مرر

design of the surface of the second

يستناه لناله

JANKING BURNING

IGNITION AT TIMELMINI . 4.63

States and the states of

	EXPLOSION	EXPLOSION	EXPLOSION	FLASH FIRE	POOL BURNING
CELL #	PEAK OVERPRESSURE	IMPULSE	DISTANCE	THERMAL RAD'N	THERMAL RAD'N
	N/#7	PI\$/MZ	<u>KN</u>	J/M25	J/M25
1	3371.	528.	2.28	1000 (PERIN) = U.U.	0.0 IEFF(MIN) # 0.0
	4437.	671.			Q_
3	5931.	864.	1.35	9.	0.
	536¢.	791.	1.49		
5	3172.	502.	2.38	0.	0.
6	1427.	264.	3.61	·····	0.
7	1003.	201.	4.28	0.	Ŏ .
3	1049.	192.	5.24	0.	· · ·
9	1071.	187.	5.91	J.	0.
10	\$53.	169.	6.30	0.	0.
11	620	116	19.50		0
12	652.	121.	7.98	0.	· ·
13	1033.	182.	6+02	0.	0.
14	1090.	190.	5.85	g.	0.
15		157.	6.62		0.
16	1217.	234.	3.87	0.	ů.
17	2551 •	433	2.66		0
18	4332.	653.	1.34	0.	0.
19	1235949.	11013		124458.	0
20	9092 -	1476.	1.01	0.	0.
21	3464	540.	2.23	0	0
22	1937.	336.	3.15	0.	J.
23	1559	209.	3.43	<u> </u>	0
24	1917.	189.	5.17	0.	0.
25		197	4.35	0	0
26	937.	185.	4.72	0.	0.
		196.			0
24	930.	100.	5.40	.	0.
		1 34/.	7. 54	0.	0
.10	129.	135.	7.33	0.	0.
		!!!?•	9.26	°•	0
22	620.	110.	10.12	0.	0.
					<u>×</u> ·
35	420	114	10.43	0.	0.
				^* -	
30	A 20.	116.	7+10	v .	.
			7 40		
19	850.	153	6.71	0. 0.	0.
	1133	196			
41	1120.	159.	5.19	0.	0.
	1272.				0
43	1620.	263.	3.62	0.	0.
44	1697.		3.54		
45	3219.	508	2.35	9.	9.
46	46 86 .	705			0
47	5630.	327.	1.47	<u>0</u> .	0.
49	16253.	1772.	0.62		9.
· •				**	~~
49	2399.	399-	2.82	0.	0.

TABLE 7-2g

RESULTS OF BOTH FLASH FIRE AND EXPLOSION



110

TABLE 7-2h

CELL BY CELL ASSESSMENT OF INJURY AND DAMAGE CAUSED BY EXPLOSION

المسمسة سأتقلقه ججادا كالاجر

. + 1. Alexandra and and and a state of the second and a state of the second and a state of the second and a second as
BLAST DEATHS (E1) IMPACT DEATHS (E2) ALAST INJURY (E3) IMPACT INJURY (E4) FRAGMENT INJURY (E4) MULTIPLE INJURY (EM) STRUCTURAL DMG (S1) GLASS BREAKAGE (S2) SUMMARY	X - PEOPLE - 6 IN OUT IN 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0UT 1N OUT 0 3 0 0 0 0	PEOPLE -		IN OUT	EOPLE - A	IN OUT	OPLE - 8
BLAST DEATHS (E1) IMPACT DEATHS (F2) ALAST INJURY (E3) IMPACT INJURY (E4) FRAGMENT INJURY (E4) STRUCTURAL DMG (S1) GLASS BREAKAGE (S2) SU4MARY 	IN DUT IN 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 10 001 0 0 0 0 0 0 0	<u>r in</u>	<u>0</u>	IN OUT	IN OUT	IN OUT	tii Out
SUMMARY					~ ~	A A	A A	
ALAST INJURY (E3) IMPACT INJURY (E4) FRAGMENT INJURY (E5) MULTIPLE INJURY(E5) STRUCTURAL DMG (S1) GLASS BREAKAGE (S2) SUMMARY		0 0	a 0	ŏ			Å	
INPACT INJURY (E4) FRAGMENT INJURY(E5) MULTIPLE INJURY(EN) STRUCTURAL DMG (SL) GLASS BREAKAGE (S2) SU4MARY	<u> </u>		5 0		0 2			
FRAGMENT INJURY(ES) MULTIPLE INJURY(EN) STRUCTURAL DMG (SL) GLASS BREAKAGE (S2)		0 0 0	<u>o</u>	<u> </u>		Q		
NULTIPLE INJURY(EN) STRUCTURAL DMG (SL) GLASS BREAKAGE (S2) SU4MARY	0 0 0	0 0 0) 0	1	0 41	0 1011	0 0	0 0
SUMMARY	0 0 0	0_0) 0	0	<u> </u>	029	00	
SUMMARY	STRUCTURES	•	STRUCTURES	•	- STA	UCTURES	STRU	CTURES
GLASS BREAKAGE (S2)	<u> </u>	0 0			30 4	19 8044 800		<u> </u>
SU4MARY	68 1368	0 84	920	ŏ	70 9	72 0	8 23	3
SU4MARY								
	LNG SPILL, M ALL CELLS	EDIUM, 200M3 -	8347					
BLAST DEATHS (EL)	PEOPLE 921					· · · · · · · · · · · · · · · · · · ·		
INPACT DEATHS (E2)	5							
1 WDACT IN 11124 (123)	<u>•1</u>							
FRAGMENT INJUNY(FS)	1031							
MULTIPLE INJURY(EM)	29							
		TOTAL DEAT	HS	921			<u></u>	
		TOTAL INJU	IPIES	1101				
	0 - STRUCTURE	5 - 5				<u> </u>		
STRUCTURAL DHG (S1)	9861	702 3200						
GLASS BREAKAGE (S2)	5071	0						
·	·····	TOTAL STRU	IC TURES	986				

منافقة استركار

CONTINUATION OF CELL BY CELL ASSESSMENT OF INJURY AND DAMAGE CAUSED BY EXPLOSION AND THE SUMMARY FOR ALL CELLS

and a second and a second and a second se

TIME = 5.6 MIN	CELL # 19		
	A - PEOPLE - #		
FL FIPE DEATHS (F1) FL FIRE BURNS (F2)	IN DUT IN 0 67 0 1ST_DEGREE BURNS	<u>617</u>	
	STRUCTURES		
FL FIRE IGNTN (F3)	25 140 22120	00	
•			
	LNG SPILL,	MEDIUM, 20083 . 83MT	
SUMMARY	ALL ČELLS		
FL FIRE DEATHS (E1)	PEOPLE		
		TOTAL INJURIES	0
	# - STPUCTUR	ES - 8	
FL FIRE IGNTN (F3)	140	2212000	
		TOTAL STRUCTURES TOTAL VALUE	140 2212000

1.1.1

:

÷

1.48-1.4

لطنيك قعاريه بمالقد ويج

î

ų

11. X 1. Z 11

م السالة (

Ę.

ļ

Jan Andrew Martine

TABLE 7-2j

CELL BY CELL ASSESSMENT OF INJURY AND DAMAGE CAUSED BY FLASH FIRE AND THE SUMMARY FOR ALL CELLS

and and and the second statements were stated and the second state of the second state of the second state of t

Hierarchical Source Code	Source Designation Printed	Source of Value
0		Value is missing in the Chemical Properties File.
1	DEFAULT	Value is taken from the default file.
2	EST PROP	Value taken from the Chemical Properties File is estimated.
3	CHM PROP	Value taken from the Chemical Properties File is an accepted value.
4	COMPUTED	Value is computed by the program.
5	USER	Value is supplied by the user.
6	SYSTEM	Value is computed by the program and will override a user-supplied code.

المتكرك والعراق أليان كالإنكارات والمرارين

1977-1824

A REAL AND A REAL PROPERTY OF A REAL PROPERTY.

TABLE 7-3

DESIGNATION DENOTING SOURCE OF VALUE FOR THE VARIABLE

The source designation printed indicates the source of the value printed out for a given variable. Each value has associated with it a hierarchical source code; for a given variable a value with a numerically higher source code will override (replace) a value with a numerically lower source code.

113

ſ				nitio:	of	est Cas	e									Result	5									
ł		Carlo Pa	rareters.	Invire	ment	al Para	neters	Int. Para.	To 1	gnition	Portion of			Exp	loston			Flash F	ine .		P00	Burnin	9	Tox	icity	
	Case	. <u>• • • •</u>	Acount In tank retric tons	Wird Speed r/s	bind Dir.	Stab. Class	Ign. Type	Tire Step: At min.	Cr Sim Time Time	End Of Wlation Distance	Evaporated Hass Condusting	Yield tons of ThT	Deaths	injuries Ø	Structures Damaged MS	Structures with Broken Windows #	Geaths	Injuries # Cells Assessed	Structures Ignited HS	Amount in Ppol	Flaue Height	Burning Tine cin.	Structures Ignited MS	Deaths	lajuries T	G.: e
		111 113 113 114 114	20,753 20,753 20,753 93 83		53 53 59 50	4 4 4 4	E F E F	1	4.2 4.2 4.2 5.6 5.6	1,010 1,010 1,010 1,350 1,350	12.1 12.1 12.1 39.2 39.2	40,700 41,200 41,200 657 657	984 1,238 921	77.542 77.753 1,101	129.5 131.1 17.0	29,490 29,628 5,871	534 617	1	2.2 2.2	7,590 7,200 7,200	455 456 456	1.1 1.0 1.0	0 0 0			11 12 13 14 15
		183 113 113 113 113 113	83 83 0.30 0.30 0.30 0.60		53 53 50 50 50 53	4 4 4 3	E F F E	1			1 MC 1 MC 1 MC 1 MC 1 MC															16 17 18 19 19
		UNE UNE	0.6J 23	4	53 53	•	F	;	:		DNI									43	18	0.4	9			H.
	y ;	Mothyl Alcungi) 		53	4	P	1				ļ								117	18	117	0			P 1
114	A) 12 13 11	Ationia Antonia Antonia Antonia Unia Ituonia	16.140 16.340 15.340 251 251		53 53 53 53 53 53 53	4	E F N E F	1 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	11.8 11.8 20 25 5	2,800 2,800 7,200 3,600 3,600	7.9 7.9 Dr.1 Dr.1	4,110	т	11,419	68.3	15,720	т	т	9.23	15,200 15,200	35 146 146	3.8 19.5 19.5	0 0	6.726 6.720 18.151 4.574 4.564	2.2.2.2.2 2.2	11 月 石 石 石 石
	• • • • •	i cona Posta Posta	251 8.2 8.2 8.2	4	53 53 53 53	4	N E F N	2 1 1 2	130 115 120	7,205 3,600 3,600 7,200	ON I ON I													6,033 5 0 0	, un u un u	
		Interne Interne Interne Interne	- 1.090 - 1.090 - 1.090 - 15	401115	53 53 53 53	4 2 4 3 3		2 2 2 2 2	84 84 90 84 90 84 90 84	7.200 7.200 7.200 7.200 7.200 7.200														13.679 18.112 9.955 1.720 340)37 \$,946 6+5 8.7.4 7,425	
	59 57	Childrane Critorane	C.9 D.9	4	53 53	4	N	2	87) 24	7.200 7.200		Į												277 C	5.748 2	6 67

TABLE 7-4

TEST RUNS FOR THE URBAN GRID

These test runs were made to exercise the VM. These results do not necessarily represent the results that would be obtained from a more complete VM, since the model in its current state of development (a) does not treat all damage mechanisms rigorously, (b) does not model all aspects of spill development with sophistication sufficient to guarantee very precise results, and (c) does not give a completely realistic treatment to population density and shifts of density with time of day and season of the year.

33. S

COLUMN HEADING	ABBREVIATION	MEANING
Cargo	LNG	Liquefied natural gas.
Ignition Type	E	Ignition source set to produce an explosion.
	F	Ignition source set to produce a flash fire.
	P	Pool burning initiated at the first time step.
	N	Ignition not allowed (for NH ₃) or not possible (for Cl ₂).
Portion of Evaporated Mass Combusting	DNI	Fuel cloud did not ignite.
Deaths or Injuries	T	Deaths or injuries would have occurred but the population of the cell was already dead from toxic inhalation.
Toxicity/Injuries	NA	Toxic injuries are not assessed in the VM for this substance (NH3).

115

KEY TO TABLE 7-4

		1						E	plosion			
		Hind			Portion	Explosive				Structures	Plas	h Fire
Case	Amount (Metric	Direction (Toward)	Igr Time	Distance	of Mass Combustible	Yield (Tons of	Deaths	Injuries	Structural Damage	Broken Windows	Deaths	Structures Ignited
Numbe	r Tons)	(*)	(min)	(m)	(7)	TNT)	(1)	(#)	(M\$)	······	(7)	(15)
LI	20,750	50	4.2	1,010	12.1	40,700	984	77,542	129.5	29,490		
L2. 1	3 20,750	53	4.2	1,010	12.1	41,200	1,238	77,753	131.1	29,628	534	2.2
L4, I	5 83	50	5.6	1,350	39.2	657	921	1,101	17.0	5,871	617	2.2
L6, L	7 83	53	l	DID NOT	IGNITE							{
L8, 1	9 0.8	50		DID NOT	IGNITE]		
L10, L	11 0.8	53		DID NOT	IGNITE							

TABLE 7-5

LNG SPILLS

and the second second second second

Distance is the distance from the spill location to the center of the puff at the time that ignition occurred.

سأعقاقه والمنابعة والمعاد فاحدا عطاسهمان الاستكرامه

wind toward 50° and then with the wind toward 53°. When the vapor cloud from the very large spill detonated at cell 19, the resulting explosion was calculated to have been immense - equivalent to about 40,000 tons of TNT. Such explosions are known to be unlikely; moreover, they are not known to be physically possible. Bear in mind that there are few LNG ships in operation and few planned compared to the total number of all ships or even all tankers operating. These ships will call at few ports and are expected to be subject to unusual operating constraints in the port calls they do make. There are very few ways in which any spill, much less a major spill, can occur from such specially designed vessels, and exceptional precautions are taken to preclude such an event. Although the probability of such a rare event as an LNG spill is difficult and expensive to estimate reliably, it is known to be very low relative to the total number of LNG ships (say, 42 to 85) which will operate in U.S. waters (a small part of their total operating time) over the full life cycle of LNG importation (say, 15 to 25 years). The next point to bear in mind is that, even if an LNG spill were to occur, it is far more likely that the material would burn in the immediate vicinity of the ship than that a cloud of vapor would disperse away from the ship. Even if the material did disperse, it would enter a populated area only if the wind direction carried it to that area rather than away from such an area. Even if an LNG vapor cloud were to approach a populated area, it appears far more likely that it would be ignited when only the leading edge of the cloud was in that area and while the bulk of the cloud was still over water. If the cloud were to move over a populated area without ignition, as it was warmed by the air and ground it would tend to rise (at ambient temperatures LNG is lighter than air) and to disperse harmlessly into the atmosphere. Even if the cloud were to be ignited while over a populated area and before its natural buoyancy carried it up and away from people and buildings, it is known that it could burn in a flash fire or deflagration; it is not known whether unconfined parts of the cloud could detonate. In this respect, methane (the major component of commercial LNG) is less dangerous than some other hydrocarbon vapors, such as propane or ethylene oxide. A possibility that has not been fully researched (and would be very expensive to research) is that a detonation from a <u>confined</u> explosion of a portion of the methane cloud (see Appendix C5) could accelerate the deflagrative combustion wave in portions of the unconfined cloud to supersonic speeds, thus producing detonation(s) there. Also, it is not known how much of the unconfined cloud would be detonated or how much of the cloud would be impacted by the initiating detonation wave.

त्रिकार्थने विश्वमिति के विश्वमित के प्राप्त के विश्वमित के प्राप्त के कार का प्राप्त के प्राप्त के प्राप्त कि

hander and Sur at A.

Julua .

Cases L1 and L2 in Table 7-4, then, represent hypothetical events which may not be even physically possible and which are definitely extremely unlikely ever to happen. Why, then, should such cases be explored at all ir a model? If it can be established that even very rare types of accidents will not result in major damage, then unnecessarily expensive safety precautions and/or scientific research on safety problems can be avoided. Where major accidents cannot be conclusively ruled out, then careful consideration of safety precautions, accident probability estimates, and/or further scientific research is indicated. Even so, no final decisions can be made on the basis of this model alone. If the risks (probability times consequences) are apparently comparable to other dangers to the public, then deciding factors may be the the benefits of an activity and/or cost-effective application of additional safety measures.

117

One final caution is that extensive scientific research can be timeconsuming as well as expensive; frequently decisions are time-limited such that it is unfeasible to obtain all of the scientific input desirable when it is needed. All production and transportation of hazardous materials involve some risk; the key question is whether drastic constraints affecting the economics of these materials involve even greater risks.

The reason that there were more deaths for the very large spill with the wind toward 53° is that the ignition time varied by two seconds and this changed the yield slightly. The extra yield was just enough to increase the effect in cell 48 from 2% dead to 13% dead. As cell 48 is a very densely populated cell, this accounts for the excess deaths in the case where the wind was blowing toward 53°. With the wind toward 50°, the flash fire was assessed to have killed 58% of the population in cell 19, and the radiation level there was high enough to have caused first-degree burns on the remainder of the persons in the cell if they had been exposed.

. . المامي قال قال لا عمالة أن

1

A LOCATION AND A LOCATION

The LNG spills of medium size are of interest, because of the fact that ignition occurred with the wind toward 53° and did not occur when the wind was toward 50°. In the case which ignited, the center of cell 19 was 21 m off the x axis, and, in the case where it failed to ignite, the center of cell 19 was 95 m off the x axis. Early in the dispersion of a puff of gas, when the diffusion coefficients are still small, this difference can be enough for the concentration to reach the flammable range in one case and fail to reach it in another. The concentration was evaluated at 6 minutes when the puff center was only 25 m from the cell center, so adjusting the time step would have no effect on the case which failed to ignite. For the neutral stability condition, at 1400 m, σ_v is approximately equal to 65 m thus in one case the cell center was about 0.3 σ_y from the puff center and in the case with the wind from 50° the cell center was about 1.5 σ_y from the puff center, so the failure to ignite is understandable. For the 50° case, cell 46 was only 30 m from the x axis but was 3037 m from the spill location, and, by the time the puff had traveled that far, it was too diffuse to ignite. The LNG spills of small size resulted in a vapor cloud which was not concentrated enough to ignite at the first downwind cell.

Chlorine Spills

For chlorine, the results of seven computer runs are shown in Table 7-6. For all three spill sizes, there are results for a wind speed of 5 m/s and unstable conditions as well as for a wind speed of 4 m/s and neutral conditions. In addition, the medium size spill was run for a wind speed of 4 m/s and unstable conditions. With stability class 4 (neutral), the values of σ_y and σ_z are smaller than they are for stability class 3 (unstable), so the resulting puff is smaller and more dense after traveling the same distance. Whether the smaller, denser puff affects more people depends on the size of the spill. Table 7-6 shows that the small spill of 900 kg is calculated to have killed 277 people and injured over 5000 people in the neutral case, but, in the unstable case, the more diffuse puff was too dilute to cause any deaths and is calculated to have injured only two people. For the very large spill, on the other hand, there is so much chlorine that the more diffuse puff is calculated to have caused more deaths as well as more injuries. Even though the concentration is lower near the center of the

Amount ric Tons)	Speed (m/s) 4 5 4 5 4 5 4	Stability Class 4 3 4 3	Time (min) 10 8 16	Distance (km) 2.4 2.4	Killed (#) 921 1.823	Injured (#) 0 5.724
ric Tons)	(m/s) 4 5 4 5 4	Class 4 3 4 3	(min) 10 8 16	(km) 2.4 2.4	(†) 921 1.823	(#) 0 5 724
1,090	4 5 4 5 4	4 3 4 3	10 8	2.4 2.4	921 1.823	0
1,090	4 5 4 5 4	4 3 4 3	10 8 16	2.4	921 1.823	0 5 724
	5 4 5 4	3 4 3	8	2.4	1.823	6 724
	4 5 4	4	16			2,734
	5 4	3		4.8	7,328	17
	4	-	20	4.8	10,932	4,943
		4	30	7.2	13,628	937
	5	3	24	7.2	18,152	4,946
15	4	4	10	2.4	921	0
	4	3	10	2.4	921	0
	5	3	8	2.4	0	921
1 1	4	4	20	4.8	7.327	1
	4	3	20	4.8	1.020	5.768
1 1	5	3	16	4.8	340	6.446
	4	4	30	7.2	9,995	865
	4	3	30	7.2	1.020	6.334
↓	5	3	24	7.2	340	7,425
0.90	4	4	10	2.4	277	645
	5	3	8	2.4	o	0
	4	4	20	4.8	277	5.111
j į	s	3	16	4.8	Ó	2
	Ā	-	30	7.2	277	5.748
_	s	3	26	7 2		2
	0.90	5 4 4 5 0.90 4 5 4 5 4 5	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

TABLE 7-6

CHLORINE SPILLS

The various cases for each spill size are listed here in order of distance traveled to facilitate the comparison of the effect of wind speed and stability class on the effects of the spill. The distance is the distance from the spill location to the center of the puff for the time listed.

611

and have a server and a set of the states of the states of the second states of the second states and the second states of the second states

محصفهم بالاستخاذ بالمحافظ والمعاقصة والأسالية

مريوري مراجع

AND THE ADDRESS OF A DECK OF A DECK OF A DECK OF A DECK OF A DECK OF A DECK OF A DECK OF A DECK OF A DECK OF A

more diffuse puff, it is still more than high enough to cause 100% deaths there, so the more widespread puff causes fatal and nonfatal injuries over a wider area. Because of the large number of casualties assessed for the very large spills of chlorine, it should be mentioned once again that they are hypothetical and represent only the use of realistic demographic data for a theoretically possible incident.

والتفوع فالمستحملان والقام متقاط فالماري والمعتم ومرارع والمعاوم مستحما والمستحم والمراري والمعادية والملاف والمتلاف

A DESCRIPTION OF THE OWNER OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OWNER OF THE OWNER OF THE OWNER OWNE

For the medium-sized spill, one may compare two runs with the same stability with different wind speeds and two runs with the same wind speeds but with different stability classes. Examination of these three runs in detail points out some interesting features of the VM and of atmospheric dispersion. First, note that at cell 19, 2.4 km from the spill location, the case with the wind speed equal to 5 m/s has 921 injuries whereas the other cases have 921 deaths. As all the chlorine was released as gas in 17.3 s, this is not due to any difference in travel time. (The release time is 17.3 s for all three spill sizes due to the fact that the submodel involved [MODI and EVDRP] assumes that all the liquid chlorine breaks up into individual drops which do not occupy a large fraction of the volume of the water and liquid chlorine mixture.) The difference is due to the fact that the puff passes over cell 19 while it is very small, so that the exact time at which the concentration is calculated is important. For the 4 m/s, stability class 3 case, the concentration at 6 minutes was 312 ppm. At this time, the puff center was at x = 1440 m, very close to the center of cell 19 at x = 1412 m, y = 95 m. For the 5 m/s, stability class 3 case, the highest concentration calculated was at 4 minutes, when the puff center was at x = 1200 m, about 200 m from the center of cell 19. Of course, at 4.8 minutes the puff center was at x = 1440 in this case, but the time step was 2 minutes, so the concentration was not calculated then. Thus the smaller number of deaths with the wind at 5 m/s is due primarily to the large value used for the time step.

This artifact in the assessment can, of course, be eliminated by using very short time steps or by calculating the factor determining toxic lethality ($/c^n dt$ as explained in Chapter 6) at a given geographical point by analytic means. A short time step requires more computer time, on which there are economic constraints. The analytic calculation of the factor determining lethality at a geographical point appeared to be a level of detail for which the VM is not ready at this time. The problem is most severe when the puff is small shortly after the release of the cargo gas. As the puff expands with distance, the problem becomes less severe. Running the VM with 10 s time steps would have caused the assessment of identical numbers of dead and injured for the 4 m/s case and the 5 m/s case with unstable conditions, but the computation time for each run would have increased by a factor of twelve. Cost is roughly proportional to computation time, so the cost would have increased by about an order of magnitude.

Comparing the two medium-sized spills which have the same wind speed, but different stability classes, it is clear that the smaller dense puff resulted in more dead but fewer injured. Cells 44, 46 and 26 were close enough to the x axis (i.e., close enough to being directly downwind) so that the higher concentrations in the neutral stability case caused 100% deaths to be assessed, whereas in the unstable case the lower concentrations resulted in the assessment of many injuries but only a few deaths.

Anhydrous Ammonia Spills

The results of the ammonia runs are shown in Table 7-7. Each of the three spill sizes was run with ignition possibly at the cell centers, and with ignition precluded. In evaluating test runs in which the ammonia did burn, the reader should note that ammonia is difficult to ignite under most conditions; however, it will burn or even explode under certain conditions. There is no known record of an unconfined ammonia cloud explosion.

The small spill of 8.2 tons caused no toxic deaths and did not ignite. Toxic injuries are not assessed for ammonia, as discussed in Chapter 6. When ignition was possible, the time step, Δt , was set to 1 minute since ignition usually occurs fairly soon if it is going to occur. With no ignition sources at the cell centers, it was desirable to run the program for 30 minutes so $\Delta t = 2$ minutes was used. In Table 7-7, the case with ignition possible at $\Delta t = 1$ minute is listed first.

The very large spill of over 16,000 metric tons of anhydrous ammonia ignited, not at the first cell in its path (cell 19) but later at cell 46. The reason for this is the evaporation time and the manner in which the puff model treats long release times. It took 20.8 minutes for all of the ammonia to escape in gaseous form, but the puff submodel starts the puff moving from the spill location when the first gas escapes and increases the arount of cargo gas in the puff as it escapes. Thus, as the puff moves away downwind, the mass in the puff increases. In this case, the mass in the puff increased fast enough to more than compensate for the spreading of the puff; the concentration increased from the time the puff was near cell 19 (6 min), so that by the time it was close to cell 46 (12 min) ignition was possible. At the time of ignition at 11.8 minutes, 7.9% of the spilled mass was combustible. In the case that the ignition set off an explosion, this amount of ammonia is equivalent to 4110 tons of TNT. The resulting damage is shown in Table 7-7. There were no deaths from explosive effects because the people in cells 46 and 47, 210 m and 490 m, respectively, from the blast center, were already dead from the toxic vapors. Explosion injuries were assessed as far away as 4350 m at cell 16, where one person was calculated to have received injuries from flying fragments. Seven buildings 6500 m from the blast center in cell 12 were assessed as having broken windows. In the case of flash fire, structures were ignited only in cell 46.

For the very large spill without ignition, it is interesting to note that the 2-minute time step resulted in more deaths from toxic vapors at 10 minutes after gas release started than the run with $\Delta t = 1$ minute, but the number of deaths at 11.8 minutes in the one case was the same as the number of deaths at 12 minutes in the other case. This is due to the use of Δt in calculating the dosage. In this example, the concentration at cell 47 was increasing so that the VM had, by 11.8 or 12 minutes, assessed all of the outdoors population in that cell as dead. Changing the size of the time step does not always have an effect; Table 7-7 shows that for the medium-sized spill the number assessed as dead is the same at 10 and at 14 minutes for $\Delta t = 1$ and $\Delta t = 2$.

]		ll i				Explosion			
Case Number	Amount Spilled (Metric Tons)	Time (min)	Distance (km)	Toxicity Deaths (#)	Injuries (#)	Structural Damage (M\$)	Structures with Broken Windows (#)	Flash Fire Structures Ignited (M\$)	۵t (min)
A1, A2	16,340	10 11.8	2.4 2.8	1,412 6,726	11,419	68.35	15,720	9.23	1
A3	16,340	10 12 30	2.4 2.8 7.2	2,177 6,726 18,151		NO IGNITIO	N ALLOWED		2 2 2
A A, AS	251	10 14	2.4 3.4	921 4,584		DID NO	I IGNITE		1
A6	251	10 14 30	2.4 3.4 7.2	921 4,584 8,033		NO IGNITIO	 n allowed 		2 2 2
A7. A8	8.2	ANY	ANY	NONE		DID NO.	I IGNITE		1
A9	8.2	ANY	ANY	NONE		NO IGNITIO	A ALLOWED		2

1.

a bit water, water also have a second to be the second of the an indiana and the second of a second or one and in

......

TABLE 7-7

AMMONIA SPILLS

No explosion deaths and no flash fire deaths or injuries are listed because inhalation of toxic ammonia fumes had killed those people in cell 19 who would have been killed by the explosion or killed or injured by the flash fire. Toxic injuries are not listed because toxic injuries are not assessed for ammonia (see Chapter 6 and Appendix E). Distance is the distance from the spill location to the center of the puff.

. . . . مُلاه كاذ مَسْ وَأَوْسَا هِ

122

Immediate Pool Burning - Methanol, Gasoline, and LNG

Methyl alcohol and gasoline evaporate much more slowly than do LNG and ammonia, with the result that the concentration at the first downwind cell (cell 19) is so low that even the maximum credible spill does not result in ignition. As the puff moves downwind, the amount of vapor it contains increases. but the dispersion coefficients increase as well, and the concentration does not approach the lower flammable limit concentration at later times either. For these two cargoes, the spill location was moved 30 m upwind from the center of the river cell, and a very short time step was used; however, even in these cases the concentration was too low to ignite. Finally, the VM was reprogrammed so that the user may set a flag which indicates that the simulation should proceed immediately to pool burning after the spill is complete. 1

1 A DUM

-

Self-Row in caretor

1997 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 -

्यत्र प्रतिकारित महित्यत्रिका विवयत्ति । स्वत्यत् प्रतिकार्त्त का स्वतं क्षेत्र का स्वतं का स्वतं का स्वतं का स

The results of runs for medium-sized spills of LNG, methyl alcohol and gasoline with immediate pool burning are shown in Table 7-8. The radiation given is for cell 3, the cell in which the spill occurred. For these spills, the location of the spill was chosen to be the center of cell 3. In the radiation calculation, entering a zero for the distance from the center of the pool to the observer causes the flux to be equal to infinity; thus, in cases like this, the program removes the observer to a distance equal to the pool radius plus 10 m. In the simulation, there are no structures in the river cell in which the pool burning takes place, and the nearest land cell center is about 1.2 km away; the radiation levels there were not high enough to ignite any structures, so no damage is assessed in each of these three cases.

It may seem strange that the burning time varies so widely among these three cases whereas the radiation intensity varies very little. This is partially due to the ways in which some of the submodels operate. First, it is assumed that the pool stops spreading when ignition occurs. Second, a simplifying assumption in the calculation of the radiation flux is that the flux is primarily a function of the adiabatic flame temperature. Finally, the burning rates of large pools (in terms of reduction of depth per unit time) were found for these three materials, and the depth of the pool was used with these data to give the burning time. Since the total energy radiated through a right circular cylinder around the flame is the product of the cylinder's area, the radiation flux, and the burning time, the amount of energy radiated is not calculated in a direct way from the energy contained in the spilled cargo pool.

Let us compare the results of LNG, which has the shortest burning time, with those for methyl alcohol which has the longest burning time. The available energy in the pool at the instant of ignition is given by the product of the heat of combustion with the mass of the cargo in the pool. Thus, assuming that the combustion is complete, one finds that the LNG pool contained 3.5×10^{12} joules, whereas the methyl alcohol pool contained 1.8×10^{12} joules. The size of the cylinder around the flame was chosen to be such that the radiation flux calculated for cell 3 may be used. In these examples, the spill was exactly at the center of cell 3, and the radiation level could not be calculated there since this point was inside the burning pool. Therefore, the radiation flux was calculated at a point 10 m beyond the edge of the pool, and the cylinder had a radius 10 m greater than the pool radius. The area of such a cylinder is $40,700 \text{ m}^2$ for the LNG case, but only 2650 m² for the methyl alcohol pool since the methyl alcohol pool was smaller and the flame was shorter. Thus, 0.044×10^{12} joules were radiated away in the LNG case, and 0.71×10^{12} joules were radiated in the methyl alcohol case. The two cases differ in radiated energy, then, by a factor of about 16, not the factor of over 200 which one might get if only the burning times were considered. Still, in the methyl alcohol case 39% of the available energy is calculated to have been radiated away, whereas in the LNG case this figure is only 1.2%. The remainder of the available energy in both cases is convected away.

.....

衝

ميكيني الكلاكساني على مد

، متأ**مدة** « - «مستخفهم»، «ا**م علاقاً» مثلغاً** في

The long burning time calculated for methyl alcohol results from the fact that the pool of methyl alcohol spreads more slowly than the pools of LNG or gasoline. Thus, when ignition occurred after one time step of 60 s, the methanol pool was about 2 cm thick, which is the reason why the methyl alcohol burning time is so long. For pools which are less than a millimeter thick at ignition, the assumption of no further spread during burning is reasonable. But, in the case of methyl alcohol, this assumption gives a pool which retains a radius of 13.7 m for almost 2 hours while it burns. It also results in a burning time which is much longer than the time that it would take the pool to evaporate, as calculated by the HACS submodel MODR.

Conclusions

Marine States of the States

These test runs show the great range of injuries and damage that may result from spills of hazardous cargoes. The size of the spill and the material spilled primarily determine the order of magnitude of the results, but the wind direction and atmospheric stability also play important roles. It is noteworthy that the methyl alcohol and gasoline spills did not evaporate fast enough to form a cloud concentrated enough to ignite at the first grid point downwind. This point was 1400 m away; however, computer experiments showed that ignition of the vapor cloud would not occur for distances as short as 30 m. These results may correctly simulate physical phenomena for spills of this type, but further validation seems to be required. Thus further investigation of the experiments and theory relating to this problem may be appropriate.

Because the very large spills of LNG, ammonia, and chlorine were assessed to have killed and injured thousands of people, it should be remembered that in this model approximations have been made in such a way as to maximize the potential damage wherever feasible and that the probability of spills this large in a densely populated area has not been taken into account.

Ignition

Since a change in the wind direction of only 3° in the medium-sized spills determined whether or not the vapor cloud would ignite, consideration should be given to making the ignition less dependent upon the wind direction. At the present time in many cases, the center of the puff must pass very closely to a cell center for ignition to occur. This is due to the fact that ignition in the VM is possible only at the cell centers at this time and

Case Number	Cargo	Amount in Tank (MT)	Amount Spilled (MT)	Pool Radius (m)	Flame Height (m)	Burning Time (min)	Radiation Flux (J/m2/s)
L13	LNG	83	65	50.7	106	0.43	42,405
ы	Methyl Alcohol	95	93	13.7	17.8	117	37,867
Gl	Gasoline	70	68	37.1	54.9	5.8	37,325

TABLE 7-8

INMEDIATE POOL BURNING

There are no deaths or injuries from the pool burning because it is assumed that there is no resident population in the river or harbor cell where the pool burning occurs; furthermore, the nearest land cell is far enough away that the radiation from the pool does not cause any injuries or the ignition of any structures there. The radiation flux is calculated at a point 10 m beyond the edge of the pool.

125

that these are rather coarsely spaced with respect to the puff size during the early stages of dispersion. A quick solution to this problem would be to subdivide the census tracts presently in use. This should solve the ignition problem, as well as making the assessment results less markedly dependent on small variations in direction.

For the long run, a more sophisticated solution may be in order. One such solution would be to assign an ignition probability, which would have the units of l/area/time, to each census tract or subdivision thereof. Thus the product of the ignition probability with the area in which the concentration is in the combustible range would have the units of l/time. For each time step, this product would be multiplied by the duration of the time step and added to the sum of similar products from previous time steps. This sum would increase with each time step, and, when a threshold value, such as one-half is reached, ignition would occur. This method would more realistically simulate both slow-moving clouds hanging over an area with sparsely distributed ignition sources, as well as clouds which move quickly over an area in which ignition sources are dense.

194**4**

مستعادمات المكتفقان ولتمالك وشوا

Resource Distribution

The fact that the VM currently treats the resources in each cell as if they were all located at the cell center makes the assessment of damage dependent on slight changes in the spill location or the wind direction, or both. This is due directly to the coarseness of the grid of cells. Census tracts are often more than 1 km across and are less than 500 m across only in the most densely populated areas. Although the use of whole census tracts as cells has been adequate for this developmental work, subdivision of census tracts will be necessary if the development of the VM proceeds much further.

Ideally, the resources would be described in a grid system which has a scale smaller than the horizontal diffusion coefficient of the air dispersion model for the conditions prevailing at the time of interest. Since the dispersion coefficients differ widely with the stability class and distance, no single scale can be ideal for all conditions. For the most stable conditions within a few hundred meters of the source, the horizontal dispersion coefficient is less than 10 m, and cells on this scale are computationally infeasible; even so, consideration might be given to making the cells smaller along the river or harbor than they are a kilometer or so inland.

Time Step Selection

The test runs have shown that the VM, at its current stage of development, is sensitive to the choice of time step under certain conditions. If development proceeds, this could be addressed by a number of actions. An increase in the density of the cells and a more sophisticated treatment of the ignition problem will largely resolve the current situation in which the concentration of the puff at the cell center is not evaluated during the time that this concentration is near its maximum; however, it may be necessary to introduce an internal limit in time step duration and to use a short time step for the early stages of the dispersion and a longer time step later on. These procedures will also remedy the problem for toxic gases where the concentration is not evaluated for representative values.

Validation

Extensive validation of the VM by reference to actual accidental spills of hazardous materials has not been feasible. The large disasters happen only infrequently and are usually not well documented. Smaller spills, on the order of the medium size or smaller, happen more often, but they too are usually not well documented and, fortunately, they commonly occur in rural areas. Moreover, upon close examination, almost every accident seems to have features which make it atypical. الستطرف حداك

р 1

ŋ

÷.

- manual a

.

dan all that he are

1.11

For example, consider the case of the freight train derailment in Crete, Nebraska, in February of 1969 [36]. In some respects this case is ideal for study since the accident involved a substantial amount of ammonia (111 m³) and occurred near the center of town. On the other hand, the accident took place at 0630 when the temperature was very low (-15.5°C) and there was no wind. Thus very few people were out of doors, and all of the houses were tightly closed. These conditions contributed to the fact that only 6 persons were killed and 28 seriously injured from ammonia inhalation in this accident.

The Phase I submodels could be validated individually or as a group by field experiments. Some work of this type has been performed and more is currently underway; however, the cost of large-scale field experiments is high in resources and time. The Phase II submodels are generally based on extrapolated animal experiments. In some instances, reports of accidents have contributed to the evolution of assessment procedures. Nevertheless, in any event the validation of Phase II submodels by planned experiments is ethically unacceptable.

Cost

and the second second second second second second second second second second second second second second secon

The cost of running the VM will vary depending upon the computer system used and the scenario simulated. The test runs were made on an IBM 360/65 computer. Running times were generally between 15 and 35 CPU seconds. Typically 1000 to 2000 lines of output were produced. The typical cost of a run was between three and eleven dollars. A cost of five dollars was average.

Closure

The examples in this chapter were presented to allow the reader to see the type of information obtained by running the VM and to judge the reasonableness of the results in view of the approximations inherent in the development of the VM. Neither the probability of a spill nor the probability of any other factors has been considered in the VM, since it is a <u>damage</u> assessment, <u>not</u> a <u>risk</u> assessment tool. The probability of

^[36] U. S. Department of Transportation, National Transportation Safety Board. Railroad Accident Report (Chicago, Burlington, and Quincy Railroad Company, Train 64 and Train 824, Derailment and Collision with Tank Car Explosion, Crete, Nebraska, Feburary 18, 1969).
U. S. Department of Transportation, Washington, D.C., February 24, 1971. Report No. NTSB-RAR-71-2.

the events comprising a simulation scenario, as well as the validity of the approximations and assumptions made in the course of VM development, should always be borne in mind when the test runs are contemplated, interpreted, or discussed by the reader. मिति हो की कि रोकि राजित का ता कि की से कार्यकार की कि दि ही हैं। अने राजित की मित की कि का की की कि कि कि कि कि

CHAPTER 8

لللغيالي شطيع يعلق المطلعة المعطيم والمعالمة المسالم

्राम् क्रियन्त्र विद्याप्रकारित्यक्षित्र व्याप्तका स्वकृत्य प्रदेशित् के विद्विति क्रिये क्रिये क्रिये विद्याप

تعقيته بعقار كالمائلا كالانفال بتارك بب

SENSITIVITY ANALYSIS

Plan of the Analysia

The concept of sensitivity analysis has been extensively discussed in the literature related to control systems and other physical systems [37, 38]. Although the theoretical concepts considered are relatively sophisticated, simplification can be made to apply these methods to a sensitivity analysis of the VM.

Typically one considers a system with n output variables, Y_{i} , i = 1, 2,..., n, and with m input variables, X_{j} , j = 1, 2, ..., m. Then we may write

 $Y_i = f_i(X_i)$

to represent the functional dependence of the n output variables on the m input variables.

Consider now the differential change in one of the output variables, dY_1 . By the chain rule we have

$$d\mathbf{Y}_1 = \frac{\partial f_1}{\partial \mathbf{X}_1} \quad d\mathbf{X}_1 + \frac{\partial f_1}{\partial \mathbf{X}_2} \quad d\mathbf{X}_2 + \ldots + \frac{\partial f_1}{\partial \mathbf{X}_m} \quad d\mathbf{X}_m$$

The factors, $\partial f_i / \partial X_j$, are usually termed the sensitivity coefficients and give, for small changes in the input variables, a measure of the effect on the output variable.

For a detailed sensitivity analysis, each of the sensitivity coefficients must be considered and its variation over the range of the several variables must be determined. One way to summarize the output variation would be to consider functions such as

$$g_{j} = \sum_{i=1}^{n} \left(\frac{\partial f_{i}}{\partial x_{j}}\right)^{2}$$

which gives the "change in arc length" generated in the n dimensional space of output variables by incremental changes in the jth input variable.

In the present stage of the development of the VM, such a rigorous sensitivity analysis is considered to be: (a) too detailed, exacting, and time consuming considering the priorities of other tasks in the program and (b) too abstract and devoid of physically tenable measures to assist the USCG. Therefore, a simpler, less sophisticated, less conventional form of sensitivity analysis has been chosen.

^[37] Radanovic, L. (ed.). Sensitivity Methods in Control Theory. Pergamon Press, New York, 1966.

^[38] Tomovic, R. Sensitivity Analysis of Dynamic Systems. McGraw-Hill, New York, 1963.

In this modified analysis, a square grid with a uniform density of population and structures was used. Therefore, input variables such as spill location and wind direction, which change only the location but not the magnitude of the damage, should have no effect. This is true as long as the area in which the damage or injuries occur always falls within the grid, and as long as the mesh of the grid is small with respect to the size of the damage mechanism.

For the sensitivity analysis computer test runs, a 15 x 15 grid with a spacing of 500 m was set up. Each grid point represented 1000 people (all assumed to be outdoors) and 100 structures. The spill took place at the central point, and the wind was in the positive x direction (toward the east). For a distance of 1 km, the horizontal dispersion coefficient for the puff is less than 10 m for the very stable condition and about 85 m for the unstable condition. Thus the grid spacing is not small with respect to the size of the puff at this distance, so the wind direction and spill location will influence the damage assessment. Consider, as an example, the case of a toxic gas with stable atmospheric conditions. The cloud will be small and concentrated, and the wind will carry it down the x axis directly over each grid point along the axis, killing all of the people at each point. If the release point is moved half a grid space to the north (in the positive y direction), the wind will carry the cloud down the corridor between the grid points, and no deaths at all may be assessed. This effect, due to changing the spill location, is completely artificial and results from the coarseness of the grid. In theory, or for an infinitely fine mesh, the wind direction and spill location have no influence on the damages assessed for a resource which has a uniform density. Therefore, since a grid spacing smaller than the size of the puff is not economically feasible, and since the wind direction and spill location do not affect the damage assessment for a very fine mesh, the spill location and wind direction are not changed in this analysis, even though they will affect the assessment due to the coarseness of the grid.

The plan for these computer runs was to vary only the size of the spill, the wind speed, and the stability class, but it became necessary to vary the time step in certain cases as well. The puff model was used for all runs, and the time step was 1 minute unless otherwise stated. As methyl alcohol and gasoline evaporated so slowly that they would not ignite, and since these cargoes are not currently treated as toxic in the VM, the sensitivity analysis test runs were made only for chlorine, anhydrous ammonia, and liquefied natural gas (LNG). For each cargo, the standard or reference run was a medium-sized spill with neutral stability and a wind speed of 4 m/s. After this reference run had been made, six other runs were made, and in each of them only one parameter varied from the reference case. The stability was changed to more and less stable conditions, and the wind speed was reduced from 4 to 2 m/s and then increased to 10 m/s. Finally, the size of the spill was changed to the very large size and then to the small size. States - Minutes

Results

1. Toxic effects

The results of the runs which concentrated on the toxic effects are given in Table 8-1. Since there were only 7 cells downwind of the spill on the x axis, the 7000 dead for the first (reference) ammonia run represent the maximum number that can be killed unless the puff is large enough to affect the cells off the x axis. Had the grid been larger and the runs extended for a longer period of time, the casualties would have been greater since this spill was causing 100% deaths when it reached the edge of the grid. The puff is less dense and more widely distributed in the

Cargo	Amount (MT)	Wind Speed (m/s)	Stability Class	Dead (#)	Injured (#)	Notes
NH ₃	251 251 251 251 251 251 16,340 8.2	4 4 4 2 10 4	N U S N N N N N	7,000 4 7,000 7,000 2,949 13,042 2,175		Reference case tevap = 7.8 min. At = 1 min. Lethalities in off-axis cells
	15 15 15 15 15 15 1,090 0.90	4 4 4 2 10 10 4 4	N U S S N N N N N N	7,000 4,580 3,000 7,000 7,000 3,000 6,000 7,000 3,865	0 2,460 850 0 924 0 0 3,135	Reference case $\Delta t = 1 \text{ min.}$ $\Delta t = 20 \text{ sec.}$ $\Delta t = 1 \text{ min.}$ $\Delta t = 20 \text{ sec.}$

TABLE 8-1

SUMMARY OF TOXIC EFFECTS

For these runs, a square grid with cells 5C0 m apart was used. The wind blew in the direction of the positive x axis, and the time step was 1 minute unless otherwise indicated. Each cell contained 1,000 people outdoors and 100 structures. A square grid structure comprised of 15 x 15 cells was used with the spill always at the center, so there were 7 cells downwind of the spill. The stability classes are denoted as follows: N=neutral, U=unstable, and S=stable. unstable case, and the concentrations just barely reached the lethality threshold. (No injuries are assessed for ammonia, as discussed in Chapter 6.) The medium-sized ammonia pool takes 7.8 minutes to evaporate, so the puff is more than halfway down the x axis before it contains the entire amount of the spilled cargo. Thus the vapor cloud did not contain all of the spilled ammonia when it passed over the first grid points; this partly explains why very few deaths were assessed with unstable conditions, since the concentration at the center of the puff is lower than with other stability conditions. Use of the puff model, as described in Chapter 3, requires that the vapor liberated at later times be added to the puff at its downwind position, without passing over the intervening area. This anomaly is produced by the lack of an extant air dispersion model to treat cases between instantaneous and continuous releases. The results for the unstable case were not affected by the value of the time step. - Milandar

÷

عممت 2. بعد باروطار عد

. -- El.

يتنعدك فيقفقهمه بواح

1

Annalisi kana ana a

Ï

1

1

ţ

The cases with stable conditions and the slow wind speed caused 100% deaths on the x axis as might be expected. The low number of assessed fatalities in the fast wind speed case is due to the use of a time step of 1 minute which is too large for a wind speed of 10 m/s. Although this run reported no fatalities in the first cell downwind because the puff center was 100 m away when the concentration was calculated, a short run with a time step of 10 s found 89% dead for this cell. Since the evaporation is completed just before the puff leaves the grid, and the run with $\Delta t = 1$ minute calculated 100% dead when the puff center was at the center of the sixth cell on the x axis at 3 minutes, it is clear that if the time step had been 50 s to insure evaluation of the concentration when the puff center was exactly at each cell center, the number of fatalities would have been over 6000.

The very large spill caused over 13,000 deaths because it was so big that the puff was capable of causing toxic concentrations in the cells adjacent to the x axis. The first cell off the axis to have deaths assessed was at x = 2 km, and the cells at 3.0 and 3.5 km had 100% fatalities. As the evaporation took almost 21 minutes for this size spill, fatalities would have been caused for at least 10 km downwind if the grid had extended that far. The small spill caused considerably fewer deaths than did the reference case, which is to be expected.

The medium-sized chlorine spill is less than one tenth the mass of the medium-sized ammonia spill, but chlorine is much more toxic than ammonia, so the chlorine spill is also assessed to have killed everyone on the x axis. The unstable case resulted in 4580 deaths, many more than the ammonia spill with unstable conditions. The difference is primarily due to the fact that all of the chlorine evaporates in only 17.3 s, so that the entire mass spill is present in the vapor cloud when it reaches the first grid point. The stable case was assessed to have killed all persons in cells on the x axis only if a time step of 20 s is used. With $\Delta t = 1$ minute, the assessment was much less. For the case with the wind at 10 m/s, even with $\Delta t = 20$ s, the cell center was 100 m from the puff center for the first cell, which was too far. With $\Delta t = 10 \text{ s}, 25 \text{ s}, \text{ or } 50 \text{ s}, 100\%$ in all the cells on the x axis would have been assessed as killed. Even the small spill of less than a ton of chlorine killed almost 4000 people. A run with an extra long grid showed that the reference case was capable of 100% lethalities more than 7 km downwind on the x axis. There were no deaths in the cells adjacent to the axis, but there were some injuries.

To sum up, the toxic deaths show a greater dependence on stability class than they do on wind speed, a result which was expected. This assumes that the length of the time step is reduced as the wind speed is increased, otherwise too infrequent evaluation of the gas concentration will result in inaccurate assessment of the injuries. For the medium-sized spills of chlorine and ammonia, the unstable condition resulted in many fewer casualties than did the neutral condition. For a very large spill, however, this may not be the case. اللهاية بالتحالية والمحدرة أوقاع مستعطية المتقادية والمحدد والم

متحمد معارية فكر المتحدث من عنه الله الله المن المتحدث من من من المعالية المتحدث المن من من من المن المن المن ا

Ă

2. Effects of Explosions and Fires

'n

The results of the sensitiv ty runs for the flammable cargoes being considered, LNG and ammonia, are shown in Table 8-2. In the case of ammonia, the concentration levels necessary to cause death in a couple of minutes or less are below the lower flammable limit concentration, so the inhabitants of the cell at which the ignition occurs are already dead before the explosion or flash fire occurs. Thus there are no injuries or deaths from the explosion or flash fire for ammonia in the ignition cell. And the very large spill is the only ammonia spill which causes a large enough explosion to affect people outside the ignition cell.

The slow rate at which the ammonia evaporates is the reason that the vapor cloud failed to ignite in three of the medium-sized spills. Ignition did occur in the stable case where the puff is smaller and more concentrated and in the slow wind case where more of the liquid was able to evaporate before the puff reached the first cell downwind. The very large spill is interesting in that it did not ignite at the first cell downwind, but did ignite at the third cell. This was due to the continued evaporation of the ammonia as the puff moved downwind. The increase in mass countered the increasing size of the dispersion coefficients and caused the concentrations on the x axis to increase with time.

The failure of the LNG spills to ignite for the unstable mixing condition and the small spill is not unexpected. The case with stable conditions resulted in a smaller explosion than the reference case because the smaller values of the dispersion coefficients allowed less of the methane to be in the flammable concentration range. In the low wind speed case as in the reference case, evaporation was complete and no liquid pool remained when the cloud ignited at the first downwind cell. In the high wind speed case, however, ignition occurred when the evaporation was little more than half completed; consequently, even though a larger portion of the vapor exploded, the resulting blast was less powerful than the one in the reference or low wind speed cases. Note that the blast in the very large spill case treated here is much smaller than it was in the test cases with an actual grid of census tracts. This is due to the fact that here ignition occurred about 310 m from the spill location, whereas in the case reported in Chapter 7 the puff moved just over 1000 m before ignition. This extra movement allowed time for additional LNG to evaporate, and for the cloud to disperse more widely, so the portion of the methane in the range of flammability was greater.

The LNG spill for the 10 m/s wind did not ignite with the usual time step of 60 s. A value of Δt such as 10 s, 25 s, or 50 s, which causes the concentration to be calculated when the puff center is exactly at the first grid point, does result in ignition. With $\Delta t = 60$ s, the puff center and cell center did not coincide at an evaluation time until 5 minutes after the

]	1	1		Ex	<u>plosion</u>			<u> </u>	ish Fire
1	f	Wind		Toxic			Structural	Fraction of	Yield		Structures
1	Amount	Speed	Stability	Deaths	Deaths	Injured	Damage	Mass Exploding	(Short Tons	Deaths	Ignited
Cango	(MT)	(m/s)	Class	(#)	(#)	(#)	(#)	(%)	of TNT)	(#)	(#)
1		[
NH ₂	*251	4	N	7,000		DIC) NOT IGNITE				
3	251	4	U U	4	1	DIC	D NOT IGNITE				
1	251	4	S	1,000			100	2	3.5		0
	251	2	N	1,000			100	8	52		0
}	251	10	5 N	2,949) DIC	NOT IGNITE				
ļ	16.340	4	N	3.000		6.550	593	8	1.020		25
	8.2	4	Ň	2,175		DIC	NOT IGNITE	-			
		ł	1	Į							
LNG	*83	4	l N	1	1,000	20	125	13	215	10	25
	83	4	U	1		010	NOT IGNITE				
	83	4	S	ł	1.000	0	100	3	54	0	0
1	83	2	N		1.000	20	125	13	215	10	25
Į	83	1 10	N	1	1.000	10	112	25	159	19	25
1	20.750	4	N	F	1.000	28.456	958	8	2,500	183	25
	0.8	i i	l N	i i	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	1	NOT IGNITE		-,		
1					1						
[L							

*Reference test runs.

TABLE 8-2

SUMMARY OF THE EFFECTS OF EXPLOSIONS AND FLASH FIRES

The grid and abbreviations used are described in the caption to Table 8-1. For the explosions of ammonia, all those people who were in a position to be killed or injured by the explosion had already been assessed as dead from toxic effects prior to ignition. For the ignition of structures by flash fire, only 25 structures in the cell in which ignition occurs are set on fire because the program assumes that only 25% of the structures in a cell are ignitable in this fashion.

134

الاستطارية والألف مع مراسب مطالبها والعرب المراجب المسترقين المراجب المراجب والمراجب والمتحد المراجب والمراجب وال

spill when the puff center was at the sixth cell downwind of the spill location. At this time, the concentration at the center of the puff was not great enough to cause ignition. (It is not true that the puff center and the grid point must coincide for ignition; however, when the fuel concentration is above the lower flammable limit only near the center of the puff, then the center of the puff must come very close to the ignition source if ignition is to occur.)

Summary and Conclusions

والأنفاذ فالمقاطر والمحاول والمقاطعة والالتقاد والمقادية ومورد ومحاولة المتعاولة والمعاولة والمتعادية والمتعادية

It is clear that, of the three parameters varied, the size of the spill had the greatest effect on the damage assessed. This was partially obscured in the case of the very large spills of toxic cargoes because when the vapor clouds from these spills leave the grid, they have concentrations far above that needed to cause death in a few minutes. The small spills of the toxic substances caused many fewer fatalities than did the medium-sized spills. The small spills of the flammable cargoes also caused many fewer deaths than did the medium-sized spills, but here one of the primary reasons was the failure of the vapor cloud to ignite. Unlike the toxic case, for flammable materials the relationship of input parameters to damage caused was discontinuous, with the discontinuity occurring at the ignition threshold. The small spills did not result in clouds dense enough to ignite at the first grid point, so there is no damage from fire and explosion for these cases.

Of the remaining two variables, the stability class had a more marked effect than did the wind speed. In the toxic case, the number killed was lower for unstable case atmospheric conditions than for the reference case with neutral stability. For ammonia, changing to unstable conditions almost eliminated the fatalities. For chlorine, the reduction was less pronounced. The effects of stable conditions do not show up as well in this sensitivity analysis because of the small number of cells downwind from the spill location. A note of caution is in order, however. For the very large spills, the change from neutral to unstable conditions may result in more deaths and injuries. This is due to the fact that an increase in dosage above the lethal level has no effect. For a big spill in which the concentration might be above the lethal concentration in the cells adjacent to the x axis, fewer people may be assessed as killed if the lethal concentration of toxic gas is contained near the axis and does not extend to the adjacent cells. Since the populations of the cells on the x axis are all killed anyway, increasing the concentration there will have no further effect.

Finally, we come to the variations in wind speed. For the materials which evaporated before the puff reached the first grid point with the 10 m/s wind, the wind speed had no effect on the damage assessed as long as the time step was reduced accordingly. For cargoes which took longer to ignite, the increased wind speed could mean the difference between ignition and no ignition. For ammonia, for example, the release of the gas was slow enough so that only for the 2 m/s case was the concentration high enough to ignite when the puff reached the first grid point. It should be pointed out that the wind speed and the stability condition are not unrelated. The most stable atmospheric conditions are invariably associated with calms or very light winds. Thus, specification of strong winds and stable conditions is not realistic.

CHAPTER 9

THE HISTORICAL SURVEY SUBTASK⁺

This chapter describes the work performed on the VM under the historical survey subtask. This subtask consisted mainly in researching and compiling data about unconfined vapor cloud explosions, data that would be helpful in the development of the VM, but that might not be readily available. Ongoing research activities and plans related to VM development were also surveyed.

In the following text, Section 1 presents a summary of the major findings of the historical survey subtask. Section 2 is a review of private sector safety research and development plans relevant to the development of the VM. Section 3 is a listing of respondents interviewed or contacted. In addition, Appendix F contains a detailed account of fourteen case studies of accidental explosion incidents. In the following text, numbered cases (such as Case F.9) refer to the descriptions provided in Appendix F.

1. OVERVIEW

2.

This section presents an introduction to and a summary of the major findings of this study, as well as a discussion of the major data gaps and recommendations for filling them.

1.1 Objectives and Scope of the Study

This study was undertaken to seek real world verification of the predictions of the VM, by collecting available data on actual vapor cloud fires and explosions.

The initial thrust of the investigation was to visit private, mostly industrial, organizations which had experienced or investigated actual vapor cloud explosions and to ask for access to their data. It was known that some such studies existed but were largely unpublished. Such studies are typically <u>not</u> published, because the details intrude too far into areas the organizations consider private and confidential. Nevertheless, it was expected that the strictly safety aspects of internal reports would usually be shared in face-to-face conferences where confidential information could be kept private.

It turned out that some very instructive studies were indeed obtained in this fashion, but overall a disappointingly small number of really usable case studies were found. A number of commonly-listed "gas" explosions, upon review, turned out to have occurred inside a reactor or

Preceding page blank

said in the fight of the second second first of

137

[†]The historical survey reported in this chapter was performed under contract to ECI by Dr. John A. Brown of John Brown Associates, Inc., Berkeley Heights, New Jersey.

inside a building and were not at all the unconfined vapor cloud events of interest here. Some companies would talk only under seal of privacy, and their data could be used only to enhance the perspective on the reportable cases. A few companies would not talk at all.

But mostly, the detail desired simply does not exist. Industrial disasters have seldom been investigated in anything like the detail lavished on, say, an airline disaster. Industrial emphasis is placed on proximate causes and on prevention of recurrences, and relatively minor attention is paid to such elements as the exact amount of fuel participating in an explosion or the resulting damage profiles. Reconstruction of industrial explosion disasters in detail comparable to that provided by the VM would require a vastly greater investigative effort than private organizations are willing or able to mount. In the end, only two industrial explosion case studies were selected for presentation herein, and this study was extended to include the more numerous and better-documented studies of transportationrelated vapor cloud accidents. In all, nineteen visits and interviews were carried out, and fourteen case studies were selected as sufficiently instructive for inclusion in this report. Lessons learned are summarized in Section 1.2, and the cases are reported in detail in Appendix F.

The existence of several previously unlisted industrial explosions was discovered during the course of investigations supporting this study, but visits to the investigators of these explosions could not be arranged in the time remaining between discovery of the incident and the termination of this program. They are listed in Table 9-1. It is not recommended that they be pursued. The two case studies presented herein are by far the most detailed of those seen, and even they are sketchy by VM standards. Conferences with numerous respondents with first or secondhand knowledge of the unexplored cases indicate that further case studies are not likely to add materially to the picture.

It must not be inferred from the remarks on privacy that important safety data are being withheld in any significant number of cases. They are not. Details impacting litigation and liability were sometimes withheld, but every respondent saw to it that the key safety information was made available, even if permission was not given to use the actual case study or to cite the source.

TABLE 9-1. ADDITIONAL MISHAP INCIDENTS MENTIONED DURING INTERVIEWS AND RECOMMENDED FOR SPECIFIC EXPLORATION WITH THE RESPECTIVE RESPONDENTS LISTED

•	"Vapor cloud 800' x 200' x 20'" Cities Service, Lake Charles, La., "1967 or 68".
٠	Ethylene oxide tank car Dow Chemical, Midland, Mich.
•	"Several incidents" PPG Industries, Lake Charles, La.
•	Butadiene explosion Union Carbide, Texas City, Texas, 7 Sept 70.
•	Isoprene plant explosion Goody ear, Beaumont, Texas, 27 Nov 74.

1.2 Findings and General Conclusions

The extraction of <u>quantitative</u> generalizations from records of such capricious events as accidental fires and explosions should be approached with extreme caution and with application of extremely wide confidence limits, but a number of clear <u>qualitative</u> conclusions do emerge from this study. A show a straight a

ano kanziskai 14.

1

くのとうないろうと

As explained in Chapter 4, the VM models the combustion of the vapor-air cloud as either a deflagration, producing no significant overpressure, or a detonation, producing a shock wave of potentially destructive overpressure. This atrict dichotomy between deflagration and detonation is based on the classical Chapman-Jouget theory of combustion waves. As mentioned in Chapter 4, more modern studies indicate the possibility that deflagration (subsonic) combustion may, under certain circumstances, produce damagegenerating, finite amplitude blast waves. In this Chapter, the terms "explosion" and "blast" are used to refer to damage-causing overpressures without regard to the nature of the combustion (deflagrative or detonative) which gave rise to the overpressures. Thus "deflagrative air blast" refers to an event causing explosion damage but thought to originate from deflagrative, rather than detonative, combustion.

The Results of Vapor Cloud Ignitions Vary Widely

The consequences of vapor cloud ignitions range from a huge but simple fire through deflagrative air blasts to true gas detonations without any apparent way to make confident predictions. Fireballs ranging up to hundreds of feet in height and diameter are almost universally reported, and about half the incidents involve air blasts which break windows and strip metal sheets from nearby buildings but do not strip leaves from trees or shatter heavy structures. Detonations are rare; the four reported herein form a disproportionately large sample from the population. They are over-reported because their disproportionate destructiveness has attracted more attention and more detailed reporting than have the results of the lesser deflagrative blasts. Spills studied ranged from a few thousand pounds of hydrocarbon to more than 100,000 pounds, but there was no visible relationship between the size of the spill and any tendency to deflagration vs. detonation. No clear trigger mechanism for detonation was identified, either: two of the cases were attributed to burning under confinement; another is thought to have been a case of burning to detonation in the open; and the other case is thought to have been initiated by the shock wave from an exploding diesel engine.

Probably neither the upper nor the lower end of the damage spectrum has been found. The largest spill on which a report was found was approximately 100,000 pounds, which is still far less than the half million or so <u>barrels</u> of LNG carried by some of the larger ocean-going tankers. Calculations by Burgess et al. [39] offer hope that flammable spill clouds may have a maximum size limited in each case by diffusion and mixing rates; but the maximum blast possible -- if there really is such a limit -- has not been established. At the other end of the scale, small spills have received little study because they do little damage;

^[39] Burgess, D., J.N. Murphy, M.G. Zabetakis, and H.E. Perlee. Volume of flammable mixture resulting from the atmospheric dispersion of a leak or spill. To be submitted to the Fifteenth Symposium (International) on Combustion, The Combustion Institute, 1974.

and there is a widespread feeling that many of them are never even reported at all.

It may be that the overwhelming majority of all spills fail to ignite. That would be consistent with the reluctance of chemical plant operators to install pilot lights to insure the immediate ignition of spills and to prevent the growth of vapor clouds, but data on the point are not at hand.

Devastation is Never Total

والمتار ومقتله والمقالع وموجرا المارية والطراب التروي والمتكا المقاطرين والتساوية والمرار

One of the most striking features of aerial photos of disaster areas such as Crescent City (Case F.9) or Decatur (Case F.3) is the apparent <u>normalcy</u> of the scene. Houses and commercial buildings all look intact, automobiles are parked normally all along the streets, trees and shrubbery look untouched, and no smoke or flame is in evidence. Using high magnification, one can see broken windows and an occasional wrinkled roof; but one has to look hard. Damage there surely is, but the community is <u>not</u> destroyed as it is by, say, a tornado. त्रायात्रीय प्रतिकारणकार्यकार विदिक्षिति देवीक्षितिर्द्रतात्रव्यकार्यन्त्र त्यात्रिका का त्यात्र त्यात्र व्यक्ति

Į

Moreover, most of the damaged buildings are repairable at relatively minor cost. In fact, they typically get repaired, and the people go on living in them. Even in Decatur, where 67 residences were posted as unsafe, 90% of the damage was termed "minor." It was typified by broken windows and loose ceiling tiles, not by collapsed buildings.

The damage that does occur is distributed very unevenly. One house will suffer severe structural damage while the adjacent house is virtually untouched. Some windows in a given wall will be broken, whereas others in the same wall are intact, even deep within the general window breakage zone. The damage counts and the claim maps do not give a clear picture of this, because they concentrate on the damaged buildings and do not list the much larger numbers of essentially undamaged buildings intermingled with them.

The spottiness of blast damage is a puzzling but typical feature of explosions. It does not seem to be due to variations in the strength of the structures, but rather to random fluctuations in the strength of the advancing blast wave front. No one has yet modeled that in detail.

There is Seldom Severe Blast Damage Beyond Half a Mile

For spills of up to about 100,000 pounds of fuel, there appears to have been little severe blast damage at distances of more than about half a mile:

Case No.	Approximate amount of fuel spilled	Approximate limit of severe blast_damage	Location
F.2	5,000 lb. ethylene	1500 feet	Longv iew, Texas
F,4	27,000 lb. vinyl chlorid monomer	e 1500 feet	Climax, Texas
F.1	100,000 lb. cyclohexane	l/4 mile	Flixborough, England
F.5	100,000 lb. propylene	1/2 mile	East St. Louis, Illinois
F.3	117,000 lb. propane	1/2 mile	Decatur, Illinois
F.6	130,000 1b. propane	2 miles	Franklin County, Missouri

The exception of Case F.6 was a rare <u>detonation</u> rather than the usual deflagrative blast, and it is the only case of its kind so far known. Since it did happen, two miles might be taken as a more conservative limit; but half a mile is the more typical limit. Another exception is the occurrence of caustics -- the focusing of blast waves, by atmospheric inhomogeneities such as temperature inversions, back down into an anomalous area of severe damage well outside the general area of severe damage. However, caustics are usually an order less severe than the central damage area, and they do not seriously invalidate the generalization.

الهوالة والمشتر مسترد المجاهر المعطين المالا بالمشاملة والمناها والمسمان

and the second second second second second second second second second second second second second second second

5

Since the data are few, the above conclusion should be reviewed in the light of each new incident as it occurs. One should also keep in mind Burgess' point [40] that the size of a flammable or detonable cloud is a stronger function of wind and weather than it is of the size of the spill. Nevertheless, the severe blast damage actually observed in actual explosions has usually not exceeded about half a mile. The half mile damage limit may not hold for the more destructive detonative combustion cases.

The Central Fireball is Deadly

الرجايي يتواديها المراجع المربية المحاطات فالمطعفين فيتعاطينا والمراجع والمراجع

All the cases studied, plus a number of others which were not detailed enough for inclusion here, featured an enormous fireball which may or may not have been accompanied by a blast. In several cases (notably Cases F.13 and F.14), people were enveloped in the fireball, and they were almost always killed. The few who survived were severely burned. Buildings which were enveloped in such fireballs were ignited and usually burned to the ground. Fence posts and telephone and power poles were charred, and wires were melted.

The fireballs also radiate enormous amount of thermal energy and can ignite fires hundreds of feet beyond their edges. Report after report tells of firemen severely burned or forced to seek shelter hundreds of feet from fireballs, and a firemen's training film cited in Case F.11 warns that firemen have died from burns received as far away as 250 feet from large fireballs. In Case F.8, paint was blistered 600 feet from a fireball. In Case F.12, the paint on a fire truck was scorched and plastic light components were warped 300 to 450 feet from a fireball, and a bystander was burned 600 feet from the fireball. In Case F.10, a fireman 1600 feet from a fireball had to stop work and cover his head with his coat.

It has sometimes been anticipated that "flash fires" in stoichiometric fuel-air clouds would be so evanescent that little if any damage or injury would result from them. This study cannot address that point, but the real vapor cloud fires analyzed in this study all exhibited intensely hot, deadly, central fireballs. It would appear that such a fireball is an inherent feature of any spill large enough to cause

 ^[40] Burgess, D.G., and M.G. Zabetakis. Detonation of a Flammable Cloud Following a Propane Pipeline Break -- the December 9, 1970 Explosion in Port Hudson, Missouri. U.S. Bureau of Mines, Washington, D.C., 1973. Report No. RI 7752.

concern. As Burgess has illustrated in some deta. [40], the vapor cloud from a large fuel spill should consist of three concentric somes: an inner some too rich to burn, a shell wherein the composition is within the upper and lower flammable or explosive limits, and the rest of the world where the composition is too lean to burn. Ignition could occur only within the flammable shell, and the flame would travel outward to the lean limit and inward to the rich limit. At the rich limit, the flame would stop and continue to burn there as a diffusion flame at the interface between the rich cloud and fresh air brought in by turbulence. Fireballs do form. The cloud of heated gas would be expected to rise like a balloon, and fireballs have been reported with heights of up to 1,000 feet (Case F.9).

Saturnan said.

रूप्रदेशीयत्रयः ः । इत्यदिन्द्रः गोकेः 'रुक्रेन्द्रीयि में स्वलः स्वत्यर्थर्थये प्रभवत्र स्वत्रात्रीयव्यविद्या

In the context of the VM the combustion of that portion of the vaporair cloud premixed between flammable limits is modeled. However the turbulent diffusion flame resulting from the combustion of the rich portion of the cloud, which is believed to produce the deadly fireball, is not treated at present in the VM. This phenomenon is extremely complex and to be modeled would have required more time and resources than were available. Furthermore the great damage potential of this mechanism, as indicated herein, was not known at the time that the fire and explosion models were formulated.

A Firestorm May Follow a Large Gas Explosion

In World War II, one of the most fearsome effects of firebombings was the widespread firestorm which sometimes ensued. A ground fire of sufficient extent creates an overwhelming updraft which draws in high winds from all directions. The resulting flaming whirlwind depletes the oxygen of the air, and it is said that more people died from suffocation than from burns in World War II firebombings.

Something of that sort happened in Case F.6. "In the seconds following detonation, a firestorm was observed to 'roll' in a generally east to west direction -- up the sloping terrain toward [the] highway" [40]. A footnote explained "firestorm" as: "That is, a diffusion flame with very high winds which consumed the remainder of the propane."

In a sense of course, every billowing fireball is a small firestorm; but we are talking here about the much worse area firestorm that results from the coalescing of many fireballs or the burning of a very <u>large</u> vapor cloud. Case F.6 is the only example encountered wherein a firestorm was specifically mentioned; but it was also the largest fuel spill encountered, and the incidence of firestorms would be expected to increase with the size of the spill. Thus an increase in spill size may change the possible class of events that follow from a spill, rather than just changing key parameters in a fixed set of consequences.

Exploding Tank Segments Can Fly up to Half a Mile

If an accident involves liquefied fuel gas tanks enveloped in a fire, still another hazard mechanism exists: the internal pressure in the tanks builds up to the bursting point, and the tanks rupture explosively. They not only then release enormous quantities of hot gas to make a fireball, but the tank segments fly off like rockets for distances up to half a mile, inflicting impact damage where they land and setting new, spot fires there. In Case F.9, exploding LPG tankcars hurled blazing segments up to 1750 feet into the heart of Crescent City where they smashed buildings and set fires. In Case F.10, the same thing happened in the center of Laurel, Mississippi. It happened in the single-tanktruck fire of Case F.12 on the New Jersey Turnpike. The overpressure safety values on such tanks are not adequate to relieve the excess pressure and prevent the explosions. It is truly a rocket effect; long, cylindrical, tank segments fly farther than shorter segments and much farther than random fragments hurled by the rupture explosion.

This hazard is an important one, because such tanks are to be expected where gaseous fuels are being handled. The effect where a tank hits is comparable to that from wartime demolition and incendiary bombs.

1.3 Data Gaps

This study, although instructive, has raised as many questions as it has answered. One wants more detail on the cases analyzed, one wants more cases, and one wants information on larger explosions and on marine spills.

ł

The case studies found in this survey have typically had disappointingly little detail on damages and injuries. They have been heavy on the dollar value of damages and on numbers of deaths, because those data fulfilled the purposes for which they were made; but they have been relatively light on the fine details needed to break new ground in the understanding of explosion effects. The more recent case studies -particularly those by the National Transportation Safety Board -- have contained more detail as awareness of its value has emerged; but one needs much more yet. One can find map plots of buildings made unsafe, for example; but one also needs to know just what kind of damage made each one unsafe, and which side of the house faced the blast. One also needs to know the condition of the adjacent houses which, being unmentioned, were presumably not such damaged. One needs an actual count of broken windows, house by house and street by street. One needs to know how many houses were in a given area as well as how many were damaged. One needs similar detail on injuries.

Much of this detail is gone forever, but much of it could still be reconstructed by sifting through raw insurance and newspaper archives, by study of the stacks of record photographs which were made in some cases and of city maps, and by field work at the explosion site. Such indepth studies were far beyond the resources provided for this program.

One wants more cases, to test the tentative conclusions drawn from the ones analyzed, and yet the data scatter seen thus far make one doubt the value of simply more of the same. There are more industrial explosion cases sitting in people's files, but their collection is not strongly recommended unless they can be studied in much greater depth than was possible on this program.

One would like data on larger explosions (both deflagrative and detonative explosions). This study found and analyzed explosions of up to 100,000 pounds of fuel (about 50 MT \cong 83 m³), but data on a million-pound blast resulting from a spill would add perspective (1 million pounds \cong 500 MT \cong 830 m³). Theoretical considerations suggest that the blast damage would by no means increase proportionately, but the point needs confirmation.

One would like data on spills which did not result in ignition, because they ought to contain clues on how to prevent ignition. Such events are not major accidents, of course, and seldom get much attention. The unignited Pensacola spill of hot cyclohexane is a fortunate exception in that some detail was recorded and in its frightening similarity to the Flixborough spill which did ignite. It would probably repay deeper study even at this late date. One would also like to know -- if only for perspective -- how many nondisastrous spills there are per year in comparison to the number of disastrous ones.

Further Studies that Appear Worthwhile

Although there are more industrial and transportation explosion cases which could be collected, a further canvass is not strongly recommended. The chances are that they would just be more of the same and would not add greatly to the picture. It would be far more productive to go back and reinvestigate selected, already identified accidents in much greater detail, and the following cases are recommended.

• The June 1974 cyclohexane explosion at Flixborough

This explosion is fresh enough that it is still under intensive study, and much of the physical evidence is still available. It is one of the largest disasters in terms of dollar value of damage, and it offers an unusually dense gridwork of damage effects to indicate ranges of effects. It is uniquely instructive in that it appears to have involved a rare open-air detonation, and in that it can be compared to an almost exactly similar spill that did not ignite. The European investigation appears to be impressively thorough and would provide an unusually complete picture.

• The July 1972 propane explosion in Decatur, Illinois

This explosion is still under investigation by the National Transportation Safety Board, and their file of observations and photographs is unusually extensive. It is fresh enough that local memories and records will still be vivid and accurate for still more detail. It is accessible for on-the-scene, streetby-street review of the patterns of damage and injuries. Occurring as it did in a densely built, urban area, it offers a rich lode of damage data with valid, side-by-side comparisons available.

• The 1944 LNG spill in Cleveland, Ohio

This accident is thirty years old, so there is nothing fresh about it; but by the same token there is also no longer anything sensitive about it and existing data ought to be freely available. Morever, it was investigated by Zabetakis, of the Bureau of Mines, which suggests that the record will be thorough and comprehensive. It is also of interest because it was a large spill of LNG in an urban area. • The 1969 LPG tankcar explosions in Laurel, Mississippi

This accident is recommended partly because of the unusually large number of buildings suffering damage and partly because of the existence of a very tantalizing claims map at Southern Railways. Two different kinds of claims are plotted on it within a radius of about five miles, but no one was found who remembered what they were! The information could doubtless be recovered from Southern's files, and their claims office has offered generous cooperation. The map also does not indicate the density of undamaged buildings, but this too could readily be reconstructed from local information. المتعاطية والمستحدث والمتعارين والمستحد والمتعالية والمستحدة والمتعارين والمستحدث والمتعارين والمستحد والمستحد

.

ात्र विदियोग् सम्बद्धाः स्वाम्याद्वारः अन्तद्वीति दिवित्र सन्दिनि स्वित्यान् जिन्द्

2. PRIVATE SECTOR SAFETY RESEARCH AND DEVELOPMENT PLANS

Literally <u>no</u> planned, privately sponsored, explosion safety research and development programs were found in the course of this survey. The study was not exhaustive of course, but five of the largest fuel and chemical companies were included, and four major R&D companies who have been actively attempting to market such programs were included; and it is unlikely that so large a sample would fail to uncover any significant programs under consideration. Respondents interviewed on the point included:

Fuel and chemical companies and associations:

- Exxon Chemical Company USA
- El Paso Algeria Company
- Institute of Gas Technology
- Texas Eastman Company
- Monsanto Company

R&D companies:

- Calspan Corporation
- Science Applications, Inc.
- Systems, Science and Software
- Aerotherm-Acurex Corporation

Some commercial companies have been contemplating safety R&D programs, but their managements are currently taking the position that safety R&D is not a proper unilateral activity. They point out that any useful results therefrom would be instantly shared with the community rather than being used to advance the company's competitive interests, and that therefore such R&D is not a proper use of funds derived from one company's earnings. In the same breath, though, they go on to say that they would consider joining in, and contributing to the funding of, multicompany safety R&D programs, perhaps at the trade association level. There is precedent for such joint R&D. Trade associations such as the American Gas Association and the American Petroleum Insitute have sponsored R&D on common problems, using members' funds, for many years, and their work is well known. There are less formal associations, too. For example, an Ad Hoc Committee of 25 major chemical companies was formed in 1971 under the leadership of W. H. Doyle, then of Factory Insurance Association and since retired, to support work on the effect of water sprays on the burning of flammable gases. This work has been completed and will soon be published, probably first at an AIChE Loss Prevention Symposium. Two members of the Steering Group were interviewed by the writer; and it is the writer's opinion that this Ad Hoc organization, or one descended from it, could be induced to support additional safety R&D, given some meritorious proposals and assurance of broad participation.

Ideas for safety R&D abound. A detailed report on ideas which have not crystallized into plans is beyond the scope of this study, but they include the following.

- Modeling of the formation and diffusion of gas clouds.
- Modeling of deflagrative explosions.

the interaction of the second second

K

Ì

.

į

1

- Determination of the TNT equivalent of large gas explosions.
- Study of cloud control and flame suppression.

and the part of the second

- Study of ignition limits and extinguishants.
- Analysis of gas container vulnerabilities.
- Modeling of fireballs and radiation fluxes.
- Development of pool fire models.
- Study of transition from deflagration to detonation.
- A more quantitative understanding of caustics.
- Study of the electrostatic ignition of gas clouds.
- Study of flame arrestors.
- Research on atmospheric stability factors.
- Research on maximum cloud and fireball size.

In the course of the present study, the writer was shown a number of proposals and capability documents on the above subjects. They were mostly "not for citation" and none of them was scheduled for early implementation. and the second second

The American Gas Association has a <u>recommended</u> plan based on an industry-wide poll of LNG research needs, both foreign and domestic. It recommends over \$5 million in safety R&D. A detailed analysis of the plan is beyond the scope of this study, but the chart that follows illustrates its general outlines.



دكم بلغ ويعادكه كالطيلال الالعب

رهاد واليكور الجزارة فإسجاده كميتهم منابعها مدارهه بملتز متدابخ يعارمه

AGA LNG Research Plan. (After Sarkes [41])

The Plan is available from AGA, and a review of it was published at the Fourth International Conference on Liquified Natural Gas in Algiers on 24-27 June 1974 [41].

Ŭ

With research ideas aplenty and with potential co-sponsors not unwilling to participate on a team basis, what is mainly needed is some leadership and the selection of a <u>limited</u> number of <u>specific</u> projects. Some seed funding would probably be needed to attract industrial funding, but the most pressing need is for organization.

1,47

^[41] Sarkes, L.A. A survey of LNG technological needs in the USA - 1974 to beyond 2000. Presented at the Fourth International Conference on Liquified Natural Gas, Algiers, 24-27 June 1974.
3. **RESPONDENTS INTERVIEWED**

The following individuals and organizations were canvassed and interviewed as to their accident experiences and R&D plans.

• Exxon Company USA, Houston, Texas

Personal visit. Discussion of gas spills in Exxon facilities and possible R&D programs.

• El Paso Algeria Corporation, Houston, Texas

Telephone interview; a visit could not be scheduled due to host's travel commitments. No spill incidents on record and no R&D plans.

• LP Gas Administration, Oklahoma City, Oklahoma

Personal visit. Search through files of investigations of local gas spills and resulting fires and explosions.

• Monsanto Company, St. Louis, Missouri

Personal visit. Discussions of cyclohexane spills like Flixborough, spread of gas clouds, and ad hoc committee safety R&D.

• Professor Roger Strehlow, University of Illinois

Personal visit. Discussions of East St. Louis and Decatur explosions, NT: 3 studies, calculation of TNT equivalents and further information sources.

• Institute of Gas Technology, Chicago, Illinois

Telephone interview, following a visit to the Washington office. A Chicago visit could not be scheduled. No accidents on record and no R&D plans.

Texas Eastman Company, Longview, Texas

Personal visit. Discussions of ethylene cloud explosions and water sprays. Range of blast damages.

• Calspan Corporation, Buffalo, New York

Personal visit. Search of clipping tiles of accidents. Discussion of potential R&D programs.

• Systems, Science & Software, La Jolla, California

Personal visit, telephone interview and correspondence. Discussions of industry R&D plans and S^3 proposals. Literature exchange.

• Science Applications, Inc., La Jolla, California

Personal visit plus an escorted visit to Enviro Control. Discussion of R&D programs.

- Aerotherm Division of Acurex Corporation, Mountain View, California Telephone interview plus evening conference at JANNAF-AIAA meeting. Safety R&D proposals.
- Bureau of Mines Explosive Research Center, Pittsburgh, Pennsylvania

Telephone interview plus correspondence. Discussions of gas explosions and specific incidents. References to additional contacts.

• Naval Surface Weapons Center, Dahlgren, Virginia

Personal visit, with Coast Guard personnel. Discussions of Navy data on cloud travel and instrumentation. Introduction of Coast Guard to Navy computerized data base.

• Exxon Research and Engineering Company, Florham Park, New Jersey

Multiple telephone interviews. Discussions of explosion in Bayway facility not for citation because in litigation. Discussions of past safety R&D on LNG.

• American Gas Association, Washington, D.C.

Telephone interview; visit could not be scheduled. Discussion of Gas Industry Research Plan - 1974-2000.

• National Transportation Safety Board, Washington, D.C.

Study of files on recent accidents still under investigation and collection of issued reports. Discussion of characteristics of accidents and evolosions.

• Mr. W. H. Doyle, retired, formerly of Factory Insurance Association

Telephone interview; visit could not be scheduled in time available. Chairman of the 25-company, ad hoc committee that sponsored the study of water sprays.

• Southern Railway System, Washington, D.C.

Personal visit. Further details on the Laurel, Mississippi, tank car explosions.

• Federal Railway Administration, Washington, D.C.

Personal visit. Additional details on Climax, Texas, vinyl chloride tank car explosion.

• Mr. D. H. Slater, Cremer and Warner, Consulting Engineers, London, England

Correspondence. Additional details on the Flixborough cyclohexane explosion.

The following additional respondents were listed for interview, but visits could not be arranged in the time remaining for this study.

- Naval Weapons Center, China Lake, California
- Stanford Research Institute, Menlo Park, California
- Dow Chemical Company, Midland, Michigan
- Factory Mutual Research Corporation, Norwood, Massachusetts
- Eglin Air Force Base, Florida

Į

ŗ

• BuMines Enforcement and Safety Administration, Washington, D.C.

مناه فلغ الجم دي من من ما يتكلي الله المن المن أن من المحمس منه المالة المالة الكلم منهم منه المالة الكلم منه م

The second second second second second second second second second second second second second second second s

- Air Force Office of Scientific Research, Washington, D.C.
- National Fire Protection Association, Boston, Massachusetts
- Cities Service Company, Lake Charles, Louisiana
- PPG Industries, Lake Charles, Louisiana
- Union Carbide Company, South Charleston, West Virginia
- American Petroleum Institute, Washington, D.C.
- American Chemical Society, Washington, D.C.
- Manufacturing Chemists Association, Washington, D.C.

CHAPTER 10

THE - T

1

. 24

ويعينك أليميا والمسارمين

CONCLUSIONS AND RECOMMENDATIONS

Conclusions

The primary conclusion of this study is that the concept of the VM is feasible for use in risk analysis. The material required to operate such a model, viz., the input data and the predictive submodels, is either available or attainable. Although some aspects of the modeling are not at the highest level of sophistication, the VM yields results that appear to be consistent with the historical survey of actual accidental spills. Technical details of conclusions drawn from the operation of the VM are given in Chapters 7 and 8. Some of the more significant conclusions are listed below.

- The injuries and damage resulting from the detonation of a vapor cloud are highly dependent upon the exact nature of the air dispersion. This dependence upon the spill characteristics (location of spill, rate of gas release) and the atmospheric conditions (temperature, wind, stability) is a realistic reflection of the complexities of the actual world and is not considered to be an inaccuracy or quirk in the modeling used.
- The injuries resulting from the release of toxic gas into the air are highly dependent upon the nature of the air dispersion, just as in the case of explosion. This is a realistic aspect of the simulation.
- In general the parameters that are input to the VM have a certain level of importance in determining damage; three such parameters, listed in descending order of importance, are:
 - (1) spill size
 - (2) stability class
 - (3) wind speed

1

For flammable and toxic cargoes there is a spill size below which the VM does not assess any damage even though some may occur in reality; this minor anomaly is primarily the result of using in the VM a grid cell system of finite spacing.

The simulation shows that for each flammable cargo spilled under a given set of environmental conditions, there is a critical spill size such that spills smaller than this do not cause any damage. This occurs because by the time mass has gone from the spill into the air in a large enough amount to cause damage, the fuel-air cloud is too diffuse to ignite. That is, this critical size is that for which the evaporation rate is slow enough relative to the dispersal rate, that the fuel concentration is never above the lower flammable limit except within a few tens of meters of the spill site. This aspect of the simulation is only qualitatively correct. In the real world, the damage-causing potential of a spill decreases as the spill size decreases, but it does not necessarily go to zero. The finite size of the grid cells with their associated ignition points and the limitations on the correct application of the Phase I submodels only on geographical scales over tens of meters cause the simulation to predict zero damage for spills below a certain size when, in fact, a small damage-causing potential could be present.

المشقدة بالمعمدين

Similarly for each toxic cargo spilled under a given set of environmental conditions, there is a critical spill size such that spills smaller than this are not capable of causing significant injury. This occurs because toxic material is dispersed to below an injurious concentration level at distances close to the spill. That is, the evaporation rate is slow enough relative to the dispersal rate so that the concentration is never above the threshold for inhalation injuries except within a few tens of meters of the spill site.

Recommendations

and the second states of

Jan Statistic Street, or a

ż

1

2

ģ

;

Recommendations are given for improvements to specific submodels, aspects of submodels, overall model architecture, and data sources, collection, and analysis. Improvements to the VM should be considered in view of (1) need, and (2) feasibility. The need for an improvement is determined by the magnitude of changes in output produced by making the improvement. The greater the change effected, the greater is the need. The feasibility of an improvement is determined by the facility, both technical and financial, with which the improvement can be made. Improvements, which are more tenable technically and lower in cost, are more feasible.

On this basis recommendations arising from this study may be grouped in four categories.

- (1) Recommendations for improvements of greatest need without regard to feasibility. These improvements will have a large effect on the output of the VM.
- (2) Recommendations for improvements of lesser, but significant, need and unquestionable feasibility. These improvements will have a significant effect on the output of the VM and are both technically and financially feasible.
- (3) Recommendations for improvements of questionable significance or feasibility, but not both. These improvements are either (1) of significant effect, but questionable feasibility, or (2) of questionable significance, but unquestionable feasibility.
- (4) Recommendations for improvements of questionable significants and feasibility.

The degree to which any of these improvements can be implemented is, of course, largely dependent on the amount of resources, time and level of effort, that is allocated. In addition, however, improvements, especially to specific aspects of modeling and to overall model structure, are further constrained by factors inherent to the VM.

- Improvements requiring an extensive increase in the amount of data input are undesirable since an exorbitant cost for data collection, manipulation, and computer storage will make usage of the VM impractical.
- Improvements requiring an extensive increase in computer running time and/or storage requirements will make operation of the VM too expensive to be useful; the use of certain very accurate, but costly, finite difference models is therefore precluded.

Within the context of these constraints, recommendations for further work to improve the VM are listed below.

Recommended Improvements of Greatest Significance:

(1) The VM will be modified to assess deaths and injuries suffered by indoor sheltered populations due to all relevant damage mechanisms. Consideration will be given to the varying degrees of shelter afforded by different types of structures. For injury caused by inhalation of toxic gases, assessment of injury to the indoor population will involve considerations of seepage of the toxic substance into structures.

For injury caused by explosion, assessment of injury to the indoor population will involve the degree of shelter afforded by a particular building type and also the degree of shelter afforded by the placement of the people in the building.

Injury to persons inside buildings as a result of flash fire or pool burning is much less likely to occur as a primary damage mechanism than injury to sheltered populations from explosion or toxic substances.

The treatment of death and injury to people inside structures is feasible and will have a significant effect on the damage calculation performed by the VM.

(2) The damage assessment models should consider secondary (nonimmediate or not direct) damage mechanisms; for example, a spill of flammable liquid ignites producing a moderate fire hazard, but this spill-based fire ignites a nearby petroleum refinery producing fires and explosions that cause calamitous damage.

- (3) An effort should be made to restructure the VM and the rerequired submodels to allow simulation of phenomena actually occurring simultaneously as such. Thus, for example, spilling, evaporation, and air dispersion are currently modeled as a sequence of events, even though all three may happen at the same time. Restructuring Phase I of the VM in this way appears to be feasible, although not necessarily easy. For certain simulations, a quite different picture will result if events are modeled simultaneously instead of sequentially.
- (4) The movement of vulnerable resource populations with time of day, season of the year, and other factors should be researched and modeled to give a more realistic description of the resources at risk. For example, the census tract data currently used give low population densities to commercial etc., even though the population density in such locations can be very high at selected times. It appears to be feasible to account for population shifts caused by work, recreation, and travel. Shifting of the vulnerable resources will, of course, strongly affect the damage assessment.
- (5) An alternative to locating the ignition sources at discontinuous points (cell centers) should be considered. A treatment is desired which would consider ignition potential as a true function of area, rather than the current treatment which considers ignition sources to be concentrated at the center of a grid cell. By describing ignition sources in a manner other than locating the sources at discrete sites (the cell centers), a more realistic assessment of time to ignition and distance between ignition point and spill may be obtained. Such a treatment of ignition sources is feasible and is expected to have a significant effect on damage assessment results by changing the time for ignition.

Recommended Improvements of Definite Peasibility and Significant Need:

- (1) The flash fire should be modeled so that damage beyond the flame location is allowed. The calculation of radiation levels at locations distant from the flame is definitely feasible. Some sources report significant damage to both personnel and property at locations moderately far from fires.
- (2) Modeling should be added to permit accurate treatment of subsurface spills. The release of cold, soluble chemicals underwater is of special interest. Such modeling is definitely feasible. The ability to treat this additional class of spills will enhance the utility of the VM.

(3) The VM should be modified to treat the ingestion of water containing toxic concentrations of pollutant. Estimating a specific level of injury or percent of population injured is not possible since the quantity and rate of ingestion are highly variable and unpredictable. What is required is some indication of the toxic hazard presented by a given concentration of spilled substance. One possible approach is to establish levels of hazards. The VM should be modified to inform the user when and where these hazard levels exist in the water body. This modification is definitely feasible and will make the output of the VM significantly more useful. فكالالا والمعارفة والمعالية و

بالمليان

والمسارك والمستعلم والمستعمل والمست

· • •

Ş

• •

- (4) The VM should incorporate a better treatment of the ignition of structures. A more precise method to account for shielding from thermal radiation should be devised. The current criterion that 25% of the structures are ignited in a cell subject to radiation sufficient to cause ignition should be replaced by a criterion that has a variable percentage ignited and the percentage ignited is to be calculated on the basis of physical principles. This improvement is feasible and will significantly affect the damage assessed to structures.
- (5) Further toxicological considerations are required so chat additional toxic substances may be treated by the VM. Since the additional toxic substances may be other than irritant gases, the assessment procedures used for the added substances may differ from those used previously for NH₃ and Cl₂. The additional toxicological considerations are feasible and will enhance the applicability of the VM.
- (6) The water mixing model should be expanded to include water mixing and reaction. Such modeling is certainly feasible, but separate models may be required for each class of reactions. The inclusion of this modeling in the VM will increase the number of substances for which spills can be simulated.
- (7) The capability to treat spills of dissolvable solids, liquids, and gases should be added to the VM. The treatment of substances with finite solubilities is feasible. The ability to simulate substances of this type will enhance the value of the VM.
- (8) Advantages may result from treating the population density of vulnerable resources in a continuous rather than a discrete, cell-based manner. The use of discrete cells requires that a greater number of smaller cells be used in order to obtain greater accuracy for a simulation over the same geographical area. Since the damage simulated is often highly localized, only part of the detailed Gata base, maintained and used at considerable cost, is actually required for a given simulation. It is feasible to treat vulnerable resources in a manner such that the geographical area impacted by damage is defined and the vulnerable resource data for that area only are retrieved. Such a change in the treatment of vulnerable resource data will make the VM more precise and more efficient.

(9) Computational efficiency may be increased by the use of lookahead techniques. Preliminary calculations may be made to estimate the region impacted by damage mechanism. Detailed calculations and data retrieved need only be performed for the region estimated to be affected. Savings in computer time and cost will result. Such look-ahead techniques are certainly feasible.

3297737314995

14.14

÷.

*

:

(10) The VM should be modified to model the heating, rupture, and release of pressurized cargo in a fire. These phenomena could result in significantly different damages than release of a cargo without additional heating. The modeling is feasible.

Recommended Improvements of Questionable Feasibility or Need, But Not Both:

លេះដែរដែរ ស្រុះដឹងដែរសារីដ

- (1) The analytical and experimental background applicable to "the roiling fireball." The roiling fireball seems to be the very rich, but gas phase, portion of a flammable vapor cloud that results in a burning process significantly different from the diffusion flame described by the pool burning model or the combustion of the premixed cloud described by the flash fire model. The roiling fireball should be investigated to determine whether and how this phenomenon might be modeled because as suggested by the historical survey, this may be a significant damage mechanism. The roiling fireball is distinguished from the flash fire by: (1) combustion in the flash fire is rapid, compared to combustion in the fireball; and (2) the burning mixture in the fireball is rich and is supported by turbulent diffusion, whereas the burning mixture in the flash fire is premixed within flammable limits. It is recognized that this phenomenon may also induce significant overpressures. Significance is moderate to great; feasibility is questionable.
- (2) The VM should be modified to treat damage caused by the inhalation of toxic combustion products. Although some limitations of the ability to treat injury from toxic combustion products may not yield to analysis, the model should be modified at the very least to inform the user of the existence of this damage mechanism. Significance is moderate; feasibility, questionable.
- (3) Injury by asphyxiation should be addressed. Although an extensive effort in this area is not envisioned, some means of determining the seriousness of this damage mechanism is desired. The assessment approach should take into account the time varying concentration of the asphyxiant. For substances which exhibit both asphyxiant and toxic effects (e.g., dichlorodifluoromethane), the problem of combined effects needs to be addressed. Significance is indeterminate; feasibility, definite.

- (4) The air dispersion model should be made more realistic by incorporating the ability to treat the arbitrary time variation of source strength and source location. Significance is unknown; feasibility, definite.
- (5) The pool burning submodel should include a more realistic treatment of simultaneous spreading and burning. Feasibility is definite; significance is unknown.
- (6) The VM should be modified so that the release and migration of heavy insolubles on river beds can be modeled. Feasibility is definite; significance is unknown, since heavy insolubles would appear to offer less of a threat to people and property than other classes of substances.
- (7) The effects of water wave action and of wind on the spread of a surface spill should be considered. Feasibility is definite; significance, unknown.

- (8) The air dispersion model should be extended to include the dispersion of reacting chemicals. Feasibility is definite; significance is questionable, since spills of relatively few cargoes would seem to result in the air dispersion of a reacting substance.
- (9) Air dispersion by gravity spreading in calms should be modeled. Feasibility is definite; significance, unknown.
- (10) Micrometeorological effects, and meandering effects, presently treated analytically, should be implemented in the computer models. Feasibility is definite; significance, unknown.
- (11) Instead of computing damages at a given sequence of cell centers, techniques capable of yielding isodamage contours would be, in some respects, more appealing. Feasibility is definite; significance, unknown.
- (12) Buoyancy considerations should be incorporated into the air dispersion modeling. Feasibility is definite; significance, unknown.
- (13) Consideration of topological features should be incorporated into the air dispersion models. Feasibility is definite; significance, unknown.

Recommended Improvements of Questionable Feasibility and Need:

(1) The effect of explosion on the liquid spill should be considered; it is conceivable that streams of flaming liquid could be sprayed long distances by a vapor cloud exploding over a spill.

- (2) The assessment of damage from inhalation of toxic fumes should consider that the toxic concentration varies in space and time in a stochastic manner; the receptor response to these concentration variations may be different from the currently modeled response to an average concentration.
- (3) Further consideration should be given to the significant ways in which diffuse explosions differ from conventional explosions, both in physical characteristics and in damage phenomena.
- (4) The effects of precipitation should be assessed, and modeled as required, for the air dispersion, surface spreading, and water mixing submodels.

Recommended Improvements to Data:

موالمستناك كمحكاكاتكان وحادكمكانك لالالور

A state of the second s

ţ

The following are data items that may be available from investigations of accidental spills and are useful for improving the VM modeling; in all cases it is desired that the data gathered be as precise and complete as possible. These recommendations are included at the direction of the Risk Analysis Advisory Board of the USCG.

- I. In the case of damage from toxic materials:
 - A. Source and Spill Development Data
 - 1. Chemical and physical nature of the spilled substance
 - 2. Quantity spilled
 - Local weather conditions prevailing at the time of spill

 a. wind speed and direction } each as a function of
 - b. temperature 'height, if available
 - c. inversion characteristics
 - d. humidity and precipitation
 - 4. Visual or photographic observations of the travel of a visible toxic cloud
 - B. Receptor Data
 - 1. Best possible estimate of population at risk
 - 2. More precise definition of:
 - a. nature of effects on receptors
 - b. duration of effects on receptors
 - r. severity of effects
 - 4. treatment provided affected receptors
 - e. state of health of receptors prior to incident
 - f. age of receptors
 - 3. Effects on nonhuman receptors animals and plants

II. In the case of damage from fire and/or explosion:

Α. Source and Spill Development Data

- 1. Chemical and physical nature of the spilled substance
- 2, Quantity spilled
- 3. Sequence of spillage and ignition
- 4. Data
 - a. wind speed and direction each as a function of
 - } height, if available b. temperature
 - c. inversion characteristics
 - d. humidity and precipitation
- 5. Observations of cloud size and travel before ignition

B. Receptor Data

- 1. Damage maps
 - a. location of dead and cause of death
 - b. location of injured and severity of injuries
 - c. location and severity of physical damage
- 2. Extensive photography

Closing Remark

The listing of such a large number of recommendations was intended to point the way toward areas related to the VM for which additional effort is believed to have the greatest potential for benefit. This list of recommendations is not, nor is it intended to be, an indictment of the VM. To the contrary the VM, in this its first stage of development, is believed to be a useful, practical tool for use in the risk analysis of marine spills. Further development of the VM will make this already functional tool more useful, more precise, and more widely applicable.

REFERENCES

- Dunn, W.A., and P.M. Tullier. Spill Risk Analysis Program Phase II Methodology Development and Demonstration. Operations Research, Inc., Silver Spring, Md., August 1974. USCG Tech. Report #840.
- [2] Raj, P.P.K., and A.S. Kalelkar. Assessment Models in Support of the Hazard Assessment Handbook. Arthur D. Little, Inc., Cambridge, Mass., January 1974. USCG Report No. CG-D-65-74 (NTIS AD 776617).
- [3] Hazard Assessment Computer System (HACS) User Manual. Arthur D. Little, Inc., Cambridge, Mass., December 1974.
- [4] Raj, P.P.K., J. Hagopian, and A.S. Kalelkar. Prediction of Hazards of Spills of Anhydrous Ammonia on Water. Arthur D. Little, Inc., Cambridge, Mass., January 1974. D.O.T. Report No. CG-D-74-74 (NTIS AD 779400).
- [5] Slade, David H. (editor). Meteorology and Atomic Energy. U. S. Atomic Energy Commission, July 1968.
- [6] Tyron, G.H. (editor-in-chief). National Fire Protection Association, Boston, 1969.
- [7] Eckert, E.R.G., and R.M. Drake. Heat and Mass Transfer. McGraw-Hill, N. Y., 1959. Ch. 14.
- [8] Pasquill, F. The estimation of the dispersion of windborne material. Meteorology Mag. 90(1063):33-49, 1961.
- [9] Pasquill, F. Atmospheric Diffusion. Van Nostrand, London, 1962.
- [10] Turner, D. Bruce. Workbook of Atmospheric Dispersion Estimates. Environmental Protection Agency Publication No. AP-26, Revised 1970, pp.37-38.
- [11] Stewart, N.C., H.J. Gaie, and R.N. Crooks. The atmospheric diffusion of gases discharged from the chimney of the Harwell Reactor BEPO. Int. J. Air. Pollution 1:87-102, 1958.
- [12] Cramer, H.E. Engineering estimates of atmospheric dispersal capacity. Amer. Ind. Hyg. Assoc. J. 20(3):183-189, 1959.
- [13] Nonhebel, G. Recommendations on heights for new industrial chimneys. J. Inst. Fuel 33:479-513.
- [14] Fay, J.A. Unusual fire hazard of LNG tanker spills. Combustion Sci. Tech. 1:47, 1973.
- [15] Feldbauer, G.W., et al. Spills of LNG on Water Vaporization and Downwind Drift of Combustible Mixtures. Esso Research & Engineering Company, Report No. EE61E-72 (Released by the Am. Petroleum Institute, Re 6232), March 1973.

Preceding page blank

- [16] Strehlow, Roger A. Unconfined vapor-cloud explosions an overview. In Proceedings of the International Symposium on Combustion, 1972.
- [17] Brown, John A. A Study of the Growing Danber of Detonation in Unconfined Gas Cloud Explosions. John Brown Associates, Inc., Berkeley Heights, N. J., December 1973.
- [18] Kinney, Gilbert F. Explosive Shocks in Air. The Macmillan Co., New York, 1962, p.11.
- [19] Strehlow, Roger A. Equivalent Explosive Yield of the Explosion in the Alton and Southern Gateway Yard, East St. Louis, Illinois, January 22, 1972. Engineering Experiment Station, College of Engineering, University of Illinois, Urbana, June 1973. AAE TR 73-3, UILU-ENG-73 05-05.
- [20] Kinney, Gilbert F. Engineering Elements of Explosions. Naval Weapons Center, November 1968. NWC TP-4654.
- [21] Kirk, P.L. Fire Investigation. John Wiley & Sons, Inc., New York, 1969, p.193ff.
- [22] Hawkins, S.J., and J.A. Hicks. A New Explosives Technique for Synthesizing a Wide Range of Pressure Waveforms in Air, Part 1: Approximate Theory of Air Blast from Extended Explosive Charges. Ministry of Technology, Explosives Research and Development Establishment, Waltham Abbey, Essex, October 2, 1968. Report No. ERDE 9/R/68.
- [23] Woolfolk, R.W., and C.M. Ablow. Dependence of the blast wave from an explosion on the energy release rate. In Proceedings of the Fifteenth International Symposium on Combustion, August 1974.
- [24] Strehlow, R.A., L.D. Savage, and G.M. Vance. On the measurement of energy release rates in vapor cloud explosions. Combustion Science and Technology 6:307-312, 1973.
- [25] Strehlow, R.A., and A.A. Adamczyk. On the Nature of Non-Ideal Blast Waves. Engineering Experiment Station, College of Engineering, University of Illinois, Urbana, April 1974. AFOSR-TR-0834.
- [26] Karim, G.A., and P. Tsang. Flame propagation through atmospheres involving concentration gradients formed by mass transfer phenomena. Presented at ASME-CSME Fluids Engineering Conference, Montreal, Quebec, May 12-15, 1974.
- [27] Lewis, B., and G. von Elke. Combustion, Flames and Explosions of Gases. Academic Press, Inc., New York, 1951.
- [28] Kutateladze, S.S., and V.M. Borishanskii. A Concise Encyclopedia of Heat Transfer (J.B. Arthur, trans.). Pergamon Press, New York, 1966.
- [29] U.S. Department of Commerce, ureau of the Census. 1970 Census of Housing, Block Statistics, September 1971.

[30] Finney, D.J. Probit Analysis, 3rd edit. Cambridge University Press, 1971.

Ł

1

1

1

[31] Kowytz, T.A., et al. Arch. Environ. Health 14:545, 1967.

Sector Sec

11-1-1-1-1-

- [32] Chasis, H., J.A. Zapp, J.H. Whittenberger, J.L. Helm, J.J. Doheny, and C.M. MacLeod. Chlorine accident in Brooklyn. J. Occup. Med. 4:152, 1947.
- [33] Joyner, R.E., and E.G. Durel. Accidental liquid chlorine spill in a rural community. J. Occup. Med. 4:152-154, 1962.
- [34] Weill, H., G.M. Schwarz, and M. Ziskind. Late evaluation of pulmonary function after ecute exposure to chlorine gas. Am. Rev. Respiratory Disease 99:374-379.
- [35] Longinow, A., G. Ojdrovich, L. Bertram, and A. Wiedermann. People Survivability in a Direct Effects Environment and Related Topics. IIT Research Institute, Chicago, May 1973.
- [36] U.S. Department of Transportation, National Transportation Safety Board. Railroad Accident Report (Chicago, Burlington and Quincy Railroad Company, Train 64 and Train 824, Derailment and Collision with Tank Car Explosion, Crete, Nebraska, February 18, 1969). Report No. NTSB-RAR-71-2, Washington, D.C., February 24, 1971.
- [37] Radanovic, L. (editor). Sensitivity Methods in Control Theory. Pergamon Press, New York, 1966.
- [38] Tomovic, R. Sensitivity Analysis of Dynamic Systems. McGraw-Hill, New York, 1963.
- [39] Burgess, D., J.N. Murphy, M.G. Zabetakis, and H.E. Perlee. Volume of flammable mixture resulting from the atmospheric dispersion of a leak or spill. Advance copy, to be submitted to the Fifteenth Symposium (International) on Combustion, The Combustion Institute, 1974.
- [40] Burgess, D.G., and M.G. Zabetakis. Detonation of a Flammable Cloud Following a Propane Pipeline Break -- the December 9, 1970 Explosion in Port Hudson, Missouri. U.S. Bureau of Mines, RI 7752, 1973.
- [41] Sarkes, L.A. A survey of LNG technological needs in the USA 1974 to beyond 2000. Session VIJ, Paper 1, Fourth International Conference on Liquified Natural Gas, Algiers, 24-27 June 1974.

APPENDIX A

PHASE I FLOW CHARTS WITH NARRATIVE

This appendix described the Phase I Vulnerability Model (VM) flow diagrams. In general, available submodels from HACS are used wherever possible. Where submodels are still to be developed, they are so indicated.

The inadvertent release of all or a portion of a hazardous cargo is characterized by many quantities, most of which are listed in the attached symbol list. In addition, many quantities describing the ambient air and water conditions must be given. The cargo may be either solid, liquid, or gas, or a combination of gas and liquid. (This last condition will usually be the case for a tank containing a fluid under pressure or at reduced temperature or both.) The puncture, rupture or vent may occur above or below the water line, so we must consider the possibility that the cargo will escape in three different phases and that it may escape into air or into water.

Because of their extremely rare occurrence, the following possibilities are not considered.

1. A fluid cargo is being transported at less than ambient air pressure.

2. The fluid cargo is such that it changes to the solid state (freezes) upon contact with ambient conditions (air or water).

3. The cargo is lighter than air in its liquid or solid phases.

Because the vent will almost always be within 15 m of the water surface $(\pm z)$ we assume the following:

a. For the case of a gas venting into water, no significant advection by the current will occur before the gas reaches the water surface. (This assumption is reasonable because the rise of the gas to the water surface will be very fast compared to the advection processes being considered.)

b. For the case of a liquid venting into air, no significant advection by the wind will occur before the liquid reaches the water surface, and no significant evaporation of the cargo will occur in the time it takes the liquid to reach the water surface. (This assumption is also reasonable because of the relatively short time required for the liquid to fall to the surface. The exception would be a rupture in a tank holding a liquid cargo under pressure, which results in a fountain of the liquid cargo, directed upwards. This is extremely unlikely, however, as a rupture in the top of the tank will result in the escape of the cargo in gaseous form in virtually every case, and a rupture in the side of the tank below the gas-liquid interface would result in a fountain directed primarily in the horizontal direction.)

Preceding page blank

The flow chart is entered at the top of chart 1, at "SPILL". The first thing that must be done is to enter the many quantities needed to describe the cargo, vessel location, ambient conditions and the puncture or rupture. For some common cargoes, the physical and chemical properties such as critical point temperature, viscosity, concentration at lower flammability limit, and molecular weight may be obtained from the HACS Properties File. But other quantities such as the temperature, pressure and amount of the cargo will have to be supplied by the user. The difference between the two types of information needed is distinct. The properties of a cargo are inherent, and will not change from spill to spill, so they may be stored in the computer memory once and recalled when needed. The quantities which the user must supply each time are those which may vary with the spill, such as location of the spill, size of the vent, and the ambient conditions.

With this information available, the first task is the calculation of the rate of escape of the cargo, and the phase of the cargo which escapes. For a tank containing both gas and liquid phases, the height of the vent z_v would have to be compared with the height of the phase change of the gas-liquid interface z_{pc} . Further, if the vent is located below this interface, the escaping cargo may change from liquid to gas when sufficient cargo has escaped to lower the interface level to the vent level. Generally the vent or puncture area will be given, and the rate of release calculated, but the user may stipulate an instantaneous release. A solid spill must be assumed to be instantaneous at this time, as there is no extant subroutine to calculate rate of a solid spill. For liquids and gases, the HACS/CHRIS model is being used. This model takes the tank walls to be either isothermal or adiabatic, and for the venting of a gas, the subroutine uses either choked (sonic flow) or unchoked (subsonic flow) equations as appropriate.

At this point it might be mentioned that the tie-points 1 through 8 represent intermediate conditions, which might be called meta-stable states. The cargo may exist, of course, in three phases or states -- gas, liquid or solid. Since cargo spills upon land are outside the scope of this study, the cargo can be in the air, in the water, or it can occupy space at the interface between the air and the water. This last case is considered worthy of separate treatment because the spread of the cargo will be significantly different in this case than it will in the other two cases. The three cargo phases and the three locations lead to nine possible combinations, but the case where a gaseous cargo is in the water is considered too transient. This first chart, then, treats the escape of the cargo and its rapid transition to one of the eight intermediate states. The states, represented by tie-points 1 through 8, are not necessarily the final disposition of the cargo, but states in which the cargo remains for several minutes at least. This is long enough that transport of the cargo by diffusion, the wind, or the current must be considered. The processes on chart 1, on the other hand, generally take place in several seconds, so that advection during this period may be ignored at this time.

GASEOUS RELEASE

كالمح سناتان والمناسل وتوجك فالملك

For the case where all or part of the escaping cargo is in the gaseous phase, the flow diagram on chart 1 is followed to the left, to point Gl, where different paths are followed depending upon whether the vent is above or below the water. In the former case, at G2 we determine whether some of the escaping gas will condense to the liquid phase, forming a mist or fog comprised of droplets of the cargo. The escaping gas has been assumed to have expanded to the ambient fir pressure, P_a , by a throttling process, without mixing with the air. If the gas is considered to be a perfect gas, in such a throttling process the final temperature of the gas is the same as its initial temperature, and the expanded gas will be superheated and dry. The gas at the edge of the cloud will soon come to equilibrium with the air temperature, however, and condensation of the cargo will occur there if the vapor pressure of the cargo at the ambient air temperature is less than the air pressure, that is, if $P_{VC}(T_A) < P_A$.

If it is determined that liquid drops of the cargo form a mist, the portion of the cargo in each phase is calculated (at G3), and for each quantity one proceeds to specific models through tie-points 1 and 3. In the case that no mist forms, all the escaped cargo is in vapor form, and at G4 it is asked whether the gaseous cargo is heavy enough so that it mixes only slightly into the air and primarily floats upon the water surface. This is an unlikely case, but it is possible for exotic gases or for very cold gases. If the gas mixes with air, we proceed to chart 2 through tie-point 1. If the gas is very dense, one proceeds to chart 3 through tie-point 2.

Returning to Gl, in the case where the gas is escaping below the water line of the vessel, at G5 we calculate the amount of the escaping cargo which goes into solution as the gas bubbles rise to the surface. The portion going into solution is treated as liquid in water, through tie-point 4 to chart 5, and the portion remaining as a gas escapes to the surface. For this gas at G6, we ask if it mixes into the air or remains at the interface, and proceed to tie-points 1 or 2 accordingly, as in the case of an escape of gas directly into the atmosphere. At the present time, the calculation of the portion of the cargo going into solution has not been considered, and we temporarily assume that all the escaping gins reaches the water surface. It should be pointed out that it is not possible for the water pressure to force the escaping gas into liquid phase. If the water pressure outside the puncture were great enough to liquify the gaseous cargo, then the water pressure would force water into the cargo tank while liquifying the cargo in the tank, and very little cargo would escape.

LIQUID RELEASE

If the escape rate calculation indicates that some or all of the cargo escapes in liquid form, one proceeds to Ll where the vent height, z_V , is compared with the height of the water surface, z=0, to determine whether the leak is above or below the water. If the liquid cargo is escaping into air, we assume that it reaches the water surface without advection or loss, as discussed above. Next, at L3, we determine whether the liquid floats or sinks. If it sinks, we have the liquid in water case, and proceed to chart 5 through tie-point 4.

For the case in which the liquid cargo escapes uncerwater, at L4, it is asked whether the cargo floats or sinks. As in the case where the cargo was released into the air, the original temperature of the cargo in the tank is used in calculating the density of the cargo, for it is assumed that a spill of sufficient mass to be of interest will take some time to come to ambient temperature, and the temperature change of the cargo is considered on the flow charts which follow this first one. Most cargoes will have a density at least 10% different from that of water, but in the cases where they do not, the water temperature and salinity may be used to calculate the density of the water more precisely. If the liquid cargo sinks, we again proceed to tie-point 4. If the cargo floats, one considers that case further through tie-point 5.

SOLID RELEASE

The case of a spill of a hazardous cargo which is in solid phase has been given a lower priority than the case of a cargo in a fluid state. Therefore many of the calculation subroutines are in undeveloped condition, but the decision pathways are fairly clear regardless. There is currently no routine available to calculate the rate of a solid phase spill, so it will be declared to be instantaneous. The spill rate calculation for solid cargoes may be developed later as time and interest dictate. Since a cargo consisting of fist-sized chunks will spill at a very different rate from a cargo consisting of grains about the size of sand grains, the spill rate calculation will have to take into account the size range of the particles which make up the cargo.

Having decided that all or part of the spill is solid, at S1 the decision is made as to whether the puncture is above or below the water line of the vessel. If the cargo escapes into air, at S2 we wish to calculate how much of the cargo is carried away by the wind, and how much falls to the water surface. At present this subroutine has not been designed, so it is assumed that all of the escaping cargo falls to the water surface. This will be inaccurate only for a cargo consisting of very fine particles of a light material when a high wind is present.

The cargo having fallen to the water surface, at S3 it is decided whether the particles float or sink. One then proceeds to follow the case of solid in water in more detail on chart 8 through tie-point 7, or the solid on water surface case on chart 9 through tie-point 8. If the solid cargo is escaping below the water line, at S4, the density of the cargo is ascertained, and in the case that it sinks, one proceeds to tie-point 7. If the cargo particles rise to the surface and float, one should calculate the portion lost by going into solution during the brief period it takes the particles to reach the water surface. This has not been implemented yet, however, so currently there is no such loss, and all the escaping cargo arrives at the surface and is treated later through tie-point 8. The route to the liquid in water case through tie-point 4 is shown, however.

GAS IN AIR

For the case of a gaseous cargo escaping into the air, the detailed events are followed on chart 2 through the-point 1. First the air temperature, stability or mixing characteristics, and wind velocity are used to calculate the dispersion of the cargo. At present this calculation assumes that all the escaping cargo has been released instantly. This model has been discussed in some detail in chapter 2 of this report, so we proceed to the decision at G11 on the reactivity of the cargo. This information will come from the properties file, or will be entered by the user for cases of unusual spills where the properties file does not have this information. For many, if not most, of the hazardous cargoes, the cargo will be reactive. At G12 on the basis of flammability limiting concentrations and ignition temperatures, it is decided whether the situation is such that combustion is possible. If it is, it is next determined whether an ignition source is present, and if so we proceed through tie-point 9 to the fire and explosion model, which is discussed in detail elsewhere in this report. If the gaseous cargo is not combustible, or if it has not been ignited, the path followed is to G13 where a series of calculations determine what reactions will take place, how much heat is liberated, what reaction products are formed and at what rates. This model is still in the germinal stage. Finally, the spread and dispersal of the reaction products must be determined. While the mathods to be used for this are much the same as for the dispersal of the cargo originally, the use of these methods to trace the spread of the reaction products has not been implemented at this time. This will be necessary ut some point, for there are a number of cases, though not very common, where the reaction products are more toxic or dangerous than the original cargo.

GAS ON WATER SURFACE

The case for a heavier-than-air gas resting on the water surface has not been programmed at this time, but on chart 3 the calculations which should be performed are indicated. For the case of a gas that is heavier than air primarily because of its low temperature, the calculation of the spread of the gas should be similar to the calculations used to determine the spread of a cryogenic liquid, and should present no major difficulties. The rate of heat input to the gaseous cargo should also have much in common with the flow of heat to a cryogenic liquid spill. As the cargo attains a density not too different from that of air the cargo can be expected to mix into the air, and with this rate calculat[,]d, at G21, one then goes to the gas-in-air calculations, which have just been discussed. While the case of a gas which owes its very high density to its very low temperature will be more common, the case of gases which are inherently much heavier than air, even at ambient temperature and pressure may also be treated by this model. As in the case of a cold gas, the spilled cargo will eventually disperse into the air, but the rate of such a dispersal will depend not upon the rate of warming of the gas, but primarily upon the wind speed and local turbulence conditions. The calculation at the top of chart 3 should consider this type of loss into the air as well as heat transfer and gravity-induced spreading upon the water surface.

LIQUID IN AIR

Chart 4 takes up the case of a liquid in air - a mist or fog whose droplets are composed of the cargo in liquid phase. The dispersion calculation will be the same as for the dispersion of gas in air, with the addition of a subroutine to calculate the loss of the cargo in liquid phase due to evaporation and rainout. Rainout may occur either due to the incorporation of the drops into existing precipitation, or when the density and size of the drops allows them to grow in size while decreasing in number until the surviving drops fall due to their own weight. The increase of the number of drops might be precipitated by a reduction in the termperature in the cloud, which would reduce the vapor pressure of the cargo, and lead to more condensation. In most cases, however, evaporation will reduce the amount of cargo in the liquid phase. The mixing of the originally concentrated cargo plume with ever larger amounts of air during the course of dispersal is the cause of this. The amount of cargo per unit volume in the plume will have to be more than the air can absorb at saturation in order for mist droplets to form. The mixture of the original volume of the spill into a larger amount of air will increase the amount of cargo that the air can hold in vapor phase, so, in the absence of a marked temperature decrease during the spreading of the plume, evaporation of the droplets can be expected.

After the calculation of the rates of phase change for the cargo, at L11, the portion evaporating is treated through tie-point 1, and the portion raining-out is treated through tie-point 3 or 4 depending upon the density of the cargo in its liquid phase. The cargo remaining in liquid phase is next assessed for its combustibility at L12. If it is combustible, and if it ignites, then one proceeds to the fire and explosion model through tiepoint 9. If the cargo is incombustible or is not ignited, one proceeds to the evaluation of toxicity, air pollution, reduced visibility (which may be a navigation hazard). and other damage.

LIQUID IN WATER

Chart 5 takes up the case of ε liquid in water - entry from the main flow chart by means of tie-point 4. Since the behavior of the spilled liquid will depend on whether it mixes with water or not, the immediate question concerns the cargo's miscibility. A miscible liquid will mix into the water and disperse much like a gas dispersing into air. An immiscible liquid, on the other hand, will maintain its own separate identity and stay separate from the water. If lighter than water, the immiscible liquid will remain on the surface and that possibility is treated through tie-point 5. Here only immiscible liquids which have a density greater than that of water are considered. They will, of course, sink to the bottom, and there, if the current is strong enough, and the cargo is not too heavy and viscous, significant transport of the immiscible cargo along the bottom may occur. Thus the transport of the spilled liquid will be considerably different for the immiscible spill than it is for the miscible spill, and different calculations are indicated at L22 and L23.

The considerations of boiling of the liquid spill and of reactions are not significantly dependent upon the miscibility of the spill, so no differentiation has been shown here. If the liquid has a boiling point lower than the water temperature, all of the spill will eventually escape into the air, and this path is indicated by the left branch from the decision point at L24. If no boiling takes place, the possibility of reactions is considered, at L26, and the gaseous reaction products escape from the water and are treated through tir-point 1. The unreacted cargo and other reaction products go on to produce water pollution and possibly toxicity hazards.

LIQUID ON WATER SURFACE

For a liquid floating on the surface of the water, the development of the spill is continued on chart 6 through tie-point 5. This case is presently programmed and running in abbreviated form because common hazardous cargoes such as gasoline and liquified natural gas fall into this case, and it was given top priority. After determining whether the cargo is miscible or not, at L31, the spread of the cargo is calculated. At L34 it is determined whether the spreading miscible spill boils at ambient conditions. If so, one proceeds to the gas in air model after calculating the rate of gas formation at L35. If the spill does not boil, the cargo may still pass to the gaseous state by evaporation, so this is considered at L36, and the possibility of reactions is also considered at this time. The portion of the cargo evaporating and the gaseous reaction products go to the gas in air model via tie-point 1. For the portion of the spill remaining on the water surface, some may be lost by going into solution in the water, but this calculation is not available yet, so it is assumed that all of the spill which does not evaporate or react remains in the pool on the surface. At L37 we ask whether this pool can burn, and if so whether it is ignited. (Ignition would normally occur from an explosion or flash fire resulting from the spread of the cargo in vapor phase.)

For the liquid which does not mix with water at L43 it is determined whether this liquid boils at ambient temperature. If the answer is yes (as it would be for a liquid natural gas spil!, for example) at L44 the rate of escape of the cargo in the gas phase is calculated, and the consideration of the problem shifts to the gas in air model. If the spill does not boil, loss by evaporation is calculated at L45 (this would be the case for gasoline), and the escaping gas is again treated by the gas in air model. Reactions are considered at L45 for the unevaporated portion remaining. The possibility of fire is considered at [4] for the pool of cargo and/or reaction products floating.

SPILL OF SOLID CARGO

The case of a spill of a solid phase cargo is taken up on charts 7, 8, and 9 depending on whether the spill has come to be temporarily in the air, on the water, or in the water. We have excluded from consideration the possibility that a solid cargo may be combustible if it is in the water or on the water surface. While these possibilities do exist, they are extremely rare, and other considerations have higher priorities. The case that a solid cargo suspended in the air (in small particles) may explode or burn, is explicitly considered, however, since substances usually harmless, such as wheat flour, have caused explosions when the conditions were right.

The calculation of the spread of fine particles of the solid cargo in the air, reached through tie-point 6, on chart 7 is much the same as the dispersion calculation used for the spread of a gaseous cargo, with the addition that the loss of cargo due to fallout (with or without cgglomeration) and the rainout must be considered. For a solid in water (chart 8), the calculation of transport in the water will be much like that for an immiscible liquid, except that the settling rate will be different. The loss of the solid cargo through reactions and solution are next considered at S22. For liquid and gaseous reaction products, and the portion of the solid going into solution, the further consideration proceeds through tie-points 1 and 4 as appropriate. Chart 9 takes up the last of our 8 possibilities, the spill of a solid phase cargo which floats upon the surface by the water. After a calculation of the spread and transport of the cargo by winds and currents. \$31, the rates of disappearance due to sublimation, going into solution, and reactions are calculated. Liquid reaction products and cargo going into solution are treated by the liquid in water model through tie-point 4, and the gaseous reaction products and sublimed cargo go to the gas in air model through tie-point 1. None of the calculations for a solid spill has been programmed at present, but the knowledge needed for many of the subroutines is available.

المعالمة والمستعلقات

i

Ì

:

. . .

•

۰.

SYMBOL LIST FOR FLOW CHARTS WITH NARRATIVE

x, y, z : Cartesian coordinates

+z is in the upward direction, z=0 at the water surface

+x is in the direction toward which the wind is blowing

t:

time, t=0 at the commencement of the spill

- subscripts: c cargo s spill a air
 - w water
 - G gas phase of the cargo
 - L liquid phase of the cargo
 - S solid phase of the cargo

Cargo information supplied by user

T Temperature at which the cargo is stored

- P_c The pressure above ambient air pressure at which the cargo is stored
- S State of cargo (gas, liquid or solid) for a cryogenic or pressurized cargo a code will indicate whether the cargo is partly liquid and partly gaseoue
- z Height of the phase change (liquid-gas interface) for a fluid cargo which exists in the tank in both phases
- V Volume of Largo potentially affected

M Mass of cargo potentially affected

<u>Comment</u>: If only one of a number of tanks or holds on a vessel is ruptured or punctured, V_c and M_c are the vous and mass of the affected tank. Note that V_c and M_c will not necessariling e equal to the amount spilled. The nature and location of the puncture may preclude all of the cargo in the tank from escaping.

 P_c Density of cargo

<u>Comment</u>: V_c , M_c , and ρ_c are inter-related, and all three need not be given in most cases. From T_c and P_c , and other known constants, ρ_c can be calculated for a liquid or gas, and thence M_c if V_c is known, which will usually be the case. For a solid cargo, ρ_c will be the density of the individual grains or churks, and $M_c \neq \rho_c V_c$ because the packing fraction will have to be taken into account, but in most cases M_c will be known and V_c will not be needed.

Cargo information obtained from the properties file

1995 - S. S. M.

Th Boiling point of the cargo at atmospheric pressure

Freezing point of cargo at atmospheric pressure T_f

T_{FP} Flash Point temperature

 $\chi_{\rm L},\chi_{\rm U}$ Concentration of the cargo in gas phase at the Lower and Upper Flammability limits

P Vapor pressure, as a function of temperature

λ Solubility

Miscibility ίΩ.

Spill information supplied by user

Mass or Volume of spill for an instantaneous spill M_g or V_g

Area of puncture, rupture, or vent - needed only if the spill is not Α, instantaneous

Height of puncture, rupture, or vent 2 .,

Comment: If the spill is not to be instantaneous, then the spill rate as a function of time may be calculated from the area of the vent and the cargo conditions.

Information about Ambient Conditions supplied by the user

Air Temperature T Pa Air Pressure H Relative Humidity v, Wind vector Sa Atmospheric Stability Class Water Temperature T_ ٧₆ Current vector

N Salinity - may be needed to calculate the water density ρ_W





10.1

المتركب والمتركين

والمتعلية وستعقشهم والأقتار بداعيت

and the

4

and she was not



177

CHART 3



LIQUID IN AIR

. . .

. .

47.97

a ser ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a ser a se

- -

.

•

المتصفحات المقار

سيندر وتصدقكم

-

-. -. -.

~T3





i de la companya de l

والأكبة الاعتراد

.

ne Chen ann an ann an Anna an Anna an Anna an Anna an Anna an Anna an Anna an Anna Anna an Anna Anna Anna Anna



CHART 7

Same in the set

.

ł



SOLID IN WATER

Sanas 4 vodalaten deletere soniotroches e voix deletere en serie de contrar britan den den den en en en en en en e

11.02



CHART 9

المكملية بالمربع بالمعالية والمراجع والمراجع والمراجع

.....

3.7

: • •

And the second second second

. .

SOLID ON WATER SURFACE



APPENDIX B1

MODIFICATION OF THE GAUSSIAN PUFF MODEL

Since the Gaussian puff model is primarily intended for use with low concentrations, when applied very close to a large source the unmodified Gaussian puff equation may give a concentration of the vapor-phase cargo which is greater than the density of the pure cargo vapor at ambient atmospheric pressure and temperature. This is not physically possible, of course, and the computer model has been modified to preclude such a concentration being given.

In the case when the unmodified Gaussian puff equation would give a concentration greater than pure cargo at ambient conditions, what is actually present is a glob of nearly pure cargo, surrounded by a volume in which the concentration decreases from the density of pure gaseous cargo towards zero. It is reasonable to assume that the concentration decreases in a Gaussian manner from the surface of the glob of pure or nearly pure cargo.

To treat this mathematically, let us first transform from the physical x,y,z coordinate system into one which is scaled by the diffusion coefficients:

$$\mathbf{x}' = \frac{\mathbf{x} - \mathbf{U}\mathbf{t}}{\sigma_{\mathbf{x}}}, \quad \mathbf{y}' = \frac{\mathbf{y}}{\sigma_{\mathbf{y}}}, \quad \mathbf{z}' = \frac{\mathbf{z}}{\sigma_{\mathbf{z}}}$$
 (B1-1)

where U is the wind speed. Since the point of liberation of the gas is at the water surface, z=0, in this coordinate system, let the glob be a hemisphere centered at x'=y'=z'=0. Let us take the radius of this hemisphere to be R', and assume that it is pure cargo-gas. If C_p is the density of the cargo in gas phase at ambient temperature and pressure, and $(r')^2 = (x')^2 + (y')^2 + (z')^2$ is the square of the distance from the center of the puff to the observation point in the new coordinate system, then the concentration C at x',y',z' will be

 $C = C_{p} \text{ if } r' \leq R'$ $C = C_{p} \exp \left(-\frac{(r'-R')^{2}}{2}\right) \text{ if } r' > R'$ (B1-2)

The term $(r' - R')^2$ in the exponent is the correct distance factor, but it appears unusual. It is clear that the Gaussian function depends on the square of the distance from some point, line, or surface. The usual Gaussian diffusion equations are written with respect to a point (puff mode!) or a line (plume or continuous source model). In these cases, the distance from the point or line may be conveniently expressed in the Gaussian coordinate system, by the familiar equations. Here, however, we have a case where it is the distance from the surface of a sphere which is important, and this is expressed conveniently only in spherical coordinates.

Preceding page blank

The reader familiar with the diffusion equation or time-dependent heat flow problems may have noticed that (B1-2) is not the solution to

 $\frac{\partial c}{\partial t} = K \nabla^2 C \tag{B1-3}$

with

 $C = C_{p} \quad \text{for} \quad r \leq r_{1}$ $C = 0 \quad \text{for} \quad r > r_{1}$ $At t = 0 \quad (B1-4)$

which is the rigorous statement of the problem of the dispersion of material which forms a sphere at t = 0. (The equation for the diffusion of particles is identical to the equation for the transfer of heat by conduction, and the thermal problem antedates the diffusion problem, so the treatment in many books is in terms of heat flow. Analytic solutions to the diffusion equation for a number of simple cases may be found in standard texts such as Kreyszig [B1] or Landau and Lifshitz [B2]).

While this problem could be solved for the concentration as a function of time and space for the early period of the puff's dispersal, the problem of matching this solution to the Gaussian puff solution at a later time when the puff model is appropriate would remain. This matching would require shifting from the solution to (B1-3) to the empirically determined Gaussian puff concentration over an arbitrary period of time by means of interpolation of spline functions. To make a smooth transition, the matching would be quite complex. Since the Gaussian puff should need modification to avoid unrealistically high densities only for a short time after the puff begins dispersing, this complication was not considered worthwhile. The chosen solution (B1-2) approximates the theoretical solution to (B1-3) well enough for the use the VM will make of it, and the transition to the Gaussian puff distribution is trivial.

Further, the type of flat-topped profile of concentration as a function of radial distance which (B1-2) gives, is in accordance with

[B2] Landau, L.D., and E.M. Lifshitz. Fluid Mechanics. Pergamon, London, 1959. (See Chapter 5 and 6.)

[[]B1] Kreyszig, E. Advanced Engineering Mathematics. John Wiley & Sons, Inc., New York, 1967. (See Section 9.5 and 9.6)
cther work. Figure B1-1 shows a profile calculated by Lind (B3), which clearly shows a region of constant concentration. While this concentration is not 100% as it is in (B1-2), it is clear from the figure that very close to the spill a concentration close to 100% would occur. Lind's method of calculation is based upon that used by Esso, which was experimentally verified [B4].

In equation (B1-2), R' marks the boundary between the hemisphere of pure cargo vapor and the region in which the vapor is mixed with air. The value of R' is determined by the necessity of conserving mass:

$$M = \iiint Cdv = \iiint C(\frac{dv}{dv'}) dv' = \iiint C(\sigma_x \sigma_y \sigma_z) dv'$$
(B1-5)
all
space space space

where dV = dxdydz, and dV' = dx'dy'dz'.

It may seem that the concentration used should be converted to the primed coordinate system, but this is not necessarily the case. It is incorrect to transform the limits of integration, the unit volume, and the concentration or density. However, one may define $C' = C \sigma_X \sigma_y \sigma_z$ to be the new density if one so desires*.

- [B3] Lind, C.D., Explosion Hazards Associated with Spills of Large Quantities of Hazardous Materials - Phase I. Report No. CG-D-30-75, United States Coast Guard, Dept. of Transportation, 18 October 1974.
- [B4] Feldbauer, G.W., et al. Spills of LNG on Water Vaporization and Downwind Drift of Combustible Mixtures. Esso Research and Engineering Company, Report No. EE61E-72 (Released by the American Petroleum Inst., Re 6232), March 1973.
- That this new integral is correct may be demonstrated by a simple example. In the x,y,z system let the density be some constant value, C_0 , in a rectangular region with slides $\sigma_x, \sigma_y, \sigma_z$, respectively, and zero outside this region. Clearly:

$$M = \int_{0}^{\sigma_{x}} \int_{0}^{\sigma_{y}} \int_{0}^{\sigma_{z}} C_{0} dxdydz = C_{0} \sigma_{x} \sigma_{y} \sigma_{z}$$

Taking care to transform the limits of integration, in the x', y', z' system we have

$$M = \int_{0}^{1} \int_{0}^{1} \int_{0}^{1} C_{0}\sigma_{x}\sigma_{y}\sigma_{z}dx'dy'dz' = C_{0}\sigma_{x}\sigma_{y}\sigma_{z}$$



FIGURE B1-1. Sea Level Concentration Profile, 10,000 m³ Spill. [B3]

It will be easiest to do the integral in spherical coordinates in the new coordinate system so equation (B1-5) becomes:

$$M = \int \int \int C (\sigma_x \sigma_y \sigma_z) (r')^2 s'n \theta' dr' d\theta' d\phi'$$
(B1-6)
$$\phi'=o \theta'=o r'=o$$

المعار<mark>ك أو المحمد المحم المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد ا</mark>

1

ŝ

19 金融 \$10 mm,小型展现 自己的时候,一个人的人们也会是我们的自己的事件的。 19 mm,这些人们的事件。

.

where the integral over θ' only goes from 0 to $\pi/2$ because the gas can only occupy the +z half-space.

The φ' and θ' integrals can be done at once, leaving

$$\frac{M}{\sigma_{x}\sigma_{y}\sigma_{z}} = 2\pi \left[\int_{0}^{R'} C_{p}(r')^{2} dr' + \int_{R'}^{\infty} C_{p} \exp(-\frac{1}{2}(r'-R')^{2})(r')^{2} dr' \right]$$
(B1-7)

Let the second integral above be denoted by A $C_{\rm p}$, and we have

$$\frac{M}{\sigma_{x}\sigma_{y}\sigma_{z}} = 2\pi C_{y}\frac{(g^{*})^{3}}{3} + 2\pi A C_{p}$$
(B1-8)

2

The integral A is evaluated by making the substitution s = r' - R':

$$A = \int_{R'}^{\infty} \exp \left(-\frac{1}{2} (r' - R')^{2}\right) (r')^{2} dr' = \int_{0}^{\infty} e^{-\frac{s^{2}}{2}} (s^{2} + 2sR' + (R')^{2}) ds$$
$$= \int_{0}^{\infty} s^{2} e^{-\frac{s^{2}}{2}} ds + 2R' \int_{0}^{\infty} s e^{-\frac{s^{2}}{2}} ds + (R')^{2} \int_{0}^{\infty} e^{-\frac{s^{2}}{2}} ds \qquad (B1-9)$$
$$= (\frac{\pi}{2})^{1/2} + 2R' + (R')^{2} (\frac{\pi}{2})^{1/2}$$

The second of the three integrals in s is trivial, and the other two are tabulated definite integrals.* With the integral A evaluated, from (B1-8) we have:

$$\frac{M}{\sigma_{\mathbf{x}}\sigma_{\mathbf{y}}\sigma_{\mathbf{z}}} = C_{\mathbf{p}} \left[\frac{2}{3} \pi (\mathbf{R}')^{3} + 2\pi (2\mathbf{R}' + (1 + (\mathbf{R}')^{2}) (\frac{\pi}{2})^{1/2}) \right]$$
(B1-10)

which is a cubic equation in R'.

*It may seem odd that $\int_{0}^{\infty} s^{2}e^{-\frac{s^{2}}{2}}$ as $= (\frac{\pi}{2})^{1/2} = \int_{0}^{\infty} e^{-\frac{s^{2}}{2}}$ ds, but such is indeed the case. This may be proved by integration by parts. We have (cont'd)

$$(R')^3 + p(R')^2 + qR' + s = 0$$

with
$$p = \frac{\pi}{2} (2\pi)^{1/2}$$

 $q = 6$
 $s = p - \frac{3M}{2\pi C_p \sigma_x \sigma_y \sigma_z}$

In use, at each time step the concentration at the center of the puff is calculated by the usual Gaussian puff equation:

$$C = \frac{2M}{(2\pi)^{3/2} \sigma_{x} \sigma_{y} \sigma_{z}}$$
(B1-12)

If this concentration does not exceed C_p , the usual Gaussian puff equation is used. If this concentration exceeds C_p , then equation (B1-11) is solved for R', and then the concentrations are calculated from equation (B1-2) with:

$$\mathbf{r'} = \begin{bmatrix} \left(\frac{\mathbf{x} - \mathbf{Ut}}{\sigma_{\mathbf{x}}}\right)^2 + \left(\frac{\mathbf{y}}{\sigma_{\mathbf{y}}}\right)^2 + \left(\frac{\mathbf{z}}{\sigma_{\mathbf{z}}}\right)^2 \end{bmatrix}^{1/2}$$
(BJ-13)

*(cont'd)

$$\frac{d}{ds} (fg) = \frac{df}{ds} g + f \frac{dg}{ds}, \text{ and since } \int \frac{d}{ds} (fg)ds = fg, \int_{0}^{\infty} f \frac{dg}{ds} ds = \left[fg\right]_{0}^{\infty} - \int_{0}^{\infty} \left(\frac{df}{ds}\right) gds. \quad \text{Let } \frac{dg}{ds} = se^{-\frac{s^{2}}{2}} \text{ and } f = s, \text{ then } g = -e^{-\frac{s^{2}}{2}} \text{ and}$$
$$\frac{df}{ds} = 1. \quad \text{Thus } \int_{0}^{\infty} s^{2}e^{-\frac{s^{2}}{2}} ds = \left[-se^{-\frac{s^{2}}{2}}\right]_{0}^{\infty} - \int_{0}^{\infty} (-e^{-\frac{s^{2}}{2}}) ds$$
$$\text{which gives } \int_{0}^{\infty} s^{2}e^{-\frac{s^{2}}{2}} ds = \int_{0}^{\infty} e^{-\frac{s^{2}}{2}} ds.$$

Both these integrals are evaluated in tables of definite integrals, and are equal to $(\frac{\pi}{2})$. (For example, see integrals 423 and 426 on p. 304 of [B5].) L BURNING STRUCK

(B]-11)

[[]B5] C.R.C. Standard Mathematical Tables, 11th ed. C. D. Hodgeman (Editor-in-Chief), Chemical Rubber Publishing Company, Cleveland, Ohio, 1957.

In the case that an explosion or flash fire occurs while the puff is still so close to the source that the unmodified Gaussian puff model gives unrealistic values for the concentration, the VM must be able to calculate the amount of cargo vapor which explodes or burns for the modified distribution we have been discussing. In Appendix C it is shown that the mass which will explode or burn will be given by

$$M_{e} = \int \int \int CdV + \int \int \int PCdV$$

$$S_{s} = S_{u}$$

where S_u is the surface on which $C = C_u$ S_s is the surface on which $C = C_s$ S1 is the surface on which $C = C_1$ Cu = the upper flammable limiting concentration C_s = the stoichiometric concentration C_1 = the lower flammable limiting concentration

and where

$$P = \frac{0.21}{N_0} \left(\frac{C_p}{C} - 1\right)$$
(B1-15)

(B1-14)

is the fraction of the fuel which burns when the mixture is richer than stoichiometric. No is the number of moles of oxygen consumed for every mole of fuel burned.

As before it will be easiest to work in the primed coordinate system (equation B1-1) and in spherical coordinates in this system (equation B1-13). In this coordinate system the surfaces of constant concentration will be spheres. Let

 r_{i}^{1} = radius of the surface on which C = C_u $r_s = radius$ of the surface on which $C = C_s$ r'_1 = radius of the surface on which C = C₁

Values for these radii may be obtained by setting C equal to C_u , C_s , or C_1 in equation (B1-2). (Since pure fuel vapor cannot burn - there is no oxygen present - C_u must be less than C_p , and we need only consider the region in which r'is greater than R'.)' The values for the radii are easily obtained:

 $r'_{u} = R' + (2ln(C_{p}/C_{u}))^{1/2}$ $r'_{s} = R' + (2ln(C_{p}/C_{s}))^{1/2}$ (B1 - 16) $r_1^{\dagger} = R^{\dagger} + (2 \ln (C_p / C_1))^{1/2}$

So with (B1-2) and (B1-15), equation (B1-14) may be written

$$M_{e} = \int_{\phi'=0}^{2\pi} \int_{\theta'=0}^{\pi/2} \int_{r'=r_{u}}^{r_{s}} \left[\frac{0.21}{N_{0}} C_{p} \left(1 - \exp\left[- \frac{\left(r' - R' \right)^{2}}{2} \right] \right] \right] \cdot \sigma_{x} \sigma_{y} \sigma_{z} \left(r' \right)^{2} \sin \theta' dr' d\theta' d\phi' + (B1-17)$$

$$\int_{\phi'=0}^{2\pi} \int_{\theta'=0}^{\pi/2} \int_{r=r_{s}}^{r_{1}} C_{p} \exp\left[-\frac{(r'-R')^{2}}{2}\right] \sigma_{x} \sigma_{y} \sigma_{z} (r')^{2} \sin\theta' dr' d\theta' d\phi'$$

The transformation used to obtain this expression is analogous to the one used to go from (B1-5) to (B1-6). And here also the ϕ' and θ' integrals can be done at once giving

$$\frac{M_{e}}{2\pi C_{p}\sigma_{x}\sigma_{y}\sigma_{z}} = \frac{0.21}{N_{0}} \int_{r_{u}}^{r_{s}'} (r')^{2} dr' -$$

$$\frac{0.21}{N_{0}} \int_{r_{u}'}^{r_{s}'} exr_{r} \left[-\frac{(r'-R')^{2}}{2} \right] (r')^{2} dr' + \int_{r_{s}'}^{r_{1}'} exp \left[-\frac{(r'-R')^{2}}{2} \right] (r')^{2} dr'$$
(B1-18)

Now the first integral is trivial and gives $(r')^3/3$. The other two integrals may be done by considering integrals of the form

$$I = \int_{a}^{b} s^{2} \exp(-\frac{s^{2}}{2}) ds$$
 (B1-19)

which we treat by integration by parts. With $dg/ds = s \exp(-s^2/2)$ and f(s) = s, then $g(s) = -\exp(-s^2/2)$ and df/ds = 1. Since

$$\int_{a}^{b} \frac{dg}{ds} f ds = [fg]_{a}^{b} - \frac{df}{ds} g ds \qquad (B1-20)$$

we have

$$\int_{a}^{b} s^{2} \exp\left(-\frac{s^{2}}{2}\right) ds = \left[-s \exp\left(-\frac{s^{2}}{2}\right)\right]_{a}^{b} - \int_{a}^{b} - \exp\left(-\frac{s^{2}}{2}\right) ds$$

$$= a \exp\left(-\frac{a^{2}}{2}\right) - b \exp\left(-\frac{b^{2}}{2}\right) + \int_{a}^{b} \exp\left(-\frac{s^{2}}{2}\right) ds$$
(B1-21)

Now let s = r' - R'; then r' = s + R' and $(r')^2 = s^2 + 2sR' + (R')^2$ so we have

$$A = \int_{r_{u}}^{r_{s}} \exp \left[-\frac{(r'-R')^{2}}{2}\right] (r')^{2} dr' = \int_{r_{u}-R'}^{r_{s}'-R'} \exp \left(-\frac{s^{2}}{2}\right) \left[s^{2}+2sR'+(R')^{2}\right] ds (B1-22)$$

Let
$$a = r'_{1} - R'$$
, $b = r'_{3} - R'$, and $c = r'_{1} - R'$. Then (B1-22) becomes

$$A = \int_{a}^{b} s^{2} \exp(-\frac{s^{2}}{2}) ds + 2R' \int_{a}^{b} s \exp(-\frac{s^{2}}{2}) ds + (R')^{2} \int_{a}^{b} \exp(-\frac{s^{2}}{2}) ds \quad (B1-23)$$

Using (B1-21) this becomes

i

A =
$$a \exp(-\frac{a^2}{2}) - b \exp(-\frac{b^2}{2}) + \int_a^b \exp(-\frac{s^2}{2}) ds$$

+ $2R' [a \exp(-\frac{a^2}{2}) - b \exp(-\frac{b^2}{2})] + (R')^2 \int_a^b \exp(-\frac{s^2}{2}) ds$ (B1-24)

The similar terms may be combined to give

A = (1+2R') [a exp
$$(-\frac{a^2}{2})$$
 - b exp $(-\frac{b^2}{2})$] + [1+(R')²] $\int_{a}^{b} exp (-\frac{a^2}{2}) ds$ (B1-25)

Now the remaining integral has an integrand which differs from the normal curve of error only by a constant, and the normal curve of error, $\phi(x)$, is related to the error function, erf(t) by

$$\int_{0}^{t} \phi(x) dx = \int_{0}^{t} (2\pi)^{-1/2} \exp(-\frac{x^{2}}{2}) dx = 1/2 \operatorname{erf}(\frac{t}{\sqrt{-2}})$$
(B1-26)

The error function is a tabulated special function, and we may express the integral A in terms of it by using the identity

$$\int_{a}^{b} f(s)ds = \int_{0}^{b} f(s)ds - \int_{0}^{a} f(s)ds$$
(B1-27)

and so we have

$$A = (1+2R') [a \exp(-\frac{a^2}{2}) - b \exp(-\frac{b^2}{2})] +$$

$$+ [1+(R')^2] (\frac{\pi}{2})^{1/2} [erf(\frac{b}{\sqrt{2}}) - erf(\frac{a}{\sqrt{2}})]$$
(B1-28)

But r's was defined by

$$C_s = C_p \exp \left[-\frac{(r_s - R^*)^2}{2}\right] = C_p \exp \left(-\frac{b^2}{2}\right)$$
 (B1-29)

and r'_u and r'_1 were defined in an analogous manner so we have

$$A = (1+2R') [a(C_u/C_p) - b(C_s/C_p)] + [1+(R')^2] (\frac{\pi}{2})^{1/2} .$$

$$[erf(\frac{b}{\sqrt{2}}) - erf(\frac{a}{\sqrt{2}})]$$
(B1-30)

....

Later to the second

ł

The third integral in (B1-18) is the same as the second integral except for the limits so we have, by analogy:

নার দের

$$B = \int_{r_{s}}^{r_{1}'} \exp \left[-\frac{(r_{s}-R')^{2}}{2}\right] (r')^{2} dr' = (1+2R') \left[b(C_{s}/C_{p}) - c(C_{1}/C_{p})\right]$$

$$+ \left[1+(R')^{2}\right] (\frac{\pi}{2})^{1/2} \left[erf(\frac{c}{\sqrt{2}}) - erf(\frac{b}{\sqrt{2}})\right]$$
(B1-31)

......

and the mass exploding or burning, from equation (B1-18) can finally be written as:

THE CONTRACTOR

ź.

$$M_{e} = 2\pi C_{p} \sigma_{x} \sigma_{y} \sigma_{z} \left[\frac{0.21}{3N_{0}} \left((r'_{s})^{3} - (r'_{u})^{3} \right) - \frac{0.21}{N_{0}} A + B \right]$$
(B1-32)

APPENDIX B2

GAUSSIAN PLUME AIR DISPERSION MODEL

In cases where the liberation of the cargo in vapor phase takes a long time, it is more appropriate to use a Gaussian plume model (continuous source model) than a Gaussian puff model (instantaneous source model). The question of how long the gas release time should be in order to use the plume model is discussed in chapter 4 and the following section of this appendix. بالالكيمية والمكلا

The Gaussian plume model is discussed in chapter 5 of the CHRIS documentation [B6], and equation (B2-1) below is essentially equation (5.2) from p. 54 of this work. The CHRIS documentation does not discuss the possibility of the vapor release rate varying as a function of time. Equation (B2-1) allows for this possibility, but as programmed the model uses an average release rate which is not a function of time. The reason for this limitation in the program is twofold: first, if an explosion or flash fire occurs, the amount of fuel which explodes or burns must be found by integration of the concentration distribution between given concentrations. Not only is this much easier for the case where the release rate is constant, but the various subroutines make no provision for storing the release rate as a function of time, so a number of HACS subroutines would have to be modified if the average release rate was not used. Second, some of the HACS submodels which treat the release of cargo vaper do not calculate a release rate as a function of time, but output only the amount of vapor released and the time at which the release is complete.

The Gaussian plume model is undefined when the wind speed is zero. Since this is the case, and the plume model is not appropriate for use when the wind speeds are low, the VM uses the puff model whenever the wind speed is below 2 m/s.

The details of the calculation are as follows; let:

- C the concentration of the cargo in vapor phase.
- x,y,z = the position of the observer (or the cell center) in a coordinate system in which the origin is at the spill or vent location (which is assumed to be at the water surface) and the positive x axis is in the direction toward which the wind is blowing.
- t = the time at which the concentration at x,y,z is to be calculated.

t = o, when the vapor release begins.

t_v

f. - x/U = the time at which the cargo vapor observed at x,y,z at time t would have escaped or evaporated if the wind speed was steady at U and there was no diffusion in the x direction.

[[]B6] Raj, P.K., and A.S. Kalelkar, Assessment Models in Support of the Hazard Assessment Handbook, A.D. Little, Inc., Cambridge, Mass., January 1974. Dept. of Transportation Report No. CG-D-65-74, NTIS AD 776617.

- t = time at which the gas venting or evaporation is complete.
- F = the rate of liberation of the cargo in gaseous state.
- σ σ y, z = the diffusion parameters in the y and z direction, respectively. They are functions of x and of the stanospheric stability, of the Form ax^b.
 - = the wind speed. U is taken to be a constant.

U

It is clear that F_e should be evaluated at time t_v , since the rate of vapor liberation at the pool or vent at the same time that the concentration is being measured many hundreds of meters or many kilometers downwind is irrelevant. What is relevant is release rate at t_v , since the time it has taken the vapor to travel with the wind from the origin to x is just x/U.

If the conditions are suitable for use of the plume equation as opposed to the puff equation, the concentration is given by

$$C(x,y,z,t) = \frac{F_e}{\pi U \sigma_v \sigma_z} \exp \left[-\frac{1}{2} \left(\frac{y}{\sigma_y}\right)^2 - \frac{1}{2} \left(\frac{z}{\sigma_z}\right)^2\right] \text{ for } 0 \le t_v \le t_e \qquad (B2-1)$$

$$C(x,y,z,t) = 0$$
 for $t_{..} < 0$ and $tv > te$

where F_e is evaluated at t_v . In this equation it has been assumed that the gas is released at the water surface and a factor of two has been incorporated to account for the reflection about the z=0 plane, i.e., for the fact that the gas escapes into the positive-z half-space.

Next we consider how much of the cargo vapor which is dispersing in a Gaussian plume will explode or burn in a flash fire. The basic assumptions are that no fuel will burn where the concentration is less than the lower flammable concentration, and that no fuel will burn where the concentration is above the upper flammable concentration. Where the concentration is between the stoichiometric concentration and the upper flammable concentration, it is assumed that all of the oxygen will be consumed, but that not all of the fuel will be consumed. This assumption is discussed in detail in appendix C.

In addition to the items defined above, let:

- C_p = concentration or density of pure cargo vapor at ambient atmospheric temperature and pressure.
- C_s = stoichiometric concentration.
- C₁₁ = upper flammable concentration limit.
- C_1 = lower flammable concentration limit.
- ti = time of ignition.

 $x_i, y_i, z_i = coordinates of ignition source.$

coordinate of the source-end of the plume:

 $x_e = 0$ if $t_1 \leq t_e$

1. AN -

...

 $x_e = U(t_i - t_e)$ if $t_i > t_e$.

No = number of moles of oxygen consumed per mole of fuel.

Me = mass of cargo consumed in flash fire or explosion.

P

X

portion of fuel Surned when the fuel concentration is greater than stoichiometric. and the second second second second second second second second second second second second second second second

Solution and the second second

Let us begin by finding an expression for the curve in the yz plane where $C = C_g$. The concentration is given by

$$C = \frac{2F_e}{2\pi U\sigma_x \sigma_y} \exp\left[-\frac{1}{2}\left(\frac{y}{\sigma_y}\right)^2 - \frac{1}{2}\left(\frac{z}{\sigma_z}\right)^2\right]$$
(B2-2)

Where the factor of two in the numerator comes from the assumption that the source is at z=0, and the escaping gas can only go into the positive-z half-space. With $C=C_g$, rearranging (B2-2) we have

$$\left(\frac{y}{\sigma_y}\right)^2 + \left(\frac{z}{\sigma_z}\right)^2 = 2 \ln \left(\frac{F_e}{\pi U \sigma_y \sigma_z C_s}\right)$$
(B2-3)

This is the equation for an ellipse. The mathematics will be simplified if we transform to a coordinate system in which we have a circle rather than ellipse. Let

$$y' = \frac{y}{\sigma_y}$$
, $z' = \frac{z}{\sigma_z}$, $(r')^2 = (y')^2 + (z')^2 = (\frac{y}{\sigma_y})^2 + (\frac{z}{\sigma_z})^2$ (B2-4)

If $r' = R'_{s}$ on the circle in the primed coordinate system where C=C_s, then

$$(R_{g}')^{2} = 2 \ln \left(\frac{F_{e}}{\pi U \sigma_{y} \sigma_{z} C_{g}}\right)$$
 (B2-5a)

Likewise,
$$R'_{1}$$
 given by $(R'_{1})^{2} = 2 \ln (\frac{F_{e}}{\pi U \sigma_{y} \sigma_{z} C_{1}})$ (B2-5b)

is the radius of the circle on which $C=C_1$, and R'_u , given by

$$(R'_{u})^{2} = 2 \ln \left(\frac{F_{e}}{\pi U \sigma_{y} \sigma_{z} C_{u}}\right)$$
(B2-5c)

is the radius of the circle on which $C=C_u$.

Now the fraction of fuel consumed when the fuel concentration is greater than stoichiometric is

$$P = \frac{0.21}{N_0} \left(\frac{C_P}{C} - 1\right)$$
(B2-6)

as discussed in appendix C.

The mass of cargo vapor which explodes or burns in a flash fire is then

$$\begin{array}{cccc} x_{i} & C = C_{1} & C = C_{s} \\ M_{e} = \int \left[\int \int Cd_{y}d_{z} + \int \int PC d_{y}d_{z} \right] d_{x} \\ x_{e} & C = C_{s} & C = C_{u} \end{array}$$
(B2-7)

where the limits on the y and z integrals are the curves on which the concentration is equal to C_1 , C_5 , or C_u . The y-z integrals will be most easily done in the primed coordinace system in plane polar coordinates. The first integral is

$$C=C_{1} \qquad C=C_{1} \qquad \sigma'=R_{1}' \qquad +\pi/2 \qquad F_{e} \qquad e^{-\frac{(r')^{2}}{2}\sigma_{y}\sigma_{z}r'dr'd\phi'} = \iint_{\sigma'=R_{s}'} \qquad \phi=-\pi/2 \qquad F_{e} \qquad e^{-\frac{(r')^{2}}{2}\sigma_{y}\sigma_{z}r'dr'd\phi'} = \frac{F_{e}}{2} \qquad e^{-\frac{(R_{s}')^{2}}{2}\sigma_{z}\sigma_{z}r'dr'd\phi'} = \frac{F_{e}}{2} \qquad (B2-8)$$

But with R_1^{\dagger} and R_2^{\dagger} given by (B2-5) we have

$$C_{1} \int Cdydz = \frac{F_{e}}{U} \left(\frac{\pi U \sigma_{y} \sigma_{z}}{F_{e}}\right) (C_{s}-C_{1}) = \pi \sigma_{y} \sigma_{z} (C_{s}-C_{1})$$
(B2-9)
$$C=C_{s}$$

Using (B2-6) the second y-z integral in (B2-7) is

$$C=C_{s} \qquad C=C_{s}$$

$$\int\int PCd_{y}d_{z} = \int\int \frac{0.21}{N_{0}} (C_{p}-C) \sigma_{y}\sigma_{z}d_{y}d_{z}'$$

$$C=C_{u} \qquad C=C_{u}$$
(B2-10)

$$= \frac{0.21}{N_0} \int \int \int C_p \sigma_y \sigma_z r' dr' - \int \int \frac{F_e}{\pi U \sigma_y \sigma_z} e^{-\frac{(r')^2}{2}} \sigma_y \sigma_z r' dr' d\phi'$$

In this expression, the ϕ' integral and the r' integral are trivial, and the second r' integral is similar to the integral evaluated in (B2-8) and (B2-9), we we have

$$\sum_{j=1}^{C=C_{g}} PCd_{y}d_{z} = \left(\frac{0.21}{N_{0}}\right) \pi\sigma_{y}\sigma_{z} \left[C_{p} \left(\frac{(R_{s}^{\prime})^{2} - (R_{u}^{\prime})^{2}}{2}\right) - (C_{u}-C_{s})\right]$$
(B2-11)
C=C₁₁

Thus (B2-7) becomes

$$M_{e} = \iint_{x_{e}} \pi \sigma_{y} \sigma_{z} \left[(C_{s} - C_{1}) + (\frac{0.21}{N_{0}}) \frac{C_{p}}{2} ((R_{s}^{\dagger})^{2} - (R_{u}^{\dagger})^{2}) - (\frac{0.21}{N_{0}}) (C_{u} - C_{s}) \right] d_{x}$$
(B2-12)

Let $\sigma_v = ax^b$, $\sigma_z = cx^d$, then

$$M_{e} = \pi A \iint acx^{b+d} dx = \pi A \left(\frac{x_{1}^{b+d+1} - x_{e}^{b+d+1}}{b+d+1}\right) ac \qquad (B2-13a)$$

$$x_{e}$$

where

$$A = C_{g} - C_{1} + \frac{0.21}{N_{0}} [C_{p} \log_{e} (C_{u}/C_{s}) + C_{s} - C_{u}]$$
(B2-13b)

In practice, all the quantities in (B2-13) will be known at the time it is calculated that the ignition occurs, except R'_g and R'_u . These quantities may be calculated from (B2-5), however. In the CHRIS/HACS models, σ_y and σ_z are not always given by an expression of the form ax^b . However, in most cases the plume will not be more than several km long when it is ignited, and expressing σ_y and σ_z in ax^b form for such a range of distance is an acceptable approximation.

Appendix B3

SCALE ANALYSIS

Consideration of the lengths and times which are characteristic of air dispersion enables one to come to some useful conclusions. Scale analysis is used in this section of appendix B to show that the advection of a gaseous cargo by the wind will be much more effective in spreading the gas than will eddy diffusion alone, and it is used to indicate how a suitable time step might be chosen, and how to decide whether the plume or puff model is the more appropriate. Finally, it is shown that certain spills are too small to be treated accurately by the VM.

Let us define the following scale quantities:

L = 100 m = horizontal scale of surface roughness.

V = 30 m = vertical scale of surface roughness.

- H = 3000 m = scale of distance over which a hazardous condition might exist.
- U = 5 m/s = typical wind speed
- K = 1000 m^2/s = horizontal eddy diffusivity on the scale of H.
- t₋ = duration of gas release.

From these basic parameters we may derive several other quantities:

 $t_h = H/U = 10$ min = transport time for scale length H.

 $t_k = H^2/K = 2.5$ hr = diffusion time for scale length H.

 $t_1 = L/U = 20$ s = transport time for the scale length L.

In fluid dynamics it is customary to characterize the roughness of a surface by a representative length. Most surfaces, over which a fluid may flow, have surface imperfections of a random nature. In a pipe the surface roughness is caused by scratches, pits, and protuberances on the inner surface. For atmospheric flows the surface roughness is caused by topological features of the ground. In an urban area the vertical scale of surface roughness would be approximately the average height of the buildings. The horizontal scale of surface roughness would be close to the average distance between individual buildings or between groups of adjacent buildings of similar height. The surface roughness lengths have been taken to be characteristic of an urban area, for this is the area in which a spill would be most objectionable. For grassland or the open sea, the parameters L and V would have values about 1/100 of the values used above. The value of K, the horizontal turbulent diffusion coefficient, is taken from the data of Richardson as presented on p. 93 of [B7].

[[]B7] Slade, D. H., ed. Meteorology and Atomic Energy. U.S. Atomic Energy Commission, TID-24190, 1968.

The quantities above are representative values, and only give the orders of magnitude involved. This is sufficient for the problems addressed here. For example, comparison of t_h and t_k indicates the relative importance of wind advection and turbulent diffusion in dispersal of a substance in the atmosphere. For almost any wind speed above zero, the transport by the wind will be more important in spreading the material than will diffusion. For any scale distance H and wind speed U, the ratio of the wind transport time to the diffusion time is given by the dimensionless ratio K/H/U. For the values used above, this ratio is 1/15.

Choice of Time Step

The choice of the time step to be used in the air dispersion submodel of the VM must be based on the scale of the problem as well as the computational and cost factors. Only the input from scale considerations are considered here. Because the air dispersion model is not intended for use on the scale of the surface roughness, a time step, Δt , on the order of t_1 is not appropriate. Although the VM will function perfectly well for any size time step, the use of a time step as small as t_1 or smaller may easily lead one to make unwarranted inferences about the accuracy of the VM. There is little if anything to be gained by the use of time steps less than t_1 .

On the other hand, the VM is intended for use on the scale H, and is reasonably accurate on such a scale. To get some detail on events on this scale, Δt must be chosen smaller than t_h . There are no definite rules, but typically one might choose Δt to be between $t_h/10$ and $t_h/5$. Interpretation will be easiest if Δt is an even number of convenient units. For the scale distances and wind speed listed above, $t_h = 10$ min, so either $\Delta t = 1$ min or $\Delta t = 2$ min would be reasonable choices. To follow a cloud of toxic gas over long distances, one might wish to increase Δt somewhat.

Plume Model or Puff Model

The characteristic times defined above are useful in determining whether the plume or the puff model is the most appropriate. Clearly the puff model is inappropriate if the time of gas release is as long or longer than the time needed to traverse the scale of distance over which damage may reasonably be expected. Thus if the time over which gas is released, t_r , is greater than t_h , the plume model should be used. It is also evident that the puff model is called for if t_r is less than t_1 . For the case where t_r is between t_1 and t_h , there is no obvious dividing line between the cases for which the plume model is appropriate and the cases for which the puff model is appropriate. The line might well be drawn in the region between $3t_1$ and $t_h/3$, however. If Δt is chosen to be about $t_h/10$, then the plume model may be specified if t_r is greater than $3\Delta t$ or $5\Delta t$.

Lower Limit of Applicability

For a spill to be capable of causing damage, and also to be validly treated by the VM, it must exceed a certain size. When the vapor cloud

from a small spill is concentrated enough to be hazardous, it may be smaller than surrounding structures, and so the air dispersion submodel will not predict its motions and spreading correctly. If, when the cloud has expanded to a size such that the air dispersion submodel will treat it adequately, the concentration of cargo vapor is so low that the event is no longer of interest, then such a spill is too small to be considered by the VM. If the glob of vapor which is of interest is considerably smaller than the scale of the roughness of the surface, then this glob, if near the surface, will be inaccurately treated by the air dispersion models. Since vapors are no longer a threat when considerably elevated, the VM treats only vapors near the ground or water surface.

The second second second second second second second second second second second second second second second s

いたいたいとうないないないないというないないないです。

The scales L and V above were chosen for urban areas. For suburban areas, these scale lengths must be reduced by a factor of 2 or 3. For open fields or the sea surface, reduction by a factor near 100 is in order. Since most of the vulnerable resources are concentrated in urban areas, the lengths used above were chosen.

It is clear from the way in which the gaussian models were developed, and from observations of the eddies around buildings, that a glob of vapor which is much smaller than the buildings among which it is found will not be treated accurately by the VM. It follows then, that a glob of vapor with a radius of L/10 = 10 m is one for which the VM is inapplicable. Since we are interested in spills and vapors near the ground, let us consider a hemisphere of radius 10 m, which has a volume of 2100 m^3 .

An idea of the minimum amount of material for which the model is applicable can be obtained by calculating how much of the substance this hemisphere would contain at some lower significant concentration. For flammable substances this concentration is taken to be the lower flammable limiting concentration, also known as the lower flammable limit. For toxic substances, the concentration which causes immediate irritation has been used as the significant concentration.

As a rough estimate of the lower limit of the amount of a cargo which must be spilled in order for the dispersion model to be applicable, the amount of the cargo necessary to create a hemisphere of radius 10 m with a uniform concentration at the significant level has been calculated for each of the primary cargos. The results are given in Table B3-1.

This table is interpreted as follows: For the case of LNG, it takes 53 gallons to create a hemisphere 10 m in radius at the lower flammable concentration. A hemisphere this small cannot be treated accurately by the VM in an urban area because of the size of the surrounding buildings. If the cloud were to expand to a size which the VM does treat accurately, then the vapor would be so diffuse that it could not ignite and the danger of explosion or fire is not present. Thus, a spill of 53 gallons = 0.20 m^3 of LNG is too small to be treated by the VM.

Since the volume of a hemisphere is proportional to the cube of its radius, a spill of 53,000 gallons = 200 m^3 is necessary to create a hemisphere of radius 100 m at the lower flammable concentration. This hemisphere

is on the same scale as the surface roughness, and therefore represents the order of magnitude of the spill large enough to be treated adequately by the VM. Between 0.20 m³ and 200 m³, the air dispersion submodel becomes more and more applicable as the size of the spill is increased, but no definite line can be drawn to separate a region of applicability from a region of inapplicability.

كسناها ومشطيك وجريب

and the second second

Thus the volumes in the two righthand colurns of table B3-1 represent spill sizes for which the VM is not applicable. The VM is certainly applicable for a spill of 1000 times as much cargo as fisted in these columns, however, and for spills between these two sizes the VM must be applied with caution.

TABLE B3-1

Volume of Liquid for the Five Priority Cargos Which Is Necessary to Create a Hemisphere of Radius 10 m at the Significant Concentration

a multiplication and

Cargo	Significant Concentration	Mass in Hemisphere	Volume in Liquid Phase	
		kg	m	gallons
LNG	5.3%	80	0.20	53
Methyl Alchohol	7.3%	210	0.27	71
Gasoline	1.4%	130	0.19	50
Chlorine	45 mg/m ³	0.084	0.000056	0.015
Anhydrous Ammonia	700 mg/m ³	1.5	0.0018	0.48

APPENDIX C1

ľ

IGNITION SOURCE CONSIDERATIONS

As a further refinement to the fire and explosion modeling it has been suggested that gradations within the classes of ignition sources be considered. Such gradations would permit the user to specify at a given location a type of ignition source that would be capable of igniting some flammable materials but not others.

The approaches to implementing some type of ignition source gradation in the VM is based on a somewhat unconventional use of the concept of flashpoint. According to one standard text:

> Flash Point of a liquid is the lowest temperature of the liquid at which it gives off vapor sufficient to form an ignitible mixture with the air near the surface of the liquid or within the vessel used. By "ignitible mixture" is meant a mixture within the flammable range (between upper and lower limits) that is capable of the propagation of flame* away from the source of ignition when ignited. Combustion is not continuous at the flash point. This term applies mostly to flammable liquids, although there are certain solids, such as camphor and naphthalene, that slowly sublime (change from a solid to a vapor) at ordinary room temperature and therefore have flash points while still in the solid state. [C-1]

*By "propagation of flame" is here meant the spread of flame from layer to layer independently of the source of ignition. A gas or vapor mixed with air in proportions below the lower limit of flammability may burn at the source of ignition, that is, in the zone immediately surrounding the source of ignition, without propagating (spreading) away from the source of ignition.

Thus flashpoint is seen to be a property of the fiammable substance; flashpoint is related to two more fundamental properties of the substance, viz. the limits of flammability and the dependence of vapor pressure on temperature. Nevertheless flashpoint gives a measure of the ease by which a particular substance may be ignited. The lower the flashpoint, the easier it is to ignite the material.

Since flashpoint is a commonly accepted measure of flammability, it seems reasonable to "turn-the-tables" and use flashpoint as the basis for

[[]C-1] Fire Protection Handbook, p. 4-8 ff. G. H. Tryon, ed.-in-chief. National Fire Protection Association, Boston, 1969.

a measure of ignition source strength.

At the present time it seems undesirable to perform the data preparation and modeling required to characterize real-world ignition sources, their ability to ignite a certain hazardous cargo, and their distribution spatially and temporally. Instead a less detailed model, that is more easily developed and used, is desired. Again the user should be able to keep all parameters of a simulation fixed except the type of flammable substance and have an ignition for one substance, but no ignition for another. Basing the ignition source strength on flashpoint permits the user to have this capability. Powerful ignition sources will ignite all combustible materials; weak ignition sources will only be able to ignite very flammable substances, i.e. those with very low flashpoints.

فكتعاشك تتماسين مستحاذها اللاعلام وترافع للمتعاقفة فتمعد

تعلك بأشاله معد

1 T

One approach to providing such gradations would be to designate an ignition temperature for each ignition source. Those materials (generally flammable liquids) that have a flashpoint less than or equal to the given ignition temperature would be ignited (provided of course that the vaporair mixture had a concentration within the ignitable range), while those materials with a flashpoint above the given ignition temperature would not be ignited. This rather unconventional application of the concept of flashpoint is primarily an attempt to allow the user to specify ignition sources of different strengths; thus two simulations having the same input data, except for the type of substance spilled, will yield ignition in one case but not in the other. Flashpoint, of course, is a material property encompassing a set of physical and chemical parameters such as volatility, ignitibility, specific heat, and vapor density. The use of flashpoint to characterize ignition sources is recognized to be an artifice without a clear phenomenological basis. However, it does allow the user of the VM, albeit in an artificial manner, to vary the strength of the ignition sources specified. Another approach to providing gradations of the fire-type ignition source would be to characterize ignition sources on the basis of the standard classifications of flammable liquids. Table C1-1 shows the ICC and NFPA classifications for flamma'le liquids. An ignition source given a certain level designation would be capable of igniting all those substances whose flashpoints fell within the limits for that level and would be capable of igniting all those substances whose hazard level was greater. For example, using the NFPA classification an ignition source designated moderate could ignite methyl alcohol (flashpoint = +65°F, within the moderate range) as well as gasoline (flashpoint = -50° F, within the high range); however, it would not ignite spirits of turpencine (flashpoint = 95°F, within the slight range) nor would it ignite olive oil (flashpoint = 437°F, within the combustible range).

For the Vulnerability Model, the NFPA classification of flammable substances is used as the basis for ignition source differentiation. The flashpoint of flammable substances is not currently included in the properties file; consequently this information will be provided as input data until the properties file is revised. Guidelines will be provided to the user suggesting which type of ignition source to use depending on the land use or some other indicator of human activity. For computerization the ignition sources are denoted by a single digit code as indicated in Table C1-2. An ignition source designated by "0" means no ignition regardless of the flammability class of the spilled substance. For ignition source codes 1 through 4 ignition will occur provided the flashpoint of the material spilled is less than or equal to the upper bound of the NFPA range indicated. This relationship between ignition source code and the flashpoint of substances subject to ignition is given in Table Cl-3. If the sign of the code is positive, then the ignition source is understood to cause conflagration only; i.e., the flash fire submodel is to be selected. If the sign of the code is negative, then the ignition source is understood to cause detonation; i.e., the explosion submodel is to be selected.

į

Ι,	ICC:			
		flash point	fire hazard	
		< 80°F 80°F-350°F > 350°F	high moderate slight	
11.	NFPA:			
	_	flash point	fire hazard	
		< 20°F 20°F-70°F 70°F-200°F > 200°F	high moderate slight combustible	

and the second second second second second second second second second second second second second second secon



CLASSIFICATION SYSTEMS FOR FLAMMABLE LIQUIDS [C-2]

	NFPA Classification		
	Flash point of	Fire hazard of	
	substances ignited	substances ignited	Ignition potential
Code	by the source	by the source	of the source
0			none
1	< 20°F	high	low
2	20°F-70°F	moderate	moderate
3	70°F-200°F	slight	high
4	> 200°F	combustible	all combustibles

TABLE C1-2

NUMERICAL DESIGNATION OF IGNITION SOURCE STRENGTH USED IN THE VM

Ignition Source	Flash point of Sp	Flash point of Spilled Material		
Code	Ignition	No Ignition		
0	none	all		
1	F.P. ≤ 20°F	F.P. > 20°F		
2	F.P. < 70°F	F.P. > 70°F		
3	F.P. <200 [°] F	F.P. > 200°F		
4	all combustibles	only non-combustibles		

TABLE C1-3

RELATIONSHIP BETWEEN IGNITION SOURCE CODE AND FLASH POINT OF SUBSTANCE IGNITED OR NOT IGNITED

Destate and the fam

1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997

[[]C-2] Dangerous properties of industrial materials, 3rd ed., p. 198. N.I. Sax, ed. Van Nostrand Reinhold Co., New York, 1968.

APPENDIX C2

THE TAKE TO STATE THE STATE

CALCULATION OF THE AMOUNT OF FUEL CONSUMED

Three ways of calculating the mass of fuel consumed were considered. An upper bound on the quantity consumed is provided by assuming that all the fuel at a concentration above the lower limit of flammability burns completely; i.e.

$$\mathbf{m_{e_1}} = \int C(\mathbf{x}, \mathbf{y}, \mathbf{z}) d\tau \qquad (C2-1)$$

where

C (x,y,r,t) = the concentration of the fuel (kg/m³) dt = element of volume V₁ = the region enclosed by the surface, C (x,y,z,t) = K_L (lower explosive limit*) in the half-space z > 0.

A second estimate of the quantity of fuel consumed is obtained by assuming that all the fuel at a concentration between the upper and lower flammability limit is consumed; for this case

$$m_{e_2} = \int_{V_2} C(x,y,z,t) dt$$
 (C2-2)

where V₂ is the region enclosed by the two surfaces, C $(x,y,z,t) = K_L$ and C $(x,y,z,t) = K_{II}$ (upper explosive limit) on the half-space $z \ge 0$.

A lower limit of the quantity of fuel consumed is obtained by assuming (1) only fuel at concentrations within the flammability range is consumed and (2) the fuel is not necessarily consumed completely, but is consumed only to the extent that sufficient oxygen is present. Thus, for this case

$$m_{e_3} = \int_{V_3} C(x,y,z,t) dt + \int_{V_4} F(C) C(x,y,z,i) dt \qquad (C2-3)$$

where

 V_3 = the region enclosed by the surfaces C (x,y,z,t) = K_L and C (x,y,z,t) = K_s (stoichiometric) in the half space $z \ge 0$.

 V_4 = the region enclosed by the surfaces C (x,y,z,t) = K_U and C (x,y,z,t) = K_a in the half space $z \ge 0$.

and F (C) is a weighting function giving the fraction of the fuel present that enters into the combustion reaction. That is, if the fuel-air mixture is richer than stoichiometric then we desire to compute the weighting

Although some authors distinguish between flammable and explosive limits of concentration, many other authors, including several standard references do not. Certainly for many cases of diffuse vapor clouds the distinction may well be impossible to make. Therefore, for the purposes of the VM, explosive and flammable limits are taken to be identical.

function, F (C), which is the mass fraction of fuel present that is consumed, i.e.

$$\mathbf{F}(\mathbf{C}) = \frac{M_{\mathbf{C}}}{M_{\mathbf{p}}} \tag{C2-4}$$

 $M_{\rm C}$ = the mass of fuel consumed and $M_{\rm D}$ = the mass of fuel present.

These masses are, of course, proportional to the number of moles in each category, i.e.

 $M_{\rm r} = n_{\rm c} M_{\rm f}$ (C2-5a)

 $M_{\rm p} = n_{\rm p} M_{\rm f} \tag{C2-5b}$

where

The second s

Ę

-

ت المحتم م

aliati in a

 n_c = number of moles of fuel consumed n_p = number of moles of fuel present M_f = molecular weight of the fuel.

Then the mass fraction consumed can be written as,

$$F(C) = \frac{n_c}{n_p}$$
(C2-6)

The combustion reaction of the fuel can always be written in the form,

1 mole fuel + L moles $0_2 \rightarrow$ reaction products.

Since air is 21% oxygen by weight, if one mole of fuel requires L moles of of oxygen for complete combustion (i.e. a stoichiometric mixture), then one mole of fuel will require (L/0.21) moles of air. Phrased another way, each mole of reacting air will consume (0.21/L) moles of fuel. Hence the number of moles of fuel consumed, n_c , can be related to the moles of air present in the rich mixture by,

$$n_c = (\frac{0.21}{L}) n_a$$
 (C2-7)

where ,

 $n_a =$ number of moles of air present in the rich mixture (all the 02 is presumed consumed).

Thus the mass fraction is written as,

$$F(C) = (\frac{0.21}{L}) \frac{n_a}{n_p}$$
 (C2-8)

The quantities n_a and n_p represent the number of moles of each component present in the two-component fuel-air mixture. The total number of moles present is given by,

 $n = n_a + n_p \tag{C2-9}$

where,

n = the total number of moles in the two component mixture.

CH STREET, LOW SHOWS

Now assuming that the perfect gas law holds for the fuel, the air, and the mixture of the two, we may write for a system containing hypothetical fuel only,

The area warded to the state of

OPENING LINES, JIL NESS

but his she want the

$$P = \frac{\rho_f RT}{M_f}$$
(C2-10)

where

- P = the total pressure
- T = the temperature
- R = the perfect gas law constant

 ρ_f = the density of the fuel at temperature, T, and pressure, P

But the perfect gas law may also be written, for an elemental volume V of the mixture as,

$$P = \frac{nRT}{V}$$
(C2-11)

or

 $n = \frac{PV}{RT}$

hence by the use of equation (C2-10), where identical temperatures and pressures are assumed for the mixture and the system containing only fuel,

$$\pi = \frac{V \rho f}{M_{\rm f}}^{\star} \tag{C2-12}$$

Now for an elemental volume of the mixture the mass of fuel present is gi en by,

 $M_{\rm p} = CV \tag{C2-13}$

hence the number of moles present is given by,

$$n_{\rm p} = \frac{M_{\rm p}}{M_{\rm f}} = \frac{CV}{M_{\rm f}} \tag{C2-14}$$

Rearranging equation (C2-9),

Also note that

$$n = \frac{V \rho a}{M_a}$$

This is a consequence of Aragandro's Law.

and substituting equations (C2-12) and (C2-14) in the result yields,

$$n_{a} = \frac{V \rho_{f}}{M_{f}} - \frac{CV}{M_{f}}$$

$$= \frac{V}{M_{f}} (\rho_{f} - C)$$
(C2-15)

Note that when $C=\rho_f$ the mixture is pure fuel and the number of moles of air as given by equation (C2-15) is zero. Using this result and equation (C2-14) the mass fraction, as given by equation (C2-8) can be written as.

F (C) =
$$(\frac{0.21}{L})$$
 $\frac{\frac{V}{M_{f}}(\rho_{f} - C)}{\frac{CV}{M_{f}}}$

17

or

1.1

marin 's'r.

ĉ

こうしょう あい

and series for

and a second for

(C2-16)

F (C) =
$$(\frac{0.21}{L}) (\frac{\rho_f}{C} - 1)$$

Now that the weighting function, F(C), has been determined the third estimate, m_{e_3} (the estimate ultimately used) for fuel participating in rapid combustion, can be computed.

Thus equation (C2-3) becomes

$$\mathbf{m}_{e_3} = \int_{\mathbf{v}_3} Cd\tau + \frac{0.21}{L} \left[\rho_f \, \overline{\mathbf{v}}_4 - \int_{\mathbf{v}_4} Cd\tau \right]$$
(C2-17)

where

 \overline{V}_4 is the volume of region V_4 .

To simplify the treatment of the volume integrals that occur in equations (C2-1), (C2-2), and (C2-17), only surface spills will be considered. This is consistent with restrictions already made in this program. Furthermore, the general behavior is expected to be the same regardless at what height the spill occurs. For a surface spill, the concentration according to the CHRIS model is given by,

$$C(x,y,z,t) = \frac{2m}{(2\pi)^{3/2}\sigma_x\sigma_y\sigma_z} exp - \left[\frac{(x-Ut)^2}{2\sigma_x^2} + \frac{y^2}{2\sigma_y^2} + \frac{z^2}{2\sigma_z^2}\right] (C2-18)$$

where

 $\sigma_x \cdot \sigma_y, \sigma_z$ = the variances of the Gaussian concentration profile in the respective directions (m)

m = the total mass of spill in the vapor phase x,y,z, = coordinates U = wind velocity t = time By performing the coordinate transformation,

$$\left[\frac{\mathbf{x}-\mathbf{U}\mathbf{t}}{\sigma_{\mathbf{x}}}\right] = \mathbf{r} \sin\theta \cos\phi$$

$$\left[\frac{\mathbf{y}}{\sigma_{\mathbf{y}}}\right] = \mathbf{r} \sin\theta \sin\phi \qquad (C2-..9b)$$

$$\left[\frac{\mathbf{z}}{\sigma_{\mathbf{z}}}\right] = \mathbf{r} \cos\theta \qquad (C2-19c)$$

the volume integrals bounded by surfaces of constant concentration can be transformed to a more convenient form. Thus, for example, for equation (C2-1) we get,

$$me_{1} = \frac{2m}{(2\pi)^{1/2}} \int_{0}^{rL} e^{-r^{2}/2} r^{2} dr \qquad (C2-20a)$$

where

$$r_{L} = \left[2 \ln \left(\frac{2m}{(2\pi)^{3/2}\sigma_{x}\sigma_{y}\sigma_{z}K_{L}}\right)\right]^{1/2}$$
(C2-20b)

and K_L = concentration at the lower explosive limit (kg/m^3) .

If the quantity

$$\frac{2\mathfrak{m}}{(2\pi)^{3/2}\sigma_{x}\sigma_{y}\sigma_{z}K_{L}}$$

is less than one, the logarithm is negative; i.e. there is no real solution for rL. What this means physically is that the variances $({}^{\sigma}_{X}, {}^{\sigma}_{Y}, {}^{\sigma}_{Z})$ are so large that the concentration everywhere in the vapor plume is less than KL. For such a case the exploding mass is obviously to be taken to be zero. On the other hand when the variances are small (this is the case soon after the time of release of the vapor), rL becomes very large. In such a case

$$\frac{1}{(2\pi)^{1/2}} \int_{0}^{1} e^{-r^{2}/2} r^{2} dr = \frac{1}{2} \quad \text{for} \quad r_{L} >> 1$$

so that $m_{e_1} \stackrel{\sim}{\underline{}} m$.

In other words almost the entire mass of vapor released burns or explodes. Since the explosive yield of spills never seems to be greater than 10 percent of the total yield available from all the material spilled[C3], the approximation given by m_{e_1} may grossly overestimate the yield. The overestimation will occur at times soon after the spill when most of the mass is still undiffused and therefore contained within the lower explosive limit contour. Therefore, m_e is rejected as a suitable approximation to the mass of vapor that enters in the explosive reaction.

[[]C3] Brown, John A. A study of the growing danger of detonation in unconfined gas cloud explosions. John Brown Associates, Inc., Berkeley Heights, New Jersey. December 1973.

The second estimate of exploding mass given by equation (C2-2) gives more reasonable answers; however this estimate also seems to give values which are generally too high. Thus equation (C2-2) becomes,

$$m_{e_2} = \frac{2m}{(2\pi)^{1/2}} \int_{r_0}^{r_L} e^{-r^{2/2}} r^2 dr \qquad (C2-21a)$$

where r_{I} is given by equation (25-b) and

$$r_{U} = \left[2 \ln \left(\frac{2m}{(2\pi)^{3/2}\sigma_{x}\sigma_{y}\sigma_{z}K_{U}}\right)\right]^{1/2}$$
(C2-21b)

and $K_{U} = \text{concentration at the upper explosive limit } (kg/m³)$

Equations (C2-20b) and (C2-21b) may be combined to yield,

$$r_{L}^{2} - r_{U}^{2} = 2 \ln \left[\frac{K_{U}}{K_{L}}\right]$$
 (C2-22)

From equation (C2-22) it is clear that the integration limits in equation (C2-21a), viz. r_L and r_U , are not independent. In a sense m_{e_2}/m , the fraction of available mass that this estimate gives as exploding, can be treated as a function of the lower limit of integration r_U , with the explosive limits ratio, [KU/KL] a parameter. For any given r_U the integration interval (hence m_{e_2}/m) is increased as the ratio, [KU/KL], is increased. This is what one would expect physically.

If r_U is taken to be very large, r_L is also very large and the integral is small. The case of ry large corresponds to times soon after the spill when very little mass is contained in the region between the explosive limits. As stated previously, whenever the argument of the square root in equations (C2-20b) and (C2-21b) is negative the radius, ry or r_L , is to be set equal to zero. Thus for very diffuse vapor clouds both ry and r_L approached zero. In that case the integral also zero. Thus the estimate me2 has the desirable property that very rich or very lean vapor clouds both tend to have small explosive yield, while at some intermediate point the explosive yield is maximum. This behavior is what is expected to occur.

It appears that the largest yield is obtained when $r_U = 0$. Values of m_{e_2}/m_{are} plotted as a function of KU/KL in Figure C2-1. Values of KU/KL for four flammable substances of special interest are given in Table C2-1. As shown in Figure C2-1, when KU/KL is large (e.g. for acetone KU/KL = 32) nearly the entire mass of vapor is within combustible limits. Even for substances less pernicious than acetone, for example, methanol with KU/KL = 5, more than 60% of the vapor explodes. This estimate gives yields which are several times more than the five to ten percent of mass spilled that has been observed. Consequently it seems appropriate to use the lowest estimate, m_{e_3} .

By using the transformation of coordinates given by equations (C2-19), equation (C2-17) may be rewritten as,



Ammonia	1.563
Gasoline	5.429
Methane	2.642
Methanol	5.00

KU/KL

·· ·

. .

The second second second second second second second second second second second second second second second s

TABLE C2-1

RATIOS OF UPPER EXPLOSIVE LIMIT CONCENTRATION, KU, TO LOWER EXPLOSIVE LIMIT CONCENTRATION, KL, FOR FOUR SUBSTANCES OF SPECIAL INTEREST.

$$\frac{m_{e_3}}{m} = 2 [S(r_L) - S(r_S)] - 0.42 [S(r_S) - S(r_U)]$$

$$+ \frac{0.21}{L} - \frac{\rho_f \sqrt{4}}{m}$$
(C2-23)

where

$$r_{S} = \left[2 \ln \left(\frac{2m}{(2\pi)^{3/2}\sigma_{x}\sigma_{y}\sigma_{z}}K_{S}\right)\right]^{1/2}$$
(C2-24)

いいたいまでありていている」と言いている

<u>ىلىكىكەر ئىلغانىكە ئەلمەر ئەلمەرلىكىكەر ھەر بىل يەھەر دەمەرلەركىكە بىلەر مەتتىرىلى ئەلمەر مەمەيكەر بەلمەك بە</u>

 K_S = stoichiometric concentration (kg/m³)

and where

$$S(r) = \int_{-\infty}^{r} \frac{1}{(2\pi)^{1/2}} e^{-\frac{x^{2}}{2}} x^{2} dx \qquad (C2-25)$$

or

$$S(r) = P(r) - \frac{1}{\sqrt{2\pi}} r e^{-\frac{r^2}{2}}$$
 (C2-26)

where

$$P(r) = \int_{-\infty}^{r} \frac{1}{(2\pi)^{1/2}} e^{-\frac{x^{2}}{2}} dx \qquad (C2-27)$$

is the cumulative normal probability function. Now from equation (C2-21b)

$$\frac{2\mathfrak{m}}{(2\pi)^{3/2}\sigma_{x}\sigma_{y}\sigma_{z} V_{U}\rho_{f}} = e^{r_{U}^{2}} \qquad (C2-28)$$

since

 $K_{U} = V_{U} S_{f}$

where VU is the volume fraction of vapor at the upper explosive limit. Furthermore, the volume of the ellipsoidal shell comprising V4 is given by,

$$\overline{v}_4 = \frac{2\pi}{3} \sigma_x \sigma_y \sigma_z (r_s^3 - r_t^3)$$
 (C2-29)

Combining these two results gives,

$$\frac{\frac{2}{r} \nabla_4}{m} = \frac{2 e^{-r_U^2} (r_S^3 - r_U^3)}{3 (2\pi)^{1/2} v_U}$$
(C2-30)

hence the mass ratio given by equation (C2-23) becomes,

$$\frac{me_3}{m} = 2[S(r_L) - S(r_S)] - \frac{0.42}{L} [S(r_S) - S(r_U)] + \frac{0.21}{L} \frac{2 e^{-r_U^2} (r_S^3 - r_U^3)}{3(2\pi)^{1/2} v_U}$$
(C2-31)

Again this estimate appears to give a maximum value when $r_U = 0$. For that condition and for methanol (KU/KL = 5, KU/KS = 2.9723) meg/m has a value of 0.216. This value is still higher than observed explosions; however, the value of meg/m = 0.216 is favorably smaller than the value meg/m = .641 obtained for KU/KL = 5. Therefore meg should be used as the estimate of mass exploded because (1) this estimate exhibits a maximum value as expected physically, (2) the apparent maximum value for a typical explosive vapor is conservative; i.e. the explosive yield predicted is probably larger than that actually obtained, and (3) this estimate depends upon the stoichiometric concentration as well as the upper and lower explosive limits; therefore, it is physically more realistic.

APPENDIX C3

EQUIVALENCY OF CHANGE IN HELMHOLTZ FREE ENERGY AND CHANGE IN ENTHALPY FOR ESTIMATING THE YIELD OF VAPOR-AIR EXPLOSIONS

To show the close agreement between the change in Helmholtz free energy and the change in enthalpy, proceed as follows. By definition,

H = E + PV

(C3-1)

(C3-2)

where,

H = enthalpy E = internal energy P = pressure V = volume

Differentiation of equation (C3-1) yields,

dH = dE + P dV + V dP

and for an isobaric process this becomes

dH = dE + P dV

The usually accepted standard heat of combustion is the change in enthalpy resulting from the combustion of a substance, in the state that is stable at 25°C and atmospheric pressure, with the combustion beginning and ending at a temperature of 25°C. Thus the tabulated values for change in enthalpy resulting are for an isobaric process. Thus equation (C3-2) may be written as,

 $\Delta H = \Delta E + P \Delta V \tag{C3-3}$

According to Kinney [C4], the energy yield of an explosion is given approximately by the change in Helmholtz free energy, $-\Delta A$; $-\Delta A$ is computed using,

 $-\Delta A = -\Delta E + T \Delta S$ (C3-4)

where,

 ΔE is the heat of explosion measured at constant volume

T is the temperature and

 ΔS is the change in entropy for the isothermal process.

[[]C4] Kinney, Gilbert Ford. Explosive Shocks in Air, p. 11. The MacMillan Co., New York, 1962.

The quantity E has the same meaning in equation (C3-3) and (C3-4). Combining equations (C3-3) and (C3-4) gives

$$-\Delta A = -\Delta H + P \Delta V + T \Delta S$$

or

$$(-\Delta A + \Delta H) = P \Delta V + T \Delta S \qquad (C3-5)$$

Equation (C3-5) states that the difference between $-\Delta A$ and $-\Delta H$, the alternative quantities used to compute explosive yield, is given by (P ΔV + T ΔS).

Consider the combustion of methane by the reaction,

$$CH_4 + 2 O_2 \neq CO_2 + 2 H_2O$$
 (C3-6)

Since three moles of gaseous reactants combine to produce three moles of gaseous combustion products, there is no change in volume, i.e., at the same temperature and pressure (by definition the reactants and products are brought to 760 mm Hg at 25°C), volumes of gas containing equal number of moles are of equal volumes (Avogadro's Law [C5]). Thus for the reaction given by equation (C3-6), ΔV in equation (C3-4) is zero.

Now for a mixture of ideal gases, the entropy is given by,

$$S = R \sum_{k=1}^{N} n_{k} (\sigma_{k} - \ln P - \ln X_{k})$$
(C3-7)

where

$$\sigma_{k} = \frac{1}{R} \left[\int \frac{C_{p_{k}} dT}{T} + s_{o_{k}} \right]$$
(C3-8)

and

S = entropy of mixture
nk = number of moles of gas species "k"
P = pressure
Xk = mole fraction of gas species "k"
Cpk = specific heat of gas species "k"
R = gas constant

 S_{OL} = standard molar entropy at STP

[[]C5] Hougen, O.A., K.M. Watson, and R.A. Ragatz. Chemical Process Principles, Part I, p. 305. John Wiley & Scns. Inc., New York, 1959.

Equation (C3-7) now can be used to calculate the entropy on both sides of the reaction (C3-6) so that the change in entropy can be computed. Note that the terms

ł

And the state of the second second

$$- R \sum_{k=1}^{2} n_k \ln P$$

.

and

$$= R \sum_{k=1}^{2} n_k \ln X_k$$

are identical on either side of the reaction. Hence,

$$\Delta s = s_{R} - s_{L} = s_{\circ CO_{2}} + 2 s_{\circ H_{2}O} - s_{\circ CH_{4}} - s_{\circ O_{2}}$$

(where S_R , S_L are the entropy on the right and left hand sides respectively of the reaction (C3-6).

Following through the computation with tabulated values gives,

$$\Delta S = -1.28 \frac{cal}{g-mole^{\circ}K}$$
(C3-9)

hence,

$$T\Delta S = -381.44 \frac{cal}{g-mole}$$
 (C3-10)

Referring again to equation (C3-5) we see that the difference between $-\Delta A$ and $-\Delta H$ is -381.44 cal/g-mole. However the value of ΔH for the combustion of methane is -212.798 x 10³ cal/g-mole. In other words, for this reaction the error made in using $-\Delta H$ to represent $-\Delta A$ is of the order of 1/10 of one percent.

Now consider the combustion of n-octane by the reaction,

$$C_8H_{18} + \frac{25}{2}O_2 + 8CO_2 + 9H_2O$$
 (C3-11)

Again using equation (C3-7) we obtain,

$$\Delta S = 107.385 \frac{cal}{g-mole-K}$$
(C3-12)

In the combustion of n-octane, however, the number of moles of gas changes, hence the term $P\Delta V$ in equation (C3-5) is not zero. From the perfect gas law,

PV = nRT

(C3-13)

it can be shown for a reaction in which the reactants and product are brought to the same temperature and pressure, that

 $\Delta V = \Delta n R T$

(C3-14)

where Δn is the change in the number of moles in the gas phase by virtue of the reaction. For the reaction given by equation (C3-11)

 $\Delta n = -3.5$

. «»، « تخطيف خلافاً الأقسم بالقديات» ، «الما الرئيسة فقاطة والمحمولية»، « « المحمد معاملة الألفانية في المتكلفة فل

1.1.1.1

No. of Street, or other

Thus the total difference between $-\Delta A$ and $-\Delta H$ is

2.993 x 10⁴ <u>cal</u> g-mole

But the heat of combusion, $-\Delta H$, for n-octane is 1.307 x 10⁶ cal/g-mole. In other words, the error is of the order of two percent.

These rather lengthy calculations have been performed to demonstrate that the change in enthalpy is indeed a good approximation to the change in Helmholtz free energy. The energy available from the terms $P\Delta V$ and $T\Delta S$ are very important for the condensed phase explosives with which Kinney [C4] was mainly concerned; however, for gas mixture explosions, with which we are concerned, these terms are negligible.
APPENDIX C4

CORRECTION TO HEATS OF COMBUSTION

Correction for Water Heat of Vaporization

The value for heat of combustion, - ΔH , used in Chapter 4 to determine explosive yield (equation 4-2), is for final products that include water vapor; most handbooks give - ΔH for final products that include <u>liquid</u> water. These values for - ΔH must be corrected for the heat of vaporization of water.

This correction is given by,

 $-\Delta H_{\omega} = -\Delta H - n_{\omega} \lambda_{\omega}$

(C4-1)

where

- ΔH_{ω} = heat of combustion corrected for water vaporization

- ΔH = handbook value for heat of combustion
 - n_w = number of moles of water formed per mole of fuel burned
 - λ_{ω} = heat of vaporization of water.

For the combustion of methane as given in equation (C3-6) the correction factor is -19.43×10^3 cal/g-mole or about a nine percent correction. For the combustion of n-octane as given by the reaction in equation (C3-11) the correction factor is 0.08745×10^6 cal/g-mole or about a seven percent correction.

Correction for Fuel Heat of Vaporization

The value for heat of combustion, used in Chapter 4 to determine explosive yield (equation 4-2), $-\Delta H$, is for an initially vaporized (gaseous phase) fuel.

Heats of combustion are usually reported for the substance in the normal state at 25°C. Since the reaction of interest is for a substance normally a liquid at 25°C, then unless specified otherwise, the handbook value must be corrected to account for the heat of vaporization of the fuel. This correction is given by,

 $-\Delta H_f = -\Delta H - \lambda_f$

(C4-2)

whe re

- ΔH_f = heat of combustion corrected for fuel vaporization λ_f = heat of vaporization of the fuel.

Since methane is gaseous at 25°C and 760 mm H_g, no correction is required for the combustion of methane. For n-octane, which is a liquid at standard conditions, the correction factor is -9221 cal/g-mole or about 0.7 percent. For substances less easily vaporized the correction factor will, of course, be larger.

APPENDIX C5

化过去分子 医变性的

enclassic 🕋

the second second second

DAMAGE POTENTIAL OF CONFINED EXPLOSIONS

The following two charts give a measure of the damage potential of confined explosions. The charts consider the energy yield available when confined spaces are filled with combustible mixtures. The charts do not address the more sophisticated consideration of what seepage rates into and out of structures will produce combustible mixtures in the presence of a time varying atmospheric concentration. Neither do the charts address the additional hazard from fragments expected for confined, rather than unconfined explosions.

The first chart gives the explosive yield as a function of volume resulting from the ignition of a stoichiometric mixture of the materials noted, presuming the volume is completely filled with a stoichiometric mixture.

The second chart shows the effect on energy yield as a function of volume when the concentration is varied away from stoichiometric; of course for any concentration other than stoichiometric the energy yield is less given the same enclosed volume of vapor-air mixture. The substance is methanol. Ratios of 0.6 and 3 correspond approximately to the lower and upper explosive limits, respectively.



volume - al



್ಷ-೧೯೮೫ ಮುಗಗಳನ್ ವರ್ಷ

228

APPENDIX D

DAMAGE TO VULNERABLE RESOURCES FROM FIRE AND EXPLOSION

In this appendix further detail is provided about the basis for damage assessment in the event of fire or explosion. Since the physical phenomena causing the damage in a fire are quite different from the damage-causing phenomena in an explosion, the damage assessment for fire is discussed separately from that for explosion. Furthermore under each of the separate discussions for fire and explosion, damage to each type of vulnerable resource is treated separately; people and structures are the two types of vulnerable resource considered. Damage from explosion will be described first since damage from this source is likely to be more serious than from either a flash fire or pool burning.

EXPLOSION DAMAGE

In assessing damage from explosions that result from spills of hazardous materials, considerable use is made of studies performed for and by the military; these studies assess damage from both conventional and nuclear weapons.

As mentioned in Chapter 4 of this report, the blast wave resulting from a diffuse explosion behaves differently than a standard condensed phase explosion (like TNT) and certainly behaves differently than a nuclear explosion. Nevertheless, the assumption that diffuse and condensed phase explosions are similar yields small errors compared to those errors given by some other assumptions. Since much of the damage assessment analysis is presented in terms of basic blast wave parameters (sav overpressure rather than the TNT yield), if the analyses becomes available to predict the behavior of diffuse explosions, these predicted values may be used with the preexisting data to provide a more accurate damage assess-In the absence of a convincing and complete analysis of diffuse ment. explosions, the approach will be to treat the diffuse explosions as condensed phase explosions and assess damage accordingly. This approach is certainly state-of-the-art and is consistent with approximations in modeling made elsewhere in the VM.

Damage to Personnel

In choosing models to describe damage from explosions several sources were consulted, but primary among these is a recent document [D1] prepared for the Department of Defense Explosives Safety Board which is a compendium of assessment techniques for damage from conventional explosions. In this and other references, damage to personnel is classified into three categories:

[D1] Fugelso, L.E., L.M. Weiner, and T.H. Schiffman. Explosion effects computation aids. General American Research Division, General American Transportation Corporation, Niles, Illinois, June 1972. GARD Project No. 1540, AD903279 L - (the limited distribution denotation on this document has recently been lifted).

- primary damage direct blast effects (interaction between the blast wave and personnel only, with no other intervening or associated factors)
- (2) secondary damage damage from missiles and fragments
- (3) tertiary damage damage from translation and subsequent collision with an obstacle.

These categories of blast injury are based on the mechanism of damage and are independent of which of the basic blast wave parameters (peak overpressure or dynamic impulse) is used to assess damage of that category.

The discussions that follow, taken largely from references [D1] and [D2] explain in more detail, the nature of these categories of blast injury to personnel.

(1) Direct Blast Injury

When the human body is exposed to the environmental pressure variation accompanying a blast wave, the wave is transmitted in complex patterns throughout the body but, more important, the body wall is pushed violently inward. As a consequence of this implosive effect, very high transient internal pressures occur. These often exceed the external pressures by considerable amounts. Severe disruptive forces are produced at junctions of tissues of different densities such as bone with soft tissue or in air-containing organs such as the lungs and abdominal viscera. Only for "long" duration typical blast waves is the severity of the injury sustained roughly proportional to the magnitude of the peak overpressure; otherwise, and in addition, the hazard is a function of the duration and the rate and character of the rise of the pressure pulse. In general, except for the ears and sinuses, the human body is far more resistant to direct blast injury than are rigid structures such as buildings.

While eardrum rupture may occur at peak overpressure as low as 5 psi, the best value of the peak overpressure for 50 percent probability of rupture appears to be between 15 and 20 psi. Though painful, eardrum rupture is not serious. However, infection, fracture and displacement of the ossicles, including the foot plate of the stapes, may result in impaired hearing and require specialized treatment.

Blast injuries to the chest begin to occur at about 10-12 psi for "long" duration waves. These include bruising of the soft tissues of the chest wall adjacent to the ribs and rupture of small vascular elements of the pulmonary

[[]D2] Department of the Army. Nuclear Handbook for Medical Service Personnel. TM8-215, 1969.

tissue. Rupture of pulmonary vessels is always associated with interstitial hemorrhage and/or bleeding into the airways. Also, the risk of pulmonary edema is considerable and serious. With severe tearing and rupture, air emboli may occur with grave, frequently fatal, cardiac or cerebral complications. Clinical signs of pulmonary injury may be misleadingly mild until heart failure and edems appear: the heart failure and pulmonary edema are the primary treatment problems. . .

Blast injuries of the abdomen may include rupture of the liver and spleen, and rarely in airblast, perforations of the intestine. Organ perforations, especially in the lower ileum, cecum, and colon are much more common in underwater blast. Also, in airblast hemorrhages of the mesentery and gut wall are almost invariably present. Frequently there is abdominal wall rigidity without perforation of hemoperitoneum, and the significance of the signs and symptoms are difficult to assess. . . ([D2], pp. 6-7)

The primary cause of lethality from direct blast effects is lung hemorrhage. Data on direct blast injury to personnel have been obtained by experimentally determining overpressure-duration relationships for animals, and extrapolating these to humans. That is, the level of injury depends upon both peak overpressure level <u>and</u> the duration of the overpressure. For large-scale conventional explosions and most probably for all diffuse explosions, the duration of the blast wave may be considered "long". Thus it is current practice to use the free field (side on) overpressure, associated with various levels of lethality at infinitely large durations to assess deaths from direct blast effects. Table D-1 shows the relationship between overpressure and lethality from direct blast effects. The data in table D-1 were used to derive the probit equation E1:

 $Probit = -77.1 + 6.91 \log_{p} (Pp)$

(D-1)

where Pp is the peak overpressure measured in N/m^2 .

The main nonlethal injury resulting from direct blast effects is eardrum rupture. Unlike the lungs for which overpressure and blast wave duration together determined damage, eardrums are damaged in response to overpressure alone since the characteristic period of the ear vibration is small compared to the duration of a blast wave from even low yield explosions. The relationship between the overpressure and the probability of eardrum rupture is given in Table D-2. These data were used to generate the probit equation E3:

$$Probit = -15.6 + 1.93 \times \log_{2} (Pp)$$
(D-2)

where Pp is the peak overpressure measured in N/m^2 .

	Pe Over	eak pressure
Damage	(psi)	<u>(N/m²)</u>
Threshold (1% Lethality)	14.5	100,000
10% Lethality	17.5	120,000
50% Lethality	20.5	140,000
90% Lethality	25.5	175,000
99% Lethality	29.0	200,000

A CANAN AND A CANANA AND A CANANA AND A CANANA AND A CANANA AND A CANANA AND A CANANA AND A CANANA AND A CANANA

į

TABLE D1

CRITICAL OVERPRESSURES FOR LUNG DAMAGE TO HUMANS

The relationship between overpressure and death for direct blast effects. It is assumed that the pressure pulse duration is infinite and that lung hemorrhage is the operative cause of lethality. This table is based on data given in "Explosive Effects Compulation Aids" [D1].

Probability of	<u>rree-r</u> Overr	ressure
Eardrum Rupture	(psi)	(N/m^2)
Threshold (1%)	2.4	16,500
10%	2.8	19,300
50%	6.3	43,500
90%	12.2	84,000

TABLE D2

PROBABILITY OF EARDRUM RUPTURE

The relationship between peak overpressure and probability of eardrum rupture. This table is based on data given in "Explosive Effects Computation Aids [D1].

(2) Indirect Blast Injury

The transfer of momentum by a blast wave to objects in its path can result in injury from secondary missiles (both penetrating and nonpenetrating) or from displacement of the human body resulting in subsequent severe impact or decelerative tumbling; these are secondary and tertiary blast effects respectively. The injuries which result include wounds, such as contusions and fractures, which result from being thrown against an object. In addition, crush injuries from falling debris, should they occur, would be particularly more common in urban areas and less common in the open. Certain kinds of indirect blast injuries, such as violent decelerations or sharp blows to the head from blunt debris, are known to produce significant lethality just as does direct blast injury to the lung. However, the magnitude and severity of indirect hazards are very much dependent on the conditions of exposure, range, and explosive yield.

لللغاز مرزهيا كمكأكك كمرتما والملته ومنازلوا الأكافيك كالأطب للكمار

쀩

(a) Secondary Blast Injury

The missiles involved in secondary damage in the VM result from normal environmental debris (e.g., pebbles), damaged structures (e.g., broken glass), and fragments of cargo containers (e.g., ship hull fragments). Studies of military weapons have modeled missiles generated purposefully by weapon design (such as fragmenting shell casings) [D1]. Nevertheless nuclear weapon effects research has considered the effects of missiles resulting from debris set in motion by the nuclear blast waves [D3]. These studies have considered personnel damage from both penetrating and nonpenetrating missiles. The injuries which may result are lacerations and punctures from penetrating missiles or contusions, fractures, and internal injuries from large nonpenetrating objects. Penetrating missiles are light (10 grams or less in mass) compared to nonpenetrating missiles capable of causing injury (thousands of grams in mass). At the current stage of development the interest is for the VM to model diffuse, rather than condensed phase explosions. The occurrence of heavy missiles, capable of causing serious nonpenetrating woulds, is expected to be less common in diffuse explosions than in condensed phase or nuclear explosions. Therefore, the only kind of secondary damage currently considered by the VM is the nonlethal damage caused by penetrating missiles. Table D-3 gives damage criteria for both penetrating and nonpenetrating missiles. The critical parameter used to assess damage is the impact velocity of the missile. Another source [D2] gives data for penetrating missiles that are virtually the same, as shown in Table D-4.

For use in the VM the damage criteria based on missile impact velocity must be related to a blast wave parameter; it is clearly beyond the scope of the VM to treat the detailed motion of the thousands

[[]D3] White, Clayton S. The nature of the problems involved in estimating the immediate casualties from nuclear explosions. Lovelace Foundation for Medical Education and Research, Albuquerque, New Mexico, March 1971. CEX 71.1.

Kind	Critical	Related Impact Velocity
of Missile	Organ or Event	ft/sec
Nonpenetrating	Cerebral Concussion:	
10-1b object	Mostly "safe"	10
J	Threshold	15
	Skull Fracture:	
	Mostly "safe	10
	Threshold	15
	Near 100 per cent	23
Penetrating		
10-gm glass	Skin Lacetration:	
fragments	Threshold	50
	Serious Wounds:	
	Threshold	100
	50 per cent	180
	Near 100 per cent	300

TABLE D3

TENTATIVE CRITERIA FOR INDIRECT BLAST EFFECTS INVOLVING SECONDARY MISSILES

The relationship between medical damage from secondary missiles and missile characteristics. Based on data from reference [D3],

P Injury Pr	eak Over- essure-PSI	Impact Velocity (meters/sec)	Range 1KT	in Kil 10KT	ometers 100 KT
(Penacrating Missiles-10 Gram Glass Fragment)					
Skin laceration threshold	1-2	15 m/sec	1.8	4	9
Serious wound threshold	2-3	30 m/sec	1.2	2.5	5.5
Serious wounds near 50% probability	4-5	55 m/sec	.75	1.8	4
Serious wounds near 100% probability	7-8	90 m/sec	.55	1.3	2.5

TABLE D4

INJURY CRITERIA FOR PENETRATING MISSILES

Relationship between secondary effect damage and missile characteristics for three different sizes of nuclear weapons. Based on data from reference [D2]. of missiles that an accidental explosion could cause. The impact velocity is most properly related to the dynamic overpressure impulse, rather than the peak overpressure. The dynamic overpressure impulse, J, is defined as the integral of pressure over time for the duration of the positive phase of a blast wave; i.e.:

$$J = \int_{t_a}^{t_o} P(\tau) d\tau$$

where.

- P(T) = the overpressure as a function of time at a given location
 - t_a = the arrival time of the blast wave at the given location
 - t_o = the time at which the positive phase of the blast wave ends.

Figure D-1 further illustrates this definition. The dynamic overpressure impulse is called simply "the impulse" elsewhere in this report ("dynamic overpressure impulse" is used by Fugelso [D1]; "impulse" and "impulse per unit area" are used by Kinney [D4]).

The relationship between missile impact velocity and the impulse of the blast wave is given by,

$$M V_{i} = C_{D} A J$$

where,

- M = mass of the missile
- V_{i} = impact velocity
- C_D = drag coefficient of the missile, taken to be unity here
- A = presented area of the missile
- J = impulse

Now we may also write

 $M = \rho \overline{V}$

(D-6)

(D-5)

(D-4)

where,

[[]D4] Kinney, Gilbert Ford. Explosive Shocks in Air, p. 11, The Macmillan Co., New York, 1962.



FIGURE D-1

Pressure-time graph of two typical blast waves at a given observation point. Blast wave "A" decays faster and shows a smaller impulse than blast wave "B" even though their durations and the peak overpressures are identical. ρ = density of the missile

 \overline{V} = volume of the missile

But

$$\overline{V} = \frac{4\pi}{3} r^3 \qquad (D-7)$$

where we have assumed a spherical fragment of radius, r, and

$$A = \pi r^2 \tag{D-8}$$

Thus we have from equation (D-5),

 $J = \frac{MV_1}{C_D A}$

or combining with equations (D-6), (D-7), and (D-8),

$$J = \frac{\rho(\frac{4\pi}{3}r^3) V_1}{1 \cdot \pi r^2} = \frac{4\rho r V_1}{3}$$
 (D-9)

Furthermore from (D-7) and (D-6)

$$r = \left[\frac{3M}{4\pi\rho}\right]^{1/3}$$
(D-10)

Now for M = 10 grams and $\rho = 2.65 \text{ g/cm}^3$, a typical density for glass, then,

r = 0.9658 cm.

hence from equation (D-9),

 $J = (3.4125 \text{ g/cm.}^2) V_{i}$ (D-11)

Substituting values for impact velocity, V_1 , from Table D-4 into equation (D-11) generates a corresponding set of values for impulse as given by Table D-5. The data given in Table D-5 are the criteria used to establish the probit equation E5:

$$Probit = -27.1 + 4.26 \log_{e} J$$
(1)-12)

where J is in $N-S/m^2$.

An assumption implicit in equation (D-5) of the preceding derivation is that no forces other than those of the blast wave are exerted on the missile. That is the forces involved when the missile breaks away from its pre-blast position, slides or rolls along the ground, or impacts intervening objects are all neglected. Furthermore, it is assumed that missiles are always available to be set in motion by the blast wave.

	dyne-e	N-q	THE DEVEL
Injury	cm ²	<u>m</u> 2	psi-msec
Skin laceration threshold	5120	512	74.2
Serious wound threshold	10240	1024	148.4
Serious wounds near 50% probability	18770	1877	272.1
Serious wounds near 100% probability	30710	3071	445.3

• • •

5

t

TABLE D5

DAMAGE CRITERIA FOR PENETRATING MISSILES IN TERMS OF CRITICAL IMPULSE

Relationship between penetrating missile damage and free field impulse level. Equation D-11 and the data given in Table D4 were used to obtain these values.

These values were obtained by assuming a missile of 10 grams mass.

Also all personnel (not inside buildings) in a region traversed by a blast wave of sufficient strength are assumed to suffer injury from missiles; the density of flying fragments and the target area presented by people are not factors affecting the probability of injury in this analysis. Evidently then the estimate of fragment injury given by the probit equation equation, E5 (D-12), is too large. Since there appears to be no facile method for correcting this assessment, the overestimate of damage given by equation (D-12) is currently used in the VM.

(b) Tertiary Blast Injury

Injury to personnel by tertiary effects is related to the maximum translation velocity that the body attains subject to the blast. Injury is caused when the body set in motion by the blast wave strikes an obstacle. The probability of lethality for body impact has been determined as a function of the impact velocity. Lethality criteria have also been determined for impact of the head against an obstacle. Table D-6 shows the impact velocity associated with several levels of lethality.

Again the problem is to relate the damage criteria in table D-6 based on impact velocity to some free field, blast wave parameter. One reference [D1] indicates that a method that may be used to relate impact velocity and free field impulse is the same as that used for missiles in the above. Similar to equation (D-5) we write,

 $J_{1} = \frac{MV_{1}}{C_{D}A}$ (D-13)

where,

- $J_1 = critical$ impulse for impact damage
- M = mass of the body
- V_1 = critical velocity for impact damage
- C_D = drag coefficient of the body, taken here to be unity
 - A = cross-sectional area of the body presented to the blast

For each impact velocity, V_1 , of interest a corresponding impulse, J_1 , may be calculated using equation (D-13). We may calculate J_1 assuming a typical grown male adult, using a weight of 155 pounds and an area of 8 square feet. Table D-7 (taken from [D1]) gives both the impact velocity, Vm, that causes a certain level of damage and the dynamic impulse corresponding to that velocity for the grown male adult described above. Note, however, that the critical levels for lethality from body impact given in Table D-7 are significantly different from those levels given in Table D-6. The disagreement is considerable and places some doubt on the assessment procedure. The levels given in

Condition Critical Organ or Event	Related Impact Velocity <u>ft/sec</u>
Skull Fracture	
Mostly "safe" (0%)	10
Threshold (10%)	13
50 per cent	18
Near 100 per cent	23
Total Body Impact	
Mostly "safe" (0%)	10
Lethality Threshold (10%)	21
Lethality 50 per cent	54
Lethality near 100 per cent	138

State State

2. . . ~ • •

.

TABLE D6

TENTATIVE CRITERIA FOR INDIRECT (TERTIARY) BLAST EFFECTS INVOLVING IMPACT

Relationship between impact velocity and damage to the head and body. Based on data in reference [D3].

Lethality	V m	Body J _c	v	Head Jc
Threshold	20	83.6	13	54.3
50%	26	108.6	18	75.2
99%	30	125.4	23	96.1

TABLE D7

LETHALITY DUE TO CRITICAL IMPACT VELOCITY AND CRITICAL IMPULSE (V in ft/sec and J in psi-msec)

Relationship between impact lethality and impact velocity or impulse. From reference [D1].

Table D-7 are preferred since the publication from which it comes is more recent and the levels given are more conservative. The data given in Table D-7 was used to generate a probit equation. It was assumed that for any given impact, there is a 20% chance of head impact and an 80% chance of body impact; the casualty rates were calculated on this basis assuming that the entire population outdoors is subject to impact for a given blast wave strength.

This assessment procedure, however, gave unrealistic results. Far too many people were killed for the explosive yields and distances considered. The critical impulse levels for a given lethality rate shown in Table D-7 were found to be an order of magnitude lower than expected from observed explosions. The cause for the overestimation of damage from a given impulse level is that certain mitigating factors have not been considered. Some of the factors reducing the likelihood of death are: (1) Friction of the body sliding and rolling along the ground reduces the peak velocity attained on exposure to the blast; (2) During the passage of the blast wave the body may tumble, thereby reducing the area presented to the wave and the momentum absorbed from it; (3) Not all the momentum absorbed by the body is converted to velocity of impact, since both rotational and translational acceleration of the body may occur; (4) Sublethal impact with the ground or obstacles may prevent the body from ever attaining a lethal impact velocity. A model for impact injury that gives consideration to these various factors has been developed for use in civil defense studies [D5]. A plane rigid body subject to a blast environment simulates the trajectory of a person and determines the impact velocities experienced. Lethality is then determined from the data given in Table D-0, since the impact velocity and part of body impacted are predicted by the model. Since at the present time the VM does not model individual damage receptors in a locally detailed environment, this detailed model of impact injury is inappropriate for use in the VM. However, repeated operation of the impact injury model for a given size nuclear weapon showed that mortality could be parameterized on the basis of peak overpressure and distance to a vertical obstacle (wall); the parameterized results for the surface burst of a one megaton nuclear weapon are given in Table D-8. The survivability fractions given in the table are related to probability of death by,

% dead = 100% [1-S]

(D-14)

where S is the survivability value given in Table D-8. Similar data for initially prone bodies was also available, but not used.

By use of the explosion scaling laws (cf. Kinney [D6]) the peak overpressures for a 1MT blast given in Table D-8 may be converted to

[[]D5] Longinow, A., G. Ojdrovich, L. Bertram, and A. Wiedermann. People Survivability in a Direct Effects Environment and Related Topics. IIT Research Institute, Chicago, May 1973.

[[]D6] Kinney, G.F. Engineering Elements of Explosions. Naval Weapons Cencer, November 1968, NWC TP-4654.

				SURV	'IV≜BIL	. ITY FO	R STAN	ning P	CSITIC	N5				
OVERPRESSUR	£				t	ISTANC	F. TO -	ALI. (F	T)					
	0	5	10	15	20	25	30	35	40	45	50	55	A0	NO WALL
2	1.00	.90	1,0°	1.00	1,00	1.00	1,00	1.00	1,00	1,00	1,00	1.05	1,00	1.00
3	1.00	.92	.96	.96	1,00	1.00	1.00	1.00	1,00	1.00	1,00	1.00	1.00	1.00
•	1.00	. 53	.92	. 91	1,00	1.05	t.on	2.00	1,00	1,00	1,09	1,00	1.00	1,00
5	1.02	، ۴2	, ċ t	. 57	,68	. 50	.25	.75	,75	,75	,15	. 75	,75	,75
6	.70	.67	.07	.23	, 36	.00	.50	,50	, 51	, 51	, >1	• >1	,51	,51
7	.50	,47	,53	•11	.20	.03	.25	.25	, 25	,25	,25	. 25	,25	.49
8	.31	.26	,37	• 0 G	, 05	.06	,03	.04	•00	• • • •	.00	.00	.00	, 47
9	.20	.20	,25	.00	.03	.03	.02	, 92	.00	• 20	• UC	.00	.00	,23
10	.10	.13	.14	.00	.01	.01	.00	.uc	,00	,00	, uo	.00	.00	100
11	•01	.:"	.10	. 20	.00	• 6 0	.00	• Su	•00	•00	.00	. 110	.00	, 30
12	.00	. 63	•04	- 00	.00	.00	.00	.00	.00	- 00	,00	. 00	.00	• 00
13	.00	•65	<u>ו</u> 04	.00	.00	•00	•0¢	.00	.00	.00	•0C	.50	.00	•00
14	.00	.01	.02	• 00	.00	.06	.00	.00	.00	. 00	, uo	.00	.00	•07
19	.00	.01	.03	.00	.00	.00	.00	,00	.00	.00	.00	.00	, OC	. 20
18	. 00	.00	.03	.00	,00	.on	.00	.00	.00	.00	.00	. 90	.00	.00
17	.::	.01	.01	•00	.00	. 20	.00	.00	.00	.00	.00	.00	.00	. un
18	.00	.01	.00	-00	.00	•01	.00	.01	.00	.00	.00	.04	.00	.07
19	.10	. ac	.00	. no	.00	•00	.00	.00	.00	•00	.00	ւսո	.00	.or
20	• 21	.03	.00	•00	.00	• 0:1	.00	. ၁0	٥٥.	•00	.00	. 69	.ບລ	•20

. . .

TABLE D8

Relationship between damage from impact and the parameters, overpressure, and distance to obstacle, for the surface burst of a one megaton nuclear weapon. Percent deaths = $100 \times (1 - survivability)$

242

impulse levels; thus deaths from impact may be related to impulse. Table D-9, derived from the data in Table D-8, gives the relationship between deaths from impact and impulse level for a wall distance of ten feet (3.3 meters). The spread in overpressure between the threshold of damage and the threshold of complete lethality is largest for a wall distance of ten feet; thus the dose-response relationship for this set of data is least discontinuous. Ten feet is a reasonable impact distance for urban areas. Furthermore the change in wall distance is not that significant in determining the relationship between impact injury and overpressure (hence, impulse). For all these reasons the data for an impact surface ten feet distant were used to generate Table D-9 and subsequently the probit equation, E2. Based on the data in Table D-9 the probit equation, E2, is:

 $Probit = -46.1 + 4.82 \log_{0} J$ (D-15)

where J is $\ln N-S/m^2$.

ŧ

This probit equation gives estimates of deaths from impact that are much more in accord with observed events than does the probit equation based on the less sophisticated model used first.

In addition to death, tertiary effects can cause nonlethal injuries -- mainly broken bones. Table D-10 gives a variety of nonlethal injuries (as well as lethal injuries) and the impact velocities at which such injuries occur. As previously indicated, the range data for various sizes of nuclear weapons are unimportant; the critical impact velocities are significant for the VM. It is interesting that the lethality levels from this source [D2] agree reasonably well with those given in Table D-7 for body impact.

Using equation (D-13) to relate impulse level to impact velocity, one can generate a new table relating injury to impulse level by using the values given in Table D-10. Again, the calculations are for a 155 pound adult male with an effective area of 8 square feet. These computations are compiled in Table D-11. Notice, however, that the impulse levels for injury is below the levels for death given in Table D-9. The reason for this, as before, is that the data in Table D-ll are for an ideal transfer of momentum from the blast wave to the body, while the data in Table D-9 are for the more realistic model of momentum transfer. Obviously the injury criteria given in Table D-11 are unacceptable. Unfortunately the study [D5] using the more realistic model for impact damage does not include a parametric study of nonlethal injury, as it did for death. In order to circumvent this problem, an assumption was made about the relationship between the ideal model and the more realistic model of injury from impact. The more realistic model gives levels of impact for a given percent lethality that are approximately 50 times larger than the levels given by the ideal model for the same percent lethality. It was assumed that the relationship ! tween the ideal and more realistic model would be the same regardless of whether lethal or nonlethal injury was considered. Therefore the impulse levels given in Table D-11 for ideal momentum transfer were multiplied by the same proportionality factor that held for lethal injury levels in order to give the data shown in Table D-12.

Probability of Death	Impulse Level
(%)	$(N-s/m^2)$
1	18,000
8	28,600
31	37,300
63	45,100
86	49,700
100	60,800

TABLE D9

RELATIONSHIP OF DEATH BY IMPACT TO IMPULSE

Relationship between impact injury and impulse level. Based on results for a surface burst of a one megaton nuclear weapon and a distance to impact of ten feet (3.3 meters).

	Pesk over- Impact velocity		Rang	e in kilome	tera
anjury	Pressure-PSI	(meters/sec)	1 KT	10 KT	100 KT
(PHYSICAL DISPLACEMENT OF MAN WITH IMPACT WITH HARD SURFACE).					
Mostly safe (whole body)	8-5	S m/sec	.75	1.8	4.5
Fractured feet and legs	4-6	4 m/sec 4.3 m/sec	.65 .6	1.6	4 3.6
Skull fracture near 50% probability	5-7 6-8	5.5 m/sec	.55	1.4	3.5 3.1
Skull fracture near 100% probability.	6-9	7 m/sec	.49	1.2	3
Lethality near 100% probability (whole-body)	8-11	9.1 m/sec	.45	1.1	2.8

* In 3 meters of travel.

. . . .

ie. .

TABLE D10

TENTATIVE CRITERIA FOR PRODUCTION OF INDIRECT BLAST INJURIES WITH RANGES FOR VARIOUS YIELD WEAPONS (FAST RISING, LONG DURATION OVERPRESSURE IN AIR)

Relationship between impact velocity and injury. Taken from reference [D2, p. 9].

Injury	Impul	se
	psi-msec	$\frac{N-s}{m^2}$
Mostly safe (whole body)	41.15	283.8
Fractured feet and legs	58.98	406.8
Lethality threshold (whole body)	82.30	567.6

TABLE D11

CRITERIA FOR NONLETHAL INJURIES FROM IMPACT

Relationship between nonlethal impact injury and impulse, calculated assuming a 155-pound male adult with an effective cross-sectional area of 8 square feet.

Impulse
(N-s/m ²)
13,000
20,000 28,000

はなながら、それないには、たいとうかいたか。 しんしかい アー・アード・アード

and the Res 1

•••••

TABL: D12

.....

RELATIONSHIP BETWEEN NONLETHAL INJURIES FROM IMPACT AND IMPULSE

Relationship between percent injury and impulse level based on data for ideal momentum transfer and an assumption of proportionality between the results on the ideal and more realistic impact model.

÷

The probit equation derived from the data in Table D-12 and used to make the damage assessment is given by:

Probit =
$$-39.1 + 4.45 \log_e J$$

ALLER AND ALLER AND A STREAM AND A

(D-16)

Share and a state of the second second second second second

11

where J is in $N-S/m^2$.

This concludes the discussion of explosion injury to the vulnerable resource, people.

Explosion Damage to Structures

To begin the discussion of explosion damage to structures, it is worthwhile to raise some important considerations concerning the nature of the interaction between the blast wave and structures. The following two paragraphs are largely based on a discussion given by Kinney ([D6], p. 25ff).

Damage to a structure from blast comes from motion of the structure as imparted by forces of the blast wave. In principle, an analytic solution for structural motion can be obtained from the equation of motion expressing the relation between structural mass, structural acceleration, and the unbalance between the driving force of the blast wave and the resistance of the structure. The driving force is a transient one, given as the product of structure cross-section area and a blast-wave overpressure. The resistance of the structure depends or. its mechanical features. However, for dynamic situations, this is seldom known precisely and indeed perhaps is not capable of being known. Furthermore, even if both the transient driving force of the blast wave and the dynamic resistance of the structure were known, the mathematical form of the equation of motion is not conducive to a simple solution, but rather calls for numerical or analogue methods. Hence, only in simpler situations is a precise solution for structural motion in response to blast to be obtained.

As an alternative to an exact solution for structural motion, various empirical estimates of the damage potential of blast have been used. The most common of these is based on the peak overpressure in the free-field blast wave. For example, it may be stated that a peak overpressure of such and such psi causes major structural damage. It should be recognized that such a statement even if correct, can at best be only a crude approximation. It ignores the fact that the damage potential of blast is a function of two individual items, the transient blast loading plus the dynamic response of the structure; two such aspects are always involved in assessment of damage potential.

Since the VM just treat a large number of disparate structures whose mechanical characteristics are probably unknown (possibly difficult or impossible to obtain), it seems prudent to use the less sophisticated approximation involving only a single parameter related to the blast wave. Although details of structural response could be considered eventually, especially for large or important buildings, for now the single level blast wave characterization given by reference [D1] is used. The damage levels for structures are given in Table D-13. These data are for frame structures subject to a blast from an explosion equivalent to 500 tons of TNT. To obtain the probit equation for glass breakage it was assumed that the 1% level for structual damage corresponds to the 90% level for glass breakage. The probit equations for damage to structures from explosion are thereby obtained. For overall structural damage the probit equation, S1, is given by:

Probit = $-23.8 + 2.92 \log_{e} Pp$

(D-17)

and for glass breakage alone the probit equation, S2, is given by:

Probit = $-18.1 + 2.79 \log_{e} Pp$

(D-18)

where Pp is in N/m².

This concludes the treatment of explosion damage to vulnerable resources.

FIRE DAMAGE

A COLORED IN

-

Alexandra and Party of

Damage to personnel and damage to property from both flash fire and pool burning are similar enough to be treated together. For both personnel and property, the damage criteria consist of both a thermal radiation factor and a time factor. To cause a certain level of damage, say second-degree burns to bare skin, a thermal radiation level must be exceeded and the radiation must persist for a specified time. In general, as the radiation level becomes higher, the time required to cause a certain level of damage becomes smaller.

Both the flash fire and pool burning models were designed so that the ouput from Phase I is both a radiation level and a duration (time). For the flash fire, since radiation level changes rapidly with time, an effective radiat on level and time are given. Essentially, an equivalent square time pulse of radiation is substituted for the actual time-varying radiation pulse.

For damage to structures, the basic concern is whether ignition occurs. Since surface treatment, geometrical position, and other factors play such an important role in the determination of ignition, the problem is simplified by considering only the ignition of wood. In a classic paper [D7] Lawson and Simms developed two empirical relations for the ignition of wood.

These investigators experimentally determined the intensity of radiation required to ignite wood spontaneously both with and without a pilot flame one-half inch from the surface of the material. The expression relating time for ignition, t, to the critical ignition intensity, I, is the following:

[[]D7] Lawson, D.I., and D.L. Simms. The ignition of wood by radiation. Brit. J. Appl. Phys. 3:288-292, 1952.

		Peak Overpressure	
Target	Damage Level	psi	(N/m^2)
Frame Structure			
	Threshold Glass Breakage (1%)	0.25	1,700
	Threshold Structure Damage (1%)	0.90	6,200
	50% Structural Damage	3.00	20,700
	Total Damage (99%)	5.00	34,500

and the second

TABLE D13

DAMAGE LEVELS FOR SELECTED TARGETS

Relationship between structural damage and peak overpressure. Based on data [D1] for frame structures exposed to a blast from an explosion of 500 tons TNT equivalent yield.

$$(I - I_n) t^{2/3} = 0.025 \times 10^6 (K\rho s + 68 \times 10^{-6})$$
 (D-19)

where

I = thermal radiation intensity $(cal/cm^2/sec)$

 $I_{\rm p}$ = critical intensity for pilot ignition (cal/cm²/sec)

K = thermal conductivity (cal/cm²/sec/°C)

 ρ = density (g/cm³)

s = specific heat of the material (ca1/g/°C)

t = time to ignition (sec)

The corresponding expression for spontaneous ignition was found to be

$$(I - I_n) t^{4/5} = 0.05 \times 10^6 (K \rho_B + 35 \times 10^{-6})$$
 (D-20)

where

 $I_s = critical$ intensity for spontaneous ignition (cal/cm²/sec) Values of I_p and I_s for various types of wood are given in reference [D7].

Pilot ignition implies the presence of an open flame near the irradiated wood, while spontaneous ignition implies the absence of a nearby flame. In the context of the VM, the flash fire provides a flame nearby the irradiated wood, while pool burning, as a general rule, does not. Thus equation (D-19) was chosen to provide a basis for an ignition criteria in the case of flash fire, while equation (D-20) was used for the case of pool burning. The product of constants Kps was chosen to be the average of those materials tested by Lawson and Simms; likewise the critical intensities, I_s and I_p , were taken to be the average for those materials tested.

 $I_{\rm p} = 0.32 \text{ cal/cm}^2/\text{sec} = 13,400 \text{ Joules/m}^2/\text{sec}$ (D-21)

 $I_{g} = 0.61 \text{ cal/cm}^2/\text{sec} = 25,400 \text{ Joules/m}^2/\text{sec}$ (D-22)

$$K\rho s = 5 \times 10^{-6} \text{ cal}^2/\text{cm}^5/(^{\circ}\text{C})^{-6}/\text{sec}$$
(D-23)

For these average values, equation (D-19) and (D-20) become respectively,

$$(I - 1.34 \times 10^4 \text{ J/m}^2/\text{s}) t^{2/3} = 8050 \text{ J/m}^2/\text{s}^{1/3}$$
 (D-24)

and

$$(I - 2.54 \times 10^4 \text{ J/m}^2/\text{s}) t^{4/5} = 6730 \text{ J/m}^2/\text{s}^{1/5}$$
 (D-25)

From these equations the following criteria for ignition of structures are obtained.

.

Por ignition from flash fire:

1. The vapor concentration at grid cell center must be between the limits of flammability for the spilled substance (this to assure the presence of a pilot flame).

2. The radiation intensity, I_r , must exceed the value

$$I_p = 1.34 \times 10^4 \frac{Joules}{m^2 - s}$$

3. The effective duration of the radiation, t_{eff} , must exceed the value given by:

$$t_{p} = \left[\frac{7.22 \times 10^{5} \frac{J}{m^{2} - s}}{(I_{r} - I_{p})}\right]^{3/2}$$
(D-26)

i.e.,

If $t_{eff} \ge t_p$ then there is ignition of structures

If $t_{eff} < t_p$ then there is <u>no</u> ignition of structures

For ignition from pool burning:

1. For every grid cell look up the radiation intensity, ${\rm I}_{\rm r},$ at the cell center.

2. The radiation intensity at the cell center must exceed the value,

$$I_{g} = 2.54 \times 10^{4} \frac{\text{Joules}}{\text{m}^{2} - \text{s}}$$

3. The duration of the pool burning, t_b , must exceed the time given by, 5/4

 $\mathbf{t}_{\mathbf{g}} = \left[\frac{6.10 \times 10}{(\mathbf{I}_{\mathbf{r}} - \mathbf{I}_{\mathbf{g}})} \right]$ (D-27)

4. If $t_b \ge t_s$ then there is ignition of structures

If $t_b < t_s$ then there is no ignition of structures where in both cases,

 I_r = radiation incensity at the cell center

If these ignition criteria are satisfied at the cell center (Gas' fire or pool burning), one-fourth of the structures in the cell one assumed to ignite. This assumption was made to account for the fact that some structures will be shielded from thermal radiation by others.

As with inanimate objects, burn damage to people depends on a combination of radiation level and duration. The relation between level and duration has been the subject of considerable attention from the nuclear weapon effects community. In Table D-14, taken from reference [D3], the radiation level required to cause certain damage is given. The thermal radiation is the thermal radiation intensity integrated over time and is parameterized, not by pulse duration, but by weapon yield. For our purposes, we require a thermal radiation intensity level and an effective duration.

A Minute Shift and success of Shift had a set

A standard reference [D8] provides a formula to calculate the effective duration of the thermal pulse of a nuclear weapon from a knowledge of the yield. By calculating the thermal pulse duration and dividing into the integrated radiation values displayed in Table D-14, the desired data are obtained. Table D-15 gives the critical radiation intensity levels required to cause various levels of damage, for various sizes of weapons. The effective time durations given at the bottom margin of the table are the calculated time durations of the thermal pulse for the weapon size indicated.

From the data in Table D-15, it may be deduced that the effects of thermal radiation are generally proportional to II^{43} , where t is the time and I is the radiation intensity. For the 20KT, 1MT, and 20MT bomb data, the critical levels of radiation intensity were converted to MKS units and the dosage, in the quantity II^{43} , was calculated. The results thus obtained for lethal levels are presented in Table D-16. From these data the probit equation, Fl, for death from burns in a flash fire is calculated:

Probit = $-14.9 + 2.56 \times \log_{P}(t1^{4/3} \times 10^{-4})$ (D-28)

where t is in seconds and I is in $Joules/m^2/sec$.

For nonlethal burns, we are interested only in the threshold. Averaging the three values given in Table D-15 for first-degree burns, the criterion obtained by a curve fit is:

 $tI^{1+15} = 550,000$

where I is in Joules/m²/sec and t is in seconds. A power law of 1.15 fits this data better than the 4/3 power found for the lethal levels. If the value $t_{eff}I^{1+15}$ exceeds 5.5 x 10^5 then first-degree burns are presumed to occur in those cells where the vapor concentration is within flammable limits at the cell center. Burn deaths from pool burning are assessed by probit equation (D-28) except that actual burning time and radiation level are used instead of effective values as in the case of flash fire.

[[]D8] Glasstone, S., ed. The effects of nuclear weapons, p. 357 ff., USAEC, April 1962.

Critical event Thermal radiation in cal/cm for indicated explosive yiel		m ² eld			
	20 kt	100 kt	<u>l Mt</u>	10 Mt	20 Mt
First degree burn	2.5		3		4
Second degree burn	4.5	5	6.5	9	10
Lightly clothed (summer)					
Few if any injuries	2.5		3		4
Significant injury threshold	4	4,5	6	8.5	9,5
Lethality					
Threshold	5	6.0	8	10.0	11
Near 50 per cent	9	11.0	14	18.0	20
Near 100 per cent	20	24	31	40	43
Burns due to hot debris		No biological criteria available,			
an å hot, dust-laden air.		but probably a serious problem for large-yield explosions.			

مشعبة فكفنا للامعان فالمالة فأذفت كنا

TABLE D14

TENTATIVE BIOMEDICAL CRITERIA FOR THERMAL RADIATION

Relationship between biological damage and thermal radiation. This data [D3] is for thermal pulses from nuclear weapons; thermal radiation the time integral of the radiation intensity.

• 	<u>20 kt</u>	100 KT	<u>1 MT</u>	10 MT	20 MT
lst degree burn	1.75		. 297		.0886
2nd degree burn	3.14	1.56	.643	.281	. 221
Slightly clothed (summer) few if any injuries	1.75		. 297		.0886
Significant injury threshold	2.80	1.405	. 594	.266	. 210
Lethality					
Threshold	3.50	1.875	.792	. 312	. 243
Near 50%	6.30	3.44	1.385	.563	.442
Near 100%	14.0	7.5	3.07	1.25	.952
Effective Time Duration(s)	1.43	3.20	10.1	32.0	45.2

TABLE D15

CRITICAL RADIATION INTENSITY LEVELS AND DURATIONS REQUIRED FOR CERTAIN TYPES OF INJURY

[Thermal radiation intensity in $\frac{cal}{cm^2-s}$]

Relationship between thermal injury and the radiation intensity and duration. Based on the data given in Table D14.

Portion	Duration	Radiation	Dosage	
(\mathbf{X}) (sec)		(cal/cm ² /sec)	(Joules/m ² sec)	t1 ^{4/3}
1	1.43	3.50	146,000	1099×10^{4}
1	10.1	.792	33,100	1073 × 10^{4}
1	45.2	.243	10,200	1000 × 10^{4}
50	1.43	6.30	263,600	$2417 \times 10^{4} \\ 2264 \times 10^{4} \\ 2210 \times 10^{4}$
50	10.1	1.385	57,950	
50	45.2	.442	18,500	
99	1.43	14.0	586,000	7008×10^4
99	10.1	3.07	128,000	6546 × 10^4
99	45.2	.952	39,800	6149 × 10 ⁴

TABLE D16

RELATIONSHIP OF DEATH FROM RADIATION BURNS TO RADIATION LEVEL AND DURATION

Based on data in Table D15. The squence of data is for 20KT, 1MT, and 20MT weapons. The dot ge his been calculated for the radiation intensity $12 + 6m \ln^2/sec$.

£

State of the second

APPENDIX E

INHALATION TOXICOLOGY OF CHLORINE AND AMMONIA

Emphasis has been placed on human inhalation toxicology of chlorine and ammonia since these are clearly the most significant toxic hazards of the five substances of immediate concern in the VM. A full treatment would consider also:

- Human inhalation toxicology of LNG, methanol, gasoline, and combustion products
- Human ingestion toxicology of all five materials
- Effects on other land fauna, and on flora

- Effects on aquatic biota
- Impairment of beneficial water uses in general.

Some data and comments on these are summarized at the end of this section.

General

The Vulnerability Model translates concentration/time histories or cumulative doses into percent response of an exposed population in each grid cell. These are then converted into estimated numbers of casualties by using population numbers in each cell. The basic response data are for unprotected healthy adults in the open. The question of high risk populations - very young, very old, sick - is dealt with in a later section. Reduction of hazard by physical protection, e.g., in closed buildings with recirculated air, is not a toxicological problem but a matter of applying a proportionality factor to calculated open-air exposures.

Levels of Response

Response to increasing exposure ranges from threshold odor and irritation to death. Sublethal effects must be considered. The most drastic of these may involve permanent or long-term impairment, but transient effects are likely to be more important. They may cause temporary incapacitation and even if there is no actual harm, harm may be imagined. The number of people exposed to "threshold" effects is likely to be many more than those experiencing dangerous exposure.

Examination of the toxicological effects of chlorine and ammonia suggests three basic levels for consideration: (1) odor, (2) respiratory and eye irritation, and (3) death. These correspond to three levels of public receptor response: awareness of abnormal environment, possibly eliciting complaint; actual harassment and temporary incapacitation, leading to complaint and perhaps some real harm; and death. The toxicological properties of chlorine and ammonia are such that long-term impairment is likely to be

significant only in aggravation of pre-existing conditions such as chronic respiratory disease. This does not mean that there will be <u>no</u> long-term effects, but that they will be minor relative to the total consequences: see the following section.

Long-Term Effects

And and any set of the set

מסויול על לכלעה אלפיניי איניילו. בריות 11, 12: בעופוניים אל מוד שונה ה' הפני העופויביו ללכור לבלידות

ì

.

ŀ

1

The question of permanent harm was considered, with particular reference to chlorine. Should we include this category of effect, between the categories of temporary hospitalization and death? A search of the literature relating particularly to World War I and to major accidental spills, and discussion with two eminently qualified toxicologists [E1, E2], substantiated the view that the relative importance of this category was low unough to justify ignoring it: i.e., the percentage of permanent casualties would be very small.

The most recent paper seen was Weill et al. [E3]. They say that their data for subjects seven years after an accidental exposure to chlorine "are consistent with the prevailing clinical view that significant permanent lung damage does not result from acute exposure to chlorine gas." Reports referenced in their paper support this finding: for example, "no evidence that chlorine intoxication produced residual pulmonary disease" in the 33 most severely affected victims of a major accident; a large survey of industrial exposures did not find "any evidence of permanent damage to the respiratory tract"; another study including war casualties found that "permanent pulmonary injury was rare." The common belief in extensive permanent disability is apparently based on World War I gas casualtics, but Vedder, in The Medical Aspects of Chemical Warfare [E4], presents the view that a strong bias is introduced by the incentive of permanent pensionable status, and says that "very few individuals have any pathological basis for symptoms more than a year after even the most severe gassing" and most recoveries are very much earlier.

There are a few reports of effects other than pulmonary, more than a year after exposure: these include cardiac irregularity and anxiety reactions, but the percentage vs. total casualties is low.

Our conclusion is that the incidence of permanent effects is certainly low and probably negligible in the context of the VM study; this is more evidently true for pulmonary effects than for other effects which have received less attention. This conclusion applies here only to chlorine and to ammonia, the effects of which arc essentially acute and short-term.

- [E1] Kramer, C. G. Personal communication, 1974.
- [E2] Wands, Ralph M. Personal communication, 1974.
- [E3] Weill, H., G.M. Schwarz, and M. Ziskind. Late evaluation of pulmonary function after acute exposure to chlorine gas. Am. Rev. Resp. Dis. 99:374-379, 1969.
- [E4] Vedder, E.B. The Medical Aspects of Chemical Warfare. Williams and Wilkins Co., Baltimore, 1925.

The Significance of Concentration and Dosage

In inhalation toxicology the following variables are important:

	Usual Units	Symbols
Concentration of toxic gas (a) Peak (b) Average	mg m ⁻³ or ppm	С
Time of exposure	min	t
Breathing rate ("respiratory minute volume")	liter min ⁻¹	
Retention of inhaled material	%	
Dosage	mg min m^{-3}	Ct
Dosage lethal for 50% of those exposed	п	LCt ₅₀

Note the special usage of the term "dosage" for the product of concentration and time. It is not to be confused with "dose", which is the amount actually retained and depends on the breathing rate and percent retention. Dosage is the same for all subjects exposed to the same cloud, no matter whether they breathe fast or slow, or put on gas masks.

In the VM development we are concerned with both concentration and dosage. For example, men exposed to a concentration of 150 ppm of ammonia lacrimate immediately and continue to do so until removed to clean air, but do not risk other effects. Exposed to 15 ppm of chlorine they similarly experience immediate harassment, but in this case there is also a cumulative effect and after an hour or so at this concentration it is likely that some will die: i.e., we are concerned with the dosage (concentration times time) as well as the level of concentration. At a higher concentration, deaths occur after shorter exposure. Haber's law, Ct = k, is widely applicable to toxic gases but unfortunately not to chlorine over a wide range of concentration. The reason is that chlorine attacks by two mechanisms: immediate respiratory spasm from intense irritation, and delayed (ca. 24 hours) effects from oxidative and other destructive effects on lung tissue: the first effect is a function of concentration and the second of dosage. Hence, as concentration increases the acute effect becomes more important and progressively dominates the other: that is to say, the 50% lethal dosage (LCt₅₀) decreases.

High Risk Populations

There is no doubt that dose response estimates based on healthy adults will underestimate the sensitivity of certain sectors of the population,

notably the very young, old and sick. Kramer [E5] has commented on this in relation to chlorine, and it is interesting to note that the one death among 100 treated casualties in the 1961 Louisiana chlorine spill was an ll-month old infant [E6]. (Older siblings who were similarly exposed did not die.) The problem in attempting to assign appropriate factors is that there are virtually no sound data. Such indications as we have, come from accidents like the one just mentioned, in which there are of course no measurements of the degree of exposure. And extrapolation from animal experiments, always a questionable matter, is even less trustworthy in modeling groups such as human infants or advanced emphysema patients.

Number of the second se

The problem was discussed with Kramer (see footnote [E1]) who confirmed the absence of data but said that in-plant experience was that subjects with well-established bronchitis/emphysema are "much more sensitive" to chlorine. He remarked that the 7-8 ppm region of chlorine concentration is a critical one in which exposure may be tolerated with lasting harm, especially in susceptible subjects: i.e., the <u>concentration</u> is not above the level of tolerance, but after some time the cumulative <u>dosage</u> (and the retained dose) may hospitalize the subjects with pulmonary edema, etc.

We suggest that high risk populations be defined as (a) infants, (b) those over 70 years, and (c) others with advanced pulmonary/cardiovascular disease (as defined by some acceptable medical criterion, not yet established). These groups should be aggregated. A proposed scaling, which we have applied to specific concentrations and exposure times for chlorine and ammonia, is:

	Deaths, %		
Level of Effect	Genera. Population	High Risk <u>Population</u>	
Severe harrassment with some risk	0	25	
Lethal	3	50	
Lethal	50	100	

We recognize that these are guesses and probably on the high side, but it seems better to be pessimistic about the effects than to ignore an augmenting factor which certainly exists and may be substantial.

Estimating Dose Response Relationships for the VM

Most of the published work with toxic vapors is directed towards establishing response levels rather than dose response regression equations: e.g., odor and irritation thresholds, maximum acceptable concentrations, and exposures that are "dangerous" or "lethal". Experiments are often reported this way: e.g., in an ammonia inhalation experiment, volunteers were "all

[[]E5] Kramer, C.G. Chlorine. J. Occup. Med. 9:193-196, 1967.

[[]E6] Joyner, R.E., and E.G. Durel. Accidental liquid chlorine spill in a rural community. J. Occup. Med. 4:152-154, 1962.

of the opinion that no person would remain in such an atmosphere..." There are limited data on volunteers exposed to controlled concentrations to determine odor and irritation thresholds, but none, of course, on langerous and lethal exposures. Available data for humans are therefore estimates, based on extrapolations and accidents, plus the few low-level laboratory data.

The need in the VM is for dose response models which can translate exposure into percent effects in the exposed population. Our approach to estimating data which are experimentally unavailable is as follows:

Dose response data for lethality of chlorine were taken from four experiments with mice [E7], two with rats [E8], and one with dogs [E9]. These are plotted on Figure E1. Dose response data for nose and throat irritation by ammonia in man are also plotted [E10]. It is the slopes that interest us, because we want to use these data to help us estimate percent response at various dosage levels from estimates at one particular level. and it will be seen that the chlorine dose-response lines have similar slopes and the ammonia slopes are not greatly different. (Dose units are not shown and the lines are arbitrarily placed with respect to the dose axis, because we are not interested in, for example, the actual dose which kills a mouse; it is the slope we are using.) Slopes were averaged and composite lines for chlorine lethalicy and ammonia irritation were drawn, passing for convenience through the intersect of 50% response and unit dose. The relative doses for 90% response, vs. unit dose and 50% response, are 1.54 for chlorine and 1.78 for ammonia. These figures differ by only 0.12 or 7% from their average of 1.66. The similarity of dose response for different subjects, concentrations and effects suggests that we may justifiably use the composite chlorine lethality line, for example, to estimate dosages for various lethality rates in man from a published estimate of the 50% lethal dosage.

This proves to be not inconsistent with the estimates of "dangerous" and "lethal" exposures to chlorine in the National Academy of Sciences Guide for Short-Term Exposure [Ell]:

> Dangerous exposure is given as 14 to 21 ppm, 1/2 to 1 hour. The corresponding dosage = $\frac{14 + 21}{2} \times \frac{0.5 + 1}{2} = 13$ ppm-hr

Lethal exposure is given as 34 to 51 ppm, 1 to 1-1/2 hour. The corresponding dosage = $\frac{34 + 51}{2} \times \frac{1 + 1.5}{2}$ = 53 ppm-hr

- [E7] Weedou, R. F., et al. Toxicity of ammonia, chlorine, hydrogen cyanide, hydrogen sulfide and sulfur dioxide. Contr. Boyce Thompson Inst. II, p. 365ff., 1940.
- [E8] Silver, S.D., and F.P. McGrath. Chlorine median lethal concentration for mice. DATR 351, Edgewood Arsenal, Md., May 9, 1942.
- [E9] Silver, S.D. et al. Chlorine median lethal concentration for mice. DATR 373, Edgewood Arsenal, Md., July 17, 1942.
- [E10] American Conference of Governmental Industrial Hygienists. Documentation of threshold limit values for substances in workroom air, 3rd ed. ACGIH, Cincinnati, Ohio, 1971.
- [E11] National Academy of Sciences National Research Council, Committee on Toxicology. Guides for short-term exposures of the public to air pollutants. VIII: Guide for chlorine. NAS-NRC, Washington, March 1973. 261


DGSVCE

· wed

If we assume that "dangerous" means 5/100 dead and "lethal" means 90/100 dead, we get the chlorine lethality line of Figure El (the line at the extreme left) which agrees very tolerably with the animal data.

CARACTERISTICS TO STRUCTURE REPORT

ユアリマ

ź

1

Notes on Toxicology

Some notes are collected here which are relevant to chlorine and ammonia in the context of the VM. Data are from the NAS-NRC reports (see footnote [E11] and) [E12].

Chlorine

TLV (Threshold Limit Value - this is the upper limit for regular and indefinitely continued occupational exposure established by the American Conference of Governmental Industrial Hygienists: ACGIH, 1971.)

1 ppm time-weighted average for 8-hr working day

- 3 ppm permissible excursion for 15 min.
- STPL (Short Term Public Exposure Limit mild odor, minimal irritation, no health hazard.)
 - 1 ppm for 10 min.

0.5 ppm for 30 min.

0.5 ppm for 60 min.

- PEL (Public Emergency Limit strong odor, some irritation; these are ceilings, not averages.)
 - 3 ppm for 10 min.
 - 2 ppm for 30 min.
 - 2 ppm for 60 min.
- EEL (Emergency Exposure Limit proposed by Zielhuis [E13], for chlorine in manufacturing areas, transport and storage.)
 - 7 ppm for 5 min.
 - 5 ppm for 15 min.
 - 4 ppm for 30 min.

1

- 3 ppm for 60 min.
- [E12] National Academy of Sciences National Research Council, Committee on Toxicology. Guides for short-term exposure of the public to air pollutants. IV: Guide for ammionia. NAS-NRC, Washington, November 1972.
- [E13] Zielhuis, R.L. Tentative emergency exposure limits for sulfur dioxide, sulfuric acid, chlorine, and phosgene. Ann. Occup. Hyg. 13:171-176, 1970.

Ammonia

TLV 25 ppm

Occupational ceiling (recommended) 50 ppm

- Odor Reported threshold 5 ppm to 50 ppm: lower value is probably correct for near 100% detection, upper limit for 100% in all circumstances.
- 100-500 ppm Irritant but tolerable. 100 ppm for 8-hour working day causes irritation but no adverse effect. This is the generally accepted threshold of tolerance and in one experiment all agreed 140-200 ppm was not willingly tolerable; however, in another experiment volunteers inhaled 500 ppm/30 min.
- 400-700 ppm Strongly irritant, but infrequent brief (1 hour) exposure has no serious effect.
- 2000-3000 ppm Convulsive coughing, very irritant, may be fatal; no permissible minimal time.

Over 2500 ppm (2500-6500) Dangerous in 1/2 hour.

5000-10,000 ppm Rapidly fatal.

Long industrial experience indicates no permanent injury from lifetime working exposure below the level of intolerable acute effects.

STPL

20	₽₽¤	for	10	min.	(Note however that odor is objectionable
10	ppm	for	30	min.	to many, who may complain at any exposure, however brief, above their threshold of
10	ppm	for	60	min.	olfactory perception.)

PEL

100 ppm for 10 min. 75 ppm for 30 min. 50 ppm for 60 min.

Interpretation of Data and Selection of Levels of Exposure/Effect

There are two problems:

• Estimates of hazardous/lethal exposures for man are necessarily guesstimates, based on animal experiments, non-hazardous exposures on man, and non-quantitative accidents.

• With these highly irritant gases the immediate effect of <u>concen-</u> <u>tration</u> contributes largely and may dominate the cumulative effect of <u>dosage</u>. Hence one cannot, for example, use a constant value the 50% lethal dosage outside a limited time (or concentration) range.

The following dosage-response estimates were used.

Chlorine

Concentration	Time	Effect		
3 ppm	Any	No risk, but public complaints with some harassment (NAS Public Exposure Limit for 10 min. is 3 ppm)		
7 ррт	l hour or more	Strong to intolerable irritation, with some risk to highly susceptible sut- jects only		
20 ppm	S everal hours	50% lethal (our estimate, based on the following 1 hour figure)		
33 ppm	ca. 1 hour	50% lethal (based on the figures in the NAS-NRC report referred to in footnote [Ell])		
60 ррт	ca. 10 minutes	50% lethal (our estimate)		

Amnonia

Concentration	Time	Effect
20 ppm	Any	Odor detected by majority of population
100 ppm	Any	Irritation and complaint
500 ppm	Any	Strong to intolerable irritation, with risk to high susceptible individuals
2500 ppm to 5000 ppm		Fatalities mostly from a few in the first 5 minutes to 90% to 100% after 1 hour, depending on concentration
Above 5000 ppm		100% fatal. Heavy casualties in 5-10 minutes, and shorter exposures are unlikely in practice (Neither short time of cloud passage nor effort to escape is likely to reduce casualties much below 100% at these concentrations)

Dose response figures for irritation in man are available but seem of doubtful value: (1) the levels for odor/irritation are for majority of exposed population and there is not much benefit in estimating 25%, 50% etc. intolerably irritated; (2) the estimated effect at 2500-5000 ppm in fact corresponds roughly with slope estimates; (3) the heavy casualty level is not subject to reduction, for reasons given above in the table.

A Note on Medical Treatment

It has been suggested that a cruplete VM could take account of medical treatment. This does not appear necessary for the particular cases of inhalation effects of chlorine and ammonia, though it certainly will be in other areas. The reason is that the consequences of exposure are largely determined during the exposure and not too much can be done except for shock, coma, and respiratory arrest. For other patients, rest (perhaps with sedation) and oxygen are helpful. For chlorine especially, intermittent positive pressure oxygen has been recommended; also, cough suppressants and bronchodilators. But it seems, on the whole, that if a patient is going to die or be hospitalized, his fate is pretty well settled at the termination of exposure. Medical treatment will do some good and medical ethics demand utmost effort -- but in terms of VM casualty estimates, the difference is likely to be insignificant.

Note incidentally that references here to lethality are not limited to death during the specified exposure but include later death, usually quite early (though there may be a small proportion after several weeks).

Notes on Other Topics

Human Toxicity by Inhalation of LNG, Methanol, and Gasoline

The American Conference of Governmental Industrial Hygienists (see footnote [E10]) classes methane and related aliphatic hydrocarbons among the "simple asphyxiant, non-toxic" gases. (Pentane is given a TLV of 500 ppm, being slightly narcotic; however, the fatal concentration is estimated as 130,000 ppm.) We conclude that LNG is non-toxic in the context of the VM. However, consideration should be given to the risk of asphyxiation in high concentrations of LNG vapor.

Methanol is given a TLV of 200 ppm by the ACGIH (see footnote [E10]), indicating a low level of toxic risk. It has been left aside for immediate purposes but should be reconsidered to determine the risk that high concentrations could result in (a) a toxic dose, based on known figures for toxicity by ingestion, or (b) in asphyxia.

Gasoline presents a problem, because its toxic properties depend mainly on the aromatic hydrocarbon content, which is widely variable. Since the toxicity is not high, and no sound way is apparent for estimating a representative value, it has been set aside for the present.

Combustion products of the above substances and of ammonia: it appears likely that a combustion source generating a large amount of a toxic product such as carbon monoxide would at the same time generate so much heat - and hence convective turbulence - as to dilute the product below hazard level outside the fire hazard zone. This intuitive conclusion might be reexamined in any future work.

Human Toxicity (Ingestion)

المالك فالمسلكة

Ļ

È

i

ì

÷

f

Some toxicity figures are available, but we have not considered this problem because it is unlikely in the VM context that a large number of people would be exposed to this danger. The circumstances in which a large number of persons would swallow the material, or water contaminated by it, are not easily imagined. Nevertheless some group of people, possibly uninvolved bystanders, might be exposed to a significant danger. Therefore further study is recommended to model the potential hazard of polluted waters, so that controls on water usage, both commercial and recreational, may be scientifically formulated.

Effects on Other Land Fauna

For large animals the lethality figures for men should be good enough if estimates of economic damage are to be developed later.

Effects on Land Flora

The level of "irritation" in man appears to approximate to the level of visible damage in more sensitive plants, with some risk of economic damage.

Effects on Aquatic Biota

The range is so wide, and the biota at risk in specific places so varied, that "damage" level is not treated at present in the VM.

Impairment of Beneficial Water Uses

These seem to fall outside the scope of the model. For a specific site the VM could be used to generate an "exposure history" for specific municipal/industrial intakes, from which a "damage" could be estimated: this might often be a simple economic estimate of the cost of shutting down while acceptable intake limits were exceeded.

APPENDIX E. INHALATION TOXICOLOGY OF CHLORINE AND AMMONIA

List of References

[E1] Kramer, C. G. Personal communication, 1974.

محد الانبع حمالتك

A SAME AND A SAME AND A STRATE

- [E2] Wands, Ralph M. Personal communication, 1974.
- [E3] Weill, H., G. M. Schwarz, and M. Ziskind, Late evaluation of pulmonary function after acute exposure to chloring gas, Am. Rev. Resp. Dis. 99:374-379, 1969.

ŝ

i

ł

ä

ł

Local Second

- [E4] Vedder, E. G. The Medical Aspects of Chemical Warfare. Williams and Wilkins Co., Baltimore, 1925.
- [E5] Kramer, C. G. Chlorine. J. Occup. Med. 9:193-196, 1967.
- [E6] Joyner, R. E., and E. G. Durel. Accidental liquid chlorine spill in a rural community. J. Occup. Med. 4:152-154, 1962.
- [E7] Weedon, R. F., et al. Toxicity of ammonia. chlorine, hydrogen cyanide, hydrogen sulfide and sulfur dioxiue. Contr. Boyce Thompson Inst. II, p. 365ff., 1940.
- [E8] Silver, S. D., and F. P. McGrath, Chlorine median lethal concentration for mice. DATR 351, Edgewood Arsenal, Md., May 9, 1942.
- [E9] Silver, S. D., et al. Chlorine median lethal concentration for mice. DATR 373, Edgewood Arsenal, Md., July 17, 1942.
- [E10] American Conference of Governmental Industrial Hygienists Documentation of threshold limit values for substances in workroom air, 3rd ed. ACGIH, Cincinnati, Ohio, 1971.
- [E11] National Academy of Sciences-National Research Council, Committee on Toxicology. Guides for short-term exposures of the public to air pollutants. VIII: Guide for chlorine. NAS-NRC, Washington, March 1973.
- [E12] National Academy of Sciences-National Research Council, Committee on Toxicology. Guides for short-term exposure of the public to air pollutants. IV: Guide for ammonia. NAS-NRC, Washington, November 1972.
- [E13] Zielhuis, R. L. Tentative emergency exposure limits for sulfur dioxide, sulfuric acid, chlorine, and phosgene. Ann. Occup. Hyg. 13:171-176, 1970.

APPENDIX F

CASE STUDIES

This Appendix presents fourteen case studies of actual vapor cloud fires and explosions, to provide back-up detail for the generalization highlighted in Chapter 9. The cases illustrate three different kinds of events:

- Explosion of an open-air cloud.
- Explosive rupture of a pressurized tank, followed by a fireball.
- Fireballs without blasts.

Each kind of event poses its own special problems and hazards.

The fourteen cases are drawn from a larger pool of known and studied accidents, but they include all those which were reported in sufficient detail to be instructive. The cases which were excluded would add little to the picture.

Since this study was directed toward unpublished case studies, a large number of the sources are perforce private communications; but references to published reports are also given wherever possible. Some of the unpublished cases, notably Cases F.1 and F.3, will enter the literature when investigations still going on at this writing are completed.

F.1 CYCLOHEXANE CLOUE EXPLOSION FLIXBOROUGH, ENGLAND - 1 June 1974

ş

the second

Ŷ

F

i,

On 1 June 1974, the Nypro caprolactam plant near Flixborough, England, suffered a rupture of a 28-inch diameter pipe carrying cyclohexane at a temperature of 145°C and a pressure of 120 psi. The hot liquid poured out through both 28-inch diameter ends and flashed into a mixture of vapor and mist. 50 to 60 tons of cyclohexane were lost, and it is estimated that about 1/4 of that (15 tons) formed a white, vapor-mist cloud about 50 feet thick and many hundreds of feet in diameter. Another estimate indicated 30 tons in a cloud of about 100 meters in radius and occupying a volume of at least a million cubic feet.

The wind was about 18 knots from the southwest, and it distorted the cloud into the approximate shape shown in the following figures. The cloud boundaries shown were deduced from observed carbonization, melting and soot formation, and from the results of jet flow calculations.

Ignition occured after a delay of approximately 54 seconds, probably at the reformer furnace of the hydrogen plant; and a massive explosion resulted. Most of the explosion appeared to have been a deflagration, but it is felt that a kernel of gas about 7 meters in radius and containing about 1.4 tons of cyclohexane detonated with the force of about 15 tons of TNT. Both the fact of detonation and the size of the kernel are deduced from the degree of damage observed.

The blast was devastating. Processing units, main pipe racks, tanks and operating buildings in the cyclohexane caprolactam area were totally destroyed. Towers were blown down, twisted, crushed or bent over into piles of rubbish. Pipe racks were blown off their foundations for their full lengths and toppled over. Buildings, including two-story, reinforced concrete buildings, were crushed into piles of rubble. Exposed steel supports of the pipe racks were shifted five to ten feet off their foundations. Large vessels, towers, etc., appeared to be crushed from the top "as though some giant hand

had smashed them". There were also raging fires from liquid spillage on the northeast side of the area, but most of the damage appears to have been done by the blasts. Damage estimates exceed \$100,000,000.

Twenty-nine people were killed, most of them in the plant control room which was completely demolished. Reports available at this writing do not specify what actually killed them, but it appears that either the blast or the fireball would have been sufficient. One early, private report said that there was no eardrum damage, which would imply overpressures of less than about 6 to 12 psi and suggest fire deaths; but such a low overpressure seems inconsistent with the observed structural damage in the area, and the report needs confirmation.

Damage to neighboring structures was much less severe, and less severe than early newspaper accounts indicated. An on-site observer reported that there was damage to tile roofs, and that windows were broken in the village of Flixborough, about 1/2 to 3/4 mile to the west aton a hill but out of the line of sight from the plant; but that "the damage was not what is normally called 'severe'". Five or six dwellings in a row about half way between Flixborough and the plant had heavy damage to windows, doors and tile roofs; but their walls were all intact. Window breakage was reported up to eight miles from the plant.

The overpressure in the center of the blast has been estimated at approximately 25 - 30 psi by scaling from the observed damage patterns. The overpressure gradient was also estimated from the same observations, and the corresponding circular isobars are shown on one of the following figures. The same damage patterns were used to locate the apparent epicenter of the blast, to conclude that part of it was a detonation, and for one estimate of the cloud size and shape.

This accident is still under intensive investigation at this writing (December 1974); and there will ultimately be a very detailed report. There even appears to be a movie of the initial pipe break, cloud formation and explosion - made by an amateur who just happened to be filming the scene. Pending the complete, documented report, all conclusions should be tentative.

j.

à

Sources:

.

and the second second

Ę

- 1. Chemical & Engineering News, 10 June 74, page 4.
- 2. Testimony of Professor Sir Fredrick Warner, of Cremer and Warner, Consulting Engineers, at the court inquiry into the accident.
- 3. Several private communications and confidential memos which were shared in the interests of safety but "not for attribution because still preliminary".





A CYCLOHEXANE CLOUD WHICH DID NOT IGNITE

The Flixborough disaster is illuminated somewhat by a closely similar spill on 11 Sep 1971 at a caprolactam plant in Pensacola, Florida, which however did not ignite.

As at Flixborough, a pipe rupture released a massive amount of cyclohexane under pressure and above its boiling point in the midst of a large industrial complex. In this case, 74,000 pounds of cyclohexane were released; and "a sizeable amount" is thought to have flashed into vapor, with the rest mostly ending up as mist. It formed a dense, white cloud estimated at over 100 feet high by reference to taller equipment and approximately 2000 feet across at its maximum. The wind was "about at a dead calm" at the time of the rupture, and it "picked up to a very slight breeze from the southwest" during the event. The attached map shows a sketch of the size and shape of the cloud.

The cloud was rich enough that two trucks stalled in it from lack of oxygen. Some of it was drawn into a power house furnace, causing the stack to emit thick, black smoke. Amazingly, the cloud never did ignite; and eventually it dissipated harmlessly.

The attached sketch map shows the growth of the cloud. The area was a typical, densely built, chemical complex, with both open and closed operating areas, pipe racks, distillation towers, storage tanks, and the like very much like the complex at Flixborough, and with a gridwork of reads lacing it in both directions. Only a few of the buildings are shown on the map to give an idea of scale; there was essentially no unoccupied space in the area.

Source: Private communication from Walter B. Howard, Monsanto Company.



z

.....

ť

i I

F.2 ETHYLENE CLOUD EXPLOSION LONGVIEW, TEXAS - 25 February 1971

On 25 February 1971, a 1/2-inch pipe carrying high pressure ethylene gas broke in an industrial complex near Longview, Texas, releasing an estimated 1000 pounds of ethylene. The resulting cloud found an ignition source and exploded, breaking numerous other pipes and releasing many thousands of pounds of fresh ethylene which then formed a much larger cloud. The larger cloud apparently sufficiented a gas-driven compressor and was itself ignited. The second explosion "shook the earth". It killed four people in the immediate area and hospitalized 60 others. Fires, mostly liquid fed, raged for two and a half hours; and the ultimate damage reached \$14,000,000.

Damage was mostly confined to the industrial complex itself, and the majority of it was within 1800 feet of the blast center. A few windows were broken in the city of Longview, five miles away; but the pattern of window breakage was very erratic - a house 1-1/2 miles away was untruched. The documented damage is indicated on the following sketch map and explained in the accompanying detail notes.



DAMAGE NOTES:

- Blast site Completely demolished, but personnel were found alive behind control panels and in other protected places. Overpressure estimated at 15 psi, based on observed damage to equipment.
- A Warehouse 600 feet east. Building was 150' wide by 500' long. The sheet metal roof was buckled inward but not fallen. The 14-guage sheet metal walls were buckled inward but not detached.
- B Processing building 600 feet west. Windows blown in. Brick wall standing but ruined and had to be pulled down. Sheet metal roof buckled inward but not fallen.
- C Brick administration building 1800 feet west. About half of the windows on the blast side were broken, none on the far side. There was no brick damage and no roof damage.
- D Shop building 1200 feet southwest. Wire glass windows broken but not torn from their frames. Brick wall undamaged.
- E Open structure alcohol plant 300 feet southeast. Open part undamaged; sheet metal wall stripped from adjacent weigh building.
- F Cooling tower 150 feet southeast. Transite siding ripped off.
- G Open sided compressor shed 600 feet southeast. Transite roof stripped off.
- H Brick building 400 feet southeast. Wall OK; transite roof gone.
- I Processing building 100 feet east. Brick wall smashed and all glass gone.
 Steel frame bent but standing.
- J Warehouse 300 feet east northeast. Sheet metal walls largely torn away. Sheet metal roof buckled and parts torn away. Steel frame largely intact.
- K Brick building 100 feet west. Roof gone. Brick wall badly broken but mostly standing.

Source: Private communication from Mr. William Lauderback, of Texas Eastman Company.

F.3 PROPANE CLOUD EXPLOSION DECATUR, ILLINOIS - 19 July 1974

÷

÷

At 5:03 AM on 19 July 1974, a railroad tank car full of propane suffered a puncture in the yards of the Norfolk and Western Railroad in Decatur, Illinois. The escaping propane formed a greyish white ground fog reaching up to a height of 10 to 12 feet above the nearby boxcars and spreading toward the west, south and north. There was "very little" spread toward the west. The wind was generally from the north (020° at 10 knots); but local circulation patterns were more important. The temperature was 77°F, and the dewpoint was 68°F.

The cloud found an ignition source somewhere and exploded. A distant observer reported a fireball "100 to 150 yards in height and 200 to 300 yards in length" when he looked up at the sound of the blast. There were several more explosions during the next ten minutes, apparently as secondary pockets of vapor found ignition sources - which were plentiful by then.

There were seven fatalities, most of them very near the center of the blast; and there were 152 injuries, most of them minor glass cuts and burns.

General damage profiles are sketched on the accompanying map. Zone "A" is the area of "severe structural damage"; "B" is "scattered structural and severe glass damage"; "C" is "scattered light damage"; and "D" is an anomalous area of "light structural and heavy glass damage" well outside the general area of such damage, apparently the result of atmospheric refraction and focusing of the blast waves, a not uncommon feature of blast damage maps. 700 residences had damage, most of it minor; but 67 residences were posted as unsafe. They were concentrated in the northwestern part of Zone "A", with a few of them in Zone "B" just to the west of Zone "A" and a few of them in the lower end of Zone "A". A school in the southeastern quadrant of Zone "A" near the edge had extensive window, ceiling and partition damage; and an unfinished gymnasium addition collapsed to the ground.

Damage was characteristically erratic and spotty. Even in the central blast area, intact railroad cars were left sitting beside completely destroyed cars. Cars were sometimes split open as if there had been an internal explosion, with the shattered pieces of wood inside bearing no scorch marks. The Norfolk and Western Annex, scene of several severe injuries, was devastated in some places and practically untouched in others, despite the fact that it was near the center of a large fire.

Nor was the damage anything like total even near the center of the blast. Aerial photos show intact freight cars all over the yard, as well as several areas of destroyed cars. The industrial buildings adjacent to the railroad yar! look intact in the photos, although they doubtless had extensive window and perhaps partition damage. Trees, shrubs and automobiles all appear intact; and indeed a newspaper account noted that a row of trees alongside the railroad right-of-way all retained their leaves - excent for one 25-foot swath. The photos are to some extent deceiving - after all there were 67 houses posted unsafe - but the point remains that non-detonative gas cloud blast damage is by no means similar to, say, tornado damage. No houses were leveled, and the aerial photos give little hint of any damage at all except in a few cases (for example the school) and under high manification. People are indeed killed, and property damage is indeed widespread; but whole sections of cities are not wiped out, and repairs are made relatively quickly.

This explosion is still under active investigation at this writing (December 1974). It promises to be one of the most thoroughly investigated and documented events in the record, and it will reward additional browsing especially in the NTSB files - after the investigations are completed.

Sources:

^{1.} Private communications from personnel at NTSB.

^{2.} Newspaper clippings reviewed at Calspan Corporation



.

.

.

BLAST DAMAGE MAP - DECATUR EXPLOSION

- A Innermost zone Severe structural damage.
- B = Second zone Scattered structural damage (cosmetic) and severe glass damage.
- C Third zone Scattered light damage.

- D Hatched zone Light structural damage and heavy glass damage.
 - Dots Residences so damaged as to be "unsafe".

F.4 VINYL CHLORIDE MONOMER CLOUD EXPLOSION CLIMAX, TEXAS - 29 June 1974

At 7:45 PM on 29 June 1974, a freight train derailed at Climax, Texas, about ten miles from the city of Nacogdoches. A 32,137 gallon tank car of vinyl chloride monomer was punctured in the derailment, cearing a hole about 4x4 feet in one end. The escaping vinyl chloride vanor traveled about 1600 feet with a slight southwest breeze to the derailed locomotive units where the vapor was ignited.

The resulting air blast appears to have been a true detonation, one of the few that have been observed in unconfined gas clouds. It wrecked the locomotive, destroyed several cars, pulled some trees from the ground and completely stripmed the foliage from all vegetation in an area about 400 yards east and west of the blast area. Trees on the east side of the track were pulled *totant* the blast center, while trees on the copesite side of the track theorized that the detonation was initiated by a strong shock wave from the exploding Diesel engine, whose cylinder head was blown off by an internal explosion thought to have due to ingesting vinvi chloride monemer through the air intake.

Windows were broken in Bacegadarnes, alast o to 7 files among but otherwise there was remarkably little dama e to other than railroad property. There was no damage to numerous homes northeast of the site on the back side of a hill, and there was only sporty damage to the south. A home about threequarters of a mile away on the crest of a hill and in sight of the wreck scene received only light damage, and a home directly behind it received none. In contrast however, a house 1/4 file forther away and how the slope received heavy damage. A man about a mile to a file and a holf gray from the blast reported feeling the concussion, but there was no damage there.

He recallment and blast site was in a million rout a set is test wide at the litter, about 300 feet how east to rest. When the set little of the north ofde and about 18 feet black on the south side. For the all's rise further on hith sides.

Preceding page blank

after the air blast, the punctured tank car continued to burn torch-fashion, with the flame impinging on a so-far undamaged car of vinyl chloride in front of it. About 20 minutes later, the second vinyl chloride car exploded from internal overpressure. When this car exploded, the center section of the shell, from about 9 reet from one end of the tank to about 30 inches beyond the manuary cover place, was flattened and left diagonally across the remainder of the truck structure. One end of the tank, approximately 12 feet in length, was found about 1/2 mile from the scene of the derailment and explosion. From damaged trees in the path of the tank head, it appeared that the head was rocketed to an altitude of over 60 feet, clearing trees of that height. The trajectory of the head is uncertain, but it appears that on descent it traveled at no more than about 40 feet above the ground for about 300 yards. It landed (irst about 250 feet from the final resting place. In the initial landing, the head dug a hole about 4 feet deep and 12 feet in diameter. The head bounced a total of four times before enting to rest in a grove of small trees. The other end of the car never was round.

Two 0,000 gallen tank dars of Mitor Fuel Anti-Fuel Corporativere also ruptured by flare implacement and destroyed, but there is no report of rocheting parts from them. Two 20,000 gallon tank cars containing hegarethylere diamine solution were also "reptured in the derillrent with loss of all contents", but the report wes not mention fire impingement or explosion.

The initial blast did indite other fadings on the train, including raw rubber ingets which were blocm out of cars near the explosion and through burning as much as a mile, starting solt tires where they fell, decover, most of the thermal damage to vegetation was contined to about 161 to 156 marks to both sides of the derailment site. There was virtually no fire or blast damage beyond about 506 yards army.

Barre were no injuries reported as a result of the accident.

Sur eq. Private communication DEDDark Function, Repert Brathold Schedul Frances.

F.5 PROPYLENE CLOUD EXPLOSION EAST_ST. LOUIS, ILLINOIS - 22 January 1972.

At 6:20 AM on 22 January 1972, a large cloud of vaporized LPG (93-94% propylene and 5-6% propane) from a nunctured tank car exploded in the Alton and Southern Railroad Company's Gateway Yard in East St. Louis, Illinois, injuring more than 230 persons and doing more than \$7.5 million dollars worth of damage. There were no deaths, but 19 persons were hospitalized. The injuries were mostly from lacerations.

The spill occured when an LPG tank car containing 28,289 gallons of LPG was struck and punctu. I by the coupler of another car while rolling free in the railroad yard. The punctured car, with four other cars, rolled approximately 1300 feet at a speed of approximately 15 MPH, spewing liquid LPG as it went, until it finally came to rest as shown on the following map. The LPG vaporized into a visible cloud which traveled to the position shown and covered an area of about 5 acres when it was ignited by a source in an unoccupied caboose.

Flames first spread out horizontally, then an orange flame spread upward, and then a large fireball flared upward with explosive force. Almost immediately thereafter, a second, more severe, explosion occured. The force of the explosion damaged buildings and a number of freight cars in the area of the caboose. Some of the cars caught fire. Heat from the fires damaged some of the rails, and the force of the explosions wrapped size of the switch targets around switch scands. The office building of the yard and its roof tower were damaged and some manhole covers in the drainage system vere dislodged. Approximately 870 to 1000 homes and buildings were damaged out in the city. The accompanying damage map shows the pattern of the damage in the surrounding community, ranging from broken glass in Area t to serious structural damage in Areas 1 and 5. Prof. Roger Strehlow has analyzed the apparent yield of the explosion (from the given distance/damage data) and concluded that the explosive energy release was equivalent to thre from about 2000 to 5000 pounds of TNT. This is about 0.1 to 0.3 percent of the potential energy available if all the propylene had mixed stoichiometrically with air and detonated. Prof. Strehlow has also concluded that one or more gas detonations probably occured inside empty boxcars where the confinement was apparently sufficient to permit a deflagration-to-detonation transition.

Sources:

- 1. Private communication from Prof. Strehlow.
- 2. NTSB Railroad Accident Report NTSB-RAR-73-1.
- 3. Roger A. Strehlow, "Equivalent Yield of the Explosion in the Alton and Southern Gateway Yard, East St. Louis, Illincis, Junuary 22, 1972.", University of Illinois Report AAE TR 73-3, June 1973.



Plan of Alton & Southern Gateway Yard.



Map of Explosion Damage

F.6 PROPANE CLOUD DETONATION FRANKLIN COUNTY, MISSOURI - 9 December 1970.

At 10:20 PM on 9 December 1970, an 8-inch pipeline carrying liquid propane under pressure ruptured and spilled propane in a rural area of Franklin County, Missouri, about seven miles south of the town of New Haven. Propane, escaping from a 6-food split in the underside of the pipe, dug a 4-foot-deep, 10-foot-diameter drater and geysered 50 to 60 feet high, dreating a heavy white fog which flowed downhill and slowly filled the entire shallow valley nearby. A breeze of 5 knots was blowing from the northeast, the temperature was 34°F, and there was a well-defined temperature inversion with a very stable air condition on the ground. Approximately 756 barrels of propane (31,752 gallons) of propane was released before indition occured, for a calculated vapor volume of 1,143,072 cubic feet. This vapor, mixed with air to form a flammable mixture, covered an estimated 400,000 square feet of ground.

At 10:44 PM, 24 minutes after the rupture, the propane-air mixture detonated, generating a double boom. To the eye, the cloud "ignited simultaneously at all points in the valley!" A huge ball of fire immediately followed the blast, mushroomed upward and enveloped the entire area, colling from east to west across the valley bottom. The thermal column of fire was hundreds of feet high. The blast, the noise, and the pillar of fire passed over the area; and firebrands, broken boards, branches, rubble and debris fell on the countryside.

At the bottom of the valley, trees were snapped off like matchsticks, burned, blackened and strewn over the ground. In the open pastureland, many small evergreens were uprooted and large expanses of grass had been burned black. The uprooted trees and broken branches almost uniformly pointed toward the center of the blast. A ranch-type stone farmouse was incinerated, and debris of all varieties was scattered over the landscape. A black, charred stream bank traced one path of the fire for more than 1,000 feet to the southeast. A concrete block, two-ster: warehouse, located near the ranch-type stone house was obliterated; this area was the origin of the blast. Another farmhouse,

one-half mile southeast of the detonation origin, was destroyed; the windows were smashed, doors were broken and debris littered the interior.

Higher up on the rim of the valley, several houses along the highway also took the full brunt of the explosion. Glass-riddled walls, broken exposed roof beams and blown-in windows were in evidence. Sections of exterior brick and stone walls were torn from their sidings, and other adjacent frame houses were partially blown down.

Within a two-mile radius of the blast, major structural damage occurred to 14 houses; minor structural damage affected an additional 14 houses; and windows were damaged in a total of 37 dwellings. Within a 7-mile radius, excluding the inner 2-mile section, some 17 houses sustained minor structural damage; and 124 houses and other buildings had window damage. Within a 12-mile radius, which included the city of Washington, Missouri, 12 commercial cuilding had windows blown out, and 87 houses had broken windows. Two state policemen, on routine highway patrol about 25 miles from the scene, felt two separate concussions, heard what sounded like a sonic boom, and then saw the fireball. The City of St. Louis, Missouri, 55 miles away, felt the detonation; and a reading of 3.5^{*}was recorded on a seismograph there. The small map immediately following this page shows the locations and distances of damaged homes in the immediate area, and the larger map shows the location of all damage claims resulting from the explosion. The graph relates window damage to distance from the blast center.

There were no faculities in this event. The area was thinly settled, and the few residents there were amply warned by the roar of the escaping propade and the sight of the spreading cloud. They evacuated the valley and were all up on the valley ris when the blast and firestorm occurred. Ten persons were treated for injuries.

* The scale (Right (r)) was not specific [in the sources consulted, but a a vertical ground displacement of 2.2 X 10⁻³ nr was rentioned.



- x Pipeline break
- ★ Point of ignition
 ⊘ Apparent center of blust



A DESCRIPTION OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE





WINDOW DAMAGE - Percentage of structures with window damage within 5 miles of the blast site

This explosion is most unusual in that it is authoritatively thought to have been a detonation and not just the usual deflagrative explosion. Buildings were shattered and obliterated, rather than merely having roof and wall panels torn off. Trees were uprooted and stripped of all foliage, rather than being essentially unharmed as in the usual gas cloud blast. "The entire valley lit up at once" when the cloud ignited, rather than a flame rolling from one side of the cloud to the other as in the usual explosion report.

i

÷

. .

; 7.

ŕ

4

.........

;

į

ţ

Not all of the propane detonated, of course. There was a rolling firestorm following the blast, and a billowing fireball continued to burn at the rupture site for the next twelve hours. Burgess and Zabetakis estimated (from the distance-US-damage data) that approximately 7.5% of the propane released prior to the blast entered into the detonation with an energy equivalent to that from 50 tons of TNT. The balance burned.

The circumstances that led to detonation were a bit unusual too. The vapor cloud collected in a valley under a strong temperature inversion, where it was restrained from normal dispersion and where blast waves could be refracted back down into the flammable mixture. More importantly, the blast is thought to have grown to detonation *inside the confinement* of a concrete block warehouse where the reaction pressure was not readily relieved. It is the current general consensus that flammable clouds will not burn to detonation in the open but require some confinement; although the investigators of the Flixborough explosion $-\alpha_i^2$ Case Study F.1 - have recently challenged that view, and the Climax vinyl chloride cloud appears to have detonated (Case Study F.4). The latter may have been initiated by the shock wave from one of the Diesel engines, which apparently ingested VCM vapor and suffered an internal detonation which blew off the cylinder head. Roger Strehlow believes the East St. Louis railroad yard propylene explosion also involved a detonation inside the confinement of a boxcar (α_i^2 Case Study F.5).

Sources:

- Private discussions with Dr. David Burgess, BuMines, and Prof. Roger Strehlow, Univ. of Illinois.
- BuMines RI-7752 "Detonation of a Flammable Cloud Following a Propane Pipeline Break" 1973.
- NTSB Pipeline Accident Report, "Phillips Pipe Line Company Propane Gas Explosion, Franklin County, Missouri, December 9, 1970", NTSB-PAR-72-1.

F.7 CRUDE OIL VAPOR-MIST EXPLOSION HEARNE, TEXAS - 14 May 1972

At 12:30 AM on 14 May 1972, an 8-inch crude oil pipeline ruptured near Hearne, Texas; and crude oil sprayed into the air from a 6-inch irregular split in the top of the pipe. The sparsely populated surrounding countryside was showered with crude oil vapors and liquid. The crude oil flowed along a stream, passed through culverts beneath a railroad and a highway, and finally pooled up on a stock pond 1,800 feet away from the rupture.

Two frame houses were located 600 feet west of the leak in a fenced area which bordered the stream on the north side. One of these houses was within 50 feet of the stream; the other was 100 feet farther north. A petroleum tank-truck operation was located within 600 feet of the stock pond on which the crude oil had dammed up.

At 5 AM, 4-1/2 hours after the pipe had split, crude oil vapors were ignited by an unknown source in or near the small, single-story frame house adjacent to the crude-filled stream. The resultant explosion and fire killed one man, seriously burned another man and a young boy, and destroyed the house. An intense petroleum fire was kindled along the stream, which burned over and under the highway and the railroad culverts and ignited the crude oil in the stock pond. A fire 1800 feet long and several hundred feet high scorched the entire area from the leak site to the stock pond and killed all vegetation within this zone. The highway and railroad bridges were heatdamaged, and all highway and railroad traffic was stopped by the fire. The telephone, railroad, and pipeline communication lines were melted, and the poles were burned. 7,913 barrels of crude oil escaped and burned.

This rather sketchy case study has been included in this report partly because it was a flammable spill on water and partly because crude oil is not ordinarily thought of as dangerously flammable. However, crude oil contains an approximable percentage of volatile material, and this particular

297

Preceding page blank

crude was about 10% (by weight) lighter than hexane. Ten percent of 332,346 gallons (7,913 barrels) is 33,234 gallons of light hydrocarbons, or about one railroad tank car; and the spraying mode of the leak would strip it out into the air fairly efficiently to yield a flammable cloud much like that from a gas leak. A quiet spill onto the water of a harbor would give less afficient stripping of course, but even so the thin film of oil would yield up its light ends content fairly quickly and lead to a dangerously flammable vapor cloud.

Source: NTSB Pipeline Accident Report, "Exxon Pipe Line Company Crude Oil Explosion at Hearne, Texas, May 14, 1972", NTSB-PAR-73-2.

تنبعك فالمستكليك فاستلعاها متصافيه والافتقاص والفقائي والمقامين ومناجعها ومنيه والمتساد ومناكلتا التقالية

F.8 NATURAL GAS CLOUD EXPLOSION AND FIRE HOUSTON, TEXAS - 9 September 1969

At about 3:40 PM on 9 September 1969, a high pressure natural gas pipeline in a suburb north of Houston, Texas, burst from accidental overpressuring and filled the neighborhood with a cloud of natural gas. The initial pressure burst produced a concussion that threw several people, within a radius of 250 feet, from their chairs, couches or beds. There was a noise as of jet engines so severe that it was difficult to hear anything else, and the blowing gas created a duststorm-like condition that obscured visibility in the area.

Approximately 8 to 10 minutes later, a resident about 200 feet from the point of rupture heard a "crackling or sparking-type noise", followed by an explosion and fire. He was knocked down and suffered burns on his arm and face. Five houses in a row exploded at this time, indicating that the escaping gas had entered into these buildings. Fires broke out and raged to a height of 125 feet at the rupture point. The heat of the fires was so intense that buildings up to 250 feet away were completely destroyed. In all, thirteen houses were completely destroyed, and 106 others were damaged. The total damage was estimated to be \$500,000.00. All of the destroyed houses and those heavily dumaged were within 300 feet of the rupture point; they are located on the following sketch map. Between 300 and 600 feet from the rupture point, damage consisted mainly of cracked walls, blistered paint and broken windows. From 600 to 900 feet, damage was due to objects falling from shelves or walls. The destruction and heavy damage was concentrated to the north of the rupture site, apparently due to some protecting earth mounds to the south; but the lesser damage was consistent in all directions.

There were no deaths, partly due to swift evacuation efforts before the explosion-fire; but nine persons were injured, two seriously.

Source: National Transportation Safety Board Pipeline Addident Report, "Mobil Oil Corporation High-Pressure Natural Gas Pipeline, Near Houston, Texas September 9, 1969". NTS8-PAR-71-1.


-717.

.

SKETCH MAR OF SEVERE CAMAGE FROM BLAST AND FIRE MORTH OF HOUSTON, TEXAS, 9 SEPTEMBER 1969

F.9 LPG TANK CAR EXPLOSIONS CRESCENT CITY, ILLINOIS - 21 June 1970

At 6:30 AM on 21 June 1970, 15 railroad cars including nine tank cars loaded with liquefied petroleum gas derailed in the town of Crescent City, Illinois. The force of the derailment propelled the 27th car in the train over the derailed cars ahead, and its coupler struck the tank of the 26th car in passing, puncturing that tank. The released propane was ignited immediately by some unidentified source, possibly by sparks produced by the the derailing cars or by the overheated journal that caused the derailment in the first place. The initial ignition of the gas produced a fireball that reached a height of several hundred feet and extended into that part of the town surrounding the tracks. Several buildings were set on fire.

The safety values of other tank cars near the fire operated, releasing more propane which provided more fuel for the fire. More heat was directed against the elevated portion of the 27th car, which increased the pressures within that car until it ruptured with explosive force at about 7:33 AM. The east end of the car, which included the manway, dug a crater in the track structure and then was hurled about 600 feet eastward, where it stopped at an incresection in town. Fire in this section continued there after it came to rest. The west end of the car was hurled in a southwesterly direction for a total distance of about 300 feet. This section struck and collapsed the roof of a gasoline service station. Two other sizeable portions of the tank were hurled in a southwesterly direction and came to rest at points 600 feet and 750 feet from the tank.

At about 9:40 AM, the 28 car in the train ruptured. The south end of the car was hurled about 200 feet southward across the street where it entered a brick apartment building. The north end of the car was burled through the air in a northwesterly direction over the roofs of several houses and landed in an open field. It continued to roll and finally stopped after traveling about 1600 feet.

About five minutes later, at approximately 9:45 AM, the 30th car in the train ruptured. This car had been lying in a generally north-south position. The north end of the car, which included about one half of the tank, was propelled along the ground in a northeasterly direction for about 600 feet. This portion of the tank passed through and completely destroyed two buildings and came to rest within a third. The other end of the tank stayed in the vicinity of the general derailment site.

HY BERT LINE

(こう) たん ざいこう

į

ŗ

i inter

ł

- <u>-</u> - -

the second second second second second second second second second second second second second second second se

At about 10:55 AM, the 32nd car in the train and the following car ruptured almost simultaneously. One of them split longitudinally and did not separate into hurtling pieces. The west end of the other was hurled westward where it struck and punctured the head of the 34th car. Propane released from the puncture ignited inmediately. The other end of the 33rd car was hurled through the air. It struck the 34th car, ricocheted, and then struck the protective housing of the 35th car. The housing and valves of the 35th car broke off, permitting still more propane to escape and ignite. Fires continued to burn at the punctured cars for a total of 56 hours.

In all, 16 business establishments were destroyed and seven were damaged. Twenty-five residences were destroyed and a number of others were damaged extensively. Total injury and property damage costs amounted to approximately two million dollars. Sixty-six persons were injured; there were no deaths.

The key damage mechanism in this accident was the rocketing of burning tank cars which spread massive missiles of steel and new fireballs up to 850 feet from the derailment site. This is a characteristic result of all accidents wherein pressure vessels are exposed to massive fires, and it should be included in all disaster scenarios. The following map shows the paths of the major tank fragments in this accident.

Source: Mational Transportation Safety Board Reffrend Accident Report, "Derailment of Toledo, Peoría and Western Railread Company's Train No. 20 With Resultant Fire and Tank Car Ruptures, Crescent City, 1111nois, June 21, 1970". MTSB-RAR-72-2.



TPAJECTOPICS OF TAUX CAP FRACMENTS



DERAILMENT CONFIGURATION

F.10 LPG TANK CAR EXPLOSIONS LAUREL, MISSISSIPI - 25 January 1969

والمتعالية والمتحالية والمتحالية والمتحالية

Ī

į

ŧ

Ł

Ì

At 4.15 AM on 25 January 1969, a train carrying 15 tank cars of liquefied petroleum gas (primarily propane) derailed in the middle of the town of Laurel, Mississippi. The derailment was followed by an immediate, violent fire and explosion; and other explosions followed at intervals for the next 40 minutes as the fire caused one tank car after another to overpressure and burst.

The initial fireball immediately set fire to buildings 200 to 400 feet away, primarily residences, between the wreck and Meridian Avenue (see map) and to industrial buildings about 200 feet away on the opposite side of the tracks. Smaller, spot fires were set by burning fragments up to about 10 city blocks away.

The concussion from the explosions caused structural damage to both private and industrial buildings within about 400 feet. Broken windows were reported as far as three miles west of Laurel, but the majority of the broken glass was in the downtown area about 8 blocks away. One witness put it, "Very little glass survived within about half a mile to a mile". Damage claims were filed - and paid - from up to five miles away; but they were distributed in scattered bunches and not generally distributed. Cracked and broken plumbing was widespread in the concussion area; but such damage was thought to be a special case due to the prevalence of light-construction, frame buildings, simple-construction foundations and uncompacted soil which had already seen some creep and shifting due to heavy rains.

The major damage mechanism was impact and fire from rocketing tank car fragments and burning propane. Building and automobiles in the immediate vicinity were destroyed by direct flames or by heat radiation from the fireballs. Parts of the tank cars which were hurled through the area set fire to buildings as far as 1560 feet away as well as inflicting impact damage. A 37-foot section of POTX car 261 (see map) was propelled through the air in a southeasterly direction, striking the ground three times before it came to rest. It struck

the ground first about 1,000 feet from the wreck. After the first bounce, it carried another 300 feet over some residences without striking them; bounced again for 200 feet, and again for 100 feet, coming to rest atop a dwelling 1,600 feet from the wreck center. The resulting fire where the tank came to rest destroyed three houses. A 37-foot section of POTX car 269 was propelled through the air in a northwesterly direction striking the peak of the roof of a mill 800 feet from the track. This section then struck in the parking area and bounced end over end successively 100 feet, 200 feet and 50 feet before coming to rest in a reversed position on the rear of a dwelling 1,100 feet from the wreck. Other fragments took other trajectories, as shown on the map.

In all, some 2930 buildings suffered damage. Fifty four residences were destroyed, mostly by fire. A large transfer and storage company warehouse about 200 feet away and its contents were almost totally destroyed by fire, and a wash-and-dry laundry was totally destroyed. The structures of four other industries and businesses were heavily damaged: a hardboard fabricating plant and the Mississippi Power Company, west of the tracks; a machine shop and a wholesale grocer, east of the tracks (see map). Six public schools and five churches were damaged, all in the downtown area. Total preperty damage settlements amounted to \$6,615,195.

Two fatalities resulted from the wreck. A 65-year old minister died as a result of burns, and a 17-year old girl died about a month after the wreck. There were 976 other personal injuries, most of them burns and cuts from flying glass. Seventeen persons were hospitalized for more than a month.

The large number of personal injuries was the result of another special case which is instructive in itself. At the sound of the first explosion, people boured out into the streets instead of seeking shelter; and the streets were filled with people when the second explosion and fireball occurred. They were then trapped between the fire and the many fences that paralleled the tracks

and could not get back to shelter. There were many purely panic injuries as people were trampled in the rush and injured attempting to scale the fences. They were of course only 200 to 400 feet from the towering fireballs, and many burn injuries occurred.

There was still another unusual aspect to this accident. A National Guard explosive ordnance team punched holes in two unruptured tank cars, using shaped demolition charges, to prevent them from exploding. The wisdom of such action is still controversial, but in this case it worked. The gas ignited immediately with a sound described as "like a jet plane taking off", and about two hours later the area was declared safe for the people to return.

Sources:

dustan wardening (nr.

Į.

İ

È

ł

「こう モディモン・デートディーディ

- 1. Private communication, Mr. W. L. Millwood and his staff at Southern Railway System, Washington, DC.
- National Transportation Safety Board Railroad Accident Report, "Southern Railway Company Train 154 Derailment with Fire and Explosion, Laurel, Mississippi, January 25, 1969". October 6, 1969. (No NTSB number assigned)





F.11 VINYL CHLORIDE MONOMER TANK CAR EXPLOSIONS HOUSTON, TEXAS - 19 October 1971

At 1:44 PM on 19 October 1971, 20 freight cars were derailed at Mykawa Station, approximately 2 miles inside the southern city limits of Houston and 10 miles south of the city center. Six of the cars contained flammable vinyl chloride monomer, a liquefied gas under a pressure of approximately 40 psi at the existing temperature; and two others concained butadiene (also a figurmable, compressed gas) and acetone. The cars jackknifed in the derailment, and two vinyl chloride cars were punctured.

There was an explosion and a billowing fire before the wrecked cars had ceased to move. The two punctured cars came to rest in the burning area, and the fire enveloped large areas of the tank cars. The heat increased the internal pressure until at approximately 2:30 PM a violent rupture and massive energy release occurred when the tank of one car suddenly failed. The sudden release of approximately 100,000 pounds of vinyl chloride monomer produced a large fireball and rocketing tank fragments. Residual vinyl chloride in the other tank car ignited inside the car at that point; and the car rocketed in one piece to a point approximately 300 feet away, spewing burning, liquefied vinyl chloride from its punctures. The acetone car, which had been struck and damaged by the rocketing car, caught fire; and a house and several vehicles were ignited by the stream of burning vinyl chloride. The secondary fires burned for approximately five hours before the last one was extinguished.

Other than the train and railroad roadbed, damage was very light. A nearby residence, a fire truck, an automobile and a railroad motor truck were destroyed by fire; and several buildings in the area incurred such damages as paint blisters or broken windows. Fortunately, the area was very sparsely built and populated.

One fireman was killed; and 50 people, most of them firemen, were injured. Their approximate locations are plotted on the map. Twenty of the injured were hospitalized, mostly for burns. An autoosy was not performed on

the fireman who was killed, but he had suffered severe head injuries as well as burns. Most of the injuries appear to have been burns from fireball radiation, an observation that is consistent with the firefighters' rule of thumb that firemen have succumbed to radiation burns incurred when they were as far away as 250 feet from very large fireballs.

The following map shows the location of the burning tank cars, the injured persons and the approximate outline of the area which contained scorch marks and debris.

Source: National Transportation Safety Board Railroad Accident Report, "Derailment of Missouri Pacific Railroad Company's Train 94 at Houston, Texas, October 19, 1971". NTSB-RAR-72-6.



F.12 PROPYLENE TANK TRUCK EXPLOSION NEW JERSEY TURNPIKE - 21 September 1972

At 8:25 PM on 21 September 1972, a tractor-semitrailer (tank) carrying propylene liquid petroleum gas sideswiped a Greyhound bus (carrying no passengers) in the southbound lanes of the New Jersey Turnpike about one mile south of Exit 8. Fire, caused by friction sparks when the tractortrailor scraped the median guardrail, ignited fuel escaping from the tractor's left-side fuel tank and spread to propylene which was leaking from the cargo tank's damaged plumbing, enveloping a large part of the cargo tank in flame.

About 20 or 25 minutes after the crash, the cargo tank exploded in a huge ball of fire which inflicted burns and other injuries to 28 persons, including 7 police officers and at least one bystander who was 600 feet south of the explosion. None of these injuries was serious. Two neople trapped in an automobile wedged between the tank truck and the guardrail were killed, but it is not clear whether they were alive at the time of the explosion. Their bodies were extensively charred. A water-tank fire truck packed about 100 to 150 yards south of the explosion suffered scorched paint and warped plastic light components; an occupant said that a flaming mass case directly over the vehicle. At the time of the explosion, there were about 200 onlookers dispersed along the east and west property lines at distances ranging from 150 to 1000 feet. The locations of all 28 injured persons could not be established, but the known positions of eight of them are shown on the man following.

Pieces of the exploded tank shell rocketed. The front section, about 27 feet long or three-quarters the length of the tank was found 1,307 feet northeast of the explosion point and 402 feet east of the highway center. The rear head was 540 feet southwest, and the rear one-quarter of the cylindrical section was 229 feet southwest of the explosion point. Other tank components were found at various points along an 850-foot line generally

Preceding page blank

southeast of the explosion point. Parts of the trailer suspension and the two axles, with the tires burned off, were found along the eastern embankment of the turnpike. When first seen, the axles and wheels were glowing.

There was no air blast; the explosion was that of the bursting tank shell. Most of the fireball was produced when the expelled vapor burned in air after rupture of the tank. The tank had originally contained 7,209 gallons of propylene.

Source: National Transportation Safety Board Highway Accident Reporc, "Multiple-Vehicle Collision Followed by Propylene Cargo-Tank Explosion, New Jersey Turnpike, Exit 8, September 21, 1972". NTSB-HAR-73-4.



LOCATIONS OF EIGHT OF THE PERSONS BURNED BY THE FIREBALL

and an and a state

2

315

Ð



2000-000-09220-0009020-01 10-00-0**0**0000000-000-000-00

į



F.13 PROPANE CLOUD FIREBALL LYNCHBURG, VIRGINIA - 9 March 1972

E 4 B 5

.

At 2:30 PM on 9 March 1972, a tractor-semitrailer tank truck carrying 9,208 gallons (38,854 pounds) of LPG (mostly propane) overturned and was punctured. As the truck fell onto its right side, a 32-inch hole was torn in the LPG tank; and liquid LPG poured out and began to vaporize. It is estimated that approximately 4000 gallons of liquid were discharged before the level dropped below the opening and the flow of liquid stopped.

As the propane escaped, a visible vapor cloud containing a mixture of propane and air immediately began to form at the trailer and spread rapidly throughout the area. At the site, an embankment east of the highway rose steeply, while an embankment west of the highway consisted of a steep dropoff. A house and some outbuildings were located 200 feet west of and about 60 feet below the accident site. A fringe of trees and underbrush jutted above the level of the road.

The driver climbed out of the truck, attempted to wave oncoming motorists away and then began to run downhill away from the truck and the expanding cloud. When he had run approximately 270 feet, the vanor cloud, which had continued to expand, ignited into a fireball with an estimated radius of at least 400 feet, enveloping him and killing him. Three witnesses standing beside their automobiles approximately 450 feet away were severly burned by radiation but were not touched by the flames. The occupants of the house, hearing the crash and seeing the cloud, fied from the house and were about 420 feet from the truck when the cloud ignited. They were severly burned, and one later died; it is not clear whether they were just inside or just outside the fireball. The house, 200 feet from the wrack, was enveloped in the flames and destroyed.

There appears to have been no blast. 4.5 motorists 600 feet or so to the north (uphill, around a curve and out of the line of sight) heard only a muffled roar and saw an orange-red fireball. No concussion was felt, and

they were not injured. The actual fireball appears to have been approximately 400 feet in radius (that is, in the downhill direction; it is not thought to have been symmetrical); and there was incendiary radiation beyond that. Five people slightly more than 400 feet from the wreck site were severely burned, and an ensuing conflagration destroyed the house and outbuilding and 12 acres of woodland.

. . .

No mention is made of wind or weather in the report, and that would seem to indicate the absence of any high winds or of rain.

Source: National Safety Transportation Board Highway Accident Report, "Propane Tractor-Semitrailer Overturn and Fire, U.S. Route 501, Lynchburg, Virginia, March 9, 1972". NTSB-HAR-73-3.



₹ 8

MAP OF ACCIDENT SITE

At 10:53 PM on 22 February 1973, a 10-inch pipeline carrying natural gas liquids (NGL) at a pressure of 525 psig failed at the Austin pump station, releasing a total of 6,640 barrels (278,880 gallons) of NGL. The NGL blew a 10-foot diameter hole in the ground, sprayed into the air, flowed into ditches on both sides of a road adjacent to the station, and formed a white, fog-like vapor which enveloped the area.

Shortly after 11 PM, two cars traveling eastbound entered the vapor-rich zone and stalled. Both drivers got out of their cars, heard the roar of the escaping NGL, and smelled what they thought to be natural gas. Neither driver attempted to restart his car. Both ran castward along the road through the two-foot-deep NGL vapor. Shortly after the two cars stalled, a Dodge van carrying six adults and two small children entered the vapor-rich zone and also stalled. The passengers got out and started to walk enstward toward the two other stalled vehicles, and the driver attempted to restart his engine.

A large snark emerged from underneath the van, and flames leaped hundreds of feet into the air. The fire burned 2,400 feet east from the stalled vehicle, over and along both sides of the road, and back to the nipe rupture. The eight persons still in the vapor-rich zone and the three vehicles were completely engulfed by flames.

Four of the occupants of the van died immediately. The four other occupants were hospitalized; two of their subsequently died, and the other two suffered severe burns. The van and the two stalled automobiles were completely destroyed by the fire, and the power and concurrentiations lines on the poles next to the road were melted. The asphaltic paverent of the road had been charred; fenceposts and pasture on both sides of the road were burned black in a 2,400-foot long line.

Preceding page blank

There was no blast. Two police officers in a radio patrol car saw the fire ignite, but they neither heard any explosion nor felt any shock wave.

1.71.72.2

NUMBER OF THE OWNER OWNER OWNER O

1

1

:

1.1

i

11.21

The weather at the time of the pipe failure was cloudy, and a light wind was blowing from the southeast at 6 mph. The temperature was 44°F, and the humidity was 86 percent. By 11:30, the wind velocity had dropped to zero.

angan 2003/19977

والأجز بالأعلاء المكتبرة فبشعاء

NGL is a member of the liquefied petroleum gas family. It is liquid stripped out of natural gas by pressure condensation, and contains about 85% butane and lighter, predominately propane. It is liquid only under greater than atmospheric pressure (or refrigeration). When released to the atmosphere at normal temperatures, it will vaporize, cool the surrounding air, and cause any moisture contained therein to appear as a white fog.

Source: National Transportation Safety Board Pipeline Accident Report. "Phillips Pipe Line Company Natural Gas Liquids Fire, Austin, Texas, February 22, 1973". NTSB-PAR-73-4.



APPENDIX G

ADJUSTMENTS TO PREVENT DOUBLE COUNTING

والمراجعة والمراجعة والمراجعة

Provisions have been made in the VM to prevent double counting in three different situations. Double counting is used in this context to mean the inclusion of an element of some vulnerable resource (e.g. a person or a building) in more than one category of damage or injury. Three situations arise in which double counting will occur unless provisions are made to prevent it. The situations are:

- (1) A single damage mechanism from one event simultaneously causes injuries of differing severity (e.g. inhalation of toxic gas may cause death, non-lethal injury, or irritation).
- (2) Two or more damage mechanisms from one event simultaneously cause injuries of the same severity (e.g. an explosion cau kill people either by direct blast effects or by impact).
- (3) Different events at different times both cause damage to the same resource, and the first event so severely damages some portion of the resource that further damage is irrelevant (e.g. persons killed by toxic gas cannot be further injured by a subsequent explosion).

The problem of double counting encountered in situation (1) arises from the way the probit equations, described above, are obtained. A probit equation assesses the fraction of the vulnerable resource in a given category of damage or injury without regard to other categories of damage. For events that cause an escalating sequence of damage categories, the results of the unmodified probit equations are misleading, because that portion of the subject resource in a given damage category is also counted in all more serious damage categories. As an example, consider the case of a toxic vapor: those people counted as irritated include those people with non-lethal injuries and those killed by the vapors, since the threshold for irritation is lower than the threshold for injury or death; i.e. to be injured or killed, one has to have passed through the stage of irritation.

An unattractive feature of using the unmodified results of the probit equations is that the sum of the percentiges of resource in the various damage categories may exceed 100%.

To correct for this type of double counting, the fraction of the total resource in a given damage category is reduced by the sum of the fractions of total resource in all damage categories more serious than the given category; of course, the adjusted fraction must not fall below zero. To state this in mathematical terms let,

- P₁ = the probit equation for damage category i, where the damage becomes more serious as i increases
- and $F(P_1) =$ the fraction of the total resource calculated to be in category 1 from the probit equation, P_1

Preceding page blank

then
$$F'(P_i) = F(P_i) - \sum_{i=i+1}^{r_i}$$

where $F'(P_i)$ = the adjusted value for the fraction of the total resource in category i (a single prime(') indicates adjustment for category (1) situations); if $F'(P_i) < 0$, then set $F'(P_i) = 0$.

(G-1)

M = the total number of damage categories.

As an example consider the following case where the concentration history resulted in 4% of the population of a given cell dying:

 $F'(P_{1})$

Type of Injury	Initial <u>Percent</u>	Adjusted Percent
Lethal	4	4
Non-lethal	12	12 - 4 = 8
Irritation	100	100 - (4+8) = 88

In the first column is the percentage affected when more serious effects are included at each level, while the more easily interpreted figures are in the final column. Using the initial figures, one might conclude that 116% of the population was affected. This type of correction procedure is applied to injuries from direct blast effects:

$$F'(E3) = F(E3) - F(E1)$$
 (G-2a)

to injuries from impact:

F'(E4) = F(E4) - (F(E2)) (G-2b)

to glass breakage:

F'(S2) = F(S2) - F(S1) (G-2c)

and to sublethal toxic effects:

F'(T2) = F(T2) - F(T1) (G-2d)

$$F'(T3) = F(T3) - (F(T1) + F'(T2))$$
 (G-2e)

The problem of double counting encountered in the situations (2) arises because the probit equations assess injury without regard to other injuries occurring simultaneously from different causes. Thus some fraction of the vulnerable resource may be subject to two or more causes of damage. This may be visualized by means of a diagram in which geometrical area represents resources in a particular state. By allowing overlap, the effects of multiple mechanisms may be considered. These diagrams, known as Venn diagrams, are shown in Figure 6-1 for the cases of two and three competing mechanisms. For two damage mechanisms, Figure 6-1a shows that the resource is partitioned into four categories. For three mechanisms, Figure 6-1b shows that eight categories result. In general, for n competing simultaneous mechanisms, 2ⁿ categories will result.





A Venn diagram for two simultaneous injury mechanisms; the four resulting categories of injury are:

- (1) injury A only
- (2) injury B only

- (3) Injury A and B
- (4) no injury of either type





A Venn diagram for three simultaneous injury mechanisms; the eight resulting categories of injury are:

- (1) injury A only
- (2) injury B only
- (3) injury C only
- (4) injury A and B
- (5) injury A and C
- (6) injury B and C
- (7) injury A, B, and C
- (8) no injury of any type

Since the probit equations give the fraction of the vulnerable resource in any independent damage category, the fraction of the vulnerable resource in those categories arising from the simultaneous occurrence of damages may be calculated using the techniques of mathematical probability[G1] (particularly the complementation rule and the multiplication rule). Section of the section of the

ł

4

1

To illustrate these techniques consider the following example. Two injury mechanisms, A and B, occur simultaneously. Let

F(A) = fraction of vulnerable resources injured by mechanism A.

F(B) = fraction of vulnerable resources injured by mechanism B.

Then by the complementation rule the fraction not injured by mechanism A is given by the quantity 1-F(A) and the fraction not injured by mechanism B, is given by the quantity 1-F(B).

By use of these results and the multiplication rule the fraction of vulnerable resource in each of the four categories of injury may then be calculated; thus:

fraction injured by A only	=	F(A) [1-F(B)]
fraction injured by B only	=	F(B) [1-F(A)]
fraction injured by both A an	d B =	F(A) F(B)
fraction uninjured	=	[1-F(A)] [1-F(B)]

This procedure is illustrated by a specific numerical example shown in Figure 6-2.

At the present time the VM only treats explosions as causing damage by several simultaneous damage mechanisms. Death may be caused by either direct blast or by impact effects. Non-lethal injuries may be caused by direct blast, by impact, or by fragments.

In the case of deaths it is not of particular relevance that part of the population killed may have received two types of injury each sufficient alone to cause lethality. For this reason a separate category of deaths by multiple cause was not established for reporting out the results of Phase II. Instead the fraction of deaths from multiple cause was added to the fraction of deaths from each cause alone in proportion to the deaths caused by each effect alone. That is, the fraction of deaths due to multiple causes is given by

F(EM) = F(E1) F(E2)

[[]G1] Kreyszig, E. Advanced Engineering Mathematics. John Wiley and Sons, Inc., N.Y., 1967. Section 18.5.



In this example it is assumed that two mechanisms simultaneously cause damage to a resource or population. Let F(A) = 0.40 and F(B) = 0.20. Let the entire area of the square above represent the number of elements of the vulnerable resource.

.

If two adjoining sides of the square are segmented proportional to the fraction of vulnerable resource injured respectively by the two mechanisms, then the areas of the regions formed within the square are proportional to the fraction of the resource in each of the four damage categories shown.

F1GURE G-2

DAMAGE APPORTIONMENT DIAGRAM

Carles and the second

where

F(E1) = fraction of deaths caused by direct blast effects

F(E2) = fraction of deaths caused by impact.

The fraction of deaths due solely to one cause is given by

F(EB) = F(E1) [1-F(E2)] (G.-3a)

F(EI) = F(22) [1-F(E1)] (G-3b)

where

F(EB) = fraction of deaths due solely to direct blast

F(EI) = fraction of deaths due solely to impact.

Now the fraction of deaths due to multiple causes, F(EM), is added to the deaths due to each cause alone in proportion to their relative magnitude; i.e.

$$F''(E1) = F(EB) + F(EM) \left[\frac{F(EB)}{[F(EB) + F(E1)]} \right]$$
 (G-4a)

and

$$F''(E2) = F(E1) + F(EM) \left[\frac{F(E1)}{[F(EB) + F(E1)]} \right]$$
 (G-4b)

where F''(E1) and F''(E2) are taken to be the adjusted fractional values for deaths from direct blast and impact respectively.

In the case of injury from explosions the fact that part of the population is wounded by several mechanisms is considered important. Therefore, the fraction of the population in this category is determined and reported out in Phase II of the VM; however, no distinction is made between the different categories of multiple injury (there are four categories: (1) direct blast and impact, (2) direct blast and fragments, (3) impact and fragments, (4) direct blast and impact and fragments).

The adjusted fractional values for various types of explosion injury are calculated using the techniques of mathematical probability. The adjusted fractional values reported out by the VM are calculated by the following formulas: F''(E3) = F'(E3) [1-F(E4)] [1-F(E5)] (G-5a)

F''(E4) = F'(E4) [1-F(E3)] [1-F(E5)] (G-5b)

F''(E5) = F(E5) [1-F(E3)] [1-F(E4)] (G-5c)

$$F''(EM) = F(E5) F'(E4) [1-F(E3)] + F(E5) F'(E3) [1-F(E4)] + F'(E3) F'(E4) [1-F(E5)] + F(E5) F'(E3) F'(E4) (G-5d)$$

where,

And in the second second

÷

ł

:

.

٩,

2

1

F''(E3), F''(E4), F''(E5), F''(EM) =

the adjusted values for fraction of deaths due to direct blast only, impulse only, fragments only, and multiple injury respectively.

and

F'(E3), F'(E4) are given respectively by equations (G-2a), (G-2b).

The problem of double counting encountered in situation (3) arises because the probit equations for a given damage mechanism do not consider the reduction in the population of a vulnerable resource caused by a prior damage mechanism. In situation (3) the temporal order of the damage mechanism affecting a given vulnerable resource is crucial Figure G-3 is useful for visualizing the temporal order of various damage mechanisms. As indicated in figure G-3 diffusion of toxic gases may be followed by either flash fire or explosion; either of these events may then be followed by pool burning.

The VM models pool burning as causing no damage to people; therefore, this type of double counting may arise for the vulnerable resource "people" only when explosion or flash fire follow a significant toxic dose. The VM models toxicity as causing no damage to structures; therefore, this type of double counting may arise for the vulnerable resource "structures" only when pool burning follows explosion or flash fire.

A correct assessment of damage from a sequence of events requires the double counting correction for situation (3). However, the degree of threat imposed by each event is also of interest. For example, even though an explosion may destroy all the structures in a given region, the fact that a subsequent pool burning would also destroy these structures is of some interest to the users of the VM. Therefore, this correction is not made as each damage-causing event is simulated. Instead the uncorrected percent damage is reported out by the VM. Then after the simulation is complete, a summary damage report is generated; it is for the summary damage report that this correction for double counting is made.

For structures the correction made for the summary is given by,



Ŧ

*5**



$$F'''(PB) = F(PB) \cdot [1-G]$$
 (G-6a)

where

CALCER AND AND A COMPLEX AND A SHEEK

$$G = \begin{cases} F(FF) & \text{if flash fire} \\ F(S1) & \text{if explosion} \end{cases} \text{ precedes pool burning} \qquad (G-6b)$$

and where

F"'(PB)	=	the corrected fractional structural damcge for pool burning
F(PB)	-	the original fractional structural damage for pool burning
F(FF)	-	the fractional structural damage for flash fire
F(S1)	-	the fractional structural damage for explosion

For people the correction made for the summary is given by the following:

for explosion:

÷

F'''(E1) = F'''	(E1) [1-F(T1)]	(G-7a)
-----------------	--------------------------------	--------

$$F''(E2) = F''(E2) [1-F(T1)]$$
 (G-7b)

$$F''(E3) = F'(E3) [1-F(T1)]$$
 (G-7c)

$$F'''(E4) = F''(E4) [1-F(T1)]$$
 (G-7d)

$$F''(E5) = F''(E5) [1-F(T1)]$$
 (G-7e)

$$F'''(EM) = F''(EM) [1-F(T1)]$$
 (C-7f)

for flash fire:

$$F''(F1) = F(F1) [1-F(T1)]$$
 (G-7h)

where the triple-primed variables are the frational damage corrected for the summary and the other symbols have been previously defined.

U. S. GOVERNMENT PRINTING OFFICE 1975